.

#### **CWI Tracts**

#### **Managing Editors**

J.W. de Bakker (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.K. Lenstra (Eindhoven University of Technology)

#### **Editorial Board**

W. Albers (Enschede)
P.C. Baayen (Amsterdam)
R.C. Backhouse (Eindhoven)
E.M. de Jager (Amsterdam)
M.A. Kaashoek (Amsterdam)
M.S. Keane (Delft)
H. Kwakernaak (Enschede)
J. van Leeuwen (Utrecht)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
M. Rem (Eindhoven)
H.J. Sips (Delft)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

#### CWI

P.O. Box 4079, 1009 AB Amsterdam, The Netherlands Telephone 31 - 20 592 9333, telex 12571 (mactr nl), telefax 31 - 20 592 4199

CWI is the nationally funded Dutch institute for research in Mathematics and Computer Science.

# **CWI** Tract

On Topological and Linear Equivalence of certain Function Spaces

J.A. Baars J.A.M. de Groot

,



Centrum voor Wiskunde en Informatica Centre for Mathematics and Computer Science

86

1985 Mathematics Subject Classification: 54C35, 57N17, 57N20. ISBN 90 6196 411 3 NUGI-code: 811

.

Copyright  $\circledast$  1992, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

## **Table of contents**

#### Chapter 0. Introduction and notation

§0.1.	Introduction	5
§0.2.	Notation	7

#### Chapter 1. Tools and first applications

§1.1.	Topologies on function spaces	9
§1.2.	Linear functions between function spaces	12
§1.3.	The dual of $C_p(X)$ and $C_p^*(X)$	24
<b>§1.4</b> .	Supports and the topology of pointwise convergence	29
§1.5.	First applications	31

# Chapter 2. On the $l_p$ and $l_0$ -equivalence of locally compact spaces

\$2.1. Ordinals	44
§2.2. Derivatives and scattered spaces	51
§2.3. Factorizing function spaces	56
§2.4. Separable metric zero-dimensional compact spaces	62
§2.5. Compact ordinals	68
§2.6. $\sigma$ -compact ordinals	81
§2.7. Separable metric zero-dimensional locally compact spaces	94

#### Chapter 3. On topological equivalence of function spaces

§3.1.	Preliminaries and Q-matrices	100
§3.2.	Homeomorphic function spaces part 1	112
§3.3.	Homeomorphic function spaces part 2	123
§3.4.	Remarks	129

### Chapter 4. On the $l_p$ -equivalence of metric spaces

§4.1. $l_p$ -equivalent	t properties of metric spaces	135
-------------------------	-------------------------------	-----

§4.2. An isomorphical classification	149
§4.3. More $l_p$ -equivalent properties of metric spaces	163
§4.4. Remarks on a conjecture	174
§4.5. Partial results on $l_0$ -equivalence	182
§4.6. Partial results on $l_p^*$ -equivalence	187
References	193
Special Symbols	197
Author Index	
Subject Index	

## CHAPTER 0

## **Introduction and notation**

#### **§0.1. Introduction**

For a Tychonov space X, C(X) is the set of all real-valued continuous functions on X. The set C(X) endowed with the topology of pointwise convergence will be denoted  $C_p(X)$  (for more precise definitions see section 1.1). The spaces  $C_p(X)$  are of interest to topologists and functional analysts for various reasons.

One can consider  $C_p(X)$  as a topological ring (with the usual addition and multiplication of functions). In [40], J. Nagata proved the following

**0.1.1 THEOREM:** Let X and Y be Tychonov spaces. The spaces  $C_p(X)$  and  $C_p(Y)$  are topologically isomorphic as topological rings if and only if X and Y are homeomorphic.

In this theorem it is essential to consider topological isomorphisms. There are nonhomeomorphic spaces X and Y such that the rings C(X) and C(Y) are algebraically isomorphic (see [25]). Once we have J. Nagata's result it is natural to consider  $C_p(X)$  as a topological vector space (with the usual addition and scalar multiplication) or just as a topological space. In view of this we can state two general problems.

**0.1.2 PROBLEM:** Let X and Y be Tychonov spaces and suppose that  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic or just homeomorphic. Which topological properties  $\mathcal{P}$  satisfy: X has property  $\mathcal{P}$  if and only Y has property  $\mathcal{P}$ ?

**0.1.3 PROBLEM:** Let X and Y be Tychonov spaces. Under what conditions on X and Y are  $C_p(X)$  and  $C_p(Y)$  linearly homeomorphic or just homeomorphic?

Many topologists have worked on both problems. We will mention a few results that

are in the same spirit as the ones that will be derived in this monograph. (for a survey of recently obtained results we refer to [2]).

For example, concerning problem 0.1.2 we have for linear homeomorphisms a positive answer for pseudocompactness, compactness,  $\sigma$ -compactness (Arhangelskii [1]), and dimension (Pestov [44]). A negative answer can be obtained for local compactness, first countability, second countability, metrizability, weight and character (cf. example 2.4.10 in this monograph). A useful strategy is to find pairs ( $\mathcal{P}, \mathcal{D}$ ) of topological properties such that a Tychonov space X satisfies  $\mathcal{P}$  if and only if  $C_p(X)$  satisfies  $\mathcal{D}$ . In this way it is proved that for density (Guthrie [29]) and cardinality (Arhangelskii [2]) problem 0.1.2 has a positive answer. On the other hand there exist a compact space X and a non-compact space Y such that  $C_p(X)$  and  $C_p(Y)$  are homeomorphic (cf. chapter 3 in this monograph).

In this monograph we present our contributions to problems 0.1.2 and 0.1.3 and related problems. We do not restrict ourselves to the topology of pointwise convergence. We also consider other topologies on C(X) (mainly the compact-open topology on C(X)) and on  $C^*(X)$ , the set of all bounded real-valued continuous functions. Our results depend strongly on the results obtained by Arhangelskii in [1]. We discuss [1] in detail in section 1.2.

In chapter 1 we mainly develop tools that will be important in later chapters. However, we also present some new results. In section 1.5 we prove for normal first countable spaces X and Y such that  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic, that the set of accumulation points of X is countably compact if and only if the set of accumulation points of Y is countably compact. The first countability assumption is essential. This result is joint work with J. van Mill [5]. Furthermore we prove in this section for metric spaces X and Y such that there is a continuous linear surjection from  $C_p(X)$  onto  $C_p(Y)$ , that Y is completely metrizable whenever X is. This result is joint work with J. Pelant [7], and answers a well-known research problem of Arhangelskiĭ.

In chapter 2 we deal with function spaces of locally compact spaces. We give a complete isomorphical classification of the function spaces  $C_p(X)$  and  $C_0(X)$  (as topological vector spaces) where X is a member of one of the following classes:

(a)	compact zero-dimensional metric spaces	(section 2.4)
(b)	compact ordinals	(section 2.5)
(c)	σ-compact ordinals	(section 2.6)
(d)	separable metric zero-dimensional	
	locally compact spaces	(section 2.7).

The isomorphical classification of the function spaces  $C_0(X)$ , for X an element of class

#### §0.2. Notation

(a) or (b) is an old result (cf. [10] and [34]). The first three sections contain preliminaries which are of particular importance in this chapter. They deal with ordinals (in particular the for us important notion of a prime component), scattered spaces and factorizing lemmas on function spaces.

After the results in chapter 2, it is natural to consider non-locally compact spaces. In chapter 3 we prove for non-locally compact countable metric spaces X and Y that  $C_p(X)$  and  $C_p(Y)$  are homeomorphic. This result is joint work with J. van Mill and J. Pelant [6]. It was later extended to non-discrete countable metric spaces (see [16] or [20]), by different techniques.

In chapter 4 we consider linear homeomorphisms between function spaces  $C_p(X)$  for metric spaces X. A new tool is developed there, namely the notion of  $l_p$ -equivalent pair. This notion provides us with many properties for which problem 0.1.2 can be positively answered in the class of zero-dimensional separable metric spaces (sections 4.1 and 4.3). A complete isomorphical classification will be given for function spaces  $C_p(X)$ (as topological vector spaces) of countable metric spaces X with scattered height less than or equal to  $\omega$  (section 4.2). We indicate in section 4.4 that an isomorphical classification for function spaces  $C_p(X)$  for all countable metric spaces X seems beyond reach. Finally in this chapter some results will be given concerning the compact-open topology (section 4.5) and concerning the set of bounded continuous real-valued functions (section 4.6). We construct locally compact countable metric spaces X and Y such that  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic, while  $C_p^*(X)$  and  $C_p^*(Y)$  are not linearly homeomorphic.

AMS Subject Classification: 54C35, 57N17, 57N20.

Acknowledgments: We would like to thank J. van Mill and J. Pelant for the pleasant cooperation and their valuable contributions. We also would like to thank F. van Engelen and A.B. Paalman-de Miranda for their careful reading of the manuscript and their valuable comments.

#### §0.2. Notation

For all undefined notions and results on general topology without explicit reference we refer to [23] and [24].

Introduction and notation

Notation	Meaning
N	the set of natural numbers
Q	the set of rational numbers
IR	the set of real numbers
С	the Cantor set
$X \approx Y$	X and Y are homeomorphic
$X \sim Y$	X and Y are linearly homeomorphic
Ā	the closure of A in X
Int A	the interior of A in X
minA	the minimum of A
maxA	the maximum of A
sup A	the supremum of A
inf A	the infimum of A
$B(x, \varepsilon)$	$\{y \in X : d(x, y) < \varepsilon\}$
diam A	$\sup \left\{ d(x,y) : x, y \in A \right\}$
d(X)	the density of X
	the cardinality of X
$\mathcal{P}(X)$	the power set of X
$X \oplus Y$	the topological sum of the spaces $X$ and $Y$
$\oplus_{i=1}^{\infty} X_i$	the topological sum of the spaces $X_i$
$f \circ g$	the composition of two functions $f$ and $g$
$id_X, 1_X, 1$	the identity function of X
$\pi_i\colon \prod_{i\in I} X_i \to X_i$	the projection on the <i>i</i> -th coordinate
$\chi_A$	the characteristic function of A
11.11	the norm on a Banach space
Ker F	the kernel of a linear function $F$
span $\{v_1,\ldots,v_n\}$	the linear span of $\{v_1, \ldots, v_n\}$
$\operatorname{conv} \{v_1, \ldots, v_n\}$	the convex hull of $\{v_1, \ldots, v_n\}$
S <sup>n</sup>	the unit sphere in $\mathbb{R}^{n+1}$
$B^{n+1}$	the unit ball in $\mathbb{R}^{n+1}$
$W(\alpha)$	the set of all ordinals smaller than $\alpha$
[1, α]	the space of ordinals $\{\beta: 1 \le \beta \le \alpha\}$ with the order topology
[1, α)	the space of ordinals $\{\beta: 1 \le \beta < \alpha\}$ with the order topology
ω	the first infinite ordinal
ω <sub>l</sub>	the first uncountable ordinal
<b>א</b> _0	the cardinality of $\omega$

## CHAPTER 1

#### **Tools and first applications**

#### All spaces considered in this chapter are Tychonov.

In this chapter we introduce function spaces endowed with several topologies. Our main interest will be the topology of pointwise convergence and the compact open topology. In section 2 we present important results of Arhangelskii [1] which are among the main tools in this monograph. Section 3 deals with the topological dual of a function space endowed with the topology of pointwise convergence, and section 4 gives some more details about the results of section 2, when dealing with the topology of pointwise convergence. Finally in section 5 we give some first applications. We present topological properties which are preserved by  $l_p$ ,  $l_0$  or  $l_b$ -equivalence (resp.  $l_p^*$ ,  $l_0^*$  or  $l_b^*$ -equivalence), and properties which are not preserved by  $l_p$ ,  $l_0$  or  $l_b$ -equivalence (resp.  $l_p^*$ ,  $l_0^*$  or  $l_b^*$ -equivalence). For definitions of these notions see section 1.5.

#### **§1.1.** Topologies on function spaces

For a space X we define C(X) to be the set of all real-valued continuous functions on X and  $C^*(X)$  to be the set of all bounded real-valued continuous functions on X. C(X) and  $C^*(X)$  are vector spaces with the natural addition and scalar multiplication. For a covering  $\mathcal{K}$  of X we define a topology on C(X) by taking the family of all sets

 $\langle f, K, \varepsilon \rangle = \{g \in C(X) : |f(x) - g(x)| < \varepsilon \text{ for every } x \in K\},\$ 

where  $f \in C(X)$ ,  $K \in \mathcal{K}$  and  $\varepsilon > 0$ , as a subbase. If  $\mathcal{K}$  is a covering of X consisting of compacta, C(X) endowed with this topology is easily seen to be a topological group, whence in this case it generally suffices to consider open sets <0, K,  $\varepsilon$ >, where  $K \in \mathcal{K}$  and  $\varepsilon > 0$ .

A subset A of a space X is said to be **bounded** whenever for every  $f \in C(X)$ , f(A) is

bounded in  $\mathbb{R}$ . Note that if A is a bounded subset of X, then  $\overline{A}$  is also a bounded subset of X. If  $\mathcal{K}$  consists of all finite, compact or bounded subsets of X, respectively, we denote C(X) endowed with this topology by  $C_p(X)$ ,  $C_0(X)$  or  $C_b(X)$ , respectively. The topology on  $C_p(X)$  or  $C_0(X)$ , respectively, is often called **the topology of pointwise** convergence or the compact-open topology, respectively. We have that  $C_p(X)$  and  $C_0(X)$  are topological vector spaces.

For spaces X and Y, the notation  $X \le Y$  means that X and Y have the same underlying set and the topology on Y is finer than or equal to the topology on X. With this notation we have

 $C_p(X) \le C_0(X) \le C_b(X).$ 

**1.1.1 LEMMA:** Let X be a space and let K be a covering of X consisting of compacta. Let  $A_1, \ldots, A_n \in \mathcal{K}$ ,  $f_1, \ldots, f_n \in C(X)$  and  $\varepsilon_1, \ldots, \varepsilon_n > 0$ . Then for every  $f \in \bigcap_{i=1}^n \langle f_i, A_i, \varepsilon_i \rangle$ , there is  $\delta > 0$  such that  $\langle f, \bigcup_{i=1}^n A_i, \delta \rangle \subset \bigcap_{i=1}^n \langle f_i, A_i, \varepsilon_i \rangle$ .

**PROOF:** For  $i \le n$ , let  $\gamma_i = \max\{|f(x) - f_i(x)| : x \in A_i\}$ . Then  $\gamma_i < \varepsilon_i$ . Let  $\delta_i = \varepsilon_i - \gamma_i$ , and  $\delta = \min\{\delta_i : i \le n\}$ . We claim that this  $\delta$  suffices. Let  $g \in \langle f, \bigcup_{i=1}^n A_i, \delta \rangle$ ,  $i \le n$  and  $x \in A_i$ . Then

$$|g(x) - f_i(x)| \le |g(x) - f(x)| + |f(x) - f_i(x)|$$
  
$$< \delta + \gamma_i \le \varepsilon_i,$$

hence  $g \in \langle f_i, A_i, \varepsilon_i \rangle$ .  $\Box$ 

**1.1.2 COROLLARY:** Let X be a space and let  $\mathcal{K}$  be the covering of X consisting of all finite or compact subsets. Then

(a)  $\{\langle f, K, \varepsilon \rangle : f \in C(X), K \in \mathcal{K}, \text{ and } \varepsilon > 0\}$  is a base for C(X), and

(b)  $\{\langle f, K, \varepsilon \rangle : K \in \mathcal{K}, \text{ and } \varepsilon \rangle \}$  is a neighborhoodbase at f for  $f \in C(X)$ .

**1.1.3 EXAMPLE:** Lemma 1.1.1, and corollary 1.1.2, do not hold if  $\mathcal{K}$  consists of all bounded subsets of X.

For example let  $X = \mathbb{R}$ . Consider the identity  $id : \mathbb{R} \to \mathbb{R}$ . Note that [0, 1) is bounded in  $\mathbb{R}$  and that  $0 \in \langle id, [0, 1), 1 \rangle$ . Suppose there are a bounded  $A \subset \mathbb{R}$  and  $\varepsilon > 0$  such that  $\langle 0, A, \varepsilon \rangle \subset \langle id, [0, 1), 1 \rangle$ . Let  $f \in C(\mathbb{R})$  be defined by  $f \equiv -\varepsilon/2$ . Then  $f \in \langle 0, A, \varepsilon \rangle$ , but  $|f(1 - \varepsilon/2) - (1 - \varepsilon/2)| = 1$ , so  $f \notin \langle id, [0, 1), 1 \rangle$ . Contradiction.

For a covering  $\mathcal{K}$  of X, a topology on C(X) can also be generated by the subbase consisting of all sets

§1.1. Topologies on function spaces

$$A(K, U) = \{g \in C(X) : g(K) \subset U\},\$$

where  $K \in \mathcal{K}$  and U is open in **IR**.

**1.1.4 LEMMA:** Let X be a space and let  $\mathcal{K}$  be the covering of all finite (resp. compact) subsets of X. Then

 $\{A(K, U): K \in \mathcal{K} \text{ and } U \text{ open in } \mathbb{R}\}\$ 

is a subbase for  $C_p(X)$  (resp.  $C_0(X)$ ).

**PROOF:** First let  $K \in \mathcal{K}$ , U open in  $\mathbb{R}$  and  $f \in A(K, U)$ . For every  $x \in K$ , there is a neighborhood  $U_x$  of x and  $\varepsilon_x > 0$  such that

$$f(U_x) \subset (f(x) - \frac{\varepsilon_x}{2}, f(x) + \frac{\varepsilon_x}{2}) \subset (f(x) - \varepsilon_x, f(x) + \varepsilon_x) \subset U.$$

Since K is compact, there are  $x_1, \ldots, x_n \in K$  such that  $K \subset \bigcup_{i=1}^n U_{x_i}$ . Let  $\varepsilon = \min\{\varepsilon_{x_i} : i \le n\}$ . We claim that  $\langle f, K, \varepsilon/2 \rangle \subset A(K, U)$ . Indeed let  $g \in \langle f, K, \varepsilon/2 \rangle$ , and let  $x \in K$ . There is  $i \le n$  such that  $x \in U_{x_i}$ . Then obviously  $|g(x) - f(x_i)| < \varepsilon_{x_i}$ . So  $g(x) \in U$ . This implies  $g \in A(K, U)$ .

Second let  $f \in C(X)$ ,  $K \in \mathcal{K}$  and  $\varepsilon > 0$ . For every  $x \in K$  let  $U_x = f^{-1}((f(x) - \varepsilon/3, f(x) + \varepsilon/3))$  and let  $C_x = \overline{U}_x \cap K$ . Then each  $C_x$  is compact. Let  $V_x = (f(x) - \varepsilon/2, f(x) + \varepsilon/2)$ . There are  $x_1, \ldots, x_n \in K$  such that  $K \subset \bigcup_{i=1}^n U_{x_i}$ . We claim that  $f \in \bigcap_{i=1}^n A(C_{x_i}, V_{x_i}) \subset \langle f, K, \varepsilon \rangle$ . Indeed, for every  $i \leq n$ ,  $f(C_{x_i}) \subset \overline{f(U_{x_i})} \subset V_{x_i}$ . For  $g \in \bigcap_{i=1}^n A(C_{x_i}, V_{x_i})$  and  $x \in K$ , there is  $i \leq n$  such that  $x \in C_{x_i}$ . Then obviously  $|f(x) - g(x)| < \varepsilon$ . So  $g \in \langle f, K, \varepsilon \rangle$ .  $\Box$ 

**1.1.5 EXAMPLE:** Lemma 1.1.4 does not hold in case  $\mathcal{K}$  consists of all bounded subsets. As in example 1.1.3 consider  $\langle id, [0, 1), 1 \rangle$ . Let  $K_1, \ldots, K_n$  be bounded subsets of  $\mathbb{R}$  and let  $U_1, \ldots, U_n$  be open subsets of  $\mathbb{R}$  such that  $0 \in \bigcap_{i=1}^n A(K_i, U_i)$ . There is  $\varepsilon > 0$  such that  $(-\varepsilon, \varepsilon) \subset \bigcap_{i=1}^n U_i$ . Again let  $f \equiv -\varepsilon/2$ . Then  $f \in \bigcap_{i=1}^n A(K_i, U_i)$  and as in example 1.1.3,  $f \notin \langle id, [0, 1), 1 \rangle$ .

When dealing with the topology of pointwise convergence or the compact-open topology we will use corollary 1.1.2 and lemma 1.1.4 without explicitly referring to it.

**1.1.6 LEMMA:**  $C_p(X)$  is a dense subspace of  $\mathbb{R}^X$  with the product topology.

**PROOF:** That  $C_p(X)$  is a subspace of  $\mathbb{R}^X$  is easily seen. Let  $f \in \mathbb{R}^X$ ,  $x_1, \ldots, x_n \in X$  and  $\varepsilon > 0$ . We have to show that for

$$U = \{g \in \mathbb{R}^X : |f(x_i) - g(x_i)| < \varepsilon \text{ for every } i \le n\}$$

we have  $U \cap C_p(X) \neq \emptyset$ . For  $i \leq n$ , let  $f_i \in C(X)$  be a Urysohn function such that  $f_i(x_i) = f(x_i)$  and  $f_i(x_j) = 0$  for  $j \leq n$  and  $j \neq i$ . Let  $g = \sum_{i=1}^n f_i$ . Then for every  $i \leq n$ ,  $g(x_i) = f(x_i)$ , so  $g \in U \cap C_p(X)$ .  $\Box$ 

We define  $C_p^*(X)$ ,  $C_0^*(X)$  and  $C_b^*(X)$  similar to  $C_p(X)$ ,  $C_0(X)$  and  $C_b(X)$  using  $C^*(X)$  instead of C(X). All the observation made above for C(X) endowed with one of the defined topologies are also valid for  $C^*(X)$  endowed with this topology.

On  $C^*(X)$ , we define *the topology of uniform convergence* by the metric

 $d(f, g) = \sup \{ |f(x) - g(x)| : x \in X \},\$ 

where  $f, g \in C^*(X)$ . We denote  $C^*(X)$  endowed with this topology by  $C_u^*(X)$ . It is well-known that  $C_u^*(X)$  is a Banach space ([47, Prop. 4.1.2]). It is easily seen that  $C_0^*(X) \leq C_u^*(X)$ . For a compact space X the topology of uniform convergence and the compact-open topology coincide ([24, Th. 4.2.17]).

All results in this section are well-known. The easy examples 1.1.3 and 1.1.5 were constructed by us. For more information about topologies on function spaces we refer to [24], [37] and [47].

#### **§1.2.** Linear functions between function spaces

In this section we present results which are of fundamental importance in this monograph. In particular we present results of Arhangelskii [1] (corollaries 1.2.15 and 1.2.21).

Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be a linear function. For every  $y \in Y$ , the *support of* y in X with respect to  $\phi$  is defined to be the set supp (y) of all  $x \in X$  satisfying the condition that for every neighborhood U of x, there is  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) such that  $f(X \setminus U) \subset \{0\}$  and  $\phi(f)(y) \neq 0$ . Note that it suffices that the condition holds for arbitrarily small neighborhoods of x. For a subset A of Y we denote  $\bigcup \{ \text{supp}(y) : y \in A \}$  by supp A. Whenever  $\phi$  is a linear bijection we can consider the support of a point in Y with respect to  $\phi$  and the support of a point in X with respect to  $\phi^{-1}$ . It will always be clear from the context which support we mean. The following lemma is obvious, and is stated for reference purposes.

**1.2.1 LEMMA:** Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be a linear function. Let  $y \in Y$ . Then

(a)  $x \notin \text{supp}(y)$  if and only if x has a neighborhood U such that for  $f \in C(X)$ (resp.  $f \in C^*(X)$ ) with  $f(X \setminus U) \subset \{0\}$  we have  $\phi(f)(y) = 0$ , and

(b) supp (y) is closed in X.  $\Box$ 

**1.2.2 EXAMPLES:** (1) Let X be a space. Define  $\phi: C(X) \to C(X)$  by  $\phi \equiv 0$ . Obviously  $\phi$  is linear. By lemma 1.2.1 (a), supp  $(x) = \emptyset$ , for every  $x \in X$ .

(2) Let X be a space and let  $\lambda \in \mathbb{R} \setminus \{0\}$ . Define  $\phi: C(X) \to C(X)$  by  $\phi(f) = \lambda f$  for every  $f \in C(X)$ . Obviously  $\phi$  is linear. We claim that for every  $x \in X$ ,  $\sup p(x) = \{x\}$ . First let U be any neighborhood of x. Let  $f \in C(X)$  be a Urysohn function such that f(x) = 1 and  $f(X \setminus U) \subset \{0\}$ . Then  $\phi(f)(x) = \lambda \neq 0$ , hence  $x \in \sup p(x)$ . Second for  $y \neq x$ let U be a neighborhood of y missing x. Then for  $f \in C(X)$  with  $f(X \setminus U) \subset \{0\}$ , we have  $\phi(f)(x) = 0$ . By lemma 1.2.1 (a),  $y \notin \sup p(x)$ .

(3) Let X be a space and let  $x_0 \in X$  be fixed. Define  $\phi: C(X) \to C(X)$  by  $\phi(f) = f + f(x_0)$  for every  $f \in C(X)$ . Obviously  $\phi$  is linear. We claim that for every  $x \in X$ , supp  $(x) = \{x, x_0\}$ . Let U be any neighborhood of x. Let  $V \subset U$  be a neighborhood of x such that if  $x \neq x_0, x_0 \notin V$ . Find a Urysohn function  $f \in C(X)$  such that f(x) = 1 and  $f(X \setminus V) \subset \{0\}$ . Then  $f(X \setminus U) \subset \{0\}$  and  $\phi(f)(x) \neq 0$ . Hence  $x \in \text{supp}(x)$ . In a similar way one can prove that  $x_0 \in \text{supp}(x)$ . As in (2) one can prove that for  $y \notin \{x, x_0\}$ ,  $y \notin \text{supp}(x)$ .

Each linear function above can also be defined from  $C^*(X)$  to  $C^*(Y)$ .

The following definitions are due to Arhangelskii [1]. Let X and Y be spaces. We say that a linear function  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) is *effective* if for every  $f, g \in C(X)$  (resp.  $f, g \in C^*(X)$ ) and  $y \in Y$  such that f and g coincide on a neighborhood of supp  $(y), \phi(f)(y) = \phi(g)(y)$ . The linear function  $\phi$  is of *bounded type* if  $\phi$  is effective and for every  $y \in Y$ , supp (y) is bounded in X.

**1.2.3 LEMMA:** Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be a linear function which is not effective. Then there are  $y \in Y$ , a neighborhood U of supp (y) and  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) such that  $f(U) = \{0\}$  and  $\phi(f)(y) \neq 0$ .

**PROOF:** Since  $\phi$  is not effective, there are  $y \in Y$ , a neighborhood U of supp(y) and  $f_1, f_2 \in C(X)$  (resp.  $f_1, f_2 \in C^*(X)$ ) such that  $f_1$  and  $f_2$  coincide on U and  $\phi(f_1)(y) \neq \phi(f_2)(y)$ . Let  $f = f_1 - f_2$ . Then  $f \in C(X)$  (resp.  $f \in C^*(X)$ ). For  $x \in U$  we have  $f(x) = f_1(x) - f_2(x) = 0$  and by linearity of  $\phi$ ,  $\phi(f)(y) = \phi(f_1)(y) - \phi(f_2)(y) \neq 0$ .

**1.2.4 EXAMPLES:** (1) The linear functions in example 1.2.2 have the property that for every  $x \in X$  and for every  $f, g \in C(X)$  such that f and g coincide on supp (x),  $\phi(f)(x) = \phi(g)(x)$ , hence they are effective.

(2) Let  $X = [1, \omega_1)$  and let  $Y = [1, \omega_1]$ . Since every  $f \in C(X)$  is eventually constant, i.e., there is  $\alpha < \omega_1$  such that for each  $\beta \ge \alpha$ ,  $f(\alpha) = f(\beta)$  [24, example 3.1.27], f has a natural extension  $\tilde{f} \in C(Y)$ . The function  $\phi: C(X) \to C(Y)$  defined by  $\phi(f) = \tilde{f}$  is easily seen to be linear. We claim that  $\phi$  is not effective. It is enough to show that  $\sup (\omega_1) = \emptyset$ , since in this situation any two functions in C(X) coincide on a neighborhood of  $\sup p(\omega_1)$ . Let  $x \in X$ . then U = [1, x] is a neighborhood of x. Let  $f \in C(X)$  be any mapping satisfying  $f(X \setminus U) \subset \{0\}$ . Then  $\phi(f)(\omega_1) = 0$ , hence by lemma 1.2.1 (a),  $x \notin \operatorname{supp}(\omega_1)$ .

Note that in this situation we have  $C(X) = C^*(X)$  and  $C(Y) = C^*(Y)$ .

We will now give some general properties of effective linear functions between function spaces.

**1.2.5 LEMMA:** Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be an effective linear injection. Then supp Y = X.

**PROOF:** Suppose there is  $x \notin \overline{\operatorname{supp} Y}$ . Then there is O open in X such that  $\overline{\operatorname{supp} Y} \subset O$ and  $x \notin \overline{O}$ . Find a Urysohn function  $f \in C^*(X)$  such that f(x) = 1 and  $f(\overline{O}) \subset \{0\}$ . Since O is a neighborhood of supp Y and  $\phi$  is effective we then have  $\phi(f)(Y) \subset \{0\}$ . This implies  $f \neq 0$  and  $\phi(f) = 0$ , contradicting the injectivity of  $\phi$ .  $\Box$ 

**1.2.6 LEMMA:** Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be an effective linear function. Then for  $A \subset Y$  we have supp  $\overline{A} \subset \overline{\text{supp } A}$ .

**PROOF:** Suppose there is  $x \in \operatorname{supp} \overline{A} \setminus \overline{\operatorname{supp} A}$ . Suppose  $x \in \operatorname{supp}(y)$  for  $y \in \overline{A}$ . Find O open in X such that  $x \subset O \subset \overline{O} \subset X \setminus \overline{\operatorname{supp} A}$ . Since  $x \in \operatorname{supp}(y)$  there is  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) with  $f(X \setminus O) \subset \{0\}$  and  $\phi(f)(y) \neq 0$ . Since  $X \setminus \overline{O}$  is a neighborhood of supp A and  $f \equiv 0$  on  $X \setminus \overline{O}$ , by effectiveness of  $\phi$ ,  $\phi(f) \equiv 0$  on A. But this implies  $\phi(f)(y) = 0$ .

Contradiction.

Let X and Y be spaces. A set-valued function  $F: X \to \mathcal{P}(Y) \setminus \{\emptyset\}$  such that for every  $x \in X$ , F(x) is closed in Y is said to be *Lower Semi Continuous* (abbreviated LSC) whenever for every open  $U \subset Y$  the set  $\{x \in X : F(x) \cap U \neq \emptyset\}$  is open in X. Consequently F is LSC if and only if for every closed  $A \subset Y$  the set  $\{x \in X : F(x) \subset A\}$  is closed in X. Furthermore F is said to be *Upper Semi Continuous* (abbreviated USC) whenever for every open  $U \subset Y$  the set  $\{x \in X : F(x) \subset U\}$  is open in X. Consequently F is USC if and only if for every closed  $A \subset Y$  the set  $\{x \in X : F(x) \subset A\}$  is closed in X.

If  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) is a linear function we can consider supp:  $Y \to \mathcal{P}(X)$  as a set-valued function. We have

**1.2.7 LEMMA:** Let X and Y be spaces and let  $\phi: C(X) \to C(Y)$  (resp.  $\phi: C^*(X) \to C^*(Y)$ ) be an effective linear function such that for each  $y \in Y$ , supp  $(y) \neq \emptyset$ . Then supp is LSC.

**PROOF:** By lemma 1.2.1 (b), supp (y) is closed in X for every  $y \in Y$ . Let U be an open subset of X. Put  $O = \{y \in Y : \text{supp}(y) \cap U \neq \emptyset\}$ , and let  $y \in O$ . Then there is  $x \in \text{supp}(y) \cap U$ . Let V be open in X such that  $x \in V \subset \overline{V} \subset U$ . Let  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) be such that  $f(X \setminus V) \subset \{0\}$  and  $\phi(f)(y) \neq 0$ . Let  $W = \{z \in Y : \phi(f)(z) \neq 0\}$ . Then W is an open neighborhood of y. We claim that  $W \subset O$ . Suppose there is  $z \in W \setminus O$ , i.e.,  $\phi(f)(z) \neq 0$  and  $\text{supp}(z) \cap U = \emptyset$ . Then  $X \setminus \overline{V}$  is a neighborhood of supp(z) and  $f(X \setminus \overline{V}) \subset \{0\}$ , so  $\phi(f)(z) = 0$ . Contradiction. So  $W \subset O$  and hence the lemma is proved.  $\Box$ 

**REMARK:** If the function  $\phi$  in lemma 1.2.7 is surjective, then surely  $\sup (y) \neq \emptyset$  for every  $y \in Y$ . Indeed, if  $\sup (y) = \emptyset$  for some  $y \in Y$ , then let  $f \in C^*(Y)$  be such that  $f(y) \neq 0$ . Choose  $g \in C(X)$  (resp.  $g \in C^*(X)$ ), such that  $\phi(g) = f$ . We have that  $g \equiv 0$  on a neighborhood of  $\sup p(y)$ , so by effectiveness of  $\phi$ , f(y) = 0. This gives a contradiction. We conclude that for any effective linear surjection, supp is LSC.

In section 2.4 we will give an example of an effective linear surjection  $\phi: C(X) \rightarrow C(Y)$ , such that supp is not USC.

**1.2.8 PROPOSITION** ([1]): Let X and Y be spaces and  $\phi$ :  $C(X) \rightarrow C(Y)$  a linear function of bounded type. Let A be a bounded subset of Y. Then supp A is bounded in X.

**PROOF:** Suppose to the contrary that supp A is not bounded in X. Then there is  $f \in C(X)$  such that  $f(\operatorname{supp} A)$  is unbounded in  $\mathbb{R}$ , so there is  $\{x_n : n \in \mathbb{N}\} \subset \operatorname{supp} A$  such that  $\{f(x_n): n \in \mathbb{N}\}$  is closed and discrete in  $\mathbb{R}$ . Hence we can find an open family  $\{V_n : n \in \mathbb{N}\}$  in X such that  $x_n \in V_n$  for each  $n \in \mathbb{N}$  and  $\{f(V_n): n \in \mathbb{N}\}$  is a discrete family in  $\mathbb{R}$ . Then obviously  $\{V_n : n \in \mathbb{N}\}$  is a discrete family in X.

By induction we construct a subset  $\{y_k : k \in \mathbb{N}\}$  of A, a subfamily  $\{U_k : k \in \mathbb{N}\}$  of  $\{V_n : n \in \mathbb{N}\}$ , and a subset  $\{f_k : k \in \mathbb{N}\}$  of C(X) such that

- (1)  $f_k(X \setminus U_k) \subset \{0\}$  for every  $k \in \mathbb{N}$ ,
- (2) for  $i \neq j$  we have  $U_i \neq U_j$ ,
- (3) supp  $\{y_1, \ldots, y_{k-1}\} \cap \overline{U}_k = \emptyset$  for every k > 1, and
- (4)  $\phi(f_k)(y_k) = k + |h_k|$  for every  $k \in \mathbb{N}$ , where  $h_k = \sum_{k < k} \phi(f_k)(y_k)$  for k > 1 and  $h_1 = 0$ .

Let  $y_1 \in A$  be such that  $x_1 \in \text{supp}(y_1)$ . Let  $U_1 = V_1$ . Since  $U_1$  is a neighborhood of  $x_1$ , and  $x_1 \in \text{supp}(y_1)$ , there is  $h \in C(X)$  such that  $h(X \setminus U_1) \subset \{0\}$  and  $\phi(h)(y_1) \neq 0$ . Let

 $\lambda = \frac{1}{\phi(h)(y_1)}.$ 

Let  $f_1 = \lambda h$ . Then  $f_1 \in C(X)$  and  $f_1(X \setminus U_1) \subset \{0\}$ . Furthermore by linearity of  $\phi$ ,  $\phi(f_1)(y_1) = \lambda \phi(h)(y_1) = 1$ .

Let k > 1 and suppose we found  $y_1, \ldots, y_{k-1}, U_1, \ldots, U_{k-1}$  and  $f_1, \ldots, f_{k-1}$ . Let  $P_k = \sup\{y_1, \ldots, y_{k-1}\}$ . Since  $\phi$  is of bounded type,  $P_k$  is bounded in X. So there is  $n \in \mathbb{N}$  such that  $\overline{f(V_n)} \cap f(P_k) = \emptyset$  and hence  $\overline{V_n} \cap P_k = \emptyset$ . Since  $x_n \in \sup A$ , there is  $y_k \in A$  such that  $x_n \in \sup(y_k)$ . Let  $U_k = V_n$ . Since  $U_k$  is a neighborhood of  $x_n$  and  $x_n \in \sup(y_k)$ , there is  $h \in C(X)$  with  $h(X \setminus U_k) \subset \{0\}$  and  $\phi(h)(y_k) \neq 0$ . Let

$$\lambda = \frac{k + |h_k|}{\phi(h)(y_k)}$$

and  $f_k = \lambda h$ . Then  $f_k(X \setminus U_k) \subset \{0\}$  and by linearity of  $\phi$ ,  $\phi(f_k)(y_k) = k + |h_k|$ . To complete the inductive construction we observe that by (3) and the fact that  $x_n \in \text{supp}(y_k) \cap U_k$ ,  $U_i \neq U_j$  for  $i \neq j$ .

Since  $\{U_k : k \in \mathbb{N}\}$  is a subfamily of  $\{V_n : n \in \mathbb{N}\}$  we have by (2) that  $\{U_k : k \in \mathbb{N}\}$  is a discrete open family in X. Let  $f = \sum_{i=1}^{\infty} f_i$ . For  $x \in X$  we have a neighborhood  $U_x$  which intersects at most one member of  $\{U_k : k \in \mathbb{N}\}$ . Then by (1)  $f \mid U_x$  is a finite sum, hence  $f \in C(X)$ . For every  $k \in \mathbb{N}$ , let  $g_k = \sum_{i=1}^{k} f_i$  and

$$W_k = X \setminus \bigcup_{j > k} \overline{f_j^{-1}(\mathbb{R} \setminus \{0\})}$$

By (1) we have for every  $j \in \mathbb{N}$ ,  $f_j^{-1}(\mathbb{R} \setminus \{0\}) \subset U_j$ . So  $\{f_j^{-1}(\mathbb{R} \setminus \{0\}) : j \in \mathbb{N}\}$  is a discrete family, hence  $W_k$  is open in X. By (3), we have for j > k,  $\supp(y_k) \cap \overline{U_j} = \emptyset$ . This implies  $\supp(y_k) \cap \overline{f_j^{-1}(\mathbb{R} \setminus \{0\})} = \emptyset$ . We conclude that  $\supp(y_k) \subset W_k$ , so  $W_k$  is a neighborhood of  $\supp(y_k)$ . For j > k, and  $x \in W_k$ ,  $f_j(x) = 0$ , hence f and  $g_k$  coincide on  $W_k$ . Since  $\phi$  is effective, we then have  $\phi(f)(y_k) = \phi(g_k)(y_k)$ . But  $\phi(g_k)(y_k) = h_k + \phi(f_k)(y_k)$ , so that by (4),

 $|\phi(g_k)(y_k)| \ge \phi(f_k)(y_k) - |h_k| = k + |h_k| - |h_k| = k.$ 

We conclude that  $|\phi(f)(y_k)| \ge k$  for every  $k \in \mathbb{N}$ . But this implies that A is not bounded in Y. Contradiction.  $\Box$ 

**1.2.9 COROLLARY:** Let X and Y be spaces and let  $\phi$ :  $C(X) \rightarrow C(Y)$  be a linear injection of bounded type. If Y is pseudocompact, then X is pseudocompact.

**PROOF:** If Y is pseudocompact it is bounded, hence by proposition 1.2.8,  $\overline{\text{supp } Y}$  is bounded. By lemma 1.2.5,  $\overline{\text{supp } Y} = X$ . This implies that X is pseudocompact.

We will now give two other applications of proposition 1.2.8. The first one will be a very important tool in chapter 4. The second one will be used in section 1.5.

**1.2.10 LEMMA:** Let X and Y be normal spaces. Let K be compact and non-empty in Y and suppose  $\{V_n : n \in \mathbb{N}\}$  is a decreasing base at K in Y. Let  $\{A_s : s \in S\}$  be a locally finite family in X. Furthermore let  $\phi: C(X) \to C(Y)$  be a linear function of bounded type. Then there are  $m \in \mathbb{N}$  and  $s_1, \ldots, s_m \in S$  such that  $(\operatorname{supp} V_m) \cap \bigcup_{s \notin \{s_1, \ldots, s_m\}} A_s = \emptyset$ .

**PROOF:** If S is finite the lemma is obvious. Suppose the lemma is false for infinite S. Then there are distinct  $s_i \in S$  ( $i \in \mathbb{N}$ ) and points  $x_i \in \text{supp } V_i \cap A_{s_i}$ . Suppose  $x_i \in \text{supp } y_i$  with  $y_i \in V_i$ . Since  $\{A_{s_i} : i \in \mathbb{N}\}$  is locally finite,  $\{x_i : i \in \mathbb{N}\}$  is infinite. Let  $L = \{y_i : i \in \mathbb{N}\} \cup K$  and let  $\mathcal{U}$  be an open cover of L. Then there are  $U_1, \ldots, U_n$  in  $\mathcal{U}$  such that  $K \subset \bigcup_{i=1}^n U_i$ . Since  $\{V_n : n \in \mathbb{N}\}$  is a base at K in Y, there is  $m \in \mathbb{N}$  such that  $V_m \subset \bigcup_{i=1}^n U_i$ . So  $K \cup \{y_i : i \ge m\} \subset \bigcup_{i=1}^n U_i$ . We conclude that L is compact.

By proposition 1.2.8, supp L is bounded. It follows that  $\{x_i : i \in \mathbb{N}\}$  is also bounded. However since  $\{A_{s_i} : i \in \mathbb{N}\}$  is locally finite,  $\{x_i : i \in \mathbb{N}\}$  is a closed and discrete set. Contradiction.  $\Box$  Let X and Y be metric spaces and  $\phi: C(X) \to C(Y)$  a linear surjection of bounded type. For  $U \subset X$ , let  $T_U = \{y \in Y : \operatorname{supp}(y) \cap U \neq \emptyset\}$ . For a family  $\mathcal{U}$  of subsets of X, let  $T_{\mathcal{U}} = \{T_U : U \in \mathcal{U}\}.$ 

**1.2.11 LEMMA:** If  $\mathcal{U}$  is a locally finite open cover of X, then  $T_{\mathcal{U}}$  is a locally finite open cover of Y.

**PROOF:** By the remark following lemma 1.2.7, we have for each  $y \in Y$ ,  $\supp(y) \neq \emptyset$ . So  $T_{\mathcal{U}}$  covers Y. Furthermore by lemma 1.2.7,  $\supp$  is LSC, so for  $U \in \mathcal{U}$ ,  $T_U$  is open in Y. If  $T_{\mathcal{U}}$  is not locally finite there are  $y \in Y$ , a sequence  $y_n \to y$   $(n \to \infty)$ , and distinct  $U_n$ 's in  $\mathcal{U}$  such that  $y_n \in T_{U_n}$ . Let  $x_n \in \supp(y_n) \cap U_n$ . Since  $\{y_n : n \in \mathbb{N}\}$  is bounded, by proposition 1.2.8,  $\supp\{y_n : n \in \mathbb{N}\}$  is bounded. Hence  $\{x_n : n \in \mathbb{N}\}$  is bounded. Since X is metric we then obtain that  $\overline{\{x_n : n \in \mathbb{N}\}}$  is compact. Since  $\mathcal{U}$  is locally finite,  $\overline{\{x_n : n \in \mathbb{N}\}}$  intersects only finitely many elements of  $\mathcal{U}$ . Contradiction. This proves the lemma.  $\square$ 

Proposition 1.2.8, corollary 1.2.9, lemmas 1.2.10 and 1.2.11 are the first results in this section which are only formulated for linear functions from C(X) to C(Y) and which are not formulated for linear functions from  $C^*(X)$  to  $C^*(Y)$ . In the following example we show that proposition 1.2.8, lemmas 1.2.10 and 1.2.11 are not true for linear functions between function spaces  $C^*(X)$ .

**1.2.12 EXAMPLE:** Let  $(x_n)_{n \in \mathbb{N}}$  be a convergent sequence, say  $x_n \to x$   $(n \to \infty)$ . Let  $Y = \{x_n : n \in \mathbb{N}\} \cup \{x\}$ , and let  $X = Y \oplus \mathbb{N}$ . Define  $\phi: C^*(X) \to C^*(Y)$  by

$$\phi(f)(z) = \begin{cases} f(x_n) + \frac{1}{n}f(n) & \text{if } z = x_n \text{ for some } n \in \mathbb{N}, \\ f(x) & \text{if } z = x. \end{cases}$$

We first show that  $\phi$  is well-defined. It suffices to show that  $\phi(f)$  is continuous at x. Let  $\varepsilon > 0$ . Since f is bounded there is  $c \in \mathbb{R}$  such that  $f(X) \subset (-c, c)$ . Find  $m \in \mathbb{N}$  such that for  $n \ge m$ ,  $|f(x_n) - f(x)| < \varepsilon/2$  and  $1/m < \varepsilon/(2c)$ . Then for  $n \ge m$  we have

$$\begin{aligned} |\phi(f)(x_n) - \phi(f)(x)| &= |f(x_n) + \frac{1}{n}f(n) - f(x)| \\ &\leq |f(x_n) - f(x)| + \frac{1}{n}|f(n)| \end{aligned}$$

\$1.2. Linear functions between function spaces

$$< \frac{\varepsilon}{2} + \frac{1}{n}c < \varepsilon.$$

So  $\phi(f)$  is continuous at x. Note that since Y is compact,  $\phi(f) \in C^*(Y)$ . Obviously  $\phi$  is linear. As in example 1.2.2 we can show that for each  $n \in \mathbb{N}$ ,  $\operatorname{supp}(x_n) = \{n, x_n\}$  and  $\operatorname{supp}(x) = \{x\}$ . Furthermore for any two functions  $f, g \in C^*(X)$  which coincide on  $\operatorname{supp}(z)$  for some  $z \in Y$ , we have  $\phi(f)(z) = \phi(g)(z)$ . Hence  $\phi$  is a linear function of bounded type.

Since Y is compact, Y is bounded. However supp Y = X is not bounded. This implies that proposition 1.2.8 is not true when dealing with bounded functions. Note also that if  $\mathcal{U} = \{Y\} \cup \{\{n\} : n \in \mathbb{N}\}$ , then  $\mathcal{U}$  is a locally finite open cover of X. However  $T_{\mathcal{U}}$  is not locally finite, hence lemma 1.2.11 is also not valid when dealing with bounded functions. Similarly lemma 1.2.10 does not hold. The question remains whether corollary 1.2.9 holds. The above example does not give a counterexample since  $\phi$  is not injective.

We will now search for linear functions of bounded type.

**1.2.13 LEMMA:** Let X and Y be spaces. Suppose  $\phi: C_0(X) \to C_p(Y)$  (resp.  $\phi: C_0^*(X) \to C_p^*(Y)$ ) is a continuous linear function. Then for every  $y \in Y$ , supp (y) is compact.

**PROOF:** Since  $\phi$  is continuous at 0, there are a compact  $B \subset X$  and  $\varepsilon > 0$  such that  $\phi(<0, B, \varepsilon >) \subset <0, \{y\}, 1>$ . Suppose there is  $x \in \text{supp}(y) \setminus B$ . In this situation  $X \setminus B$  is a neighborhood of x, so there is  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) satisfying  $f(B) \subset \{0\}$  and  $\phi(f)(y) \neq 0$ . By linearity of  $\phi$  we may assume  $\phi(f)(y) > 1$ . Obviously  $f \in <0, B, \varepsilon >$ , hence  $\phi(f) \in <0, \{y\}, 1>$ . This implies  $\phi(f)(y) < 1$ . Contradiction. We conclude that supp  $(y) \subset B$ . By lemma 1.2.1 (b) supp (y) is closed and hence compact.  $\Box$ 

**1.2.14 PROPOSITION** ([1]): Let X and Y be spaces. Suppose  $\phi: C_0(X) \to C_p(Y)$ (resp.  $\phi: C_0^*(X) \to C_p^*(Y)$ ) is a continuous linear function. Then  $\phi$  is of bounded type.

**PROOF:** By lemma 1.2.13, supp (y) is bounded for every  $y \in Y$ , so it remains to prove that  $\phi$  is effective. If  $\phi$  is not effective, then by lemma 1.2.3 there are  $y \in Y$ , a neighborhood U of supp (y) and  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) with  $f(U) = \{0\}$  and  $\phi(f)(y) \neq 0$ . Let  $\delta = |\phi(f)(y)|$ . Since  $\phi$  is continuous, there are a compact subset A of X and  $\varepsilon > 0$  such that  $\phi(\langle f, A, \varepsilon \rangle) \subset \langle \phi(f), \{y\}, \delta \rangle$ . Then for every  $g \in C(X)$  (resp.  $g \in C^*(X)$ ) which coincides with f on A,  $\phi(g)(y) \neq 0$ .

Let  $B = A \setminus U$ . If  $B = \emptyset$ ,  $A \subset U$ . Since  $f(U) = \{0\}$ , 0 coincides with f on A. This gives

 $\phi(0)(y) \neq 0$ . Contradiction, so B is a non-empty compactum and  $B \cap \text{supp}(y) = \emptyset$ . Then by lemma 1.2.1 (a), there are open sets  $U_1, \ldots, U_n, V_1, \ldots, V_n$  such that

(1)  $U_i \subset \overline{U}_i \subset V_i$  for every  $i \le n$ (2)  $B \subset \bigcup_{i=1}^n U_i$ , and (3) if  $g \in C(X)$  (resp.  $g \in C^*(X)$ ) with  $g(X \setminus V_i) = \{0\}$  for some  $i \le n$ , then  $\phi(g)(y) = 0$ .

Since B is compact, there are  $\alpha_1, \ldots, \alpha_n \in C^*(X)$  such that for every  $i \le n$ ,  $\alpha_i(\overline{U}_i \cap B) = 1$  and  $\alpha_i(X \setminus V_i) \subset \{0\}$  [24, Th. 3.1.7]. Let  $\alpha = \max(\sum_{i=1}^n \alpha_i, 1)$  and  $h_i = \alpha_i/\alpha$ . Then we have

(4) 
$$h_i(X \setminus V_i) = \{0\}$$
, and  
(5)  $\sum_{i=1}^{n} h_i(x) = 1$  for every  $x \in B$ 

Let  $h_i^* = h_i \cdot f$  and  $h^* = \sum_{i=1}^n h_i^*$ . By (4) we have  $h_i^*(X \setminus V_i) = \{0\}$  so that by (3),  $\phi(h_i^*)(y) = 0$ . This means  $\phi(h^*)(y) = 0$ .

By (5), for every  $x \in B$  we have  $h^*(x) = f(x)$ . Furthermore for every  $x \in U$  we have  $h^*(x) = 0 = f(x)$ , so  $h^*$  and f coincide on A. But then  $\phi(h^*)(y) \neq 0$ . Contradiction. We conclude that  $\phi$  is effective.  $\Box$ 

**1.2.15 COROLLARY** ([1]): Let X and Y be spaces. Suppose  $\phi: C_p(X) \to C_p(Y)$  or  $\phi: C_0(X) \to C_0(Y)$  is a continuous linear function. Then

- (a)  $\phi$  is of bounded type, and
- (b) if A is bounded in Y, then supp A is bounded in X.

If moreover in X every closed and bounded subset is compact, then supp A is compact.

**PROOF:** Since  $C_p(X) \le C_0(X)$ , any linear mapping  $\phi: C_p(X) \to C_p(Y)$  or  $\phi: C_0(X) \to C_0(Y)$  is also continuous considered as a function from  $C_0(X)$  to  $C_p(Y)$ . Now apply propositions 1.2.8 and 1.2.14.  $\square$ 

By example 1.2.12 we have for bounded functions only the following corollary the proof of which is similar to the one of corollary 1.2.15 (a).

**1.2.16 COROLLARY:** Let X and Y be spaces. Suppose  $\phi: C_p^*(X) \to C_p^*(Y)$  or  $\phi: C_0^*(X) \to C_0^*(Y)$  is a continuous linear function. Then  $\phi$  is of bounded type.  $\Box$ 

**1.2.17 REMARK:** The question remains whether a result such as corollary 1.2.15 holds for continuous linear functions between the function spaces  $C_b(X)$ . In example 1.2.4 (2) we found spaces X and Y and a linear function  $\phi: C(X) \rightarrow C(Y)$  which is not effective. From corollary 1.2.15 we have that  $\phi$  considered as a function from  $C_p(X)$  to  $C_p(Y)$  (resp. from  $C_0(X)$  to  $C_0(Y)$ ) is not continuous (this can also be verified directly). Unfortunately  $\phi$  considered as a function from  $C_b(X)$  to  $C_b(Y)$  is also not continuous.

Corollary 1.2.15 will be one of the main tools in this monograph. Another important tool will be corollary 1.2.21. Before we can prove this corollary we need some other lemmas.

**1.2.18 LEMMA** ([1]): Let X and Y be spaces, and suppose  $\phi: C_u^*(X) \to C_p(Y)$  is a continuous linear function. Then  $\phi$  considered as a function from  $C_u^*(X)$  to  $C_0(Y)$  is also continuous.

**PROOF:** By linearity of  $\phi$  and since  $C_0(Y)$  is a topological vector space it suffices to prove continuity at 0, i.e., we have to prove for a compact  $A \subset Y$  and  $\varepsilon > 0$  that  $\phi^{-1}(<0, A, \varepsilon)$  is a neighborhood of 0 in  $C_u^*(X)$ . We will show that for

$$V = \{g \in C^*(X) : |\phi(g)(x)| \le \varepsilon/2 \text{ for every } x \in A\}$$

we have  $0 \in Int V$ .

Since  $\phi: C_u^*(X) \to C_p(Y)$  is continuous,  $\{g \in C^*(X) : |\phi(g)(x)| \le \varepsilon/2\}$  is closed in  $C_u^*(X)$  for every  $x \in X$ . This means that

$$V = \bigcap_{x \in A} \{ g \in C^*(X) : |\phi(g)(x)| \le \varepsilon/2 \}$$

is closed in  $C_{\mu}^{*}(X)$ .

CLAIM:  $C^*(X) = \bigcup_{n \in \mathbb{N}} n \cdot V.$ 

Let  $h \in C^*(X)$ . Since A is compact, there is  $n_0 \in \mathbb{N}$  such that  $\phi(h)(A) \subset [-n_0, n_0]$ . Find  $n_1 \in \mathbb{N}$  such that  $n_1 \ge 2n_0/\epsilon$ . Then for every  $x \in A$  we have

$$|\phi(\frac{1}{n_1}h)(x)| = \frac{1}{n_1} |\phi(h(x))| \le \frac{n_0}{n_1} \le \frac{\varepsilon}{2}.$$

This means  $h \in n_1 \cdot V$  and hence the claim is proved.

Since for every  $n \in \mathbb{N}$ ,  $n \cdot V$  is closed in  $C_u^*(X)$  and  $C_u^*(X)$  is a Banach space, there is  $n \in \mathbb{N}$  such that  $\operatorname{Int}(n \cdot V) \neq \emptyset$ . This means  $\operatorname{Int} V \neq \emptyset$ . Take an arbitrary  $g \in \operatorname{Int} V$ . Since

 $\theta: C_u^*(X) \to C_u^*(X)$  defined by  $\theta(h) = -h$  is a homeomorphism such that  $\theta(V) = V$ , we have  $-g \in \text{Int } V$ . Since  $\psi: C_u^*(X) \to C_u^*(X)$  defined by  $\psi(h) = (g+h)/2$  is a homeomorphism such that  $\psi(\text{Int } V) \subset \text{Int } V$  we have  $\psi(-g) = 0 \in \text{Int } V$ .  $\Box$ 

**1.2.19 PROPOSITION** ([1]): Let X and Y be spaces such that in X every closed and bounded subset is compact. Let  $\phi: C_0(X) \to C_p(Y)$  be a continuous linear function. Then  $\phi$  considered as a function from  $C_0(X)$  to  $C_0(Y)$  is also continuous.

**PROOF:** Let  $\phi^*$  be the restriction of  $\phi$  to the set of bounded functions. Since  $C_0^*(X) \leq C_u^*(X)$ ,  $\phi^*$  considered as a function from  $C_u^*(X)$  to  $C_p(Y)$  is continuous. Then by lemma 1.2.18,  $\phi^*$  considered as a map from  $C_u^*(X)$  to  $C_0(Y)$  is continuous. To prove continuity of  $\phi$  considered as a map from  $C_0(X)$  to  $C_0(Y)$  it is by linearity of  $\phi$  enough to prove continuity at 0. To this end let A be a compact subset of Y and let  $\varepsilon > 0$ . By propositions 1.2.8 and 1.2.14 and the assumption on the space X,  $B = \overline{\text{supp} A}$  is a compact subset of X. By the above there is  $\delta > 0$  such that for every  $f \in C(X)$  with  $|f(x)| < \delta$  for every  $x \in X$  we have  $\phi(f) \in <0, A, \varepsilon >$ . We claim that  $\phi(<0, B, \delta/2>) \subset <0, A, \varepsilon >$ . To this end let  $g \in C(X)$  with  $|g(x)| < \delta/2$  for every  $x \in B$ .

Define  $g_1: X \to \mathbb{R}$  by

$$g_{1}(x) = \begin{cases} g(x) & \text{if } |g(x)| < \frac{\delta}{2} \\ \frac{\delta}{2} & \text{if } g(x) \ge \frac{\delta}{2} \\ -\frac{\delta}{2} & \text{if } g(x) \le -\frac{\delta}{2}. \end{cases}$$

Then  $g_1 \in C(X)$  and  $g_1$  coincides on a neighborhood of *B* with *g*. By proposition 1.2.14,  $\phi$  is effective, and hence  $\phi(g) = \phi(g_1)$  on *A*. Furthermore for every  $x \in X$ ,  $|g_1(x)| < \delta$ , so  $\phi(g_1) \in <0, A, \varepsilon >$ . This means  $\phi(g) \in <0, A, \varepsilon >$ . This proves that  $\phi$  is continuous at 0.

As with proposition 1.2.8 we have that proposition 1.2.19 does not hold for linear functions between function spaces  $C^*(X)$ . We have the following

**1.2.20 EXAMPLE:** Consider X, Y and  $\phi: C^*(X) \to C^*(Y)$  as in example 1.2.12. We claim that  $\phi$  considered as a function from  $C^*_p(X)$  to  $C^*_p(Y)$  is continuous and considered as a function from  $C^*_0(X)$  to  $C^*_0(Y)$  is not continuous.

First let  $P \subset Y$  be finite,  $\varepsilon > 0$  and put  $Q = \operatorname{supp} P$ . Then

$$\phi(\langle 0, Q, \frac{\varepsilon}{2} \rangle) \subset \langle 0, P, \varepsilon \rangle$$

Second consider the open subset <0, Y, 1> of  $C_0^*(Y)$ . If  $\phi$  considered as a function from  $C_0^*(X)$  to  $C_0^*(Y)$  is continuous, there is a compact subset A of X and  $\varepsilon > 0$  such that

 $\phi(\langle 0, A, \varepsilon \rangle) \subset \langle 0, Y, 1 \rangle$ .

Find  $n \in X \setminus A$ . Let  $f = n\chi_n$ , where  $\chi_n$  denotes the characteristic function of the set  $\{n\}$ . Then  $f \in \langle 0, A, \varepsilon \rangle$  and  $\phi(f)(x_n) = 1$ . But this implies that  $\phi(f) \notin \langle 0, Y, 1 \rangle$ .

From proposition 1.2.19 we have the following important

**1.2.21 COROLLARY** ([1]): Let X and Y be spaces in which every closed and bounded subset is compact, and suppose  $\phi: C_p(X) \to C_p(Y)$  is a linear homeomorphism. Then  $\phi$  considered as a function from  $C_0(X)$  to  $C_0(Y)$  is also a linear homeomorphism.  $\Box$ 

The converse implication in corollary 1.2.21 is not true. By Miljutin's theorem (for any two uncountable metrizable compact spaces we have that  $C_0(X)$  and  $C_0(Y)$  are linearly homeomorphic, [47, Th. 21.5.10]),  $C_0(I)$  and  $C_0(I^2)$  are linearly homeomorphic (here I denotes the unit interval). However  $C_p(I)$  and  $C_p(I^2)$  are not linearly homeomorphic, since Pestov proved in [44] that whenever  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic then dim  $X = \dim Y$ .

**REMARK:** Note that in a compact or metric space the closed and bounded subsets are exactly the compacta. It is not clear to us how this property is related to other topological properties.

For linear mappings between function spaces  $C^*(X)$  we can derive a result in the spirit of proposition 1.2.19. This result (corollary 1.2.23) is a consequence of

**1.2.22 THE CLOSED GRAPH THEOREM:** Let E and F be Banach spaces and let  $\phi: E \to F$  be a linear function such that the set  $\{(x, \phi(x)): x \in E\}$  is closed in  $E \times F$ . Then  $\phi$  is continuous.

For a proof of The Closed Graph Theorem we refer to [26].

**1.2.23 COROLLARY:** Let X and Y be spaces and let  $\phi: C_p^*(X) \to C_p^*(Y)$  be a continuous linear function. Then  $\phi$  considered as a function from  $C_u^*(X)$  to  $C_u^*(Y)$  is also continuous.

**PROOF:** Since  $C_p^*(X) \le C_u^*(X)$  and  $C_p^*(Y) \le C_u^*(Y)$ , it follows directly from The Closed Graph Theorem.  $\Box$ 

In corollary 1.2.23 we can replace the topology of pointwise convergence by other topologies. However as stated above it is the only corollary we need in the sequel of this monograph.

**REMARK:** The results in this section due to Arhangelskii are not formulated in the most general form as they are in [1]. We adjusted these results and their proofs to the form in which we need them in this monograph. The original proof of lemma 1.2.18 used notions like absorbing, convex, circled and balanced spaces (for definitions see [45]). For us these notions are of no importance. Arhangelskii did not define supports for linear functions between function spaces  $C^*(X)$ . We do not know whether all other results in this section were already known to Arhangelskii.

## **§1.3.** The dual of $C_p(X)$ and $C_p^*(X)$

For a space X let L(X) be the dual of  $C_p(X)$ , i.e., the set of all continuous linear functionals on  $C_p(X)$ . For  $x \in X$  we define  $\xi_x : C_p(X) \to \mathbb{R}$  the evaluation mapping at x by  $\xi_x(f) = f(x)$ .

**1.3.1 LEMMA:** For every  $x \in X$ ,  $\xi_x \in L(X)$ .

**PROOF:** It is easily seen that  $\xi_x$  is linear. To prove that  $\xi_x$  is continuous let  $U \subset \mathbb{R}$  be open and let  $f \in \xi_x^{-1}(U)$ . Then  $\xi_x(f) = f(x) \in U$ . Find  $\varepsilon > 0$  such that  $(f(x) - \varepsilon, f(x) + \varepsilon) \subset U$ . We claim that  $\langle f, \{x\}, \varepsilon \rangle \subset \xi_x^{-1}(U)$ . Indeed for  $g \in \langle f, \{x\}, \varepsilon \rangle, |g(x) - f(x)| < \varepsilon$ , so that  $g(x) = \xi_x(g) \in U$ .  $\Box$ 

By identifying x and  $\xi_x$  we regard X as a subset of L(X) (notice that for  $x \neq y$ ,

 $\xi_x \neq \xi_y$ ). As L(X) is a vector space we are interested in a Hamel basis for L(X) (i.e., a maximal independent subset). It turns out that X is a Hamel basis for L(X), i.e.,

- (HB1) X is an independent subset of L(X) and
- (HB2) for every  $F \in L(X)$  there are  $x_1, \ldots, x_n \in X$  and  $\lambda_1, \ldots, \lambda_n \in \mathbb{R}$ such that  $F = \sum_{i=1_n} \lambda_i x_i$ .

To verify (HB1) suppose  $\sum_{i=1}^{n} \lambda_i x_i = 0$  for  $x_1, \ldots, x_n \in X$  and  $\lambda_1, \ldots, \lambda_n \in \mathbb{R}$ . Then for every  $f \in C(X)$ ,  $\sum_{i=1}^{n} \lambda_i f(x_i) = 0$ . For every  $i \le n$  let  $f_i$  be a Urysohn function such that  $f_i(x_i) = 1$  and  $f_i(x_j) = 0$  for  $i \ne j$ . So  $0 = \sum_{k=1}^{n} \lambda_k f_i(x_i) = \lambda_i$ , which proves (HB1). For (HB2) we have to do some more work.

**1.3.2 LEMMA** ([45, p. 124]): Let V be a vector space and  $\alpha, \alpha_1, \ldots, \alpha_n$  linear functionals on V. Then the following statements are equivalent:

- (1)  $\bigcap_{i=1}^{n} \operatorname{Ker} \alpha_{i} \subset \operatorname{Ker} \alpha$ .
- (2)  $\alpha \in \text{span} \{\alpha_1, \ldots, \alpha_n\}.$

**PROOF:** The implication (2) $\Rightarrow$ (1), is a triviality. We prove the implication (1) $\Rightarrow$ (2) by induction on *n*. First suppose n = 1. If  $\alpha_1 \equiv 0$ ,  $\alpha \equiv 0$  so we certainly have  $\alpha \in \text{span} \{\alpha_1\}$ . So suppose there is  $x_0 \in V$  such that  $\alpha_1(x_0) \neq 0$ . Let  $\lambda_1 = \alpha(x_0)/\alpha_1(x_0)$ . We claim that  $\alpha = \lambda_1 \alpha_1$ . To prove this, let  $x \in V$ . If  $\alpha_1(x) = 0$  we are done, so suppose  $\alpha_1(x) \neq 0$ . Then

$$\alpha_1(x - \frac{\alpha_1(x)}{\alpha_1(x_0)}x_0) = \alpha_1(x) - \frac{\alpha_1(x)}{\alpha_1(x_0)}\alpha_1(x_0) = 0,$$

so

$$x - \frac{\alpha_1(x)}{\alpha_1(x_0)} x_0 \in \operatorname{Ker} \alpha_1 \subset \operatorname{Ker} \alpha.$$

This gives

$$0 = \alpha(x - \frac{\alpha_1(x)}{\alpha_1(x_0)}x_0) = \alpha(x) - \frac{\alpha_1(x)}{\alpha_1(x_0)}\alpha(x_0) = \alpha(x) - \lambda\alpha_1(x).$$

So  $\alpha(x) = \lambda_1 \alpha_1(x)$ . This finishes the case n = 1.

Suppose we proved the implication for every n < m with m > 1.

Case 1: there is  $j \le m$  with  $\bigcap_{i \ne j} \operatorname{Ker} \alpha_i \subset \operatorname{Ker} \alpha$ .

Then by the inductive hypothesis,  $\alpha \in \text{span} \{\alpha_1, \ldots, \alpha_m\}$ .

*Case 2:* for every  $j \leq m$ ,  $\bigcap_{i \neq j} \text{Ker } \alpha_i \not\subset \text{Ker } \alpha$ .

Then for every  $j \le m$  there is  $x_j \in V$  such that  $\alpha_i(x_j) = 0$  for  $i \ne j$  and  $\alpha_j(x_j) \ne 0$ . Let  $\lambda_j = \alpha(x_j)/\alpha_j(x_j)$ . We claim that  $\alpha = \sum_{i=1}^n \lambda_i \alpha_i$ . To prove this let  $x \in V$ . For  $j \le m$  we have

$$\alpha_j(x-\sum_{i=1}^m \frac{\alpha_i(x)}{\alpha_i(x_i)}x_i)=\alpha_j(x)-\frac{\alpha_j(x)}{\alpha_j(x_j)}\alpha_j(x_j)=0.$$

This gives

$$x - \sum_{i=1}^{m} \frac{\alpha_i(x)}{\alpha_i(x_i)} x_i \in \bigcap_{i=1}^{m} \operatorname{Ker} \alpha_i \subset \operatorname{Ker} \alpha.$$

Hence

$$\alpha(x-\sum_{i=1}^m\frac{\alpha_i(x)}{\alpha_i(x_i)}x_i)=\alpha(x)-\sum_{i=1}^m\lambda_i\alpha_i(x)=0.$$

This finishes the proof of this lemma.

#### **1.3.3 THEOREM:** X is a Hamel basis for L(X).

**PROOF:** As mentioned above it is enough to prove condition (HB2). So let  $F: C_p(X) \to \mathbb{R}$  be a continuous linear functional. There is a finite subset P of X and  $\delta > 0$  such that  $F(<0, P, \delta >) \subset (-1, 1)$ . Suppose  $P = \{x_1, \ldots, x_n\}$ . We claim that  $\bigcap_{i=1}^{n} \operatorname{Ker} \xi_{x_i} \subset \operatorname{Ker} F$ . Indeed let  $f \in \bigcap_{i=1}^{n} \operatorname{Ker} \xi_{x_i}$ . Then for every  $i \le n$ ,  $f(x_i) = 0$ . Let  $\varepsilon > 0$ . Clearly  $(1/\varepsilon) \cdot f \in <0, P, \delta >$ , so that  $F((1/\varepsilon) \cdot f) \subset (-1, 1)$  or, equivalently  $F(f) \subset (-\varepsilon, \varepsilon)$ . Since  $\varepsilon$  was arbitrary we have F(f) = 0 which implies  $f \in \operatorname{Ker} F$ . Now by lemma 1.3.2,  $F \in \operatorname{span} \{\xi_{x_1}, \ldots, \xi_{x_n}\}$ .  $\Box$ 

We can define a topology on L(X) as follows. For  $f \in C(X)$  let  $L(f): L(X) \to \mathbb{R}$  be defined by L(f)(F) = F(f). The topology on L(X) is the weakest topology which makes all L(f) ( $f \in C_p(X)$ ) continuous, i.e., the topology which has as a subbase the family,

 $\{L(f)^{-1}(U): f \in C(X) \text{ and } U \text{ open in } \mathbb{R}\}.$ 

With this topology, L(X) is called the topological dual of  $C_p(X)$ . Clearly L(X) is then a locally convex topological vector space.

**1.3.4 LEMMA:** Let  $f \in C_p(X)$ . Then  $L(f): L(X) \to \mathbb{R}$  is the unique continuous linear functional that extends f.

**PROOF:** That L(f) is a continuous linear functional is obvious. For  $x \in X$  we have

L(f)(x) = x(f) = f(x), so that L(f) extends f. Since X is a Hamel basis for L(X) it follows that L(f) is unique.  $\Box$ 

In theorem 1.3.3 we derived that X is algebraically a special subset of L(X). Topologically we have

**1.3.5 PROPOSITION:** X is homeomorphic to X as subspace of L(X), and X as a subspace of L(X) is closed in L(X).

**PROOF:** First of all we show that X is homeomorphic to X as subspace of L(X). Let U be open in X and  $x \in U$ . Let  $f \in C(X)$  be a Urysohn function such that f(x)=1 and  $f(X \setminus U) = \{0\}$ . By lemma 1.3.4, f extends to a continuous linear functional  $L(f): L(X) \to \mathbb{R}$ . Let  $V = L(f)^{-1}(0, \infty)$ . Then V is open in L(X) and  $x \in V \cap X \subset U$ . Now let V be open in L(X) and let  $x \in V \cap X$ . There are  $f_1, \ldots, f_n \in C_p(X)$  and  $U_1, \ldots, U_n$  open in  $\mathbb{R}$  with  $x \in \bigcap_{i=1}^n L(f_i)^{-1}(U_i) \subset V$ . Then  $x \in \bigcap_{i=1}^n f_i^{-1}(U_i) \subset V \cap X$ . We conclude that X is a subspace of L(X).

Second we show that X as subspace of L(X) is closed in L(X). To this end let  $F \in L(X) \setminus X$ . By theorem 1.3.3 there are  $x_1, \ldots, x_n \in X$  and  $\lambda_1, \ldots, \lambda_n \in \mathbb{R}$  such that  $F = \sum_{i=1}^n \lambda_i x_i$  with  $x_i \neq x_j$  for  $i \neq j$ .

*Case 1*:  $n \ge 2$  and  $\lambda_i \ne 0$  for all  $i \le n$ .

For each  $i \le n$  find  $V_i$  open in X and  $U_i$  open in  $\mathbb{R}$  such that  $V_i \cap V_j = \emptyset$   $(i \ne j), x_i \in V_i$ ,  $\lambda_i \in U_i$  and  $0 \notin U_i$ . For each  $i \le n$  there is a Urysohn function  $f_i: X \to \mathbb{R}$  such that  $f_i(x_i) = 1$  and  $f_i(X \setminus V_i) = 0$ . By lemma 1.3.4,  $f_i$  has a continuous linear extension  $L(f_i): L(X) \to \mathbb{R}$ . We claim that  $F \in \bigcap_{i=1}^n L(f_i)^{-1}(U_i) \subset L(X) \setminus X$ . Indeed for each  $i \le n$ ,

$$L(f_i)(F) = \sum_{j=1}^n \lambda_j L(f_i)(x_j) = \sum_{j=1}^n \lambda_j f_i(x_j) = \lambda_i \in U_i$$

and for  $x \in X$  there is  $i \le n$  with  $x \notin V_i$  (since  $n \ge 2$ ) so that  $L(f_i)(x) = f_i(x) \equiv 0 \notin U_i$ . This implies  $X \cap \bigcap_{i=1}^n L(f_i)^{-1}(U_i) = \emptyset$ , which proves case 1.

Case 2:  $F = \lambda_1 x_1$ .

Since  $F \notin X$ ,  $\lambda_1 \neq 1$ . Hence there is U open in  $\mathbb{R}$  with  $\lambda_1 \in U$  and  $1 \notin U$ . Let  $f \equiv 1$  and  $L(f): L(X) \to \mathbb{R}$  be its continuous linear extension. We claim that  $F \in L(f)^{-1}(U) \subset L(X) \setminus X$ . Indeed

$$L(f)(F) = \lambda_1 L(f)(x_1) = \lambda_1 f(x_1) = \lambda_1 \in U$$

and for  $x \in X$  we have  $L(f)(x) = f(x) = 1 \notin U$ . This proves case 2 and the proposition.

The following proposition will be of importance in the last section of this chapter.

**1.3.6 PROPOSITION:** Let X and Y be spaces. Then  $C_p(X) \sim C_p(Y)$  if and only if  $L(X) \sim L(Y)$ .

**PROOF:** First suppose  $\phi: C_p(X) \to C_p(Y)$  is a linear homeomorphism. Define  $\psi: L(X) \to L(Y)$  by  $\psi(F) = F \circ \phi^{-1}$ . Then  $\psi$  is obviously a well-defined linear function. To see that  $\psi$  is continuous notice that for  $f \in C_p(Y)$  and  $U \subset \mathbb{R}$  open we have

$$\Psi^{-1}(L(f)^{-1}(U)) = (L(f) \circ \Psi)^{-1}(U) = (L(\phi^{-1}(f))^{-1}(U))$$

is open in L(X).

Define  $\theta: L(Y) \to L(X)$  by  $\theta(G) = G \circ \phi$ . In the same way we can prove that  $\theta$  is a well-defined linear mapping. As is easily seen  $\theta = \psi^{-1}$ , so that  $\psi$  is a linear homeomorphism.

Second, suppose  $\psi: L(X) \to L(Y)$  is a linear homeomorphism. Define  $\phi: C_p(X) \to C_p(Y)$  by  $\phi(f) = (L(f) \circ \psi^{-1}) | Y$ . Then  $\phi$  is obviously a well-defined linear function. In order to prove that  $\phi$  is continuous at 0 let  $P \subset Y$  be finite and  $\varepsilon > 0$ . For every  $y \in Y$  there are  $x_1^y, \ldots, x_{n_y}^y \in X$  and  $\lambda_1^y, \ldots, \lambda_{n_y}^y \in \mathbb{R} \setminus \{0\}$  such that

$$\Psi^{-1}(y) = \sum_{i=1}^{n_y} \lambda_i^y x_i^y.$$

Let  $N = \max\{\sum_{i=1}^{n_y} | \lambda_i^y | : y \in P\}$ , let  $\delta = \varepsilon/N$  and let  $Q = \{x_i^y : y \in P \text{ and } i \le n_y\}$ . We claim that  $\phi(<0, Q, \delta >) \subset <0, P, \varepsilon >$ . Indeed, if  $f \in <0, Q, \delta >$  we have for  $y \in P$ 

$$\begin{aligned} |\phi(f)(y)| &= |(L(f) \circ \psi^{-1})(y)| = |L(f)(\sum_{i=1}^{n_y} \lambda_i^{y} x_i^{y})| = |\sum_{i=1}^{n_y} \lambda_i^{y} f(x_i^{y})| \\ &\leq \sum_{i=1}^{n_y} |\lambda_i^{y}| |f(x_i^{y})| < \delta \sum_{i=1}^{n_y} |\lambda_i^{y}| \leq \epsilon \end{aligned}$$

By linearity of  $\phi$  we conclude that  $\phi$  is continuous.

Define  $\theta: C_p(Y) \to C_p(X)$  by  $\theta(g) = (L(g) \circ \psi) | X$ . In the same way we can prove that  $\theta$  is a well-defined linear mapping and as is easily seen,  $\theta = \phi^{-1}$  so that  $\phi$  is a linear homeomorphism.  $\Box$ 

We define  $L^*(X)$  similar to L(X) using  $C_p^*(X)$  instead of  $C_p(X)$ . All observations made above for L(X) are also valid for  $L^*(X)$ .

**REMARK:** The results in this section are often used in the literature on function spaces with the topology of pointwise convergence (for example [41] and [44]). We were not able to find an explicit reference for their proofs, so we provided them ourselves. For more information on dual spaces of topological vector spaces we refer to [45].

#### **§1.4.** Supports and the topology of pointwise convergence

When dealing with continuous linear functions between function spaces endowed with the topology of pointwise convergence, it is possible to give a precise description of supports (cf. lemma 1.4.1).

Let X and Y be spaces, let  $\phi: C_p(X) \to C_p(Y)$  (resp.  $\phi: C_p^*(X) \to C_p^*(Y)$ ) be a continuous linear function and let  $y \in Y$  be fixed. Notice that the function  $\psi_y: C_p(X) \to \mathbb{R}$ (resp.  $\psi_y: C_p^*(X) \to \mathbb{R}$ ) defined by  $\psi_y = \xi_y \circ \phi$  is continuous and linear. So  $\psi_y \in L(X)$ (resp.  $\psi_y \in L^*(X)$ ), the dual of  $C_p(X)$  (resp.  $C_p^*(X)$ ). For every  $f \in C(Y)$  (resp.  $f \in C^*(X)$ ) we have  $\psi_y(f) = \phi(f)(y)$ . By theorem 1.3.3 there are for  $\psi_y \neq 0$ ,  $x_1, \ldots, x_n \in X$  and  $\lambda_1, \ldots, \lambda_n \in \mathbb{R} \setminus \{0\}$  such that  $\psi_y = \sum_{i=1}^n \lambda_i x_i$  (notice that whenever  $\phi$  is a bijection,  $\psi_y \neq 0$  for every  $y \in Y$ ). This means that for every  $f \in C(X)$  (resp.  $f \in C^*(X)$ ),  $\phi(f)(y) = \sum_{i=1}^n \lambda_i f(x_i)$ . Then

**1.4.1 LEMMA:** supp  $(y) = \{x_1, \ldots, x_n\}.$ 

**PROOF:** Let  $x \in \text{supp}(y)$  and suppose that  $x \notin \{x_1, \ldots, x_n\}$ . Since  $X \setminus \{x_1, \ldots, x_n\}$  is open, there is  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) such that  $f(x_i) = 0$  for every  $i \le n$  and  $\phi(f)(y) \ne 0$ . But  $\phi(f)(y) = \sum_{i=1}^n \lambda_i f(x_i) = 0$ . Contradiction.

Now let  $i \le n$  be fixed and U an open neighborhood of  $x_i$  such that  $U \cap \{x_j : j \le n \text{ and } j \ne i\} = \emptyset$ . Let  $f \in C^*(X)$  be a Urysohn function with  $f(X \setminus U) = 0$  and  $f(x_i) = 1$ . Then  $\phi(f)(y) = \sum_{i=1}^n \lambda_i f(x_i) = \lambda_i \ne 0$ .

From lemma 1.4.1 we have the following corollary of which part (b) simplifies the notion of effectiveness in the case of the topology of pointwise convergence

**1.4.2 COROLLARY:** Let X and Y be spaces, and let  $\phi: C_p(X) \to C_p(Y)$  (resp.  $\phi: C_p^*(X) \to C_p^*(Y)$ ), be a continuous linear function. Then for  $y \in Y$ ,

(a) for every  $z \in \text{supp}(y)$ , there is  $\lambda_z \in \mathbb{R}$  such that  $\phi(f)(y) = \sum_{z \in \text{supp}(y)} \lambda_z f(z)$ ,

for every 
$$f \in C(X)$$
 (resp.  $f \in C^*(X)$ ), and  
(b) if  $f,g \in C(X)$  (resp.  $f,g \in C^*(X)$ ), coincide on supp(y), then  
 $\phi(f)(y) = \phi(g)(y)$ .

Another useful property of supports with respect to the topology of pointwise convergence is given in the following:

**1.4.3 PROPOSITION:** Let X and Y be spaces and let  $\phi: C_p(X) \to C_p(Y)$  (resp.  $\phi: C_p^*(X) \to C_p^*(Y)$ ) be a linear homeomorphism. Then for every  $x \in X$  we have  $x \in \text{supp supp}(x)$  (in other words, for every  $x \in X$  there is  $y \in \text{supp}(x)$  such that  $x \in \text{supp (y)}$ ). In particular supp Y = X.

**PROOF:** Let  $x \in X$  and suppose  $x \notin \text{supp supp}(x)$ . Since supp supp (x) is finite (lemma 1.4.1), there is a Urysohn function  $f \in C^*(X)$  such that f(x)=1 and f(supp supp(x))=0. By corollary 1.4.2 (b) it follows that  $\phi(f)=0$  on supp(x) and again by corollary 1.4.2 (b) it then follows that  $f(x)=\phi^{-1}(\phi(f))(x)=0$ , and we arrived at a contradiction.  $\Box$ 

**1.4.4 PROPOSITION:** Let X and Y be spaces and let  $\phi: C_p(X) \to C_p(Y)$  (resp.  $\phi: C_p^*(X) \to CP^*Y$ ) be a continuous linear surjection. Then supp:  $Y \to \mathcal{P}(X) \setminus \{\emptyset\}$  is LSC.

**PROOF:** This follows from corollary 1.2.15 (a) (resp. corollary 1.2.16), lemma 1.2.7 and the remark following lemma 1.2.7.  $\Box$ 

In section 1.5 we need the following

**1.4.5 LEMMA:** Let X and Y be normal spaces, and let  $\phi: C_p(X) \to C_p(Y)$  be a continuous linear surjection. Then for each closed and bounded  $K \subset X$ , the set  $L = \{y \in Y : \text{supp}(y) \subset K\}$  is closed and bounded in Y.

**PROOF:** By proposition 1.4.4, supp is LSC hence *L* is closed. If *L* is not bounded, *L* contains a closed discrete subset  $\{y_n : n \in \mathbb{N}\}$ . For each  $n \in \mathbb{N}$ , let  $t_n = n \cdot \Sigma_{z \in \text{supp}(y_n)} | \lambda_z |$ . Then  $t_n > 0$ . Let  $g \in C(Y)$  be such that  $g(y_n) = t_n$ . Since  $\phi$  is a surjection, there is  $f \in C(X)$  such that  $\phi(f) = g$ . Since *K* is bounded, there is  $c \in \mathbb{R}$  such that  $f(K) \subset [-c, c]$ .

Let  $n \in \mathbb{N}$  be such that n > c. Then

§1.5. First applications

1

$$\begin{split} \phi(f)(y_n) &|= |\sum_{z \in \text{supp}} (y_n) \lambda_z f(z)| \\ &\leq \sum_{z \in \text{supp}} (y_n) |\lambda_z| \cdot |f(z)| \\ &\leq c \cdot \sum_{z \in \text{supp}} (y_n) |\lambda_z| \\ &\leq c_n. \end{split}$$

Contradiction. This proves the lemma.

If we consider function spaces  $C_p^*(X)$  we have the weaker

**1.4.6 LEMMA:** Let X and Y be metric spaces and let  $\phi: C_p^*(X) \to C_p^*(Y)$  be a continuous linear surjection. Then for each compact  $K \subset X$ , the set  $L = \{y \in Y : \text{supp}(y) \subset K\}$  is compact.

**PROOF:** By proposition 1.4.4, supp is LSC, hence *L* is closed in *Y*. For  $f \in C^*(K)$ , let  $\tilde{f} \in C^*(X)$  be an extension of *f*. Define

 $\theta: C_p^*(K) \to C_p^*(L)$  by  $\theta(f) = \phi(\tilde{f}) \mid L$ .

If  $g \in C^*(X)$  is another extension of f, then  $\tilde{f}$  and g coincides on  $\sup L$ , hence by corollary 1.4.2 (b)  $\phi(\tilde{f}) = \phi(g)$  on L. This implies that  $\theta$  is well-defined. It follows that  $\theta$  is a continuous linear function. By corollary 1.2.23 we then have that  $\theta$  considered as a function from  $C_u^*(K)$  to  $C_u^*(L)$  is also continuous. We claim that  $\theta$  is surjective. Let  $g \in C^*(L)$  and let  $\tilde{g} \in C^*(Y)$  be an extension of g. Since  $\phi$  is surjective, there is  $h \in C^*(X)$  with  $\phi(h) = \tilde{g}$ . Let  $f = h \mid K$ . Since h extends f,  $\theta(f) = g$ , so  $\theta$  is a surjection. By [47, Prop. 7.6.2] we have for a space Z that  $C_u^*(Z)$  is separable if and only if Z is compact and metrizable. This implies that  $C_u^*(K)$  is separable and hence  $C_u^*(L)$  is separable. So L is compact.  $\Box$ 

The proofs of lemma 1.4.5 and lemma 1.4.6 are different and not reversible.

**REMARK:** Jan Pelant provided us with the description of supports when dealing with the topology of pointwise convergence. We were informed that Arhangelskiĭ knew of this description of supports.

#### **§1.5.** First applications

Let X and Y be spaces. We define X and Y to be  $l_p$ ,  $l_0$  or  $l_b$ -equivalent whenever

 $C_p(X)$  and  $C_p(Y)$ ,  $C_0(X)$  and  $C_0(Y)$  or  $C_b(X)$  and  $C_b(Y)$  are linearly homeomorphic. We say that a topological property  $\mathcal{P}$  is preserved by  $\ell_p$ ,  $\ell_0$  or  $\ell_b$ -equivalence (resp.  $\ell_p^*$ ,  $\ell_0^*$  or  $\ell_b^*$ -equivalence) if for  $\ell_p$ ,  $\ell_0$  or  $\ell_b$ -equivalent (resp.  $\ell_p^*$ ,  $\ell_0^*$  or  $\ell_b^*$ -equivalent) spaces X and Y we have X has property  $\mathcal{P}$  iff Y has property  $\mathcal{P}$ . In this section we give some topological properties which are or which are not preserved by  $\ell_p$ ,  $\ell_0$  or  $\ell_b$ -equivalence (resp.  $\ell_p^*$ ,  $\ell_0^*$  or  $\ell_b^*$ -equivalence) and we state some questions.

**1.5.1 THEOREM** [1]: The following topological properties are preserved by  $l_p$ -equivalence:

(a) pseudocompactness,

(b) compactness, and

(c)  $\sigma$ -compactness.

**PROOF:** Let X and Y be  $l_p$ -equivalent spaces.

By corollary 1.2.9 and 1.2.15 (a) we have that pseudocompactness is preserved by  $l_p$ -equivalence.

For (b) and (c) we use that by proposition 1.3.6 L(X) and L(Y) are linearly homeomorphic. For every  $n \in \mathbb{N}$  define  $h_n: X^n \times [-n, n]^n \to L(X)$  by

$$h_n(x_1,\ldots,x_n,\alpha_1,\ldots,\alpha_n) = \sum_{i=1}^n \alpha_i x_i.$$

By proposition 1.3.5, X is homeomorphic to X as subspace of the topological vector space L(X), hence  $h_n$  is continuous. Furthermore  $L(X) = \bigcup_{i=1}^{\infty} h_n(X_n \times [-n, n]^n)$ . Suppose X is  $\sigma$ -compact. Then we have that L(X) is  $\sigma$ -compact, and hence that L(Y) is  $\sigma$ -compact. By proposition 1.3.5, Y is closed in L(Y) so Y is  $\sigma$ -compact. This finishes the proof of (c). When X is compact we again have that Y is  $\sigma$ -compact and hence Lindelöf. Furthermore by (a) we have that Y is pseudocompact. Since each Lindelöf space is normal, and each normal pseudocompact space is countably compact, we have that Y is a Lindelöf countably compact space, hence Y is compact.  $\Box$ 

This theorem and the proof of (b) and (c) are due to Arhangelskii. It follows that for normal spaces countable compactness is preserved by  $l_p$ -equivalence. Whether this is true for all spaces is still an open question. By the observations in section 1.2 it was possible to give an easier proof of (a), than the original one. For (b) and (c) this is not possible unless we assume that in X and Y every closed and bounded subset is compact. For such spaces we will now derive in theorem 1.5.2 a result in the same spirit as the previous one. For a space X let  $\mathcal{K}(X)$  be the family of all compact subsets of X. We regard  $\mathcal{K}(X)$  as a poset under inclusion. Then a subset  $\mathcal{B}$  is cofinal in  $\mathcal{K}(X)$  whenever for each  $K \in \mathcal{K}(X)$ there is  $B \in \mathcal{B}$  with  $K \subset B$ . The cofinality of  $\mathcal{K}(X)$  is the cardinal defined by

 $\operatorname{cof} \mathcal{K}(X) = \min\{ |\mathcal{B}| : \mathcal{B} \text{ is cofinal in } \mathcal{K}(X) \}$ 

(cf. [21]). A space X is said to be *hemicompact* whenever  $\operatorname{cof} \mathcal{K}(X) \leq \omega$ .

**1.5.2 THEOREM:** The cofinality of the family of compact subsets of a space is preserved by  $l_p$ -equivalence in the class of spaces in which every closed and bounded subset is compact.

**PROOF:** Let X and Y be  $\ell_p$ -equivalent spaces in which every closed and bounded subset is compact and let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism. Without loss of generality we assume  $\operatorname{cof} \mathcal{K}(X) \leq \operatorname{cof} \mathcal{K}(Y)$ . Let  $\{K_i : i \in I\}$  be cofinal in  $\mathcal{K}(X)$  such that  $|I| = \operatorname{cof} \mathcal{K}(X)$ . By corollary 1.2.15 (b) and the assumption on Y we have that  $\overline{\operatorname{supp} K_i}$  is compact for every  $i \in I$ . It suffices to prove that  $\{\overline{\operatorname{supp} K_i} : i \in I\}$  is cofinal in  $\mathcal{K}(Y)$ . For this let  $A \subset Y$  be compact. Again by corollary 1.2.15 (b) and the assumption on X,  $\overline{\operatorname{supp} A}$  is compact in X. So there is  $i \in I$  with  $\overline{\operatorname{supp} A} \subset K_i$ . Then by proposition 1.4.3,  $A \subset \operatorname{supp} K_i$ .  $\Box$ 

It remains open whether in general the cofinality of the family of compact subsets is preserved by  $l_p$ -equivalence. The results in section 1.4 allow us to obtain stronger results for an even more restricted class of spaces.

**1.5.3 THEOREM:** Let X and Y be normal spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a continuous linear surjection.

(a) If X is pseudocompact, then Y is pseudocompact.

If moreover in Y every closed and bounded subset is compact, then

- (b) if X is compact, then Y is compact,
- (c) if X is  $\sigma$ -compact, then Y is  $\sigma$ -compact, and
- (d)  $\operatorname{cof} \mathcal{K}(Y) \leq \operatorname{cof} \mathcal{K}(X)$ .

**PROOF:** For part (a) we have by lemma 1.4.5 that the set  $\{y \in Y : \text{supp}(y) \subset X\} = Y$  is pseudocompact whenever X is pseudocompact. Part (b) follows from part (a) and the assumption on Y.

For (c) let  $X = \bigcup_{n=1}^{\infty} X_n$  with for each  $n \in \mathbb{N}$ ,  $X_n \subset X_{n+1}$  and  $X_n$  compact. Let  $Y_n = \{y \in Y : \operatorname{supp}(y) \subset X_n\}$ . By lemma 1.4.5 and the assumption on Y we have that  $Y_n$ 

is compact. Since for each  $y \in Y$ , supp (y) is finite we also have that  $Y = \bigcup_{n=1}^{\infty} Y_n$ .

For (d) let  $\{K_i : i \in I\}$  be cofinal in  $\mathcal{K}(X)$ . For each  $i \in I$  let  $L_i = \{y \in Y : \operatorname{supp}(y) \subset K_i\}$ . Then  $L_i$  is compact. We claim that  $\{L_i : i \in I\}$  is cofinal in  $\mathcal{K}(Y)$ . Let  $A \subset Y$  be compact. Then by corollary 1.2.15 (b),  $\overline{\operatorname{supp} A}$  is compact. Hence  $\overline{\operatorname{supp} A} \subset K_i$  for some  $i \in I$ . This implies  $A \subset L_i$ .

By using lemma 1.4.6 instead of lemma 1.4.5 we obtain for metric spaces the following

**1.5.4 THEOREM:** Let X and Y be metric spaces and let  $\phi: C_p^*(X) \to C_p^*(Y)$  be a continuous linear surjection. Then

(a) if X is compact, then Y is compact, and

(b) if X is  $\sigma$ -compact, then Y is  $\sigma$ -compact.  $\Box$ 

**1.5.5 COROLLARY:** Compactness and  $\sigma$ -compactness are preserved by  $l_p^*$ -equivalence in the class of metric spaces.  $\Box$ 

The proof of theorem 1.5.3 (d) makes use of corollary 1.2.15 (b). Since we do not have such a result for continuous linear functions between function spaces  $C_p^*(X)$  we cannot copy the proof of theorem 1.5.3 (d) to this case.

Now that we have the above theorems for  $l_p$  and  $l_p^*$ -equivalence, we become interested whether the same result hold for  $l_0$  and  $l_0^*$ -equivalence (resp.  $l_b$  and  $l_b^*$ -equivalence). First we deal with  $l_0$  and  $l_0^*$ -equivalence.

**1.5.6 LEMMA:** Let X and Y be spaces and let  $\phi: C_0(X) \to C_0(Y)$  (resp.  $\phi: C_0^*(X) \to C_0^*(Y)$ ) a continuous linear function. Then for every compact  $B \subset Y$ , supp B is compact.

**PROOF:** There are non-empty compacta  $C_1, \ldots, C_n$  in X and open  $U_1, \ldots, U_n$  in Y such that

 $0 \in \bigcap_{i=1}^{n} A(C_i, U_i) \subset \phi^{-1}(A(B, (-1, 1))).$ 

Let  $C = \bigcup_{i=1}^{n} C_i$ . Then C is compact. We claim that  $\operatorname{supp} B \subset C$ . To the contrary suppose there are  $y \in B$  and  $x \in \operatorname{supp}(y) \setminus C$ . Since  $X \setminus C$  is a neighborhood of x and  $x \in \operatorname{supp}(y)$  there is  $f \in C(X)$  (resp.  $f \in C^*(X)$ ) such that f(C) = 0 and  $\phi(f)(y) \neq 0$ . By linearity of  $\phi$  we may assume  $\phi(f)(y) = 1$ . Since  $0 \in \bigcap_{i=1}^{n} A(C_i, U_i)$  we have
$0 \in \bigcap_{i=1}^{n} U_i$ , so  $f \in \bigcap_{i=1}^{n} A(C_i, U_i)$ . This gives  $\phi(f) \in A(B, (-1, 1))$  which implies that  $\phi(f)(y) \in (-1, 1)$ . Contradiction. We now have  $\operatorname{supp} B \subset C$ , so that  $\overline{\operatorname{supp} B} \subset C$ , which implies that  $\overline{\operatorname{supp} B}$  is compact.  $\Box$ 

**1.5.7 THEOREM:** Pseudocompactness is preserved by  $l_0$ -equivalence. Compactness is preserved by  $l_0$  and  $l_0^*$ -equivalence.

**PROOF:** Let X and Y be  $l_0$ -equivalent spaces By corollary 1.2.9 and 1.2.15 (a) we have that pseudocompactness is preserved by  $l_0$ -equivalence.

Let  $\phi: C_0(X) \to C_0(Y)$  (resp.  $\phi: C_0^*(X) \to C_0^*(Y)$ ) be a linear homeomorphism. Suppose X is compact. By lemma 1.2.5 and corollary 1.2.15 (a) (resp. corollary 1.2.16),  $\overline{\text{supp } X} = Y$ , and hence by lemma 1.5.6, Y is compact.  $\Box$ 

The proof that pseudocompactness is preserved by  $l_0$ -equivalence cannot be copied for  $l_0^*$ -equivalence since the proof of corollary 1.2.9 makes use of corollary 1.2.15 (b). From theorem 1.5.7 and corollary 1.2.21, theorem 1.5.1 (a) and (b) follow for the class of spaces in which every closed and bounded subset is compact.

For  $l_b$  and  $l_b^*$ -equivalence we have

**1.5.8 THEOREM:** Pseudocompactness is preserved by  $l_b$  and  $l_b^*$ -equivalence.

**PROOF:** Let X and Y be  $l_b$ -equivalent (resp.  $l_b^*$ -equivalent) spaces and let  $\phi: C_b(X) \to C_b(Y)$  (resp.  $\phi: C_b^*(X) \to C_b^*(Y)$ ) be a linear homeomorphism. Suppose Y is pseudocompact and X is not pseudocompact. Since <0, Y, 1> is open in  $C_b(Y)$  (resp.  $C_b^*(Y)$ ) there are  $f_1, \ldots, f_n$  in C(X) (resp.  $C^*(X)$ ), bounded  $A_1, \ldots, A_n$  in X and  $\varepsilon_1, \ldots, \varepsilon_n > 0$  such that  $0 \in \bigcap_{i=1}^n < f_i, A_i, \varepsilon_i >$  and  $\phi(\bigcap_{i=1}^n < f_i, A_i, \varepsilon_i >) \subset <0, Y, 1>$ . Let  $A = \bigcup_{i=1}^n A_i$ . Then A is bounded. Since X is not pseudocompact there is  $x \in X \setminus \overline{A}$ . Let  $f \in C^*(X)$  be a Urysohn function such that f(A) = 0 and f(x) = 1. Since  $f \neq 0$ ,  $\phi(f) \neq 0$ . Let  $y \in Y$  be such that  $\phi(f)(y) \neq 0$ . Define  $g: X \to \mathbb{R}$  by  $g = f/\phi(f)(y)$ . Since  $g(A) = \{0\}, g \in \bigcap_{i=1}^n < f_i, A_i, \varepsilon_i >$ , so  $\phi(g) \in <0, Y, 1>$ . However  $\phi(g)(y) = 1$ . Contradiction.  $\Box$ 

**Question 1:** Is pseudocompactness preserved by  $\ell_0^*$ -equivalence? Is compactness preserved by  $\ell_b$  or  $\ell_b^*$ -equivalence? Are  $\sigma$ -compactness or the cofinality of the family of compact subsets of a space preserved by  $\ell_0$ ,  $\ell_0^*$ ,  $\ell_b$  or  $\ell_b^*$ -equivalence?

From now on in this section we only deal with function spaces endowed with the topology of pointwise convergence.

It is well-known that cardinality and density are preserved by  $l_p$ -equivalence [2]. By the above techniques, we can give an alternative proof which is also valid for  $l_p^*$ -equivalence:

**1.5.9 THEOREM:** The following cardinal invariants are preserved by  $l_p$ -equivalence (resp.  $l_p^*$ -equivalence),

(a) cardinality, and

(b) density.

**PROOF:** Let X and Y be  $l_p$ -equivalent (resp.  $l_p^*$ -equivalent) spaces and let  $\phi: C_p(X) \to C_p(Y)$  (resp  $\phi: C_p^*(X) \to C_p^*(Y)$ ) be a linear homeomorphism. For (a), notice that if |X| = n, the algebraic dimension of  $C_p(X)$  is equal to n, hence we have that |Y| = n. So without loss of generality we assume  $\aleph_0 \le |X| \le |Y|$ . By lemma 1.4.1,  $|\operatorname{supp} X| \le |X|$ , so by proposition 1.4.3,  $|Y| \le |X|$ . We conclude that |X| = |Y|.

For (b) notice that if d(X) is finite, then since cardinality is preserved by  $l_p$ -equivalence, d(X) = |X| = |Y| = d(Y). So without loss of generality we assume  $\aleph_0 \le d(X) \le d(Y)$ . Let  $D \subset X$  be such that  $\overline{D} = X$  and |D| = d(X). Let  $E = \operatorname{supp} D$ . By lemma 1.4.1,  $|E| \le |D|$ . To prove that d(X) = d(Y) it suffices to prove that  $\overline{E} = Y$ . By proposition 1.4.3, lemma 1.2.6 and corollary 1.2.15 (a) (resp. corollary 1.2.16),

 $Y = \operatorname{supp} X = \operatorname{supp} \overline{D} \subset \overline{\operatorname{supp} D} = \overline{E} \subset Y.$ 

We conclude that density is preserved by  $l_p$ -equivalence.  $\Box$ 

As a corollary we see that separability is preserved by  $l_p$  and  $l_p^*$ -equivalence so Lindelöfness is preserved by  $l_p$  and  $l_p^*$ -equivalence in the class of metric spaces.

**Question 2:** Are density or cardinality preserved by  $l_0$  or  $l_0^*$ -equivalence (resp.  $l_b$  or  $l_b^*$ -equivalence)?

**1.5.10 THEOREM:** Local compactness is preserved by  $l_p$ -equivalence in the class of paracompact first countable spaces.

**PROOF:** Let X and Y be  $l_p$ -equivalent paracompact first countable spaces. Suppose X is locally compact and Y is not locally compact. Since X is a locally compact paracompact space, there is a locally finite open cover  $\{X_s : s \in S\}$  of X such that for each  $s \in S$ ,

 $\overline{X}_s$  is compact. Let  $y \in Y$  be a point without compact neighborhood and let  $\{U_n : n \in \mathbb{N}\}$  be a decreasing neighborhood base at y. Then for each  $n \in \mathbb{N}$ ,  $\overline{U}_n$  is not compact.

By lemma 1.2.10 and corollary 1.2.15 (a) there is  $k \in \mathbb{N}$  and  $\{s_1, \ldots, s_k\} \subset S$  such that supp  $U_k \subset \bigcup_{i=1}^k X_{s_i}$ . Let  $L = \overline{\bigcup_{i=1}^k X_{s_i}}$ . Then L is compact.

We now have by lemma 1.2.6 and corollary 1.2.15 (a),  $\sup \overline{U}_k \subset \overline{\sup p U_k} \subset L$  so by proposition 1.4.3,  $\overline{U}_k \subset \sup p \overline{U}_k \subset \sup p \overline{U}_k \subset \sup p L$ . Since each countably compact paracompact space is compact [24, Th 5.1.20], we have that  $\overline{U}_k$  is not countably compact. Since each paracompact space is normal [24, Th 5.1.18], Y is normal, and since each pseudocompact normal space is countably compact [24, Th. 3.10.21],  $\overline{U}_k$  is not pseudocompact. Hence by normality of Y,  $\overline{U}_k$  is not bounded in Y. However L is compact so by corollary 1.2.15 (b),  $\sup pL$  is bounded in Y. Contradiction.  $\Box$ 

Theorem 1.5.10 is due to S.P. Gulko and O.G. Okunev [2]. Their proof was by different methods than ours. In section 2.4 we show that the first countability assumption is essential in this result.

Question 3: Is paracompactness essential in theorem 1.5.10?

**Question 4:** Does theorem 1.5.10 hold for  $l_0$  or  $l_b$ -equivalence? Does it hold for  $l_p^*$ ,  $l_0^*$  or  $l_b^*$ -equivalence?

Before we state our next theorem we first need the following

**1.5.11 LEMMA:** Let X and Y be spaces and  $\phi: C_p(X) \to C_p(Y)$  a homeomorphism. Suppose that  $(f_n)_{n \in \mathbb{N}}$  is a sequence in  $C_p(X)$  such that  $f_n$  converges pointwise to a discontinuous function  $f \in \mathbb{R}^X$ . Suppose  $g: Y \to \mathbb{R}$  is an accumulation point of the set  $\{\phi(f_n) \mid n \in \mathbb{N}\}$ . Then g is not continuous.

**PROOF:** Since  $\{f_n \mid n \in \mathbb{N}\}$  is closed and discrete in  $C_p(X)$  we have  $\{\phi(f_n) \mid n \in \mathbb{N}\}$  is closed and discrete in  $C_p(Y)$ .  $\Box$ 

For a space X let  $X^{(1)} = \{x \in X : x \text{ is an accumulation point of } X\}$ . We have the following

**1.5.12 THEOREM:** Let X and Y be  $l_p$ -equivalent spaces which are both normal and first countable. Then  $X^{(1)}$  is countably compact if and only if  $Y^{(1)}$  is countably compact.

**PROOF:** Suppose  $X^{(1)}$  is not countably compact and  $Y^{(1)}$  is countably compact. Since  $X^{(1)}$  is not sequentially compact, there exists a closed discrete set  $F = \{x_n \mid n \in \mathbb{N}\}$  in  $X^{(1)}$ . For every  $n \in \mathbb{N}$  let  $\{U_j^n \mid j \in \mathbb{N}\}$  be a decreasing open base at  $x_n$  and  $f_j^n$  a Urysohn function such that  $f_j^n(x_n) = 1$  and  $f_j^n(X \setminus U_j^n) = 0$ . Then  $f_j^n \to \chi_{x_n}$  pointwise, where  $\chi_{x_n}$  is the characteristic function of  $x_n$ . Notice that  $\chi_{x_n}$  is discontinuous. Furthermore let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism and let  $g_j^n = \phi(f_j^n)$ .

CLAIM: For every  $y \in Y$  and  $n \in \mathbb{N}$ , the set  $\{g_i^n(y) \mid j \in \mathbb{N}\}$  is bounded in  $\mathbb{R}$ .

Suppose not. Then there are  $y \in Y$  and  $n \in \mathbb{N}$ , such that without loss of generality for every  $k \in \mathbb{N}$  there is  $j_k \in \mathbb{N}$ , with  $g_{j_k}^n(y) \ge 2^k$ . The function  $f = \sum_{k=1}^{\infty} 2^{-k} f_{j_k}^n \in C_p(X)$ , so  $\phi(f) = \sum_{k=1}^{\infty} 2^{-k} g_{j_k}^n \in C_p(Y)$ . But then we have a contradiction since  $\phi(f)(y) = \sum_{k=1}^{\infty} 2^{-k} g_{j_k}^n(y) = \infty$ .

For every  $y \in Y$ , let  $A_y$  be compact in  $\mathbb{R}$  such that  $\{g_j^n(y) \mid j \in \mathbb{N}\} \subset A_y$ . Then  $\prod_{y \in Y} A_y$  is a compact subset of  $\mathbb{R}^Y$ . Since  $\{g_j^n \mid j \in \mathbb{N}\} \subset \prod_{y \in Y} A_y$ ,  $\{g_j^n \mid j \in \mathbb{N}\}$  has an accumulation point  $\sigma_n$ . By lemma 1.5.11,  $\sigma_n$  is discontinuous, say at  $y_n$ . Notice that  $y_n \in Y^{(1)}$ . Since  $Y^{(1)}$  is sequentially compact, without loss of generality we may assume that there is  $y \in Y$  such that  $y_n \to y$ . Let  $\{V_n \mid n \in \mathbb{N}\}$  be a decreasing open base at y. Without loss of generality  $y_n \in V_n$ .

Since Y is first countable, for every  $n \in \mathbb{N}$  there is a sequence  $(y_k^n)_k$  in  $V_n$  such that  $y_k^n \to y_n$  and

$$\sigma_n(y_k^n) \not\to \sigma_n(y_n). \tag{(*)}$$

Let  $K = \bigcup_{n \in \mathbb{N}} \bigcup_{k \in \mathbb{N}} \{y_n, y_k^n\} \cup \{y\}$ . Then K is compact. Indeed, let  $\mathcal{V}$  be an open cover of K. There is  $V \in \mathcal{V}$  with  $y \in V$ . There is  $n_0 \in \mathbb{N}$  such that  $y \in V_{n_0} \subset V$ . Then  $\bigcup_{n \ge n_0} \bigcup_{k \in \mathbb{N}} \{y_n, y_k^n\} \cup \{y\} \subset V$ . Since  $\bigcup_{n < n_0} \bigcup_{k \in \mathbb{N}} \{y_n, y_k^n\}$  is compact, we are done.

Since K is compact, we have by corollary 1.2.15 (b) that  $\overline{\operatorname{supp} K}$  is bounded in X. Since F is closed and discrete and X is normal, F is not bounded. This implies that there is  $n \in \mathbb{N}$  such that  $x_n \notin \overline{\operatorname{supp} K}$ . Then there is  $j_0 \in \mathbb{N}$  such that  $U_{j_0}^n \cap \overline{\operatorname{supp} K} = \emptyset$ . So for every  $z \in K$  and  $j \ge j_0$ ,  $f_j^n$  and the zero function on X are equal on  $\operatorname{supp} (z)$ . Then by corollary 1.4.2, we have that  $g_j^n(z) = 0$  for every  $j \ge j_0$  and  $z \in K$ . But then for every  $k \in \mathbb{N}$  we have that  $\sigma_n(y_k^n) = 0$  and  $\sigma_n(y_n) = 0$ , which gives a contradiction with (\*). This completes the proof of the theorem.  $\Box$ 

In section 2.4 we show that the first countability assumption in theorem 1.5.12 is essential. The question remains whether normality is essential.

Our last result in this section deals with the class of metric spaces. Let X be a metric space and  $\mathcal{U}$  a family of subsets of X. We define diam  $\mathcal{U}$  to be sup {diam  $U : U \in \mathcal{U}$ }. We first need the following

**1.5.13 LEMMA:** Let X be a metric space which is not completely metrizable and let  $\{\mathcal{U}_n : n \in \mathbb{N}\}$  be a collection of open covers of X such that for each  $n \in \mathbb{N}$ , diam  $\mathcal{U}_n < 1/n$ . Then there is a strictly increasing sequence  $(i_n)_{n \in \mathbb{N}}$  of natural numbers and for each  $n \in \mathbb{N}$ , there is  $U_n \in \mathcal{U}_{i_n}$  such that  $\overline{U}_{n+1} \subset U_n$ , and moreover  $\bigcap_{n=1}^{\infty} U_n = \emptyset$ .

**PROOF:** Let  $\tilde{X}$  be the completion of X. For each  $U \in \mathcal{U}_n$ , there is  $V_U$  open in  $\tilde{X}$  such that diam  $V_U < 3/n$  and  $V_U \cap X = U$ . Let  $V_n = \bigcup \{V_U : U \in \mathcal{U}_n\}$ . Then  $V_n$  is open in  $\tilde{X}$  and  $X \subset V_n$ . So  $V = \bigcap_{n=1}^{\infty} V_n$  is a  $G_{\delta}$ -subset of  $\tilde{X}$  such that  $X \subset V$ . Since X is not completely metrizable and V is completely metrizable ([24, Th. 4.3.23]), there is  $x \in V \setminus X$ .

CLAIM: There is a strictly increasing sequence  $(i_n)_{n \in \mathbb{N}}$  of natural numbers and there are for each  $n \in \mathbb{N}$ ,  $U_n \in \mathcal{U}_{i_n}$  such that  $x \in V_{U_n}$  and  $\overline{V}_{U_{n+1}} \subset V_{U_n}$ .

Let  $i_1 = 1$  and let  $U_1 \in \mathcal{U}_1$  be such that  $x \in V_{U_1}$ . Let m > 1 and suppose  $i_1, \ldots, i_{m-1}$ and  $U_1, \ldots, U_{m-1}$  are found. Let  $\delta = d(x, \tilde{X} \setminus V_{U_{m-1}})$  and let  $i_m > i_{m-1}$  be such that  $1/i_m < \delta/3$ . There is  $U_m \in \mathcal{U}_{i_m}$  such that  $x \in V_{U_m}$ . Since diam  $\overline{V}_{U_m} \le 3/i_m < \delta$  and  $x \in V_{U_m}, \overline{V}_{U_m} \subset V_{U_{m-1}}$ . This proves the claim.

Since diam  $V_{U_m} \to 0 \ (m \to \infty)$ ,  $\bigcap_{m=1}^{\infty} V_{U_m} = \{x\}$ . This implies  $\bigcap_{m=1}^{\infty} U_m = \bigcap_{m=1}^{\infty} V_{U_m} \cap X = \emptyset$ .

Furthermore for  $m \in \mathbb{N}$ , we have

$$\overline{U}_{m+1} = \overline{V}_{U_{m+1}} \cap X \subset V_{U_m} \cap X = U_m.$$

This proves the lemma.

Recall from section 1.2 that for metric spaces X and Y and a linear function  $\phi: C(X) \to C(Y)$  we defined for  $U \subset X$ , the set  $T_U = \{y \in Y : \operatorname{supp}(y) \cap U \neq \emptyset\}$ , and for a family  $\mathcal{U}$  of subsets of X the collection  $T_{\mathcal{U}} = \{T_U : U \in \mathcal{U}\}$ . We now state our last theorem in this section.

**1.5.14 THEOREM:** Let X and Y be metric spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a continuous linear surjection. If X is completely metrizable, then Y is completely metrizable.

**PROOF:** Suppose X is completely metrizable and Y is not completely metrizable. Since X and Y are metric spaces, by lemma 1.2.11 there are locally finite open covers  $\mathcal{U}_n$  of X and  $\mathcal{V}_n$  of Y,  $(n \in \mathbb{N})$ , such that

(1) diam 
$$\mathcal{U}_n < \frac{1}{n}$$
, diam  $\mathcal{V}_n < \frac{1}{n}$ ,  $\mathcal{U}_{n+1}$  refines  $\mathcal{U}_n$ , and

(2) each  $V \in \mathcal{V}_n$  intersects only finitely many elements of  $T_{\mathcal{U}_n}$ .

By lemma 1.5.13, we may assume that for each  $n \in \mathbb{N}$ , there is  $V_n \in \mathcal{V}_n$  such that  $\bigcap_{n=1}^{\infty} V_n = \emptyset$  and for each  $n \in \mathbb{N}$ ,  $\overline{V}_{n+1} \subset V_n$ . By (2), for each  $n \in \mathbb{N}$ , there is a finite subset  $\{U_1^n, \ldots, U_{m_n}^n\}$  of  $\mathcal{U}_n$  such that

(3) for  $U \in \mathcal{U}_n, V_n \cap T_U \neq \emptyset$  if and only if  $U \in \{U_1^n, \ldots, U_{m_n}^n\}$ .

We claim that for each  $n \in \mathbb{N}$ ,

(4)  $\bigcup_{j=1}^{m_{n+1}} U_j^{n+1} \subset \bigcup_{j=1}^{m_n} U_j^n.$ 

Indeed, since  $\mathcal{U}_{n+1}$  refines  $\mathcal{U}_n$ , there are for each  $j \leq m_{n+1}$ ,  $U_j \in \mathcal{U}_n$  such that  $U_j^{n+1} \subset U_j$ . Since  $V_{n+1} \subset V_n$ , and  $V_{n+1} \cap T_{U_j^{n+1}} \neq \emptyset$ , we have  $V_n \cap T_{U_j} \neq \emptyset$ . So by (3),  $U_j \in \{U_1^n, \ldots, U_{m_n}^n\}$ . This gives  $\bigcup_{j=1}^{m_{n+1}} U_j^{n+1} \subset \bigcup_{j=1}^{m_n} U_j^n$ . This proves (4).

Notice that by (3), for every  $n \in \mathbb{N}$ , supp  $V_n \subset \bigcup_{j=1}^{m_n} U_j^n$ . For each  $n \in \mathbb{N}$ , let  $y_n \in V_n$ . Then

(5) supp 
$$(y_n) \subset \bigcup_{j=1}^{m_n} U_j^n$$
.

Let  $K = \overline{\bigcup_{n=1}^{\infty} \operatorname{supp}(y_n)}$ . Since K is a closed subset of X, K is complete.

CLAIM: K is compact.

Since K is complete, it remains to prove that K is totally bounded. To this end it suffices to prove that  $\bigcup_{n=1}^{\infty} \operatorname{supp}(y_n)$  is totally bounded. Let  $\varepsilon > 0$  and let  $j \in \mathbb{N}$  be such that  $1/j < \varepsilon/2$ . For  $k \le m_j$ , let  $z_k \in U_k^j$ . Since diam  $U_k^j < 1/j$ ,  $U_k^j \subset B(z_k, \varepsilon)$ . Then by (4) and (5),  $\bigcup_{n=j}^{\infty} \operatorname{supp}(y_n) \subset \bigcup_{k=1}^{m_j} B(z_k, \varepsilon)$ . Since  $\bigcup_{n=1}^j \operatorname{supp}(y_n)$  is finite, we are done.

By lemma 1.4.5,  $L = \{y \in Y : \text{supp}(y) \subset K\}$  is a closed and bounded subset of Y, hence L is compact. Since  $\bigcap_{n=1}^{\infty} \overline{V}_n = \emptyset$ ,  $(y_n)_{n \in \mathbb{N}} \subset L$  is a sequence without convergent subsequence. Contradiction. This proves the theorem.  $\Box$ 

**1.5.15 COROLLARY:** Complete metrizability is preserved by  $l_p$ -equivalence in the class of metric spaces.  $\Box$ 

In [52], Uspenskiĭ proved that for  $l_p$ -equivalent spaces X and Y, we have that if X is metric, then Y is a  $\sigma$ -metrizable paracompact space, where  $\sigma$ -metrizable means a countable union of closed metrizable subspaces. In view of theorem 1.5.14, one could conjecture that if X is moreover completely metrizable, then Y is also Čech-complete. This is however not the case. In example 2.4.10 we give two  $l_p$ -equivalent spaces X and Y with X countable metric locally compact and Y paracompact  $\sigma$ -metrizable but not Čech-complete.

In general if  $\phi: C_p^*(X) \to C_p^*(Y)$  is a continuous linear surjection, for metric spaces X and Y, the proof of theorem 1.5.14 does not work. By example 1.2.12 we cannot make use of a lemma such as lemma 1.2.11.

Most of the proofs in this section concerning the topology of pointwise convergence depend strongly on corollary 1.2.15 (b). Since we do not have such a result for continuous linear functions between function spaces  $C_p^*(X)$ , we cannot copy these proofs. The question remains whether the results for  $l_p$ -equivalence are also valid for  $l_p^*$ -equivalence. Of course there are many more questions to ask. We made a selection in this section and we did not have the intention to be complete.

**REMARK:** For a recent survey on results obtained for  $l_p$ -equivalence we refer to [2]. Theorem 1.5.12 can be found in [5] and theorem 1.5.14 can be found in [7]. As far as we know all other results in this section (except for theorems 1.5.1, 1.5.9 and 1.5.10) are new.

·

## **CHAPTER 2**

# On the $l_p$ and $l_0$ -equivalence of locally compact spaces

The purpose of this chapter is to present isomorphical classifications of function spaces of some locally compact spaces endowed with the topology of pointwise convergence and with the compact open topology. Since ordinals play an important role in the proofs of these classifications, in section 1 we derive some (well-known) properties of ordinals. Other important notions are derivatives of spaces and scatteredness. In section 2 we will give the relevant definitions and present some preliminary results, for example the theorems of Cantor-Bendixson and Sierpiński-Mazurkiewicz. In section 3 we prove some rather general results concerning linear homeomorphisms between certain function spaces.

After these three sections we are in a position to present the first isomorphical classification. In section 4 we present a complete classification of the function spaces  $C_p(X)$  for separable metric zero-dimensional compact spaces X. It turns out that this classification is similar to the one Bessaga and Pelczyński gave in [10] for the spaces  $C_0(X)$ . In section 5 we present a complete isomorphical classification of the function spaces  $C_p(X)$ , for compact ordinal spaces. This classification is also similar to one for the spaces  $C_0(X)$  (viz. the one Kislyakov gave in [34]). In a sense, it is an extension of the classification found in section 4. In section 6 we present a classification of the spaces  $C_0(X)$  and  $C_p(X)$  for non-compact  $\sigma$ -compact ordinal spaces X. Finally in section 7 we present a complete isomorphical classification of the spaces  $C_0(X)$  and  $C_p(X)$  for separable metric zero-dimensional locally compact spaces X. This result uses the classifications found in sections 4 and 6.

We already proved that for spaces X and Y having the property that each closed and bounded subset is compact,  $C_p(X) \sim C_p(Y)$  implies  $C_0(X) \sim C_0(Y)$ . It turns out that in each of the classes mentioned above we also have the converse implication. Recall that the converse implication does not hold in general (cf. page 29).

## §2.1. Ordinals

Ordinals play an important role in this monograph, in particular prime components. In this section we will present some facts about ordinals, we will give the definitions of initial, regular and singular ordinals and of prime components, and we prove some (well-known) results. Most of these can be found in [48] and [35]. For definitions of ordinals, cardinals and related topics which are not defined or proved in this section, we refer to [48], [35] and [32]. In this section, every greek letter denotes an ordinal, and finite ordinals will be denoted by n or m occasionally.

We begin with stating some basic properties of addition, multiplication and exponentation of ordinals. Recall that addition of ordinals is associative but not commutative. If  $\alpha$  and  $\beta$  are ordinals, then  $\alpha + \beta \ge \beta$  and if  $\alpha > 0$ , then  $\beta + \alpha > \beta$ . Observe that not always  $\alpha + \beta > \beta$  because  $1 + \omega = \omega$ . Another important property of addition is the following: If  $\alpha \ge \beta$ , then there is exactly one ordinal  $\gamma$  such that  $\alpha = \beta + \gamma$ . We denote this number  $\gamma$  by  $\alpha - \beta$ . With these properties one can easily derive the following

**2.1.1 PROPOSITION:** Let  $\alpha$ ,  $\beta$ ,  $\gamma$  and  $\delta$  be ordinals. Then

(a)  $\beta < \gamma$  implies  $\alpha + \beta < \alpha + \gamma$ , (b)  $\beta < \gamma$  implies  $\beta + \alpha \le \gamma + \alpha$ , and

(c)  $\alpha < \gamma$  and  $\beta < \delta$  implies  $\alpha + \beta < \gamma + \delta$ .

Like addition, multiplication is associative but not commutative (for example  $2\cdot\omega = \omega \neq \omega \cdot 2$ ). Now let  $\alpha$ ,  $\beta$ , and  $\gamma$  be ordinals. Then  $\alpha \cdot (\beta + \gamma) = \alpha \cdot \beta + \alpha \cdot \gamma$ , but in general  $(\beta + \gamma) \cdot \alpha \neq \beta \cdot \alpha + \gamma \cdot \alpha$  (for example  $(1 + 1) \cdot \omega = \omega \neq \omega \cdot 2 = 1 \cdot \omega + 1 \cdot \omega$ ). One can now easily deduce that if  $\alpha > 0$  and  $\beta > \gamma$ , then  $\alpha \cdot \beta > \alpha \cdot \gamma$ , however, if  $\beta > \gamma$  then  $\beta \cdot \alpha \geq \gamma \cdot \alpha$  (notice that  $2\cdot\omega = 1\cdot\omega$ ).

We also have the following important

**2.1.2 PROPOSITION:** Let  $\alpha$  and  $\beta$  be ordinals. If  $\alpha > 0$  then there are ordinals  $\mu$  and  $\nu$  such that  $\beta = \alpha \cdot \mu + \nu$  with  $\nu < \alpha$ .

**2.1.3 COROLLARY:** Let  $\alpha$ ,  $\beta$  and  $\gamma$  be ordinals such that  $\beta < \alpha \cdot \gamma$ . If  $\alpha > 0$  then there are ordinals  $\mu$  and  $\nu$  such that  $\beta = \alpha \cdot \mu + \nu$  with  $\mu < \gamma$  and  $\nu < \alpha$ .

**PROOF:** By proposition 2.1.2 there are ordinals  $\mu$  and  $\nu$  such that  $\beta = \alpha \cdot \mu + \nu$  with  $\nu < \alpha$ . By proposition 2.1.1 (a)  $\beta \ge \alpha \cdot \mu$ . So if  $\mu \ge \gamma$ , then  $\beta \ge \alpha \cdot \mu \ge \alpha \cdot \gamma$ , which is a con-

§2.1. Ordinals

tradiction. Hence  $\mu < \gamma$ .

Exponentation of ordinals  $\alpha$  and  $\beta$  is defined by transfinite induction as follows:

a) If β=0 then α<sup>β</sup> = 1,
b) if β is a successor, say β=γ+1, then α<sup>β</sup> = α<sup>γ</sup>·α,
c) if β is a limit ordinal, then α<sup>β</sup> = sup {α<sup>γ</sup>: γ < β}.</li>

With this definition one can easily prove the following

**2.1.4 PROPOSITION:** Let  $\alpha$ ,  $\beta$  and  $\gamma$  be ordinals. Then (a) if  $\alpha > 1$  and  $\beta < \gamma$ , then  $\alpha^{\beta} < \alpha^{\gamma}$ , (b) if  $\alpha > 1$  then  $\alpha^{\beta} \ge \beta$ , and (c)  $\alpha^{\beta+\gamma} = \alpha^{\beta} \cdot \alpha^{\gamma}$ .

The following lemma will be used in section 2.5.

**2.1.5 LEMMA:** Let  $\alpha > 1$  and  $\beta \ge 1$  be ordinals. Then there are  $\gamma \le \beta$ ,  $1 \le \lambda < \alpha$  and  $\delta < \alpha^{\gamma}$  such that  $\beta = \alpha^{\gamma} \cdot \lambda + \delta$ .

**PROOF:** By proposition 2.1.4 (a) and (b),  $\beta \le \alpha^{\beta} < \alpha^{\beta+1}$ , so the set  $A = \{v : \alpha^{v} > \beta\}$  is non-empty. Let  $\mu = \min A$ . Notice that  $1 \le \mu \le \beta + 1$ . If  $\mu$  is a limit ordinal, then  $\beta < \alpha^{\mu} = \sup \{\alpha^{v} : v < \mu\}$  implies there is  $v < \mu$  with  $\beta < \alpha^{v}$ . This is a contradiction, so  $\mu = \gamma + 1$  for some  $\gamma$ . Since  $\mu \le \beta + 1$ ,  $\gamma \le \beta$ .

Since  $\beta < \alpha^{\gamma} \cdot \alpha$ , by corollary 2.1.3, there is  $\delta < \alpha^{\gamma}$  and  $\lambda < \alpha$  such that  $\beta = \alpha^{\gamma} \cdot \lambda + \delta$ . If  $\lambda = 0$ , then  $\beta = \delta < \alpha^{\gamma}$  which is impossible, so  $\lambda \ge 1$ .

Let  $\alpha$  be an ordinal. By  $\overline{\alpha}$  we denote the cardinality of  $\alpha$  (i.e.,  $\overline{\alpha} = |W(\alpha)|$ ) and we call  $\overline{\alpha}$  the *power* of  $\alpha$ . An ordinal  $\phi \ge \omega$  is called an *initial ordinal* if  $\phi$  is the smallest ordinal  $\beta$  such that  $\overline{\beta} = \overline{\phi}$ , i.e.,  $\gamma < \phi$  implies  $\overline{\gamma} < \overline{\phi}$ . To every initial ordinal  $\phi$  we assign the *index*  $i(\phi)$  of  $\phi$  as the ordertype of the set  $P(\phi) = \{\psi < \phi : \psi \text{ is initial}\}$ . For example  $i(\omega) = 0$  and  $i(\omega_1) = 1$ . Notice that for every initial ordinal  $\phi$ ,  $i(\phi) \le \phi$ .

The following theorem easily follows from the above definitions.

**2.1.6 THEOREM** ([35, Th. 3, p273]): If  $\psi$  and  $\phi$  are initial ordinals with  $\psi < \phi$ , then  $i(\psi) < i(\phi)$ .

As a direct consequence of this theorem we remark that to distinct initial ordinals

correspond distinct indices. So we may denote the initial ordinal  $\phi$  with index  $\alpha$  by  $\omega_{\alpha}$ . Since  $i(\phi) \leq \phi$ , it follows that  $\alpha \leq \omega_{\alpha}$ . The following theorem states that  $\omega_{\alpha}$  is defined for every ordinal  $\alpha$ .

**2.1.7 THEOREM** ([35, Th. 5, p273]): Every ordinal is the index of some initial ordinal.

An  $\alpha$ -sequence is a function  $\phi$  with domain  $W(\alpha)$ , whose values are ordinals. If  $\gamma < \beta < \alpha$  implies  $\phi(\gamma) < \phi(\beta)$ , then  $\phi$  is an *increasing*  $\alpha$ -sequence. The *limit*  $\lim_{\xi < \alpha} \phi(\xi)$  of an increasing  $\alpha$ -sequence  $\phi$  is the ordinal sup { $\phi(\xi) : \xi < \alpha$ }. We say that an ordinal  $\lambda$  is cofinal with a limit ordinal  $\alpha$ , if  $\lambda$  is the limit of an increasing  $\alpha$ -sequence  $\phi$ , i.e.,  $\lambda = \lim_{\xi < \alpha} \phi(\xi)$ .

**2.1.8 THEOREM** ([35, Th. 8, p274]): If  $\lambda$  is a limit ordinal, then  $\omega_{\lambda} = \lim_{\xi < \lambda} \omega_{\xi}$ .

**2.1.9 THEOREM** ([35, Th. 10, p274]): Let  $\lambda > 0$  be a limit ordinal. The smallest ordinal  $\alpha$  such that  $\lambda$  is cofinal with  $\alpha$  is an initial ordinal.

If we now define for every limit ordinal  $\lambda$ , the ordinal  $cf(\lambda)$  as the smallest  $\alpha$  such that  $\lambda$  is cofinal with  $\alpha$ , then by theorem 2.1.9,  $cf(\lambda)$  is an initial ordinal. For example  $cf(\omega) = \omega$ ,  $cf(\omega_1) = \omega_1$ ,  $cf(\omega_{\omega}) = \omega$  and  $cf(\omega_{\omega_1}) = \omega_1$ . Since  $\lambda = \lim_{\beta < \lambda} \beta$ ,  $\lambda$  is cofinal with itself, hence  $cf(\lambda) \le \lambda$ 

This observation leads us to the following definition. If  $cf(\omega_{\alpha}) = \omega_{\alpha}$ , we call  $\omega_{\alpha}$  a *regular initial ordinal* or shortly regular. Otherwise it is called a *singular initial ordinal* or shortly singular. For example,  $\omega$  and  $\omega_1$  are regular and  $\omega_{\omega}$  is singular. This is standard terminology of course.

We are going to prove that if  $\omega_{\gamma}$  is singular, then  $\gamma$  is a limit ordinal (cf. theorem 2.1.10). For that we first recall the following well-known fact: If  $\mathfrak{m}$  and  $\mathfrak{n}$  are cardinals which are not both finite, then  $\mathfrak{m} + \mathfrak{n} = \max(\mathfrak{m}, \mathfrak{n}) = \mathfrak{m} \cdot \mathfrak{n}$ . This allows us to derive some important corollaries.

**2.1.10 THEOREM** ([35, Th. 9, p278]): If  $\alpha$  is a successor, then  $\omega_{\alpha}$  is regular.

**PROOF:** Let  $\alpha = \beta + 1$  and  $\omega_{\gamma} = cf(\omega_{\alpha})$ . That means there is an increasing  $\omega_{\gamma}$ sequence  $\phi$  such that  $\lim_{\xi < \omega_{\gamma}} \phi(\xi) = \omega_{\alpha}$ . But then  $W(\omega_{\alpha}) \subset \bigcup_{\xi < \omega_{\gamma}} W(\phi(\xi))$ . Notice
that  $|W(\phi(\xi))| \le \overline{\omega_{\beta}}$ . So we have by the above remark and theorem 2.1.6

§2.1. Ordinals

$$\overline{\omega_{\alpha}} = |W(\omega_{\alpha})| \le |\bigcup_{\xi < \omega_{\gamma}} W(\phi(\xi))| \le \Sigma_{\xi < \omega_{\gamma}} |W(\phi(\xi))|$$
$$\le \Sigma_{\xi < \omega_{\gamma}} \overline{\omega_{\beta}} = \overline{\omega_{\gamma}} \cdot \overline{\omega_{\beta}} = \overline{\omega_{\max(\beta, \gamma)}}.$$

It follows by theorem 2.1.6 that  $\alpha \le \max(\beta, \gamma)$ , so  $\alpha \le \gamma$ . Since  $\omega_{\gamma} = cf(\omega_{\alpha}) \le \omega_{\alpha}$ , by theorem 2.1.6  $\gamma \le \alpha$  and thus  $\gamma = \alpha$ .

**2.1.11 COROLLARY:** If  $\omega_{\alpha}$  is singular, then  $\alpha$  is a limit ordinal.

The next two results are going to be used in section 2.5.

**2.1.12 PROPOSITION:**  $\alpha \cdot \omega_{\beta} = \omega_{\beta}$  for every  $0 < \alpha < \omega_{\beta}$ .

**PROOF:** By the remark on page 52,  $\overline{\alpha \cdot \omega_{\beta}} = \overline{\omega_{\beta}}$ , so we are done if we prove that  $\alpha \cdot \omega_{\beta}$  is initial. To this end, let  $\delta < \alpha \cdot \omega_{\beta}$ . By corollary 2.1.3 there are  $\mu < \omega_{\beta}$  and  $\nu < \alpha$  such that  $\delta = \alpha \cdot \mu + \nu$ . So  $\overline{\delta} = \overline{\alpha} \cdot \overline{\mu} + \overline{\nu} = \max{\{\overline{\alpha}, \overline{\mu}, \overline{\nu}\}}$ . Since  $\alpha, \mu, \nu < \omega_{\beta}$  it follows that  $\overline{\delta} < \overline{\omega_{\beta}}$ , and therefore  $\alpha \cdot \omega_{\beta}$  is initial.  $\Box$ 

Notice that proposition 2.1.12 is not true for  $\alpha = \omega_{\beta}$ . For example,  $\omega^2 \neq \omega$ .

### **2.1.13 PROPOSITION:** Let $\alpha \ge \omega$ be an ordinal. If $\overline{\gamma} \le \overline{\alpha}$ then $\overline{\alpha}^{\overline{\gamma}} = \overline{\alpha}$ .

**PROOF:** We prove this by induction on  $\gamma$ . If  $\gamma = 1$  then it is a triviality, so let  $\gamma > 1$  and suppose the proposition is true for every  $\delta < \gamma$ .

Case 1:  $\gamma$  is a successor, say  $\gamma = \delta + 1$ .

Then by the the remark on page 52,  $\overline{\alpha^{\gamma}} = \overline{\alpha^{\delta}} \cdot \overline{\alpha} = \overline{\alpha} \cdot \overline{\alpha} = \overline{\alpha}$ .

Case 2:  $\gamma$  is a limit ordinal.

Then

$$\overline{\alpha^{\gamma}} = \overline{\lim_{\delta < \gamma} \alpha^{\delta}} = |\bigcup_{\delta < \gamma} W(\alpha^{\delta})| \le \Sigma_{\delta < \gamma} |W(\alpha^{\delta})|$$
$$= \Sigma_{\delta < \gamma} \overline{\alpha^{\delta}} = \overline{\gamma} \cdot \overline{\alpha} = \overline{\alpha}$$

(by the fact that  $\overline{\gamma} \leq \overline{\alpha}$ ).  $\Box$ 

Notice that  $\overline{\alpha}^{\gamma}$  need not be equal to  $\overline{\alpha}^{\overline{\gamma}}$ . For example,  $\overline{\omega}^{\overline{\omega}} > \overline{\omega}$ , but  $\overline{\omega}^{\overline{\omega}} = \overline{\omega}$ .

**2.1.14 EXAMPLE:** There is a singular ordinal  $\omega_{\alpha}$  such that  $\alpha = \omega_{\alpha}$ . Indeed, define the sequence  $(\beta_n)$  of ordinals inductively as follows:  $\beta_0 = 0$  and  $\beta_{n+1} = \omega_{\beta_n}$ . Let  $\alpha = \sup \{\beta_n : n \in \mathbb{N}\}$ . Then  $\omega_{\alpha} = \lim_{n < \omega} \beta_n$ , hence  $cf(\omega_{\alpha}) = \omega$ . Furthermore  $\alpha = \omega_{\alpha}$ .

We now come to the following definition. An ordinal  $\rho$  is a *prime component* if it satisfies the following condition:

If  $\rho = \beta + \gamma$  for some ordinals  $\beta$  and  $\gamma$ , then  $\gamma = 0$  or  $\gamma = \rho$ .

Examples of prime components are  $\omega$  and  $\omega_1$ . The ordinals 3,  $\omega + 1$  and  $\omega$ 2 are examples of ordinals which are not prime components. Furthermore, 0 and 1 are the only finite prime components. Since prime components play a very important role in this monograph and since their properties are not well-known, we prove all the properties that we need.

**2.1.15 THEOREM** ([48, Th. 1, p279]): An ordinal  $\rho$  is a prime component if and only if for every ordinal  $\beta < \rho, \rho = \beta + \rho$ .

**PROOF**: Suppose that the ordinal  $\rho$  is a prime component, and let  $\beta < \rho$ . There is an ordinal  $\gamma$  such that  $\rho = \beta + \gamma$  (see page 50). From the definition of prime component it follows that  $\gamma = 0$  or  $\gamma = \rho$ . Since  $\beta < \rho$ , we have  $\gamma = \rho$ .

Now suppose that  $\rho$  satisfies the condition mentioned and that  $\rho = \beta + \gamma$ . Assume  $\gamma > 0$ . By proposition 2.1.1 (a) it follows that  $\beta < \beta + \gamma = \rho$ . But then  $\rho = \beta + \rho$  and we may conclude that  $\gamma = \rho$  and so  $\rho$  is a prime component.  $\Box$ 

The next theorem plays an important role in section 2.6.

**2.1.16 THEOREM** ([48, Th. 2, p278]): For every ordinal  $\alpha > 0$ , there is an ordinal  $\beta$  and a prime component  $\rho > 0$  such that  $\alpha = \beta + \rho$ , where  $\beta = 0$  or  $\beta \ge \rho$ .

**PROOF:** Let  $\alpha > 0$  be an ordinal and  $A = \{\tau > 0: \text{ there is } \beta \text{ such that } \alpha = \beta + \tau\}$ . Notice that  $\tau \in A$  implies  $\tau \le \alpha$  (because by proposition 2.1.1 (b)  $\alpha = \beta + \tau \ge \tau$ ). Let  $\rho = \min A$  (which exists because A is a non-empty subset of  $W(\alpha)$  and the last set is well-ordered) and pick  $\beta$  such that  $\alpha = \beta + \rho$ .

We prove that  $\rho$  is a prime component. Indeed, suppose that  $\rho = \mu + \nu$ , with  $\nu > 0$ . Then  $\alpha = \beta + (\mu + \nu) = (\beta + \mu) + \nu$ , so  $\nu \in A$  and thus  $\nu \ge \rho$ . Since  $\mu + \nu \ge \nu$  (proposition 2.1.1 (b)) we have  $\nu \le \rho$ , so  $\nu = \rho$  and hence  $\rho$  is a prime component.

Finally, if  $\beta < \rho$ , then by theorem 2.1.15,  $\alpha = \beta + \rho = \rho$ , so in that case we can choose  $\beta = 0$ .

**2.1.17 LEMMA** ([48, lemma p282]): Let P be a set of prime components. Then  $\sup P$  is a prime component.

§2.1. Ordinals

**PROOF:** Let  $\beta = \sup P$ . We prove that  $\beta$  is a prime component. Let  $\gamma < \beta$ . Then there is a prime component  $\rho \in P$  with  $\gamma < \rho < \beta$ . Let  $\delta = \beta - \rho$  (see page 50). Then by theorem 2.1.15,

$$\gamma + \beta = \gamma + (\rho + \delta) = (\gamma + \rho) + \delta = \rho + \delta = \beta.$$

So by theorem 2.1.15,  $\beta$  is a prime component.  $\Box$ 

By applying this lemma to the set  $P = \{\rho \le \alpha : \rho \text{ is a prime component}\}$ , we get the following important

**2.1.18 COROLLARY** ([48, lemma p282]): Let  $\alpha$  be an ordinal. Then there is a largest prime component which is less than or equal to  $\alpha$ .

In the sequel we denote the largest prime component which is less than or equal to a given ordinal  $\alpha$  by  $\alpha'$ .

**2.1.19 LEMMA** ([48, Cor. p305]): If  $\alpha > 0$  is an ordinal, then  $\alpha \omega$  is the smallest prime component larger than  $\alpha$ .

**PROOF:** We first prove that  $\alpha\omega$  is a prime component. So suppose  $\alpha\omega = \mu + \nu$  with  $\nu \neq 0$  and  $\nu \neq \alpha\omega$ . By proposition 2.1.1 (a) and (b) it then follows that  $\mu$ ,  $\nu < \alpha\omega$ . By corollary 2.1.3 there are ordinals *m*, *n*,  $\gamma_1$  and  $\gamma_2$  such that  $\mu = \alpha m + \gamma_1$ ,  $\nu = \alpha n + \gamma_2$ , *m*,  $n < \omega$ , and  $\gamma_1$ ,  $\gamma_2 < \alpha$ .

From proposition 2.1.1 (a) it follows that  $\mu < \alpha m + \alpha$  and  $\nu < \alpha n + \alpha$ . Now with proposition 2.1.1 (c) it follows that  $\alpha \omega = \mu + \nu < \alpha m + \alpha + \alpha n + \alpha$ , so by the remark on page 50,  $\alpha \omega < \alpha (m + n + 2) < \alpha \omega$ . This is a contradiction and we conclude that  $\alpha \omega$  is a prime component.

Now suppose that there is a prime component  $\rho$  such that  $\alpha < \rho < \alpha \omega$ . By corollary 2.1.3 there are ordinals *n* and  $\gamma$  with *n* finite and  $\gamma < \alpha$  such that  $\rho = \alpha n + \gamma$ . Since  $\rho$  is a prime component,  $\gamma = 0$  and it follows that  $\rho = \alpha n$ . Since  $\alpha < \rho$ , n > 1, so  $\rho = \alpha(n-1) + \alpha$ . Since  $\rho$  is a prime component and  $0 < \alpha < \rho$  we arrived at a contradiction.  $\Box$ 

**2.1.20 COROLLARY:** Let  $\alpha$  be an ordinal. Then there is  $n \in \mathbb{N}$  and  $\gamma < \alpha'$  such that  $\alpha = \alpha' \cdot n + \gamma$ .

**PROOF:** Since  $\alpha' \cdot \omega$  is the smallest prime component larger than  $\alpha'$  (lemma 2.1.19),

 $\alpha' \le \alpha < \alpha'\omega$ . So by corollary 2.1.3 there are ordinals  $n < \omega$  and  $\gamma < \alpha'$  such that  $\alpha = \alpha' n + \gamma$ . Since  $\alpha \ge \alpha'$  we have  $n \ne 0$ , hence  $n \in \mathbb{N}$ .

The following theorem will often be used in this monograph.

**2.1.21 THEOREM** ([48, Th. 1, p320]): An ordinal  $\rho > 0$  is a prime component if and only if there is an ordinal  $\mu$  such that  $\rho = \omega^{\mu}$ .

**PROOF:** Let  $\rho > 0$  be a prime component.

CLAIM: There is  $\alpha$  such that  $\omega^{\alpha} \leq \rho < \omega^{\alpha+1}$ .

Indeed, by lemma 2.1.5 there are  $\alpha \le \rho$ ,  $1 \le \lambda < \omega$  and  $\delta < \omega^{\alpha}$  such that  $\rho = \omega^{\alpha} \cdot \lambda + \delta$ . Thus  $\omega^{\alpha} \le \rho < \omega^{\alpha+1}$ , and the claim is proved.

Since  $\rho$  is a prime component and since the smallest prime component larger than  $\omega^{\alpha}$  is  $\omega^{\alpha} \cdot \omega = \omega^{\alpha+1}$  (lemma 2.1.19), it follows by the claim that  $\rho = \omega^{\alpha}$ 

For the converse implication suppose there is v such that  $\omega^{v}$  is not a prime component. Let  $\mu$  be the smallest among them. Suppose that  $\mu = v + 1$ . Then by lemma 2.1.19  $\omega^{\mu} = \omega^{v} \cdot \omega$  is a prime component, which is not true, hence  $\mu$  is a limit ordinal. But then  $\omega^{\mu} = \sup \{\omega^{v}: v < \mu\}$ . Since for  $v < \mu$ ,  $\omega^{v}$  is a prime component, by lemma 2.1.17,  $\omega^{\mu}$  is a prime component. Contradiction.  $\Box$ 

The next lemma will be used in section 2.4.

**2.1.22 LEMMA:** Let  $\alpha$  and  $\beta$  be ordinals such that  $\alpha \ge \omega$  and  $\alpha \le \beta < \alpha^{\omega}$ . Then  $\alpha' \le \beta < (\alpha')^{\omega}$ .

**PROOF:** Since  $\alpha \le \beta < \alpha^{\omega}$ , there is  $n \in \mathbb{N}$  such that  $\alpha \le \beta < \alpha^n$ . Furthermore by theorem 2.1.21  $(\alpha')^2$  is a prime component, from which we may conclude that  $\alpha' \le \alpha < (\alpha')^2$ . Since  $\alpha < (\alpha')^2$ , it is easily seen that  $\alpha^n < (\alpha')^{2n}$  (by induction and the remarks on page 50). Whence (by proposition 2.1.4 (a))  $\alpha' \le \beta < (\alpha')^{2n} < (\alpha')^{\omega}$ .

2.1.23 THEOREM ([35, Th. 9, p274]): Every initial ordinal is a prime component.

**PROOF:** Let  $\phi$  be an initial ordinal. By corollary 2.1.20 there are  $n \in \mathbb{N}$  and  $\gamma < \phi'$  such that  $\phi = \phi' \cdot n + \gamma$ . So by proposition 2.1.1 (a),  $\phi' \le \phi < \phi' \cdot (n+1)$ . By theorem 2.1.21, there is an ordinal  $\mu$  such that  $\phi' = \omega^{\mu}$ . Notice that  $\mu \ge 1$  (since  $\phi \ge \omega$ ), so there is  $\delta$ , such that  $\mu = 1 + \delta$ . By proposition 2.1.4 (c),  $\omega^{1+\delta} = \omega \cdot \omega^{\delta}$  and therefore

§2.2. Derivatives and scattered spaces

$$\overline{\phi' \cdot (n+1)} = \overline{\omega^{1+\delta} \cdot (n+1)} = \overline{(n+1) \cdot \omega \cdot \omega^{\delta}} = \overline{\omega^{1+\delta}} = \overline{\phi'}.$$

It follows that  $\overline{\phi} = \overline{\phi'}$  and since  $\phi$  is initial,  $\phi' = \phi$ .

**2.1.24 COROLLARY:** If  $\omega^{\tau}$  is initial then  $\tau$  is a prime component.

**PROOF:** Let  $v < \tau$ . Then  $\omega^{v} < \omega^{\tau}$  (proposition 2.1.4 (a)), so  $\omega^{v} \cdot \omega^{\tau} = \omega^{\tau}$  (proposition 2.1.12). So by proposition 2.1.4 (c),  $v + \tau = \tau$ , hence by theorem 2.1.15  $\tau$  is a prime component.  $\Box$ 

Finally we remark that the proofs of proposition 2.1.12, corollary 2.1.13 and proposition 2.1.24 are due to us and were included because we could not find a reference.

### §2.2. Derivatives and scattered spaces

In this section we briefly discuss some properties of derivatives of sets and scatteredness. Furthermore we formulate the well-known theorems of Cantor-Bendixson and Sierpiński-Mazurkiewicz and we present some results which we need in section 2.3 and in chapter 4.

Let X be a topological space and let  $A \subset X$ . The *derived set*  $A^d$  of A in X is defined to be the set of all  $x \in X$  satisfying the condition that for every neighborhood U of x (in X),  $U \cap A \setminus \{x\} \neq \emptyset$  (i.e., the set of all accumulation points of A in X). Notice that not necessarily  $A^d \subset A$ : for example let  $X = \mathbb{R}$  and A = (0, 1). It is well-known that  $A^d \subset \overline{A}$ and that  $A^d$  is closed in X. Now for every ordinal  $\alpha$  we define  $X^{(\alpha)}$ , the  $\alpha$ -th derivative, by transfinite induction as follows:

a)  $X^{(0)} = X$ , b) if  $\alpha$  is a successor, say  $\alpha = \beta + 1$ , then  $X^{(\alpha)} = (X^{(\beta)})^d$ , c) if  $\alpha$  is a limit ordinal then  $X^{(\alpha)} = \bigcap_{\beta < \alpha} X^{(\beta)}$ .

Notice that  $X^{(1)}$  is the derived set of X in X. Furthermore we get  $X^{(1)}$  from X by "throwing away" all isolated points of X. The above is standard notation of course.

**2.2.1 REMARK:** If A is a subspace of a topological space X, then  $A^{(1)}$  is the derived set of A in A (so  $A^{(1)} \subset A$ ) and we put  $A^d$  the derived set of A in X. Since not

necessarily  $A^d \subset A$  it follows that in general  $A^{(1)} \neq A^d$ . We claim that  $A^{(1)} = A^d \cap A$ . In particular, if A is closed, then  $A^{(1)} = A^d$ .

Indeed, let  $x \in A^{(1)}$ . Since  $A^{(1)} \subset A$ ,  $x \in A$ . Let U be a neighborhood of x in X. Then  $U \cap A$  is a neighborhood of x in A. Since  $x \in A^{(1)}$ ,  $\emptyset \neq U \cap A \cap A \setminus \{x\} = U \cap A \setminus \{x\}$  and hence  $x \in A^d$ . For the reverse inclusion let  $x \in A \cap A^d$  and let U be a neighborhood of x in A. Let V be a neighborhood of x in X such that  $V \cap A = U$ . Since  $x \in A^d$ ,  $\emptyset \neq V \cap A \setminus \{x\} = U \cap A \setminus \{x\}$ , so  $x \in A^{(1)}$ .

**2.2.2 PROPOSITION:** Let X be a space. Then for every ordinal  $\alpha$  and  $\beta$  with  $\alpha \le \beta$ (a)  $X^{(\alpha)}$  is closed in X, and (b)  $X^{(\beta)} \subset X^{(\alpha)}$ .

**PROOF:** We prove (a) by transfinite induction. For  $\alpha = 0$  it is trivial. So let  $\alpha > 0$  and suppose (a) is proved for every  $\beta < \alpha$ . If  $\alpha$  is a successor, say  $\alpha = \beta + 1$ , then  $X^{(\alpha)} = (X^{(\beta)})^d$  is closed in X. If  $\alpha$  is a limit ordinal then by the inductive hypothesis,  $X^{(\alpha)} = \bigcap_{\beta < \alpha} X^{(\beta)}$  is closed in X, and (a) is proved.

We prove (b) by transfinite induction an  $\beta$ . Notice that (b) is obviously true if  $\beta = \alpha$ , so suppose that  $\beta > \alpha$  and (b) is proved for every  $\gamma$  with  $\alpha \le \gamma < \beta$ . If  $\beta = \gamma + 1$  for some ordinal  $\gamma$ , then by the inductive hypothesis  $X^{(\beta)} = (X^{(\gamma)})^d \subset \overline{X^{(\gamma)}} \subset \overline{X^{(\alpha)}}$ . Since by (a),  $X^{(\alpha)}$  is closed, we have the desired result.

If  $\beta$  is a limit ordinal, then  $X^{(\beta)} = \bigcap_{\gamma < \beta} X^{(\gamma)} \subset X^{(\alpha)}$ .

From remark 2.2.1 and proposition 2.2.2 (a) we easily get the following

**2.2.3 COROLLARY:** Let X be a space and  $\alpha$  an ordinal. Then  $X^{(\alpha+1)} = (X^{(\alpha)})^{(1)}$ .

**2.2.4 PROPOSITION:** Let X be a space and A a subspace of X. Then for each ordinal  $\alpha$ ,

(a)  $A^{(\alpha)} \subset X^{(\alpha)}$ , and

(b) if A is open then  $A^{(\alpha)} = A \cap X^{(\alpha)}$ .

**PROOF:** We prove this proposition by transfinite induction on  $\alpha$ . If  $\alpha = 0$ , the proposition is obviously true, so suppose that  $\alpha > 0$  and that the proposition is proved for every  $\beta < \alpha$ . First suppose that  $\alpha$  is a successor, say  $\alpha = \beta + 1$ .

For (a), since by the inductive hypothesis  $A^{(\beta)} \subset X^{(\beta)}$ , by [24, Th. 1.3.4 (ii)]  $A^{(\alpha)} = (A^{(\beta)})^d \subset (X^{(\beta)})^d = X^{(\alpha)}$ .

For (b) let  $x \in A \cap X^{(\alpha)}$  and let U be a neighborhood of x in A. Since A is open, U is a neighborhood of x in X, hence  $U \cap A \cap X^{(\beta)} \setminus \{x\} = U \cap X^{(\beta)} \setminus \{x\} \neq \emptyset$  (because  $x \in X^{(\alpha)}$ ). So by the inductive hypothesis  $\emptyset \neq U \cap A \cap X^{(\beta)} \setminus \{x\} \subset U \cap A^{(\beta)} \setminus \{x\}$ , hence  $x \in A^{(\alpha)}$ . The reverse inclusion follows directly from (a).

If  $\alpha$  is a limit ordinal (a) and (b) easily follow from the inductive hypothesis and the definitions of  $A^{(\alpha)}$  and  $X^{(\alpha)}$ .

**2.2.5 PROPOSITION:** Let  $\alpha$  be an ordinal and  $X = [1, \omega^{\alpha}]$ . Then  $X^{(\alpha)} = \{\omega^{\alpha}\}$ .

**PROOF**: If  $\alpha = 0$  the proposition is obviously true, so suppose that  $\alpha > 0$  and that the proposition has been proved for every  $\beta < \alpha$ . First suppose that  $\alpha$  is a successor, say  $\alpha = \beta + 1$ . If for  $i \in \mathbb{N} \cup \{0\}$ ,  $X_i = [\omega^{\beta} \cdot i + 1, \omega^{\beta} \cdot (i + 1)]$ , then

$$X = \bigcup_{i=0}^{\infty} X_i \cup \{\omega^{\alpha}\},\$$

 $X_i$  is open in X and  $X_i \approx [1, \omega^{\beta}]$ , so by the inductive hypothesis  $(X_i)^{(\beta)} = \{\omega^{\beta} \cdot (i+1)\}$ . By proposition 2.2.4 (b),  $X_i \cap X^{(\beta)} = (X_i)^{(\beta)}$ . Since  $\omega^{\alpha}$  is an accumulation point of  $\{\omega^{\beta} \cdot (i+1) : i \ge 0\}$ , we conclude that  $X^{(\beta)} = \{\omega^{\beta} \cdot (i+1) : i \ge 0\} \cup \{\omega^{\alpha}\}$  and so  $X^{(\alpha)} = \{\omega^{\alpha}\}$ .

Now suppose  $\alpha$  is a limit ordinal. Fix  $\beta < \alpha$  and let  $\beta \le \gamma < \alpha$ . Let  $A = [1, \omega^{\gamma}]$ . By the inductive hypothesis and propositions 2.2.4 (a) and 2.2.2 (a),  $\{\omega^{\gamma}\} = A^{(\gamma)} \subset X^{(\gamma)} \subset X^{(\beta)}$ . Since  $\omega^{\alpha}$  is an accumulation point of  $\{\omega^{\gamma} : \beta \le \gamma < \alpha\}$ , it follows that  $\omega^{\alpha} \in X^{(\beta)} \subset X^{(\alpha)}$ . For the reverse inclusion let  $\xi \in X \setminus \{\omega^{\alpha}\}$ . Then there is  $\beta < \alpha$  such that  $\xi < \omega^{\beta}$ . Then  $\xi \in [1, \omega^{\beta}]$ , which is open in X, so by proposition 2.2.4 (b) and the inductive hypothesis we have  $\{\omega^{\beta}\} = [1, \omega^{\beta}]^{(\beta)} = [1, \omega^{\beta}] \cap X^{(\beta)}$ . So  $\xi \notin X^{(\beta)}$  and hence  $\xi \notin X^{(\alpha)}$ .

Let A be a subspace of X. A is **dense in itself** if  $A \subset A^d$  or equivalently  $A = A^{(1)}$ . This means that A contains no isolated points. A is **scattered** if A contains no dense in itself subsets, i.e., every subset of A contains isolated points. Again this is standard terminology.

After these definitions we can state the theorem of Cantor-Bendixson (cf. [24, p85] or [47, p148]).

**2.2.6 THEOREM** (Cantor-Bendixson): Let X be a topological space. Then there exists an ordinal  $\alpha$  such that  $X^{(\alpha)} = X^{(\alpha+1)}$ . For this  $\alpha$ ,  $X^{(\alpha)}$  is closed and dense in itself and  $S = X \setminus X^{(\alpha)}$  is scattered. In particular, X is scattered if and only if  $X^{(\alpha)} = \emptyset$ .

Furthermore, if X is second countable then S is countable.

Let X be a scattered space. By theorem 2.2.6, there is an ordinal  $\alpha$  such that  $X^{(\alpha)} = \emptyset$ . Now the *scattered height*  $\kappa(X)$  of X is defined to be the smallest ordinal  $\alpha$  such that  $X^{(\alpha)} = \emptyset$ . It is easy to see that if X is compact and scattered, then  $\kappa(X)$  is a successor, say  $\alpha + 1$  and  $X^{(\alpha)}$  contains only finitely many points. If X is second countable and scattered then  $\kappa(X)$  is countable. Notice that by proposition 2.2.5,  $\kappa([1, \omega^{\alpha}]) = \alpha + 1$ .

**2.2.7 REMARK:** Every countable compact Hausdorff space is scattered. Indeed, let X be a countable compact Hausdorff space. Then X is second countable ([24, th 3.1.21]) and regular, hence X is metrizable ([24, th 4.2.9]). So since X is countable, it is also zero-dimensional. Now suppose X is not scattered. Then by theorem 2.2.6 there is  $P \subset X$  closed, non-empty and dense in itself. Then P is separable metric zero-dimensional and compact without isolated points. Thus  $X \approx C$  [15], so P is uncountable, which is a contradiction.

We can now formulate the theorem of Sierpiński-Mazurkiewicz (cf. [47, p155] or [36]).

**2.2.8 THEOREM** (Sierpiński-Mazurkiewicz): Let X be a countable compact Hausdorff space. If  $\kappa(X) = \alpha + 1$  and  $X^{(\alpha)}$  contains m points (m finite), then  $X \approx [1, \omega^{\alpha} \cdot m]$ .

Notice that by proposition 2.2.5 it easily follows that if  $X = [1, \omega^{\alpha} \cdot m]$ , with  $\alpha$  countable and  $m \in \mathbb{N}$ , then  $\kappa(X) = \alpha + 1$  and  $X^{(\alpha)} = \{\omega^{\alpha} \cdot 1, \dots, \omega^{\alpha} \cdot m\}$ .

Now we will prove some simple results, which we will need in section 2.3. Let X be a topological space and A a nonempty closed subset of X. Let X/A be the quotient space obtained from X by identifying A to a single point, say  $\infty$  and let  $p: X \to X/A$  be the quotient map. Notice that p is closed.

The next lemma gives some results on the derivatives of X/A in terms of the derivatives of X, if A is of a special form.

**2.2.9 LEMMA:** Let X be a space,  $\alpha$  an ordinal such that  $A = X^{(\alpha)} \neq \emptyset$ , and Y = X/A. Then

(a) for every  $\beta \le \alpha$ , and  $p(X^{(\beta)}) = Y^{(\beta)}$ , and (b)  $Y^{(\alpha)} = \{\infty\}$ . §2.2. Derivatives and scattered spaces

**PROOF:** We first prove (a). Notice that by propositions 2.2.4 (b) and 2.2.2 (b),

 $X^{(\beta)} = (X^{(\beta)} \cap (X \setminus A)) \cup (X^{(\beta)} \cap A) = (X \setminus A)^{(\beta)} \cup A,$ 

for every  $\beta \le \alpha$ . In the same way  $Y^{(\beta)} = (Y \setminus \{\infty\})^{(\beta)} \cup (Y^{(\beta)} \cap \{\infty\})$ . Since  $p \mid X \setminus A : X \setminus A \to Y \setminus \{\infty\}$  is a homeomorphism,  $p((X \setminus A)^{(\beta)}) = (Y \setminus \{\infty\})^{(\beta)}$  for every  $\beta \le \alpha$ .

We prove (a) by induction on  $\beta$ . For  $\beta = 0$  this is a triviality, so let  $0 < \beta \le \alpha$  and assume it is true for every  $\gamma < \beta$ .

*Case 1*:  $\beta$  is a successor, say  $\beta = \gamma + 1$ .

Suppose  $\infty \notin Y^{(\beta)}$ . By the inductive hypothesis and since  $A \subset X^{(\gamma)}, \infty \in Y^{(\gamma)}$ , thus  $\{\infty\}$  is open in  $Y^{(\gamma)}$ . But then A is open in  $X^{(\gamma)}$ , so by proposition 2.2.4 (b) and corollary 2.2.3

$$A^{(1)} = A \cap (X^{(\gamma)})^{(1)} = A \cap X^{(\beta)} = A,$$

which gives a contradiction. Hence  $\infty \in Y^{(\beta)}$ , so by the above remarks

 $p(X^{(\beta)}) = p((X \setminus A)^{(\beta)} \cup A)$ =  $(Y \setminus \{\infty\})^{(\beta)} \cup \{\infty\}$ =  $(Y \setminus \{\infty\})^{(\beta)} \cup (Y^{(\beta)} \cap \{\infty\})$ =  $Y^{(\beta)}$ .

Case 2:  $\beta$  is a limit ordinal.

Then

$$Y^{(\beta)} = \bigcap_{\gamma < \beta} Y^{(\gamma)} = \bigcap_{\gamma < \beta} p(X^{(\gamma)}) = p(\bigcap_{\gamma < \beta} X^{(\gamma)}) = p(X^{(\beta)}).$$

This finishes the proof of (a).

By (a) we have  $Y^{(\alpha)} = p(X^{(\alpha)}) = p(A) = \{\infty\}$ , which proves (b).

With this lemma we can give a classification of X/A, for X a countable compact space.

**2.2.10 COROLLARY:** Let X be a countable compact space and let  $A = X^{(\alpha)}$  for some  $\alpha < \kappa(X)$ . Then  $X/A \approx [1, \omega^{\alpha}]$ . In other words, if  $X = [1, \omega^{\alpha} \cdot n]$  for certain  $n \in \mathbb{N}$  and  $\alpha < \omega_1$  (so  $A = X^{(\alpha)} = \{\omega^{\alpha} \cdot 1, ..., \omega^{\alpha} \cdot n\}$ ) then  $X/A \approx [1, \omega^{\alpha}]$ .

**PROOF:** The first part follows from theorem 2.2.8 and lemma 2.2.9 (b). From proposition 2.2.5 it follows that if  $X = [1, \omega^{\alpha} \cdot n]$ , then  $X^{(\alpha)} = \{\omega^{\alpha} \cdot 1, ..., \omega^{\alpha} \cdot n\}$ , which proves the second part.  $\Box$ 

We finish this section with the remark that the second statement of corollary 2.2.10 is not true if we take  $\alpha \ge \omega_1$ . For example take  $X = [1, \omega^{\omega_1} \cdot 2]$ , i.e.,  $\alpha = \omega_1$ . We first show that  $\omega^{\omega_1} = \omega_1$ . By proposition 2.1.4 (b),  $\omega_1 \le \omega^{\omega_1}$ . Since  $\omega_1$  is a prime component (theorem 2.1.23) there is  $\mu$  such that  $\omega_1 = \omega^{\mu}$ . If  $\omega_1 < \omega^{\omega_1}$  then by proposition 2.1.4 (a),  $\mu < \omega_1$ , hence  $\omega^{\mu} < \omega_1$  and we arrived at a contradiction.

Now put  $A = X^{(\alpha)} = \{\omega_1, \omega_1 \cdot 2\}$ , We prove that X/A and  $[1, \omega^{\omega_1}] = [1, \omega_1]$  are not homeomorphic. To this end, notice that  $(X/A) \setminus \{\infty\}$  contains two disjoint closed subsets *E* and *F* (namely  $E = p([1, \omega_1))$  and  $F = p([\omega_1 + 1, \omega_1 \cdot 2))$ ), such that the closures  $\overline{E}$ and  $\overline{F}$  in X/A have non-empty intersection. In  $[1, \omega_1]$  for every pair *E* and *F* of disjoint closed subsets of  $[1, \omega_1)$ , the closures  $\overline{E}$  and  $\overline{F}$  in  $[1, \omega_1]$  are disjoint ([24, Ex. 3.1.27]). Hence X/A and  $[1, \omega_1]$  are not homeomorphic (see also [8, Ex. 1]).

### §2.3. Factorizing function spaces

In this section we prove some results, which will be important tools later on. First we fix some notation and give some definitions. Let X be a space and  $A \subset X$  closed. By  $C_{p,A}(X)$  we denote the subspace of  $C_p(X)$  of all functions vanishing on A. Whenever  $A = \{a\}$  for some  $a \in A$ ,  $C_{p, \{a\}}(X)$  will be denoted by  $C_{p, a}(X)$ , so  $C_{p, \infty}(X/A)$  is the subspace of  $C_p(X/A)$  of all functions vanishing at  $\infty$ . For this kind of subspaces of  $C_p^*(X)$  we use a similar notation. Furthermore, let  $\{X_t : t \in T\}$  and  $\{Y_s : s \in S\}$  be two families of spaces. For each  $t \in T$  and  $s \in S$  let  $E_t$  be a linear subspace of  $C_p(X_t)$  and let  $F_s$  be a linear subspace of  $C_p(Y_s)$  and let  $k \in \mathbb{N}$ . We call a linear function  $\phi: \prod_{t \in T} E_t \to \prod_{s \in S} F_s \text{ a linear } k\text{-mapping if for all } (f_t)_{t \in T} \in \prod_{t \in T} E_t \text{ with}$  $f_t(X_t) \subset (-\frac{1}{k}, \frac{1}{k})$  for every  $t \in T$  we have  $(\pi_s \circ \phi)((f_t)_{t \in T})(Y_s) \subset (-1, 1)$  for every  $s \in S$ . We define  $\phi$  to be a *linear k-homeomorphism* whenever  $\phi$  is a homeomorphism such that both  $\phi$  and  $\phi^{-1}$  are linear k-mappings. Whenever there is a linear khomeomorphism between  $\Pi_{t \in T} E_t$  and  $\Pi_{s \in S} F_s$  we write  $\Pi_{t \in T} E_t \stackrel{k}{\sim} \Pi_{s \in S} F_s$ . Notice that the composition of a linear k-homeomorphism and a linear l-homeomorphism is a linear kl-homeomorphism. The definition of linear k-mapping and linear khomeomorphism can be found in [3] and can also be given for spaces of bounded continuous functions.

We now prove the following well-known theorem which will be used in the proof of proposition 2.3.2 and which will also be useful in chapter 4.

**2.3.1 THEOREM** (Dugundji [22]): Let X be a metric space and A a closed subspace of X. Then there is a continuous linear function  $\phi: C_p(A) \to C_p(X)$  such that for each  $f \in C(A)$ ,  $\phi(f) | A = f$  and  $\phi(f)(X) \subset \operatorname{conv}(f(A))$ .

**PROOF:** First suppose  $X \setminus A$  contains more than one point. Then for every  $x \in X \setminus A$ , there is a neighborhood  $V_x$  of x such that diam  $V_x < 1/2d(x, A)$  and  $V_x \neq X \setminus A$ . Let  $\mathcal{U}$  be a locally finite open refinement of the covering  $\{V_x : x \in X \setminus A\}$  of  $X \setminus A$ . Notice that by construction  $(X \setminus A) \setminus U \neq \emptyset$ .

CLAIM: If  $a \in A$  and V is a neighborhood of a, then there exists a neighborhood W of a such that if  $U \cap W \neq \emptyset$  for some  $U \in \mathcal{U}$ , then  $U \subset V$ .

Let  $\varepsilon = d(a, X \setminus V)$  and let  $W = B(a, \varepsilon/2)$ . Suppose that  $U \cap W \neq \emptyset$  for some  $U \in \mathcal{U}$ , say  $z \in U \cap W$ . Choose  $x \in X \setminus A$  such that  $U \subset V_x$  and let  $y \in U$ . Now

$$d(x, a) \le d(x, z) + d(z, a) < \frac{1}{2}d(x, A) + d(z, a) \le \frac{1}{2}d(x, a) + d(z, a),$$

hence d(x, a) < 2d(z, a). This implies

$$d(y, a) \le d(y, z) + d(z, a) < \frac{1}{2}d(x, a) + d(z, a) < 2d(z, a) < \varepsilon,$$

hence  $y \in V$ . This proves the claim.

For each  $U \in \mathcal{U}$ , define  $\lambda_U : X \setminus A \to \mathbb{R}$  by

$$\lambda_U(x) = \frac{d(x, (X \setminus A) \setminus U)}{\sum_{V \in \mathcal{U}} d(x, (X \setminus A) \setminus V)}$$

First notice that  $d(x, (X \setminus A) \setminus U)$  is defined for every  $U \in \mathcal{U}$ , because  $(X \setminus A) \setminus U \neq \emptyset$ . Second for each  $x \in X \setminus A$ , there is a neighborhood W of x which intersects only finitely many elements of  $\mathcal{U}$ . Hence if we restrict  $\lambda_U$  to W, the sum in the denominator is finite, and since  $\mathcal{U}$  covers  $X \setminus A$ , this sum is non-zero. We conclude that  $\lambda_U$  is a well-defined continuous function. Notice that  $\sum_{U \in \mathcal{U}} \lambda_U \equiv 1$ .

For each  $U \in \mathcal{U}$ , let  $x_U \in U$  and  $a_U \in A$  be such that  $d(x_U, a_U) < 2d(x_U, A)$ . Let  $f \in C(A)$ . Define  $\tilde{f}: X \to \mathbb{R}$  by

$$\widetilde{f}(x) = \begin{cases} \sum_{U \in \mathcal{U}} \lambda_U(x) f(a_U) & \text{if } x \in X \setminus A \\ f(x) & \text{if } x \in A \end{cases}$$

By similar arguments as above it is easily seen that  $\tilde{f}$  is well-defined and that  $\tilde{f} | X \setminus A$  is continuous. So continuity of  $\tilde{f}$  need only be verified at points of A. Let  $x \in A$  and  $\varepsilon > 0$ . There is  $\delta > 0$  such that for  $y \in A$  with  $d(x, y) < \delta$ , we have  $|f(x) - f(y)| < \varepsilon$ . By the claim, there is a neighborhood  $W \subset B(x, \delta)$  of x such that if  $U \cap W \neq \emptyset$  for some  $U \in \mathcal{U}$ , then  $U \subset B(x, \delta/3)$ . We claim that for  $y \in W$ ,  $|\tilde{f}(x) - \tilde{f}(y)| < \varepsilon$ . For  $y \in A$ , this is clear. So let  $y \in W \cap X \setminus A$ . Find  $U_1, \ldots, U_n \in \mathcal{U}$  such that for  $U \in \mathcal{U}$  we have  $y \in U$  if and only if  $U \in \{U_1, \ldots, U_n\}$ . Then  $\tilde{f}(y) = \sum_{i=1}^n \lambda_{U_i}(y) f(a_{U_i})$ . For  $i \leq n$ ,  $U_i \cap W \neq \emptyset$ , hence  $d(x_{U_i}, x) < \tilde{o}/3$ . This implies

$$d(x, a_{U_i}) \le d(x, x_{U_i}) + d(x_{U_i}, a_{U_i})$$
  
$$\le d(x, x_{U_i}) + 2d(x_{U_i}, A)$$
  
$$\le 3d(x, x_{U_i}) < \delta.$$

Hence  $|f(x) - f(a_{U_i})| < \varepsilon$ , so

$$\begin{split} |\widetilde{f}(x) - \widetilde{f}(y)| &= |f(x) - \sum_{i=1}^{n} \lambda_{U_i}(y) f(a_{U_i})| \\ &= |\sum_{i=1}^{n} \lambda_{U_i}(y) (f(x) - f(a_{U_i}))| \\ &\leq \sum_{i=1}^{n} \lambda_{U_i}(y) |f(x) - f(a_{U_i}))| \\ &< \sum_{i=1}^{n} \lambda_{U_i}(y) \cdot \varepsilon = \varepsilon. \end{split}$$

We conclude that  $\tilde{f}$  is continuous. Obviously  $\tilde{f}(X) \subset \operatorname{conv}(f(A))$  and  $\tilde{f}|A = f$ .

Define  $\phi: C_p(A) \to C_p(X)$  by  $\phi(f) = \tilde{f}$ . By the above we have that  $\phi$  is a well-defined function with the property that for each  $f \in C(A)$ ,  $\phi(f) | A = f$  and  $\phi(f)(X) \subset \operatorname{conv} (f(A))$ . The linearity of  $\phi$  is a triviality. To prove that  $\phi$  is continuous, it suffices to prove continuity at 0. Let  $P \subset X$  be finite and  $\varepsilon > 0$ . Let  $Q = (P \cap A) \cup \{a_U : U \in \mathcal{U}, U \cap P \neq \emptyset\}$ . Then Q is a finite subset of A. It is easily seen that  $\phi(<0, Q, \varepsilon >) \subset <0, P, \varepsilon >$ .

Now assume  $X \setminus A$  is empty or contains only one point. If it is empty, the theorem is obvious, so suppose  $X \setminus A$  contains only one point  $x_0$ . Since A is closed,  $x_0$  is isolated in X. Fix  $x_1 \in A$  and define  $\phi: C_p(A) \to C_p(X)$  by  $\phi(f)(x) = f(x)$  if  $x \neq x_0$  and  $\phi(f)(x_0) = f(x_1)$ . Then  $\phi$  is a well-defined linear function. In addition  $\phi(f) \mid A = f$  and  $\phi(f)(X) \subset \operatorname{conv} (f(A))$ . We prove that  $\phi$  is continuous. Take  $P \subset X$  finite,  $\varepsilon > 0$  and  $f \in C(A)$ . If we let  $Q = (P \cap A) \cup \{x_1\}$ , then  $\phi(< f, Q, \varepsilon >) \subset \langle \phi(f), P, \varepsilon >. \Box$ 

We now come to the important

§2.3. Factorizing function spaces

**2.3.2 PROPOSITION:** Let X be a metric space and let A be a closed subset of X. Then  $C_p(X) \stackrel{2}{\sim} C_{p,A}(X) \times C_p(A)$ .

**PROOF:** Define  $\rho: C_p(X) \to C_p(A)$  by  $\rho(f) = f \mid A$ . Notice that  $\rho$  is a continuous linear function. Because X is metric, by theorem 2.3.1 there is a continuous linear function  $\xi: C_p(A) \to C_p(X)$  such that for each  $f \in C_p(A)$ ,  $\xi(f) \mid A = f$  and  $\xi(f)(X) \subset \operatorname{conv}(f(A))$ . Notice that  $\rho \circ \xi = id_{C_p(A)}$ .

Now define  $\phi: C_p(X) \to C_{p,A}(X) \times C_p(A)$  by

 $\phi(f) = (f - (\xi \circ \rho)(f), \rho(f)).$ 

We have to prove that  $\phi$  is well-defined. Take an arbitrary  $f \in C_p(X)$ . It is obvious that  $\rho(f) \in C_p(A)$  and that  $f - (\xi \circ \rho)(f) \in C_p(X)$ . Furthermore

$$(f - (\xi \circ \rho)(f)) | A = \rho(f - (\xi \circ \rho)(f)) = \rho(f) - (\rho \circ \xi \circ \rho)(f)) = \rho(f) - \rho(f) \equiv 0,$$

so  $f - (\xi \circ \rho)(f) \in C_{p,A}(X)$ .

That  $\phi$  is continuous and linear is a triviality. We show that  $\phi$  is a linear homeomorphism. For that define  $\psi: C_{p,A}(X) \times C_p(A) \to C_p(X)$  by

 $\psi(f, g) = f + \xi(g).$ 

It is trivial that  $\psi$  is well-defined, continuous and linear. Furthermore, as is easily seen,  $\psi \circ \phi = id_{C_p(X)}$ . We show that  $\phi \circ \psi = id_{C_{p,A}(X) \times C_p(A)}$ . Take  $f \in C_{p,A}(X)$  and  $g \in C_p(A)$ . Notice that  $\rho(f) = f \mid A \equiv 0$ , hence by linearity of  $\xi$ ,  $(\xi \circ \rho)(f) = \xi(0) \equiv 0$ . So

$$\begin{aligned} (\phi \circ \psi)(f, g) &= \phi(f + \xi(g)) \\ &= (f + \xi(g) - (\xi \circ \rho)(f + \xi(g)), \, \rho(f + \xi(g))) \\ &= (f + \xi(g) - (\xi \circ \rho)(f) - (\xi \circ \rho \circ \xi)(g), \, \rho(f) + (\rho \circ \xi)(g)) \\ &= (f + \xi(g) - 0 - \xi(g), \, 0 + g) \\ &= (f, g), \end{aligned}$$

i.e.,  $\phi$  is a linear homeomorphism.

The only thing left to prove is that  $\phi$  and  $\psi$  are linear 2-mappings. We first prove it for  $\phi$ . Let  $f \in C_p(X)$  with |f(x)| < 1/2 for every  $x \in X$ . Then  $\rho(f)(A) \subset (-1/2, 1/2)$  and hence  $(\xi \circ \rho)(f)(X) \subset \operatorname{conv} \rho(f)(A) \subset (-1/2, 1/2)$ . Let  $x \in X$ . Then

$$|\pi_1 \circ \phi(f)(x)| = |f(x) - (\xi \circ \rho)(f)(x)| \le |f(x)| + |(\xi \circ \rho)(f)(x)| < 1,$$

and for  $a \in A$ ,

 $|\pi_2 \circ \phi(f)(a)| = |\rho(f)(a)| = |f(a)| < 1/2,$ 

so  $\phi$  is a linear 2-mapping. Now take  $f \in C_{p,A}(X)$  and  $g \in C_p(A)$  such that  $f(X) \subset (-1/2, 1/2)$  and  $f(A) \subset (-1/2, 1/2)$ . Then  $\xi(g)(X) \subset \operatorname{conv}(g(A)) \subset (-1/2, 1/2)$ , so for  $x \in X$ ,

$$|\psi(f, g)(x)| = |f(x) + \xi(g)(x)| \le |f(x)| + |\xi(g)(x)| < 1/2 + 1/2 = 1$$

hence  $\psi$  is a linear 2-mapping. This completes the proof of the proposition.  $\Box$ 

**2.3.3 LEMMA:** Let X be a space and let A be a closed subset of X. Then

$$C_{p,A}(X) \stackrel{i}{\sim} C_{p,\infty}(X/A).$$

**PROOF:** For every function  $f \in C_{p,A}(X)$  there is a unique function  $\tilde{f} \in C_{p,\infty}(X/A)$ such that  $\tilde{f} \circ p = f$  [24, p124]. If we now define  $\phi: C_{p,A}(X) \to C_{p,\infty}(X/A)$  by  $\phi(f) = \tilde{f}$ , then  $\phi$  is a well-defined linear bijection. Since for  $f \in C_{p,A}(X)$ ,  $y_1, \dots, y_n \in X/A$ ,  $\varepsilon > 0$ and  $x_i \in p^{-1}(y_i)$  ( $i \le n$ ) it is easily seen that

 $\phi < f, \{x_1, ..., x_n\}, \varepsilon > = < \phi(f), \{y_1, ..., y_n\}, \varepsilon >,$ 

it follows that  $\phi$  is a linear homeomorphism. That  $\phi$  is a linear 1-homeomorphism is a triviality.  $\Box$ 

From the last lemma and proposition we have the useful

**2.3.4 COROLLARY:** Let X be a metric space and let A be a closed subset of X. Then  $C_p(X) \stackrel{2}{\sim} C_{p,\infty}(X/A) \times C_p(A)$ .

The next three lemmas are used often in this monograph. The proofs are easy and left to the reader.

**2.3.5 LEMMA:** If X and Y are homeomorphic spaces, then  $C_{\nu}(X) \stackrel{1}{\sim} C_{\rho}(Y)$ .

**2.3.6 LEMMA:** If X and Y are spaces and A is a subspace of X, then  $C_{p,A}(X) \times C_p(Y) \stackrel{1}{\sim} C_{p,A}(X \oplus Y)$ .

Notice that all the given facts so far are also valid for spaces of bounded continuous functions. In lemma 2.3.7, this is only the case for the second statement as is shown in

60

section 4.6 (cf. example 4.6.6).

**2.3.7 LEMMA:** If  $X = \bigoplus_{i=1}^{\infty} X_i$  and  $Y = \bigoplus_{i=1}^{\infty} Y_i$  such that for every  $i \in \mathbb{N}$ ,  $C_p(X_i) \sim C_p(Y_i)$ , then  $C_p(X) \sim C_p(Y)$ . Moreover, if for every  $i \in \mathbb{N}$   $C_p(X_i) \stackrel{k}{\sim} C_p(Y_i)$ , then  $C_p(X) \stackrel{k}{\sim} C_p(Y)$ .

We now prove some properties of function spaces of ordinals. We use the following notation. For an ordinal  $\alpha$  we denote by  $C_{p,0}([1, \alpha])$  the subspace of  $C_p([1, \alpha])$  of all continuous functions vanishing at  $\alpha$  (i.e.,  $C_{p,0}([1, \alpha]) = C_{p,\alpha}([1, \alpha])$ ).

**2.3.8 LEMMA:** Let  $\alpha \ge 1$  and  $\beta \ge 1$  be ordinals. Then

$$C_p([1, \alpha + \beta]) \stackrel{!}{\sim} C_p([1, \alpha]) \times C_p([1, \beta]) \stackrel{!}{\sim} C_p([1, \beta]) \times C_p([1, \alpha]) \stackrel{!}{\sim} C_p([1, \beta + \alpha])$$

 $C_{p,0}([1, \alpha + \beta]) \stackrel{1}{\sim} C_p([1, \alpha]) \times C_{p,0}([1, \beta]).$ 

**PROOF:** Since  $[1, \alpha + \beta] \approx [1, \alpha] \oplus [1, \beta]$  (notice that  $h: [1, \alpha + \beta] \rightarrow [1, \alpha] \oplus [1, \beta]$  defined by  $h(\gamma) = \gamma$  if  $\gamma \le \alpha$  and  $h(\gamma) = \gamma - \alpha$  if  $\gamma > \alpha$ , is a homeomorphism) we have by lemmas 2.3.5 and 2.3.6

$$C_p([1, \alpha + \beta]) \stackrel{1}{\sim} C_p([1, \alpha] \oplus [1, \beta]) \stackrel{1}{\sim} C_p([1, \alpha]) \times C_p([1, \beta])$$

and

and

 $C_{p,0}([1, \alpha + \beta]) \stackrel{1}{\sim} C_p([1, \alpha]) \times C_{p,0}([1, \beta]). \square$ 

**2.3.9 LEMMA:** Let  $\alpha \ge \omega$  be an ordinal. Then  $C_p([1, \alpha]) \stackrel{2}{\sim} C_{p, 0}([1, \alpha])$ .

**PROOF:** Define  $\phi: C_p([1, \alpha]) \to C_{p, 0}([1, \alpha])$  by  $\phi(f)(\beta) = f(\beta - 1) - f(\alpha)$  if  $1 < \beta \le \alpha$ and  $\phi(f)(1) = f(\alpha)$ . Since  $\beta - 1 = \beta$  for  $\beta \ge \omega$ , it easily follows that  $\phi$  is well-defined. That  $\phi$  is linear is a triviality. Now take  $P \subset [1, \alpha]$  finite,  $\varepsilon > 0$  and  $f \in C_p([1, \alpha])$ . Let  $Q = \{\beta - 1: \beta \in P \setminus \{1\}\} \cup \{\alpha\}$ . It is easily seen that  $\phi(< f, Q, \varepsilon/2 >) \subset <\phi(f), P, \varepsilon >$ , hence  $\phi$  is continuous.

Now define  $\psi: C_{p,0}([1, \alpha]) \to C_p([1, \alpha])$  by  $\psi(f)(\beta) = f(1+\beta) + f(1)$ . An easy verification shows that  $\psi$  is a well-defined continuous linear function.

We are done if we prove that  $\psi = \phi^{-1}$ . Let  $f \in C_{p, 0}([1, \alpha])$  and  $\beta \in [1, \alpha]$ . Notice that  $1 + (\beta - 1) = \beta$ , so if  $\beta \neq 1$  then

$$(\phi \circ \psi)(f)(\beta) = \psi(f)(\beta - 1) - \psi(f)(\alpha)$$
  
= f (1 + (\beta - 1)) + f (1) - f (1 + \alpha) - f (1)  
= f (\beta).

Furthermore,

$$(\phi \circ \psi)(f)(1) = \psi(f)(\alpha) = f(1+\alpha) + f(1) = f(1),$$

which implies that  $\phi \circ \xi = id_{C_{p,0}([1, \alpha])}$ . Now let  $f \in C_p([1, \alpha])$  and  $\beta \in [1, \alpha]$ . Notice that  $(1+\beta)-1=\beta$ , so

$$(\psi \circ \phi)(f)(\beta) = \phi(f)(1+\beta) + \phi(f)(1)$$
$$= f((1+\beta)-1) - f(\alpha) + f(\alpha)$$
$$= f(\beta),$$

which proves that  $\psi \circ \phi = id_{C_p([1, \alpha])}$  and the lemma is proved.  $\Box$ 

Notice that  $[1, \alpha]$  is a metric space if  $\alpha < \omega_1$ , so in that case lemma 2.3.9 is an easy consequence of proposition 2.3.2 and lemma 2.3.8.

**2.3.10 REMARK:** All results stated in this section, are also valid for function spaces endowed with the compact-open topology, with the exception of lemma 2.3.3 and corollary 2.3.4. They are true for function spaces endowed with the compact-open topology under the additional assumption that A is compact.

#### §2.4. Separable metric zero-dimensional compact spaces

In [10] Bessaga and Pelczyński presented the following isomorphical classification of the spaces  $C_0(X)$ , for separable metric zero-dimensional compact spaces:

**2.4.1 THEOREM** (Bessaga and Pelczyński): Let X and Y be separable metric zero-dimensional compact spaces. Then  $C_0(X) \sim C_0(Y)$  if and only if one of the following holds:

- (a) X and Y are finite and have the same number of elements.
- (b) There are countable infinite ordinals  $\alpha$  and  $\beta$  such that  $X \approx [1, \alpha], Y \approx [1, \beta]$ and  $\max(\alpha, \beta) < [\min(\alpha, \beta)]^{\omega}$ .

§2.4. Separable metric zero-dimensional compact spaces

(c) X and Y are uncountable.

Notice that for a compact space X, we always have that X is finite, or is uncountable or is homeomorphic to  $[1, \alpha]$  for some countable infinite ordinal  $\alpha$  (by theorem 2.2.8). Also, case (c) is a direct consequence of Miljutin's theorem ([47, page 379]). Bessaga and Pelczyński's proof of (c) is different, because they were not aware of Miljutin's result (see [47, page 380]).

In this section we prove that a similar classification can be derived if we replace  $C_0(X)$  by  $C_p(X)$ . We first need to prove some properties of function spaces of ordinals.

**2.4.2 LEMMA:** Let  $\omega \le \alpha < \omega_1$  be a prime component and  $n \in \mathbb{N}$ . Then  $C_p([1, \alpha:n]) \sim C_p([1, \alpha])$ .

**PROOF:** By theorem 2.1.21 there is an ordinal  $\mu$  such that  $\alpha = \omega^{\mu}$ , so

$$C_{p}([1, \alpha \cdot n]) \sim C_{p}(\{\alpha \cdot 1, ..., \alpha \cdot n\}) \times C_{p, 0}([1, \alpha]) \quad \text{corollaries 2.2.10 and 2.3.4}$$
$$\sim C_{p, 0}([1, \alpha]) \quad \text{lemma 2.3.8}$$
$$\sim C_{p}([1, \alpha]) \quad \text{lemma 2.3.9.} \square$$

It is essential in this lemma that  $\alpha < \omega_1$  (cf. the remark after corollary 2.2.10). In section 2.5 we will show that  $C_p([1, \omega_1 \cdot 2])$  and  $C_p([1, \omega_1])$  are not linearly homeomorphic (cf. theorem 2.5.13).

**2.4.3 LEMMA:** Let  $\omega \le \alpha < \omega_1$  be an ordinal. Then  $C_p([1, \alpha]) \sim C_p([1, \alpha'])$ .

**PROOF:** By corollary 2.1.20,  $\alpha = \alpha' \cdot n + \gamma$  for some  $n \in \mathbb{N}$  and  $\gamma < \alpha'$ . By theorem 2.1.15  $\gamma + \alpha' = \alpha'$ , which implies that  $\gamma + \alpha' \cdot n = \gamma + \alpha' + \alpha' \cdot (n-1) = \alpha' \cdot n$ . So

 $C_p([1, \alpha]) = C_p([1, \alpha' \cdot n + \gamma])$   $\sim C_p([1, \gamma + \alpha' \cdot n]) \quad \text{lemma } 2.3.8 \text{ for } \gamma \neq 0$   $= C_p([1, \alpha' \cdot n])$  $\sim C_p([1, \alpha']) \quad \text{lemma } 2.4.2. \square$ 

We now come to the following result:

**2.4.4 PROPOSITION:** Let  $\omega \le \alpha < \omega_1$  be an ordinal and let  $\alpha \le \beta < \alpha^{\omega}$ . Then  $C_p([1, \alpha]) \sim C_p([1, \beta])$ .

**PROOF:** By lemma 2.1.22 and lemma 2.4.3 we may assume that  $\alpha$  and  $\beta$  are prime components. By theorem 2.1.21 there are ordinals  $\mu$  and  $\nu$  such that  $\alpha = \omega^{\mu}$  and  $\beta = \omega^{\nu}$ . Since  $\alpha \leq \beta < \alpha^{\omega}$ , by proposition 2.1.10a,  $\mu \leq \nu < \mu \cdot \omega$ .

We prove the lemma by transfinite induction on v. If  $v = \mu$  it is a triviality, so let  $v > \mu$  and suppose the lemma is true for every ordinal  $\gamma$  such that  $\mu \le \gamma < v$ .

Let  $X = [1, \beta] = [1, \omega^{\nu}]$  and  $A = X^{(\mu)}$ . By proposition 2.2.5  $\kappa(X) = \nu + 1$ , so  $\mu < \kappa(X)$ . Hence by corollary 2.2.10  $X/A \approx [1, \omega^{\mu}] = [1, \alpha]$ .

CLAIM: There are ordinals  $1 \le \gamma < \nu$  and  $n \in \mathbb{N}$  such that  $A \approx [1, \omega^{\gamma} \cdot n]$ .

Indeed, since  $\mu < \nu < \mu \cdot \omega$ , there is  $k \in \mathbb{N} \setminus \{1\}$  such that  $\mu \cdot (k-1) < \nu \le \mu \cdot k$ . So by proposition 2.2.5 and proposition 2.2.4 (a),

$$A^{(\mu \cdot (k-1))} = (X^{(\mu)})^{(\mu \cdot (k-1))} = X^{(\mu \cdot k)} \subset [1, \omega^{\mu \cdot k}]^{(\mu \cdot k)} = \{\omega^{\mu \cdot k}\},\$$

and

$$A^{(1)} = X^{(\mu+1)} \supset [1, \omega^{\mu+1}]^{(\mu+1)} \neq \emptyset$$

hence  $2 \le \kappa(A) \le \mu \cdot (k-1) + 1$ . Since  $\kappa(A)$  is a successor, there is  $1 \le \gamma \le \mu \cdot (k-1) < \nu$  such that  $\kappa(A) = \gamma + 1$ . So by theorem 2.2.8 there is  $n \in \mathbb{N}$  such that  $A \approx [1, \omega^{\gamma} \cdot n]$ , which proves the claim.

By corollary 2.3.4, lemma 2.3.9 and the claim it follows that

$$C_p([1, \beta]) \sim C_p([1, \omega^{\gamma} \cdot n]) \times C_p([1, \alpha])$$
  
 
$$\sim C_p([1, \omega^{\gamma}]) \times C_p([1, \alpha]) \qquad (\text{since } \gamma \ge 1 \text{ and by lemma } 2.4.3).$$

If  $\gamma < \mu$  then by lemma 2.3.8 and theorem 2.1.15

$$C_p([1,\beta]) \sim C_p([1,\omega^{\gamma}+\alpha]) = C_p([1,\alpha]).$$

If  $\gamma \ge \mu$  then by the inductive hypothesis  $C_p([1, \omega^{\gamma}]) \sim C_p([1, \alpha])$ , so by lemma 2.3.8 and lemma 2.4.2

$$C_p([1,\beta]) \sim C_p([1,\alpha]) \times C_p([1,\alpha]) \sim C_p([1,\alpha\cdot 2 \sim C_p([1,\alpha]). \square$$

We can now easily derive the following:

**2.4.5 COROLLARY:** Let  $\omega \le \alpha \le \beta < \omega_1$  be ordinals. Then

 $C_p([1, \alpha]) \sim C_p([1, \beta])$  if and only if  $\beta < \alpha^{\omega}$ .

(In particular if  $\alpha = \omega^{\mu}$  and  $\beta = \omega^{\nu}$  with  $\mu \leq \nu$ , then  $C_p([1, \alpha]) \sim C_p([1, \beta])$  if and only if  $\nu < \mu \cdot \omega$ ).

**PROOF:** If  $\beta < \alpha^{\omega}$  then apply proposition 2.4.4. Suppose  $C_p([1, \alpha]) \sim C_p([1, \beta])$ . By corollary 1.2.21, it follows that  $C_0([1, \alpha]) \sim C_0([1, \beta])$ . By theorem 2.4.1 this implies  $\beta < \alpha^{\omega}$ .

**2.4.6 REMARK:** If X is a separable metric compact space and A is a closed subset of X, then X/A is a separable metric compact space. This follows from the fact that the quotient map  $p: X \to X/A$  is perfect because X is compact.

We are now able to prove the classification we mentioned at the beginning of this section.

**2.4.7 THEOREM:** Let X and Y be separable metric zero-dimensional compact spaces. Then  $C_p(X) \sim C_p(Y)$  if and only if one of the following holds:

- (a) X and Y are finite and have the same number of elements.
- (b) There are countable infinite ordinals  $\alpha$  and  $\beta$  such that  $X \approx [1, \alpha], Y \approx [1, \beta]$ and max  $(\alpha, \beta) < [min (\alpha, \beta)]^{\omega}$ .
- (c) X and Y are uncountable.

**PROOF:** If  $C_p(X) \sim C_p(Y)$  then by corollary 1.2.21 we have  $C_0(X) \sim C_0(Y)$ . So by theorem 2.4.1, (a), (b) or (c) holds.

Now suppose that (a), (b) or (c) holds.

Case 1: (a) holds.

Suppose X and Y both contain m points. Then  $C_p(X) \sim \mathbb{R}^m \sim C_p(Y)$ . Case 2: (b) holds.

By corollary 2.4.5 we have the desired equivalence.

Case 3: (c) holds.

It is enough to prove that for every uncountable separable metric zero-dimensional compact space X we have  $C_p(X) \sim C_p(C)$  where C is the Cantor discontinuum. Let X be such a space.

By the Cantor-Bendixson Theorem (theorem 2.2.6) and the fact that X is second countable,  $X = D \cup S$  with D closed and dense in itself and S countable. Since X is uncountable, D is non-empty, so by the fact that C is the unique non-empty separable metric zero-dimensional compact space without isolated points ([15]), we have  $D \approx C$ . By the same characterization of C, we also have that  $(X \times [1, \omega]) \times C \approx C$ , so we can find a closed copy E of  $X \times [1, \omega]$  in D. Now

 $C_p(X) \sim C_{p, D}(X) \times C_p(D) \qquad \text{by proposition 2.3.2}$  $\sim C_{p, D}(X) \times C_p(D \oplus D) \qquad \text{since } D \oplus D \approx C \oplus C \approx C \approx D$ 

$$\sim C_{p, D}(X) \times C_p(D) \times C_p(D)$$
by lemma 2.3.6 $\sim C_p(X) \times C_p(D)$ by proposition 2.3.2 $\sim C_p(X) \times C_{p, E}(D) \times C_p(E)$ by proposition 2.3.2 $\sim C_p(X \oplus E) \times C_{p, E}(D)$ by lemma 2.3.6 $\sim C_p(E) \times C_{p, E}(D)$ since  $X \oplus E \approx E$  $\sim C_p(D)$ by proposition 2.3.2.  $\Box$ 

**2.4.8 REMARK:** From theorem 2.4.1 and theorem 2.4.7 it follows that the classification is such that for any two separable metric compact zero-dimensional spaces X and Y it follows that  $C_p(X)$  is linearly homeomorphic to  $C_p(Y)$  if and only if  $C_0(X)$  is linearly homeomorphic to  $C_0(Y)$ . In general this is not the case (see the remark after corollary 1.2.21 on page 29).

One of the steps in the proof Bessaga and Pelczyński gave of theorem 2.4.1 is proposition 2.4.4 for function spaces endowed with the compact-open topology (or the topology of uniform convergence) (cf. lemma 1 in [10]). Their proof of this result is quite different from ours. They used for example the fact that if a Banach space (and  $C_0(X)$ is a Banach space if X is a compact ordinal) is the direct sum of two closed linear subspaces E and F, then it is isomorphic to  $E \times F$ . Recall that our spaces  $C_p(X)$  are not Banach. Also, they did not use an inductive argument.

It is also possible to prove proposition 2.4.4 following the pattern of the proof of Bessaga and Pelczyński: They used the above property of Banach spaces to conclude, that if  $\beta \le \alpha < \omega_1$  then  $C_0([1, \alpha \cdot \beta]) \sim C_0([1, \alpha])$ . However, for the topology of pointwise convergence we can prove this directly by the method of corollary 2.3.4, using the fact that  $(\alpha \cdot \beta)' = \alpha' \cdot \beta'$ . All the other statements that Bessaga and Pelczyński proved, are also valid for function spaces endowed with the topology of pointwise convergence. So then we are in a position from which we can derive proposition 2.4.4 with the same arguments as the ones of Bessaga and Pelczyński.

We now give some examples which were already announced in chapter 1.

**2.4.9 EXAMPLE:** We show as announced on page 21 that in general for spaces X and Y, and an effective linear function  $\phi: C(X) \to C(Y)$  such that for each  $y \in Y$ ,  $\supp(y) \neq \emptyset$ ,  $supp: Y \to \mathcal{P}(X) \setminus \{\emptyset\}$  need not be USC.

Let  $X = [1, \omega^2]$  and  $Y = [1, \omega]$ . By theorem 2.4.7 there is a linear homeomorphism

 $\phi: C_p(X) \to C_p(Y)$ . By corollary 1.2.15 (a),  $\phi$  is effective and by proposition 1.4.3, for each  $y \in Y$ , supp  $(y) \neq \emptyset$ . We claim that supp :  $Y \to \mathcal{P}(X) \setminus \{\emptyset\}$  is not USC.

Since supp ( $\omega$ ) is finite (lemma 1.4.1), there is an infinite clopen subset U of X which misses supp ( $\omega$ ). Let  $V = X \setminus U$ . Then supp ( $\omega$ )  $\subset V$ . If supp :  $Y \to \mathcal{P}(X) \setminus \{\emptyset\}$  is USC, then there is  $n \in \mathbb{N}$  such that supp  $[n, \omega] \subset V$ . We now have by proposition 1.4.3,

 $X = \text{supp } Y = \text{supp } \{1, ..., n\} \cup \text{supp } [n, \omega] \subset \text{supp } \{1, ..., n\} \cup V,$ 

hence  $U \subset \text{supp}\{1, ..., n\}$ . Since U is infinite and supp $\{1, ..., n\}$  is finite, we have a contradiction. We conclude that supp:  $Y \to \mathcal{P}(X) \setminus \{\emptyset\}$  is not USC.

**2.4.10 EXAMPLE:** In this example we show that the first countability condition in theorems 1.5.10 and 1.5.12 is essential.

Let  $X = [1, \omega] \times \mathbb{N}$ ,  $A = X^{(1)} = \{\omega\} \times \mathbb{N}$  and Y = X/A. Then X is clearly first countable and normal. Since the quotient map p between X and Y is closed (cf. page 60), Y is normal ([24, th 1.5.20]). The proof of the following claim is standard. For the sake of completeness it will be included.

CLAIM: Y is not first countable.

Indeed, suppose  $\{U_n : n \in \mathbb{N}\}$  is a countable base at  $\infty$  in Y. Let  $n \in \mathbb{N}$ . Then  $p^{-1}(U_n)$  is open in X and  $\{\omega\} \times \mathbb{N} \subset p^{-1}(U_n)$ . So for every  $i \in \mathbb{N}$  there is  $\alpha_i^n < \omega$  such that  $[\alpha_i^n, \omega] \times \{i\} \subset p^{-1}(U_n)$ . Now let  $U = \bigcup_{i=1}^{\infty} [\alpha_i^i + 1, \omega] \times \{i\}$ . Since  $A \subset U$ ,  $\infty \in p(U)$  and  $p^{-1}(p(U)) = U$ , so p(U) is a neighborhood of  $\infty$  in Y. Hence there is  $n \in \mathbb{N}$  such that  $U_n \subset p(U)$ , so  $p^{-1}(U_n) \subset U$ . Hence  $\bigcup_{i=1}^{\infty} [\alpha_i^n, \omega] \times \{i\} \subset U$ . But then  $[\alpha_n^n, \omega] \subset [\alpha_n^n + 1, \omega]$ , which is a contradiction.

Notice that for every space Z and for every  $z \in Z$ ,  $C_p(Z) \sim C_{p, z}(Z) \times \mathbb{R}$  (It is easily seen that the function  $\phi: C_p(Z) \to C_{p, z}(Z) \times \mathbb{R}$  defined by  $\phi(f) = f - f(z)$  is a linear homeomorphism), so  $C_{p, \infty}(Y) \times \mathbb{R} \sim C_p(Y)$ . Hence

$C_p(X) \sim C_p([1, \omega] \oplus X)$	because $X \approx [1, \omega] \oplus X$
$\sim C_p([1, \omega]) \times C_p(X)$	by lemma 2.3.6
$\sim C_p([1, \omega]) \times \mathbb{R} \times C_p(X)$	by proposition 2.3.2 and lemma 2.3.9
$\sim C_p(X) \times \mathbb{R}$	as above
$\sim C_{p,A}(X) \times \mathbb{R}$	by lemma 2.3.7 and 2.3.9
$\sim C_{p,\infty}(Y) \times \mathbb{R}$	by lemma 2.3.3
$\sim C_p(Y).$	

For theorem 1.5.12 notice that  $X^{(1)}$  is not countably compact, and  $Y^{(1)} = \{\infty\}$  is countably compact. For theorem 1.5.10 notice that X and Y are paracompact (by [24, Th 5.1.3 and 5.1.33]), X is locally compact and Y is not locally compact.

Furthermore, notice that this example is a counterexample for the following statement (see also section 0.1): If X and Y are  $l_p$ -equivalent spaces, then X has property  $\mathcal{P}$  if and only if Y has property  $\mathcal{P}$ , where  $\mathcal{P}$  is one of the properties: local compactness, first countability, second countability, metrizability, weight, or character.

**2.4.11 EXAMPLE:** In this example we show that theorem 1.5.12 is not true for the  $\alpha$ -th derivative if  $\alpha$  is not a prime component.

Let  $\alpha < \omega_1$  be an ordinal which is not a prime component. Observe that in this situation  $1 \le \alpha' < \alpha < \alpha' \cdot \omega$  (lemma 2.1.19). Hence  $\omega^{\alpha'} < \omega^{\alpha} < (\omega^{\alpha'})^{\omega}$  and so by theorem 2.4.7  $C_p([1, \omega^{\alpha'}]) \sim C_p([1, \omega^{\alpha}])$ . So if we now let  $X = \bigoplus_{i=1}^{\infty} [1, \omega^{\alpha'}]_i$  and  $Y = \bigoplus_{i=1}^{\infty} [1, \omega^{\alpha}]_i$ , then  $C_p(X) \sim C_p(Y)$  (lemma 2.3.7). In addition, X and Y are normal and first countable, but  $Y^{(\alpha)} \approx \mathbb{N}$  (this follows easy from proposition 2.2.5) which is not countably compact, and  $X^{(\alpha)} = \emptyset$  which is countably compact.

This observation leads us to the following

**Question:** Let X and Y be  $l_p$ -equivalent spaces which are both normal and first countable. Let  $\alpha \ge \omega$  be a prime component. Is it true that  $X^{(\alpha)}$  is countably compact if and only if  $Y^{(\alpha)}$  is countably compact?

Finally we remark that the first part of this section (until theorem 2.4.7) is taken from [3]. The examples 2.4.10 and 2.4.11 can be found in [5].

#### §2.5. Compact ordinals

In this section we present an isomorphical classification of the function spaces  $C_p(X)$ , where  $X = [1, \alpha]$  for some ordinal  $\alpha$ . We call such spaces *compact ordinal spaces*. It turns out that this classification is similar to the one Kislyakov gave for the spaces  $C_0(X)$  (with X a compact ordinal space) in [34]. Our proof is similar to his, only some modifications are necessary.

It turns out that Kislyakov made a mistake in his proof. In this section we will identify this mistake and correct it.

Let us first present the classification of Kislyakov. For that we need some

definitions. Let X and Y be Banach spaces. We say that X and Y have *the same linear dimension*, if each of them is isomorphic to a subspace of the other. X has smaller *linear dimension than* Y if X is isomorphic to a subspace of Y, but Y is not isomorphic to any subspace of X. Notice that isomorphic spaces have the same linear dimension.

#### **2.5.1 THEOREM:** Let $\alpha$ and $\beta$ be ordinals.

If  $\alpha$  and  $\beta$  have different power, then

- (a)  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  do not have the same linear dimension and so they are not linearly homeomorphic.
- If  $\alpha$  and  $\beta$  have the same power and  $\xi$  is the initial ordinal of that power, then
  - (b) ([34]) If  $\xi = \omega$ , or  $\xi$  is a singular ordinal or both  $\alpha$ ,  $\beta \ge \xi^2$ , then  $C_0([1, \alpha]) \sim C_0([1, \beta])$  if and only if  $\max(\alpha, \beta) < [\min(\alpha, \beta)]^{\omega}$  if and only if  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  have the same linear dimension.
  - (c) ([34]) If  $\xi$  is an uncountable regular ordinal and  $\alpha$ ,  $\beta \in [\xi, \xi^2]$ , fix ordinals  $\alpha_1, \beta_1 \leq \xi$  and  $\gamma, \delta < \xi$  such that  $\alpha = \xi \cdot \alpha_1 + \gamma$  and  $\beta = \xi \cdot \beta_1 + \delta$ . Then  $C_0([1, \alpha]) \sim C_0([1, \beta])$  if and only if  $\overline{\alpha_1} = \overline{\beta_1}$  if and only  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  have the same linear dimension.
  - (d) If  $\xi$  is an uncountable regular ordinal,  $\alpha < \xi^2$  and  $\beta \ge \xi^2$ , then  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  are not linearly homeomorphic.

Notice that the case  $\xi = \omega$  in theorem 2.5.1 (b) is just Bessaga and Pelczyński's result stated in theorem 2.4.1. Furthermore theorem 2.5.1 (c) was proved by Semadeni in [46] for ordinals  $\alpha$  and  $\beta$  satisfying  $\omega_1 \le \alpha$ ,  $\beta \le \omega_1 \cdot \omega$ .

In fact Kislyakov only stated theorem 2.5.1 (b) and (c), so we will now prove part (a) and part (d). Before being able to prove this, we need to formulate the following lemma proved by Bessaga and Pelczyński in [10].

**2.5.2 LEMMA:** Let  $\alpha$  be an ordinal. If for every  $\gamma < \alpha$ ,  $C_0([1, \gamma])$  has smaller linear dimension than  $C_0([1, \alpha])$ , then  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \alpha^{\omega}])$ .

The following corollary to lemma 2.5.2 is also useful in section 2.6, and is stated without proof by Bessaga and Pelczyński in [10] and by Kislyakov in [34]. For the sake of completeness, we will present its proof.

**2.5.3 COROLLARY** ([34, lemma 1.3]): Let  $\alpha$  and  $\beta$  be ordinals. If  $\beta \ge \alpha^{\omega}$ , then  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \beta])$ .

**PROOF:** Let  $\alpha_1$  be the smallest ordinal such that  $C_0([1, \alpha_1])$  and  $C_0([1, \alpha])$  have the same linear dimension. Then for every  $\gamma < \alpha_1$ ,  $C_0([1, \gamma])$  has smaller linear dimension than  $C_0([1, \alpha_1])$ . By lemma 2.5.2 it follows that  $C_0([1, \alpha_1])$  has smaller linear dimension than  $C_0([1, \alpha_1^{\omega}])$ . Since  $\alpha_1 \le \alpha$ , it follows that  $\alpha_1^{\omega} \le \alpha^{\omega} \le \beta$ . But then it easily follows that  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \alpha])$ .

We are now able to prove theorem 2.5.1 (a) and (d).

**PROOF of theorem 2.5.1 (a):** Without loss of generality we may assume that  $\overline{\alpha} < \beta$ . First suppose  $\alpha \ge \omega$ . Since  $\overline{\alpha}^{\overline{\alpha}} = \overline{\alpha}$  (proposition 2.1.13), it follows that  $\alpha^{\omega} < \beta$ . So by corollary 2.5.3 it follows that  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \beta])$ , which implies that  $C_0([1, \alpha])$  is not linearly homeomorphic to  $C_0([1, \beta])$ .

Now suppose that  $\alpha$  is finite. As is easily seen, the algebraic dimension of  $C_0([1, \alpha])$  is finite and smaller than the algebraic dimension of  $C_0([1, \beta])$ , hence  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  are not linearly homeomorphic.  $\Box$ 

**PROOF of theorem 2.5.1 (d):** First suppose that  $\beta < \xi^{\omega}$ . Since  $\xi^2 \le \beta < \xi^{\omega} = (\xi^2)^{\omega}$ , by theorem 2.5.1,  $C_0([1, \beta]) \sim C_0([1, \xi^2])$ . Let  $\alpha = \xi \cdot \alpha_1 + \gamma$  with  $\gamma < \xi$  (proposition 2.1.2). Notice that  $\alpha_1 < \xi$ , so  $\overline{\alpha_1} < \overline{\xi}$ , because  $\xi$  is initial. From theorem 2.5.1 (c) it now follows that  $C_0([1, \alpha])$  and  $C_0([1, \xi^2])$  are not linearly homeomorphic.

Now suppose  $\beta \ge \xi^{\omega}$ . Since  $\alpha < \xi^2$ ,  $\alpha^{\omega} \le (\xi^2)^{\omega} = \xi^{\omega} \le \beta$ . So by corollary 2.5.3,  $C_0([1, \alpha])$  has smaller linear dimension than  $C_0([1, \beta])$ , which implies that  $C_0([1, \alpha])$  and  $C_0([1, \beta])$  are not linearly homeomorphic.  $\Box$ 

Now we are going to prove that the same classification holds for the spaces  $C_p([1, \alpha])$ . For that we first have to give some definitions.

For a compact space X and  $f \in C_p(X)$ , let  $||f|| = \sup_{x \in X} |f(x)|$ . Let  $\{X_t : t \in T\}$  be a family of compact spaces and for each  $t \in T$ , let  $E_t$  be a linear subspace of  $C_p(X_t)$ . By  $\prod_{t \in T}^* E_t$  we denote the linear subspace of  $\prod_{t \in T} E_t$  consisting of all points  $f = (f_t)_{t \in T}$  such that for each  $\varepsilon > 0$ , the set  $\{t \in T : ||f_t|| \ge \varepsilon\}$  is finite. If  $E_t = E$  for every  $t \in T$ , we write  $\prod_{m=1}^* E_t$  instead of  $\prod_{t \in T}^* E_t$ , where  $\mathfrak{m} = |T|$ . Notice that if T is finite, then  $\prod_{t \in T} E_t = \prod_{t \in T}^* E_t$ .

The notion of linear k-mapping (cf. section 2.3) can also be defined for the spaces  $\prod_{t \in T}^{*} E_t$ . We then have the following
§2.5. Compact ordinals

**2.5.4 LEMMA:** For each  $i \in I$  let  $\{X_{t,i} : t \in T\}$  and  $\{Y_{s,i} : s \in S\}$  be two families of compact spaces and let for each  $i \in I$ ,  $t \in T$  and  $s \in S$ ,  $E_{t,i}$  and  $F_{s,i}$  be linear subspaces of  $C_p(X_{t,i})$  and  $C_p(Y_{s,i})$ , respectively. Suppose that for each  $i \in I$ ,  $\prod_{i=T}^{*} E_{t,i} \stackrel{k}{\sim} \prod_{s \in S}^{*} F_{s,i}$ . Then  $\prod_{(t,i) \in T \times I}^{*} E_{t,i} \stackrel{k}{\sim} \prod_{(s,i) \in S \times I}^{*} F_{s,i}$ .

**PROOF:** For each  $i \in I$  let  $\phi_i: \prod_{t \in T}^* E_{t, i} \to \prod_{s \in S}^* F_{s, i}$  be a linear k-homeomorphism. Define  $\phi: \prod_{(t, i) \in T \times I}^* E_{t, i} \to \prod_{(s, i) \in S \times I}^* F_{s, i}$  by

$$\phi((f_{(t, i)})_{(t, i) \in T \times I}) = ((\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T}))_{(s, i) \in S \times I}.$$

We prove that  $\phi$  is a linear *k*-homeomorphism.

CLAIM 1:  $\phi$  is well-defined.

Indeed, let  $\varepsilon > 0$  and  $(f_{(t, i)})_{(t, i) \in T \times I} \in \prod_{(t, i) \in T \times I}^* E_{t, i}$ . It is a triviality that  $(\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T}) \in F_{(s, i)}$ . Notice that  $J = \{i \in I : (\exists t \in T)(||f_{(t, i)}|| \ge \varepsilon/k)\}$  is finite. Let  $i \in I \setminus J$ . Then for every  $t \in T$ ,  $f_{(t, i)}(X_{t, i}) \subset (-\varepsilon/k, \varepsilon/k)$ . Since  $\phi_i$  is a linear k-homeomorphism, for every  $s \in S$ 

 $(\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T})(Y_{s, i}) \subset (-\varepsilon, \varepsilon).$ 

So  $\{i \in I : (\exists s \in S)(||(\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T})|| \ge \varepsilon)\}$  is finite. Since

 $\phi_i((f_{(t,i)})_{t \in T}) \in \prod_{s \in S}^* F_{s,i}$ 

for every  $i \in I$ ,  $\{s \in S : \|(\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T})\| \ge \varepsilon\}$  is finite. But this implies that

 $\{(s, i) \in S \times I : \|(\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T})\| \ge \varepsilon\}$ 

is finite as well, which proves the claim.

Since  $\pi_s$  and  $\phi_i$  are linear and continuous, it is clear that  $\phi$  is linear and continuous.

#### CLAIM 2: $\phi$ is a linear *k*-mapping.

Indeed, let  $(f_{(t,i)})_{(t,i) \in T \times I} \in \prod_{(t,i) \in T \times I}^* E_{t,i}$  be such that  $f_{(t,i)}(X_{t,i}) \subset (-1/k, 1/k)$  for each pair  $(t, i) \in T \times I$ . Then

 $(\pi_{(s, i)} \circ \phi)((f_{(t, i)})_{(t, i) \in T \times I})(Y_{t, i}) = (\pi_s \circ \phi_i)((f_{(t, i)})_{t \in T})(Y_{t, i}) \subset (-1, 1)$ 

for each pair  $(s, i) \in S \times I$ , because  $\phi_i$  is a linear k-homeomorphism. This proves claim 2.

Now define  $\psi: \prod_{(s,i) \in S \times I}^* F_{s,i} \to \prod_{(t,i) \in T \times I}^* E_{t,i}$  by

$$\Psi((g_{(s,i)})_{(s,i)\in S\times I}) = ((\pi_t \circ \phi_i^{-1})((g_{(s,i)})_{s\in S}))_{(t,i)\in T\times I}.$$

The proof that  $\psi$  is a well-defined continuous linear k-mapping is exactly the same as the proof for  $\phi$ . Furthermore it is easily seen that  $\psi = \phi^{-1}$ , so  $\phi$  is a linear k-homeomorphism.

In the sequel we denote  $\Pi_{(t, i) \in T \times I}^* E_{t, i}$  also by  $\Pi_{t \in T}^* \Pi_{i \in I}^* E_{t, i}$  or by  $\Pi_{i \in I}^* \Pi_{t \in T}^* E_{t, i}$ . Notice that this is not the same as  $\Pi_{t \in T}^* (\Pi_{i \in I}^* E_{t, i})$ , because the latter product is not defined. Lemma 2.5.4 now gives us the following: If for each  $i \in I$ ,

$$\Pi_{t \in T}^* E_{t, i} \stackrel{k}{\sim} \Pi_{s \in S}^* F_{s, i},$$

then

 $\Pi_{i \in I}^* \Pi_{t \in T}^* E_{t, i} \stackrel{k}{\sim} \Pi_{i \in I}^* \Pi_{s \in S}^* E_{s, i}.$ 

**2.5.5 LEMMA:** Let  $\{X_t : t \in T\}$  be a family of compact spaces and let for each  $t \in T$ ,  $E_t$  be a linear subspace of  $C_p(X_t)$ . Let  $S \subset T$ . Then  $\prod_{t \in T}^* E_t \stackrel{1}{\sim} \prod_{t \in S}^* E_t \times \prod_{t \in T \setminus S}^* E_t$ .

**PROOF:** Define  $\phi: \prod_{t \in T}^{*} E_t \to \prod_{s \in S}^{*} E_s \times \prod_{t \in T \setminus S}^{*} E_t$  by  $\phi(f_t)_{t \in T} = ((f_t)_{t \in S}, (f_t)_{t \in T \setminus S})$ . It is a triviality that  $\phi$  is a well-defined continuous linear 1-mapping as well. The inverse  $\phi^{-1}$  of  $\phi$  can also be defined canonically and is a continuous linear 1-mapping as well. So  $\phi$  is a linear 1-homeomorphism.  $\Box$ 

The next lemma is the main tool in this section.

**2.5.6 LEMMA:** Let  $\gamma$  be a limit ordinal. Let  $(\lambda_{\xi})_{0 \le \xi \le \gamma}$  be a strictly increasing sequence such that

1)  $\lambda_{\xi+1} - \lambda_{\xi} \ge \omega$  ( $\xi \in [0, \gamma)$ ), 2)  $\lambda_{\xi} = \lim_{\eta < \xi} \lambda_{\eta}$  ( $\xi \in (0, \gamma]$  a limit ordinal), 3)  $\lambda_0 = 0$ .

*Then*  $C_{p,0}([1, \lambda_{\gamma}]) \stackrel{2}{\sim} C_{p,0}([1, \gamma]) \times \prod_{\xi < \gamma}^{*} C_{p,0}([1, \lambda_{\xi+1} - \lambda_{\xi}]).$ 

#### **PROOF:** Let

 $X = \{f \in C_{p, 0}([1, \lambda_{\gamma}]) : f \text{ is constant on each interval } (\lambda_{\xi}, \lambda_{\xi+1}] \ (\xi \in [0, \gamma))\}$  and

$$Y = \{ f \in C_{p, 0}([1, \lambda_{\gamma}]) : f(\lambda_{\xi}) = 0 \ (\xi \in (0, \gamma]) \}.$$

**CLAIM 1:**  $C_{p, 0}([1, \lambda_{\gamma}]) \stackrel{2}{\sim} X \times Y$ .

Indeed, define  $\phi_1: C_{p, 0}([1, \lambda_{\gamma}]) \to X$  by

$$\phi_1(f) \mid (\lambda_{\xi}, \lambda_{\xi+1}] = f(\lambda_{\xi+1}) \ (\xi \in [0, \gamma)).$$

and

 $\phi_1(f)(\lambda_{\xi}) = f(\lambda_{\xi}) \ (\xi \in (0, \gamma] \text{ a limit ordinal}).$ 

To prove that  $\phi_1$  is well-defined, we need to show that  $\phi_1(f)$  is continuous at  $\lambda_{\xi}$  with  $\xi \in (0, \gamma]$  a limit ordinal. So let  $\varepsilon > 0$ . Since  $\lambda_{\xi} = \lim_{\eta < \xi} \lambda_{\eta}$  and since *f* is continuous at  $\lambda_{\xi}$ , there is  $\eta < \xi$  such that  $f((\lambda_{\eta}, \lambda_{\xi}]) \subset (-\varepsilon + f(\lambda_{\xi}), \varepsilon + f(\lambda_{\xi}))$ . But then also

$$\phi_1(f)((\lambda_n, \lambda_{\xi}]) \subset (-\varepsilon + f(\lambda_{\xi}), \varepsilon + f(\lambda_{\xi})).$$

That  $\phi_1$  is linear is a triviality. We prove that  $\phi_1$  is continuous. Let  $P = \{\alpha_i : i \le n\} \subset [1, \lambda_{\gamma}]$  be finite,  $\varepsilon > 0$ , and  $f \in C_{p, 0}([1, \lambda_{\gamma}])$ . For  $i \le n$ , if  $\alpha_i = \lambda_{\xi}$  for some  $\xi$ , put  $\beta_i = \alpha_i$ , otherwise let  $\lambda_{\xi}$  be such that  $\alpha_i \in (\lambda_{\xi}, \lambda_{\xi+1}]$  and put  $\beta_i = \lambda_{\xi+1}$ . Let  $Q = \{\beta_i : i \le n\}$ . Then a simple calculation shows that  $\phi_1(\langle f, Q, \varepsilon \rangle) \subset \langle \phi_1(f), P, \varepsilon \rangle$ , which proves that  $\phi_1$  is continuous.

Now define  $\phi_2: C_{p,0}([1, \lambda_{\gamma}]) \to Y$  by  $\phi_2(f) = f - \phi_1(f)$ . It is easily seen that  $\phi_2$  is well-defined, continuous and linear. It follows that the map  $\phi: C_{p,0}([1, \lambda_{\gamma}]) \to X \times Y$  defined by  $\phi(f) = (\phi_1(f), \phi_2(f))$  is also well-defined, linear and continuous.

We prove that  $\phi$  is a linear 2-mapping. For that let  $f \in C_{p,0}([1, \lambda_{\gamma}])$  with  $f([1, \lambda_{\gamma}]) \subset (-1/2, 1/2)$ . Let  $\alpha \in [1, \lambda_{\gamma}]$ . Then  $|\phi_1(f)(\alpha)| < 1/2$  and therefore

$$|\phi_{2}(f)(\alpha)| = |f(\alpha) - \phi_{1}(f)(\alpha)| \le |f(\alpha)| + |\phi_{1}(f)(\alpha)| < 1/2 + 1/2 = 1,$$

which proves that  $\phi$  is a linear 2-mapping.

Now define  $\psi: X \times Y \to C_{p,0}([1, \lambda_{\gamma}])$  by  $\psi(f, g) = f + g$ . It is evident that  $\psi$  is a well-defined continuous linear 2-mapping. Furthermore one can simply derive that  $\psi = \phi^{-1}$ , and so we conclude that  $\phi$  is a linear 2-homeomorphism, which proves claim 1.

## CLAIM 2: $X \stackrel{1}{\sim} C_{p,0}([1, \gamma]).$

Indeed, define  $\phi: X \to C_{p, 0}([1, \gamma])$  by  $\phi(f)(\xi) = f(\lambda_{\xi})$ . Since by (2) the function  $\xi \to \lambda_{\xi}$  is continuous,  $\phi$  is well-defined. It is easily seen that  $\phi$  is a continuous, linear 1-mapping. Define  $\psi: C_{p, 0}([1, \gamma]) \to X$  by  $\psi(f) | (\lambda_{\xi}, \lambda_{\xi+1}] = f(\xi+1)$  for  $\xi \in [0, \gamma)$  and  $\psi(f)(\lambda_{\xi}) = f(\xi)$  for  $\xi \in (0, \gamma]$  a limit ordinal. It is a triviality that  $\psi$  is a well-defined, continuous, linear 1-mapping, which is the inverse of  $\phi$ . Whence  $\phi$  is a linear 1-homeomorphism.

CLAIM 3:  $Y \stackrel{1}{\sim} \prod_{\xi < \gamma}^{*} C_{p, 0}([1, \lambda_{\xi+1} - \lambda_{\xi}]).$ 

Indeed, define 
$$\phi: Y \to \prod_{\xi < \gamma}^* C_{\rho, 0}((\lambda_{\xi}, \lambda_{\xi+1}])$$
 by  $(\pi_{\xi} \circ \phi)(f) = f \mid (\lambda_{\xi}, \lambda_{\xi+1}]$   $(\xi < \gamma)$ .

To see that  $\phi$  is well-defined, we assume there are  $f \in Y$ ,  $\varepsilon > 0$  and  $\xi_1 < \xi_2 < \cdots < \gamma$ such that  $\|(\pi_{\xi_n} \circ \phi)(f)\| \ge \varepsilon$  for each  $n \in \mathbb{N}$ . Let  $\xi = \lim_{n \in \mathbb{N}} \xi_n$ . Then  $\xi \le \gamma$ . For each  $n \in \mathbb{N}$  find  $\alpha_n \in (\lambda_{\xi_n}, \lambda_{\xi_n+1}]$  with  $|f(\alpha_n)| \ge \varepsilon$ . Then  $|f(\lambda_{\xi})| \ge \varepsilon$ , which is a contradiction, because  $f \in Y$ . It is easily seen that  $\phi$  is a continuous linear 1-mapping.

Define  $\psi: \prod_{\xi < \gamma}^* C_{p, 0}((\lambda_{\xi}, \lambda_{\xi+1}]) \to Y$  by  $\psi((f_{\xi})_{\xi < \gamma})(\beta) = f_{\xi}(\beta)$  if  $\beta \in (\lambda_{\xi}, \lambda_{\xi+1}]$  and  $\psi((f_{\xi})_{\xi < \gamma})(\lambda_{\xi}) = 0$  if  $\xi \in (0, \gamma]$  is a limit ordinal. It is evident that  $\psi$  is a well-defined continuous linear 1-mapping, which is the inverse of  $\phi$ . Thus  $\phi$  is a linear 1-homeomorphism. The claim is now proved since  $(\lambda_{\xi}, \lambda_{\xi+1}] \approx [1, \lambda_{\xi+1} - \lambda_{\xi}]$ .

It is clear that the claims 1, 2, and 3 establish the proof of lemma 2.5.6.  $\Box$ 

**2.5.7 LEMMA:** Let  $\mu$  be an infinite ordinal and  $\gamma$  a limit ordinal such that  $\gamma = \mu$  or  $\gamma + \mu = \mu$ . Then  $C_{p,0}([1, \mu \cdot \gamma]) \stackrel{4}{\sim} \prod_{\overline{\gamma}}^{*} C_{p,0}([1, \mu])$ .

**PROOF:** By lemma 2.5.6, applied to the sequence  $\lambda_{\xi} = \mu \cdot \xi$  for  $\xi \leq \gamma$ , we have

$$C_{p,0}([1,\mu\cdot\gamma]) \stackrel{2}{\sim} C_{p,0}([1,\gamma]) \times \Pi_{\xi}^* \langle \gamma C_{p,0}([1,\mu(\xi+1)-\mu\cdot\xi])$$
  
=  $C_{p,0}([1,\gamma]) \times \Pi_{\gamma}^* C_{p,0}([1,\mu]).$ 

Now suppose  $\gamma = \mu$ . Then by lemma 2.5.5  $C_{p,0}([1, \mu \cdot \gamma]) \stackrel{2}{\sim} \Pi^*_{\overline{\gamma}} C_{p,0}([1, \mu])$ . If  $\gamma + \mu = \mu$ , then

$$C_{p,0}([1, \mu \cdot \gamma]) \stackrel{2}{\sim} C_{p,0}([1, \gamma]) \times C_{p,0}([1, \mu]) \times \Pi^*_{\overline{\gamma}} C_{p,0}([1, \mu]) \quad (\text{lemma 2.5.5})$$

$$\stackrel{2}{\sim} C_{p,0}([1, \mu]) \times \Pi^*_{\overline{\gamma}} C_{p,0}([1, \mu]) \quad (\text{lemma 2.3.8, 2.3.9})$$

$$\stackrel{1}{\sim} \Pi^*_{\overline{\gamma}} C_{p,0}([1, \mu]) \quad (\text{lemma 2.5.5}). \square$$

**2.5.8 LEMMA:** Let  $\alpha$  be an initial ordinal and  $\gamma$  a limit ordinal with  $\overline{\gamma} \leq \overline{\alpha}$ . Then there exists a subset M of  $[2, \gamma)$  consisting of successors such that

 $C_{p,0}([1, \alpha^{\gamma}]) \stackrel{4}{\sim} \prod_{\mu \in M}^{*} C_{p,0}([1, \alpha^{\mu}]).$ 

**PROOF:** Let  $\beta = cf(\gamma)$ . Since  $\beta$  is initial,  $cf(\gamma) \le \gamma$ , and  $\overline{\gamma} \le \overline{\alpha}$ , we have  $\beta \le \alpha$ .

CLAIM: There is a strictly increasing sequence  $\{\mu_{\xi} : \xi \leq \beta\}$  in  $[2, \gamma]$  such that  $\mu_{\xi}$  is a successor for each successor  $\xi \leq \beta$ ,  $\mu_{\xi} = \lim_{\eta < \xi} \mu_{\eta}$  for a limit ordinal  $\xi \leq \beta$ , and  $\mu_{\beta} = \gamma$ ,

Indeed, let  $\phi$  be an increasing  $\beta$ -sequence such that  $\lim_{\xi < \beta} \phi(\xi) = \gamma$  and  $\phi(1) \ge 1$ . Let  $\xi \le \beta$  and take  $\mu_{\xi} = \phi(\xi) + 1$  if  $\xi$  is a limit ordinal, otherwise take  $\mu_{\xi} = \lim_{\eta < \xi} \phi(\xi)$ . It is a triviality that  $\{\mu_{\xi} : \xi \le \beta\}$  is as required.

Since  $\alpha$  is a prime component (theorem 2.1.23),  $\alpha^{\mu_{\xi+1}} - \alpha^{\mu_{\xi}} = \alpha^{\mu_{\xi+1}}$  and  $\beta + \alpha^{\mu_1} = \alpha^{\mu_1}$  (by theorem 2.1.15 and the facts that  $\mu_1 \ge 2$  and  $\beta \le \alpha$ ). By applying lemma 2.5.6 to the sequence  $\lambda_{\xi} = \alpha^{\mu_{\xi}}$  for  $\xi \in (0, \beta]$  and  $\lambda_0 = 0$ , we get

$$C_{p,0}([1, \alpha^{\gamma}]) \stackrel{2}{\sim} C_{p,0}([1, \beta]) \times \Pi_{\xi < \beta}^{*} C_{p,0}([1, \alpha^{\mu_{\xi+1}} - \alpha^{\mu_{\xi}}])$$
  
=  $C_{p,0}([1, \beta]) \times \Pi_{\xi < \beta}^{*} C_{p,0}([1, \alpha^{\mu_{\xi+1}}])$   
 $\stackrel{2}{\sim} C_{p,0}([1, \beta + \alpha^{\mu_{1}}]) \times \Pi_{1 \le \xi < \beta}^{*} C_{p,0}([1, \alpha^{\mu_{\xi+1}}])$   
=  $C_{p,0}([1, \alpha^{\mu_{1}}]) \times \Pi_{1 \le \xi < \beta}^{*} C_{p,0}([1, \alpha^{\mu_{\xi+1}}])$   
 $\stackrel{1}{\sim} \Pi_{\xi < \beta}^{*} C_{p,0}([1, \alpha^{\mu_{\xi+1}}]).$ 

We applied lemma 2.5.6 for the first equivalence, lemmas 2.5.5, 2.3.9 and 2.3.8 for the third one and lemma 2.5.5 for the last one. Now take  $M = \{\mu_{\xi+1} : \xi < \beta\}$ .

**2.5.9 LEMMA:** Let  $\omega_{\gamma}$  be a singular ordinal. Then there exist  $\beta < \omega_{\gamma}$  and a strictly increasing  $\beta$ -sequence  $\phi$  such that

(a)  $\lim_{\xi < \beta} \phi(\xi) = \gamma$ , (b)  $\omega_{\phi(\xi)} \ge \beta$  for every  $\xi < \beta$ , (c)  $C_{p,0}([1, \omega_{\gamma}]) \stackrel{4}{\sim} \Pi_{\xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)}])$ .

**PROOF:** Since  $\omega_{\gamma}$  is singular,  $\gamma$  is a limit ordinal (corollary 2.1.11). Furthermore  $\gamma \leq \omega_{\gamma}$ .

CLAIM: There is an ordinal  $\beta$  and a strictly increasing  $\beta$ -sequence  $\phi$  such that  $\beta < \omega_{\gamma}$ ,  $\lim_{\xi < \beta} \phi(\xi) = \gamma$ ,  $\omega_{\phi(\xi)} \ge \beta$  for every  $\xi < \beta$ , and  $\lim_{\xi < \eta} \phi(\xi) = \phi(\eta)$  if  $\eta$  is a limit ordinal.

For the proof of the claim we consider two cases.

Case 1:  $\gamma < \omega_{\gamma}$ .

Let  $\xi_0 = \min\{\xi : \gamma < \omega_{\xi} < \omega_{\gamma}\}$  and notice that  $\xi_0 < \gamma$ . Put  $\beta = \gamma - \xi_0$  and  $\phi(\xi) = \xi_0 + \xi$ . Then  $\beta \le \gamma < \omega_{\gamma}$ ,  $\lim_{\xi < \beta} \phi(\xi) = \xi_0 + \beta = \gamma$ , and by the definition of  $\phi$ ,  $\omega_{\phi(\xi)} \ge \omega_{\xi_0} > \gamma \ge \beta$ . That  $\lim_{\xi < \eta} \phi(\xi) = \phi(\eta)$  if  $\eta$  is a limit ordinal is a triviality. *Case 2*:  $\gamma = \omega_{\gamma}$ .

Let  $\beta = cf(\omega_{\gamma})$ . Since  $\omega_{\gamma}$  is singular,  $\beta < \omega_{\gamma}$ . Let  $\phi_1$  be a strictly increasing  $\beta$ sequence such that  $\lim_{\xi < \beta} \phi_1(\xi) = \omega_{\gamma} = \gamma$ . As in the proof of lemma 2.5.8, we may assume that  $\lim_{\xi < \beta} \phi(\xi) = \phi(\eta)$  if  $\eta$  is a limit ordinal. Now put  $\phi(\xi) = \beta + \phi_1(\xi)$  for  $\xi < \beta$ . Then  $\lim_{\xi < \beta} \phi(\xi) = \beta + \omega_{\gamma} = \omega_{\gamma}$  (theorem 2.1.15 and 2.1.23) and  $\omega_{\phi(\xi)} \ge \omega_{\beta} \ge \beta$ . This proves the claim. For the following notice that  $\omega_{\phi(\xi+1)} - \omega_{\phi(\xi)} = \omega_{\phi(\xi+1)}$  (theorems 2.1.15 and 2.1.23) and that  $\beta + \omega_{\phi(1)} = \omega_{\phi(1)}$  (because  $\beta \le \omega_{\phi(0)} < \omega_{\phi(1)}$ ). By applying lemma 2.5.6 to the sequence  $\lambda_{\xi} = \omega_{\phi(\xi)}$  for  $\xi \in (0, \beta]$  and  $\lambda_0 = 0$  and by lemmas 2.5.6, 2.3.8 and 2.3.9 we obtain

$$C_{p,0}([1, \omega_{\gamma}]) \stackrel{2}{\sim} C_{p,0}([1, \beta]) \times \Pi_{\xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)} - \omega_{\phi(\xi)}])$$

$$\stackrel{1}{\sim} C_{p,0}([1, \beta]) \times C_{p,0}([1, \omega_{\phi(1)}]) \times \Pi_{1 \le \xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)}])$$

$$\stackrel{2}{\sim} C_{p,0}([1, \beta + \omega_{\phi(1)}]) \times \Pi_{1 \le \xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)}])$$

$$= C_{p,0}([1, \omega_{\phi(1)}]) \times \Pi_{1 \le \xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)}])$$

$$\stackrel{1}{\sim} \Pi_{\xi < \beta}^{*} C_{p,0}([1, \omega_{\phi(\xi+1)}]).$$

We applied lemma 2.5.5 to get the second and fifth equivalence and lemmas 2.3.8 and 2.3.9 for the third one.  $\Box$ 

**2.5.10 LEMMA:** Let  $\alpha$  be an initial ordinal and  $\gamma$  an ordinal such that  $\overline{\gamma} \leq \overline{\alpha}, \gamma \geq 2$ . Then  $C_{p,0}([1, \alpha^{\gamma}]) \stackrel{4^{4}}{\sim} \prod_{\overline{\alpha}}^{*} C_{p,0}([1, \alpha^{\gamma}])$ .

**PROOF:** First suppose that  $\gamma$  is a successor, say  $\gamma = \beta + 1$ . By lemmas 2.5.6 (applied to the sequence  $\lambda_{\xi} = \alpha^{\beta} \cdot \xi$ ,  $\xi \in [0, \alpha]$ ), 2.5.5, 2.3.8, 2.3.9 and the fact that  $\alpha + \alpha^{\beta} = \alpha^{\beta}$  if  $\beta \ge 2$  (because  $\alpha^{\beta}$  is a prime component larger than  $\alpha$ ), we have  $C_{p, 0}([1, \alpha^{\gamma}]) \stackrel{4}{\sim} \Pi_{\overline{\alpha}}^* C_{p, 0}([1, \alpha^{\beta}])$ . By lemma 2.5.4 we now have

$$\Pi_{\overline{\alpha}}^* C_{p,0}([1,\alpha^{\gamma}]) \stackrel{4}{\sim} \Pi_{\overline{\alpha}}^* \Pi_{\overline{\alpha}}^* C_{p,0}([1,\alpha^{\beta}]) = \Pi_{\overline{\alpha}}^{*2} C_{p,0}([1,\alpha^{\beta}]).$$

Since  $\overline{\alpha} = \overline{\alpha}^2$  (page 52), it follows that  $C_{p,0}([1, \alpha^{\gamma}]) \stackrel{16}{\sim} \prod_{\overline{\alpha}}^* C_{p,0}([1, \alpha^{\gamma}])$ , which is as desired.

Now let  $\gamma$  be a limit ordinal. Find  $M \subset [2, \gamma)$  as in lemma 2.5.8. Then

$$C_{p,0}([1, \alpha^{\gamma}]) \stackrel{4}{\sim} \Pi^{*}_{\mu \in M} C_{p,0}([1, \alpha^{\mu}]) \qquad (\text{lemma 2.5.8})$$

$$\stackrel{16}{\sim} \Pi^{*}_{\mu \in M} \Pi^{*}_{\overline{\alpha}} C_{p,0}([1, \alpha^{\mu}]) \qquad (\text{by the above and lemma 2.5.4})$$

$$= \Pi^{*}_{\overline{\alpha}} \Pi^{*}_{\mu \in M} C_{p,0}([1, \alpha^{\mu}])$$

$$\stackrel{4}{\sim} \Pi^{*}_{\overline{\alpha}} C_{p,0}([1, \alpha^{\gamma}]) \qquad (\text{lemma 2.5.8}). \square$$

**2.5.11 LEMMA:** Let  $\alpha$  be a singular ordinal. Then  $C_{p,0}([1,\alpha]) \stackrel{4^{5}}{\sim} \Pi_{\overline{\alpha}}^{*} C_{p,0}([1,\alpha])$ .

**PROOF:** Put  $\alpha = \omega_{\gamma}$ . By lemma 2.5.9, there are  $\beta < \alpha$  and a strictly increasing  $\beta$ -sequence  $\phi$  with limit  $\gamma$ , such that  $\omega_{\phi(\xi)} \ge \beta$  for every  $\xi < \beta$  and

$$C_{p,0}([1,\alpha]) \stackrel{4}{\sim} \Pi_{\xi<\beta}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]).$$
<sup>(\*)</sup>

Fix  $\xi < \beta$ . Notice that  $\omega_{\phi(\xi+1)} \cdot \alpha = \alpha$  (proposition 2.1.12). By lemma 2.5.6, applied to the sequence  $\lambda_{\eta} = \omega_{\phi(\xi)} \cdot \eta$  ( $\eta < \alpha$ ), we obtain

$$C_{p,0}([1,\alpha]) \stackrel{2}{\sim} C_{p,0}([1,\alpha]) \times \Pi_{\eta < \alpha}^* C_{p,0}([1,\omega_{\phi(\xi+1)} \cdot (\eta+1) - \omega_{\phi(\xi+1)} \cdot \eta])$$
  
=  $C_{p,0}([1,\alpha]) \times \Pi_{\overline{\alpha}}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]).$ 

We now have

$$\begin{aligned} \Pi_{\bar{\beta}}^{*}C_{p,\,0}([1,\,\alpha]) &\stackrel{2}{\sim} \Pi_{\xi<\beta}^{*}\Pi_{\eta<\alpha}^{*}C_{p,\,0}([1,\,\omega_{\phi(\xi+1)}]) \times \Pi_{\bar{\beta}}^{*}C_{p,\,0}([1,\,\alpha]) \\ &= \Pi_{\eta<\alpha}^{*}\Pi_{\xi<\beta}^{*}C_{p,\,0}([1,\,\omega_{\phi(\xi+1)}]) \times \Pi_{\bar{\beta}}^{*}C_{p,\,0}([1,\,\alpha]) \\ &\stackrel{4}{\sim} \Pi_{\eta<\alpha}^{*}C_{p,\,0}([1,\,\alpha]) \times \Pi_{\bar{\beta}}^{*}C_{p,\,0}([1,\,\alpha]) \\ &\stackrel{1}{\sim} \Pi_{\bar{\alpha}}^{*}C_{p,\,0}([1,\,\alpha]). \end{aligned}$$

We applied lemma 2.5.4 to get the first equivalence, and (\*) and lemma 2.5.4 to get the third one. The last equivalence follows from lemma 2.5.5 and the fact that  $\overline{\beta} < \overline{\alpha}$ .

Let  $B = \{\xi < \beta : \xi \text{ is a limit ordinal}\} \cup \{0\}$ . Notice that for every  $i \in \mathbb{N}$  and  $\xi \in B$ ,  $\overline{\omega_{\phi(\xi+i-1)} \cdot \beta} = \overline{\omega_{\phi(\xi+i-1)}} < \overline{\omega_{\phi(\xi+i)}}$  (by the choice of the sequence  $\phi$ ), hence  $\omega_{\phi(\xi+i-1)} \cdot \beta < \omega_{\phi(\xi+i)}$ , so  $\omega_{\phi(\xi+i-1)} \cdot \beta + \omega_{\phi(\xi+i)} = \omega_{\phi(\xi+i)}$  (by theorems 2.1.23 and 2.1.15). Then

$$\begin{split} C_{p,0}([1,\alpha]) &\stackrel{4}{\sim} \Pi_{\xi < \beta}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \\ &= \Pi_{\xi \in B, i \in \mathbb{N}}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \\ &\stackrel{1}{\sim} \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i-1)}]) \\ &= \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i-1)}]) \\ &\stackrel{2}{\sim} \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i-1)}]) \\ &\stackrel{2}{\sim} \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i)}]) \\ &\stackrel{4}{\sim} \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\xi \in B, i \ge 2}^* \Pi_{\beta}^* C_{p,0}([1,\omega_{\phi(\xi+i-1)}]) \\ &\stackrel{4}{\sim} \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i)}]) \\ &= \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\beta}^* \Pi_{\xi \in B, i \in \mathbb{N}}^* C_{p,0}([1,\omega_{\phi(\xi+i)}]) \\ &= \Pi_{\xi \in B}^* C_{p,0}([1,\omega_{\phi(\xi+1)}]) \times \Pi_{\beta}^* \Pi_{\xi \in B, i \in \mathbb{N}}^* C_{p,0}([1,\omega_{\phi(\xi+i)}]) \\ &= \Pi_{\xi \in B, i \ge 2}^* C_{p,0}([1,\omega_{\phi(\xi+i)}]) \end{split}$$

$$\begin{split} &\frac{1}{2} \prod_{\xi \in B, \ i \in \mathbb{N}}^{*} C_{p, \ 0}([1, \ \omega_{\phi(\xi+i)}]) \times \prod_{\beta}^{*} \prod_{\xi \in B, \ i \in \mathbb{N}}^{*} C_{p, \ 0}([1, \ \omega_{\phi(\xi+i)}]) \\ &= \prod_{\beta}^{*} \prod_{\xi \in B, \ i \in \mathbb{N}}^{*} C_{p, \ 0}([1, \ \omega_{\phi(\xi+i)}]) \\ &= \prod_{\beta}^{*} \prod_{\xi < \beta}^{*} C_{p, \ 0}([1, \ \omega_{\phi(\xi+1)}]) \\ &\frac{4}{2} \prod_{\beta}^{*} C_{p, \ 0}([1, \ \alpha]). \end{split}$$

Here we applied (\*) for the first equivalence, lemma 2.5.5 for the third, lemmas 2.3.8, 2.3.9, 2.5.4 and 2.5.5 for the fifth, and lemmas 2.5.4 and 2.5.7 for the sixth, respective-ly. To get the eighth and ninth equivalence we used lemma 2.5.5. Finally we used (\*)  $4^{5} *$ 

for the last equivalence. We conclude that  $C_{p,0}([1, \alpha]) \stackrel{4^5}{\sim} \prod_{\overline{\alpha}}^* C_{p,0}([1, \alpha])$ .

**2.5.12 LEMMA:** Let  $\xi$  be an initial ordinal,  $\alpha \in [\xi, \xi^2]$ , say  $\alpha = \xi \cdot \alpha_1 + \beta$  with  $1 \le \alpha_1 \le \xi$  and  $\beta < \xi$ . Then  $C_{p,0}([1, \alpha]) \sim \prod_{\alpha=1}^* C_{p,0}([1, \xi])$ .

PROOF: First notice that by lemmas 2.3.8 and 2.3.9

$$C_{p,0}([1,\alpha]) \sim C_{p,0}([1,\xi;\alpha_1+\beta]) \sim C_{p,0}([1,\beta+\xi;\alpha_1]) \sim C_{p,0}([1,\xi;\alpha_1]).$$

Then by lemma 2.5.7 we have  $C_{p,0}([1, \xi \cdot \alpha_1]) \sim \prod_{\alpha_1}^* C_{p,0}([1, \xi])$ . This finishes the proof of the lemma.

At this moment we are able to prove the announced classification. The following theorem states it. The reader should compare it with theorem 2.5.1.

#### **2.5.13 THEOREM:** Let $\alpha$ and $\beta$ be ordinals.

If  $\alpha$  and  $\beta$  have different power, then

(a)  $C_p([1, \alpha])$  and  $C_p([1, \beta])$  are not linearly homeomorphic.

If  $\alpha$  and  $\beta$  have the same power and  $\xi$  is the initial ordinal of that power, then

- (b) If  $\xi = \omega$ , or  $\xi$  is a singular ordinal or both  $\alpha$ ,  $\beta \ge \xi^2$ , then  $C_p([1, \alpha]) \sim C_p([1, \beta])$  if and only if  $\max(\alpha, \beta) < [\min(\alpha, \beta)]^{\omega}$ .
- (c) If  $\xi$  is an uncountable regular ordinal and  $\alpha$ ,  $\beta \in [\xi, \xi^2]$ , fix ordinals  $\alpha_1, \beta_1 \leq \xi$  and  $\gamma, \delta < \xi$  such that  $\alpha = \xi \cdot \alpha_1 + \gamma$  and  $\beta = \xi \cdot \beta_1 + \delta$ . Then  $C_p([1, \alpha]) \sim C_p([1, \beta])$  if and only if  $\overline{\alpha_1} = \overline{\beta_1}$ .
- (d) If  $\xi$  is an uncountable regular ordinal,  $\alpha < \xi^2$  and  $\beta \ge \xi^2$ , then  $C_p([1, \alpha])$  and  $C_p([1, \beta])$  are not linearly homeomorphic.

**PROOF:** Suppose (a) or (d) does not hold for some ordinals  $\alpha$  and  $\beta$ . Then by corollary 1.2.21 it also does not hold for  $C_0([1, \alpha])$  and  $C_0([1, \beta])$ . This contradicts theorem

2.5.1 (a) or (d).

We now prove (b). If  $C_p([1, \alpha]) \sim C_p([1, \beta])$ , then  $C_0([1, \alpha]) \sim C_0([1, \beta])$  (corollary 1.2.21). So by theorem 2.5.1 max( $\alpha, \beta$ ) < [min( $\alpha, \beta$ )]<sup> $\omega$ </sup>.

We now prove the converse implication. Without loss of generality we may assume that  $\alpha \leq \beta$ , so suppose that  $\beta < \alpha^{\omega}$ .

Case 1:  $\xi = \omega$ .

Then we can apply theorem 2.4.7 (b).

*Case 2*:  $\alpha$ ,  $\beta \ge \xi^2$ .

By lemma 2.1.5  $\alpha = \xi^{\gamma} \cdot \lambda + \delta$  for some  $\gamma \ge 2$  with  $\gamma \le \alpha$ ,  $1 \le \lambda < \xi$  and  $\delta < \xi^{\gamma}$ . Notice that then  $\overline{\gamma} \le \overline{\alpha} = \overline{\xi}$  and  $\alpha < \xi^{\gamma+1}$ , thus  $\beta < \alpha^{\omega} \le \xi^{(\gamma+1)\omega} = \xi^{\gamma\omega}$  (lemma 2.1.19). This implies that  $\beta = \xi^{\gamma i} \cdot \mu + \varepsilon$  for some  $i \in \mathbb{N}$ ,  $\mu < \xi^{\gamma}$  with  $\mu > 0$  and  $\varepsilon < \xi^{\gamma i}$  (corollary 2.1.3). Thus

$$C_p([1, \alpha]) \sim C_{p, 0}([1, \xi^{\gamma} \cdot \lambda + \delta]) \quad (\text{lemma 2.3.9})$$
$$\sim C_{p, 0}([1, \delta + \xi^{\gamma} \cdot \lambda]) \quad (\text{lemma 2.3.8})$$
$$= C_{p, 0}([1, \xi^{\gamma} \cdot \lambda]) \qquad (\xi^{\gamma} \text{ is a prime component and } \lambda \neq 0)$$

In the same way

 $C_p([1, \beta]) \sim C_{p, 0}([1, \xi^{\gamma i} \cdot \mu]).$ 

CLAIM 1: For every  $p \in \mathbb{N}$  and  $1 \le v < \xi^{\gamma}$  we have  $C_{p,0}([1, \xi^{\gamma p} \cdot v]) \sim \prod_{\xi \in \mathcal{C}_{p,0}}^{*}([1, \xi^{\gamma}])$ .

First suppose  $p \in \mathbb{N}$  and v = 1. We prove by induction on p that

$$C_{p,0}([1,\xi^{\gamma p}]) \stackrel{4^{p+3}}{\sim} \Pi_{\xi}^{*} C_{p,0}([1,\xi^{\gamma}]).$$
(\*)

The case p = 1 follows from lemma 2.5.10. So let p > 1. We then have

. .

$$C_{p,0}([1,\xi^{\gamma p}]) = C_{p,0}([1,\xi^{\gamma(p-1)};\xi^{\gamma}])$$

$$\stackrel{4}{\sim} \Pi^{*}_{\xi^{\gamma}}C_{p,0}([1,\xi^{\gamma(p-1)}]) \quad (\text{lemma 2.5.7})$$

$$= \Pi^{*}_{\xi}C_{p,0}([1,\xi^{\gamma(p-1)}]) \quad (\text{proposition 2.1.13})$$

$$\stackrel{4^{p+2}}{\sim} \Pi^{*}_{\xi}\Pi^{*}_{\xi}C_{p,0}([1,\xi^{\gamma}]) \quad (\text{by induction and lemma 2.5.4})$$

$$= \Pi^{*}_{\xi}C_{p,0}([1,\xi^{\gamma}]) \quad (\text{page 52}).$$

Now suppose  $p \in \mathbb{N}$  and  $1 < v < \omega$ . Then by induction on v,

$$C_{p,0}([1,\xi^{\gamma p} \cdot v]) \sim C_{p,0}([1,\xi^{\gamma p} \cdot (v-1)]) \times C_{p,0}([1,\xi^{\gamma p}]) \quad (\text{lemmas 2.3.8 and 2.3.9})$$
$$\sim \Pi_{\xi}^{*} C_{p,0}([1,\xi^{\gamma}]) \qquad (\text{by induction}).$$

80

Finally suppose  $p \in \mathbb{N}$  and  $v = \mu + n$  with  $\mu \neq 0$  a limit ordinal and n finite. Then

$$C_{p,0}([1, \xi^{\gamma p} \cdot v]) \sim C_{p,0}([1, \xi^{\gamma p} \cdot \mu]) \times C_{p,0}([1, \xi^{\gamma p} \cdot n]) \qquad (\text{lemmas } 2.3.8 \text{ and } 2.3.9)$$

$$\sim \Pi^{*}_{\overline{\mu}}C_{p,0}([1, \xi^{\gamma p}]) \times C_{p,0}([1, \xi^{\gamma p} \cdot n]) \qquad (\text{lemma } 2.5.7)$$

$$\sim \Pi^{*}_{\overline{\mu}}C_{p,0}([1, \xi^{\gamma p}]) \qquad (\text{lemma } 2.5.4 \text{ and } (*))$$

$$= \Pi^{*}_{\overline{\xi}}C_{p,0}([1, \xi^{\gamma}]) \qquad (\overline{\mu} \cdot \overline{\xi} = \overline{\xi}),$$

and the claim is proved.

By the claim we immediately get

$$C_{p,\,0}([1,\,\beta]) \sim C_{p,\,0}([1,\,\xi^{\gamma i} \cdot \mu]) \sim \Pi^*_{\xi} C_{p,\,0}([1,\,\xi^{\gamma}])$$
 and

$$C_{p,0}([1,\alpha]) \sim C_{p,0}([1,\xi^{\gamma}\cdot\lambda]) \sim \Pi_{\xi}^{*}C_{p,0}([1,\xi^{\gamma}]),$$

which proves case 2.

*Case 3*:  $\xi$  is singular and  $\alpha \in [\xi, \xi^2]$ .

CLAIM 2: If  $\alpha \in [\xi, \xi^2]$ , then  $C_{p,0}([1, \alpha]) \sim C_{p,0}([1, \xi])$ .

Indeed, by lemma 2.5.12 we have

 $C_{p,0}([1, \alpha]) \sim \prod_{m=0}^{*} C_{p,0}([1, \xi])$ 

with  $\mathfrak{m} \leq \overline{\xi}$ . By lemma 2.5.11,  $C_{p,0}([1,\xi]) \stackrel{4^5}{\sim} \Pi_{\xi}^* C_{p,0}([1,\xi])$ , so

$$C_{p,0}([1, \alpha]) \sim \Pi_{\mathfrak{m}}^{*} C_{p,0}([1, \xi])$$
  
 
$$\sim \Pi_{\mathfrak{m}}^{*} \Pi_{\xi}^{*} C_{p,0}([1, \xi]) \qquad (\text{lemma 2.5.4})$$
  
 
$$= \Pi_{\xi}^{*} C_{p,0}([1, \xi]) \qquad (\mathfrak{m} \cdot \overline{\xi} = \overline{\xi})$$
  
 
$$\sim C_{p,0}([1, \xi])$$

This proves claim 2.

CLAIM 3: If  $\beta \ge \xi^2$  then  $C_{p,0}([1, \beta]) \sim C_{p,0}([1, \xi])$ .

Indeed, notice that  $\beta < \alpha^{\omega} \le (\xi^2)^{\omega}$ . So by case 2,  $C_{p,0}([1,\beta]) \sim C_{p,0}([1,\xi^2])$ . But by claim 2,  $C_{p,0}([1,\xi^2]) \sim C_{p,0}([1,\xi])$ , which proves claim 3.

By claims 2 and 3 we finished the proof of case 3, and therefore also the proof of (b).

For (c) first suppose that  $C_p([1, \alpha]) \sim C_p([1, \beta])$ . Then  $C_0([1, \alpha]) \sim C_0([1, \beta])$ 

(corollary 1.2.21) and thus  $\overline{\alpha_1} = \overline{\beta_1}$  (theorem 2.5.1 (c)). Now suppose  $\overline{\alpha_1} = \overline{\beta_1}$ . By lemma 2.5.12

$$C_{p,0}([1,\alpha]) \sim \prod_{\alpha_1}^* C_{p,0}([1,\xi]) = \prod_{\beta_1}^* C_{p,0}([1,\xi]) \sim C_{p,0}([1,\beta])$$

and (c) is proved.  $\Box$ 

The last theorem gives a complete isomorphical classification of the spaces  $C_p(X)$ , where X is a compact ordinal space. As announced in the introduction of this section, Kislyakov gave the same classification for the spaces  $C_0(X)$  (with X a compact ordinal space). However, he made a mistake in his proof. We now will point out his mistake, and indicate how it can be corrected.

Kislyakov states the following: "Let  $\alpha = \omega_{\gamma}$ . Since  $\alpha$  is singular, it follows that  $\gamma < \alpha$  and ...." (cf. [34, lemma 3.3]). But in example 2.1.14 we gave an example of a singular ordinal  $\omega_{\gamma}$  such that  $\gamma = \omega_{\gamma}$ .

An examination of our proofs tells us that if lemmas 2.5.4, 2.5.5 and 2.5.6 hold for function spaces endowed with the topology of uniform convergence, then all the other lemmas and theorems also hold for function spaces with this topology. Kislyakov proved lemmas 2.5.4 and 2.5.6 for those function spaces (cf. resp. lemma 1.2 and lemma 3.1 in [34]). In addition, lemma 2.5.5 is very easy to prove for function spaces endowed with the topology of uniform convergence. So our proof can be copied to get a correct proof of the classification of Kislyakov. It turns out that the proof one gets in this way differs from the proof of Kislyakov at two places. First of all corollary 3.3 in [34] has to be stated in a more general form (it becomes our lemma 2.5.9 for function spaces endowed with the topology of uniform convergence) and second, the proof of lemma 3.3 of [34] (which is our lemma 2.5.11 for function spaces endowed with the topology of uniform convergence) has to be fixed (the proof for the case  $\gamma < \omega_{\gamma}$  remains the same but the case  $\gamma = \omega_{\gamma}$  has to be added).

Finally we remark that Gulko and Oskin also proved theorem 2.5.1 (b) and (d) (in [28]), independently from Kislyakov. We were inspired by [34] because [28] contains no proofs. The other results in this section are new and were never published.

### §2.6. σ-compact ordinals

In this section we give a complete isomorphical classification of the spaces  $C_p(X)$ 

and  $C_0(X)$  where  $X = [1, \alpha)$ , for ordinals  $\alpha$  cofinal with  $\omega$ . Notice that these spaces are exactly the non-compact spaces which are a countable union of compact ordinal spaces. Therefore we call such an ordinal a  $\sigma$ -ordinal. If  $\alpha$  is also a prime component or an initial ordinal, we call it a  $\sigma$ -prime component or a  $\sigma$ -initial ordinal, respectively.

**2.6.1 LEMMA:** If  $\alpha$  is a  $\sigma$ -ordinal, then every closed and bounded subset of  $[1, \alpha)$  is compact.

**PROOF:** Let  $(\alpha_n)_n$  be a strictly increasing sequence of ordinals with limit  $\alpha$ . Let A be a closed and bounded subset of  $[1, \alpha)$ . Then there is  $n \in \mathbb{N}$  such that  $A \subset [1, \alpha_n]$ . For if not, then A would contain a closed discrete subset, which is not possible because A is bounded. Since A is closed in  $[1, \alpha_n]$  it is compact.  $\Box$ 

**2.6.2 REMARK:** From lemma 2.6.1 and corollary 1.2.21 we have for  $\sigma$ -ordinals  $\alpha$  and  $\beta$  that a linear homeomorphism  $\phi: C_p([1, \alpha)) \to C_p([1, \beta))$  considered as a map from  $C_0([1, \alpha))$  to  $C_0([1, \beta))$  is also a linear homeomorphism.

Furthermore, let  $\beta$  be an ordinal with  $cf(\beta) > \omega$ . By the methods of [24, Ex 3.1.27] it easily follows that every continuous function  $f: [1, \beta) \rightarrow \mathbb{R}$  is eventually constant. But this implies that  $[1, \beta)$  is pseudocompact. By this observation it follows that lemma 2.6.1 does not hold for  $\beta$  and that  $\alpha$  is a  $\sigma$ -ordinal if and only if  $[1, \alpha)$  is a non-compact non-pseudocompact space

The following lemma is the key lemma in the proof of the classification mentioned above.

**2.6.3 LEMMA:** Let X and Y be spaces such that  $X = X_1 \oplus X_2 \oplus X_3$  and  $Y = Y_1 \oplus Y_2 \oplus Y_3$ . Suppose  $\phi: C_0(X) \to C_0(Y)$  is a linear homeomorphism such that  $\operatorname{supp} X_1 \subset Y_1$  and  $\operatorname{supp} Y_2 \subset X_1 \oplus X_2$ . Then there is a linear embedding  $\theta: C_0(Y_2) \to C_0(X_2)$ .

**PROOF:** For each  $f \in C_0(Y_2)$  we define  $f^* \in C_0(Y)$  by  $f^*(y) = f(y)$  if  $y \in Y_2$  and  $f^*(y) = 0$  elsewhere. In a similar way we define for every  $g \in C_0(X_2)$ ,  $g^+ \in C_0(X)$ . Define  $\theta: C_0(Y_2) \to C_0(X_2)$  by  $\theta(f) = \phi^{-1}(f^*) | X_2$  and  $\psi: C_0(X_2) \to C_0(Y_2)$  by  $\psi(g) = \phi(g^+) | Y_2$ .

Then  $\theta$  and  $\psi$  are continuous linear functions. Furthermore for every  $h \in C_0(Y_2)$ , we have  $\psi(\theta(h)) = h$ . Indeed, assume to the contrary that  $\phi(\theta(h)^+) | Y_2 \neq h^* | Y_2$ . Then  $\theta(h)^+ | (X_1 \oplus X_2) \neq \phi^{-1}(h^*) | (X_1 \oplus X_2)$  since  $X_1 \oplus X_2$  is a neighborhood of supp  $Y_2$ 

and  $\phi$  is effective (corollary 1.2.15 (a)). Now  $h^* \equiv 0$  on  $Y_1$ , so  $\phi^{-1}(h^*) \equiv 0$  on  $X_1$ , since  $Y_1$  is a neighborhood of supp  $X_1$  and  $\phi^{-1}$  is effective. Furthermore  $\theta(h)^+ \equiv 0$  on  $X_1$ , so that  $\theta(h)^+ = \phi^{-1}(h^*)$  on  $X_1$ . This implies  $\theta(h)^+ | X_2 \neq \phi^{-1}(h^*) | X_2$ , which is impossible because both the left-hand side and the right-hand side are equal to  $\theta(h)$ . This is a contradiction and we conclude that  $\theta$  is a linear embedding.  $\Box$ 

**2.6.4 COROLLARY:** Let X and Y be spaces such that  $X = X_1 \oplus X_2$  and  $Y = Y_1 \oplus Y_2$ . Suppose  $\phi: C_0(X) \to C_0(Y)$  is a linear homeomorphism such that supp  $Y_1 \subset X_1$ . Then there is a linear embedding  $\theta: C_0(Y_1) \to C_0(X_1)$ .

**PROOF:** Take  $X_1 = Y_1 = \emptyset$  in lemma 2.6.3.

Notice that lemma 2.6.3 and corollary 2.6.4 also hold for the spaces  $C_p(X)$  and  $C_p(Y)$ .

The strategy of the proof of the classification is as follows: First we define a class of spaces, and we prove that for every  $\sigma$ -ordinal  $\alpha$  there is a space Y in this class such that  $C_p([1, \alpha)) \sim C_p(Y)$  (lemma 2.6.6). Then we prove that if X and Y are two spaces in this class, then  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$  if and only if X = Y (corollary 2.6.15 and lemma 2.6.16). From these results we then easily derive our classification (theorem 2.6.17).

For initial ordinals  $\alpha$  and  $\beta$  with  $\alpha \ge \beta \ge \omega$  we define the following classes of spaces: Case 1: If  $\alpha$  is singular or  $\omega$  and  $\beta$  is singular or  $\omega$  then

 $\mathcal{A}^{(\alpha,\beta)} = \{ [1,\omega^{\mu}] \oplus [1,\omega^{\tau}) : \mu \text{ a prime component, } \tau \text{ a } \sigma \text{-prime component or} \\ \tau = 1, \mu \ge \tau \ge 1, \overline{\omega^{\mu}} = \overline{\alpha}, \overline{\omega^{\tau}} = \overline{\beta} \},$ 

Case 2: If  $\alpha$  is uncountable regular and  $\beta$  is singular or  $\omega$  then

- $\mathcal{A}^{(\alpha, \beta)} = \{ [1, \omega^{\mu}] \oplus [1, \omega^{\tau}) : \mu \text{ a prime component, } \tau \text{ a } \sigma \text{-prime component or} \\ \tau = 1, \omega^{\mu} > \alpha^{2}, \overline{\omega^{\mu}} = \overline{\alpha}, \overline{\omega^{\tau}} = \overline{\beta} \}$ 
  - $\cup \{ [1, \alpha \cdot \xi] \oplus [1, \omega^{\tau}) : \tau \text{ a } \sigma \text{-prime component or } \tau = 1, \xi \text{ initial, } 1 \le \xi \le \alpha, \\ \overline{\omega^{\tau}} = \overline{\beta} \}.$

Case 3: If  $\alpha$  is singular or  $\omega$  and  $\beta$  is uncountable regular then

- $\mathcal{A}^{(\alpha,\beta)} = \{ [1,\omega^{\mu}] \oplus [1,\omega^{\tau}) : \mu \text{ a prime component, } \tau \text{ a } \sigma \text{-prime component,} \\ \omega^{\tau} > \beta^{2}, \overline{\omega^{\mu}} = \overline{\alpha}, \overline{\omega^{\tau}} = \overline{\beta} \}$ 
  - $∪ \{ [1, ω<sup>μ</sup>] ⊕ [1, β·η) : μ a prime component, η = ρ·ω with ρ initial or η$  $σ-initial, ω≤η≤β, <math>\overline{ω}^{μ} = \overline{\alpha} \}.$

*Case 4*: If  $\alpha$  and  $\beta$  are uncountable regular then

- $\mathcal{A}^{(\alpha, \beta)} = \{ [1, \omega^{\mu}] \oplus [1, \omega^{\tau}) : \mu \text{ a prime component, } \tau \text{ a } \sigma \text{-prime component,} \\ \mu \ge \tau, \, \omega^{\mu} > \alpha^{2}, \, \omega^{\tau} > \beta^{2}, \, \overline{\omega}^{\mu} = \overline{\alpha}, \, \overline{\omega}^{\tau} = \overline{\beta} \}$ 
  - $\cup \{ [1, \alpha \cdot \xi] \oplus [1, \omega^{\tau}) : \tau \text{ a } \sigma \text{-prime component, } \xi \text{ initial, } 1 \le \xi \le \alpha, \\ \alpha \cdot \xi \ge \omega^{\tau}, \, \omega^{\tau} > \beta^2, \, \overline{\omega^{\tau}} = \overline{\beta} \}$
  - ∪ {[1, ω<sup>μ</sup>] ⊕ [1, β·η): μ a prime component, η = ρ·ω with ρ initial or η σ-initial, ω≤η≤β, ω<sup>μ</sup>≥β·η,  $\overline{\omega}^{\mu} = \overline{\alpha}$ }
  - $\cup \{ [1, \alpha \cdot \xi] \oplus [1, \beta \cdot \eta) : \xi \text{ initial, } \eta = \rho \cdot \omega \text{ with } \rho \text{ initial or } \eta \text{ } \sigma \text{-initial,} \\ 1 \le \xi \le \alpha, \omega \le \eta \le \beta, \alpha \cdot \xi \ge \beta \cdot \eta \}.$

Now let  $\beta \ge \omega$  be an initial ordinal.

Case 5: If  $\beta$  is singular or  $\beta = \omega$  then

 $\mathscr{B}^{(\beta)} = \{ [1, \omega^{\tau}) : \tau \text{ a } \sigma \text{-prime component or } \tau = 1, \ \overline{\omega^{\tau}} = \overline{\beta} \}.$ 

Case 6: If  $\beta$  is uncountable regular then

- $\mathscr{B}^{(\beta)} = \{ [1, \omega^{\tau}) : \tau \text{ a } \sigma \text{-prime component, } \omega^{\tau} > \beta^2, \ \overline{\omega^{\tau}} = \overline{\beta} \}$ 
  - $\cup$  {[1,  $\beta \cdot \eta$ ]:  $\eta = \rho \cdot \omega$  with  $\rho$  initial or  $\eta \sigma$ -initial,  $\omega \le \eta \le \beta$ }.

Now let

 $\mathcal{A} = \bigcup \{ \mathcal{A}^{(\alpha, \beta)} : (\alpha, \beta) \text{ as in case } 1, 2, 3 \text{ or } 4 \},\$ 

and

 $\mathcal{B} = \bigcup \{ \mathcal{B}^{(\beta)} : \beta \text{ as in case 5 or 6} \}.$ 

The class of spaces that we are currently interested in is  $\mathcal{A} \cup \mathcal{B}$ . Notice that whenever  $X = [1, \phi] \oplus [1, \psi) \in \mathcal{A}$ , then  $\phi \ge \psi$ .

For every space  $X \in \mathcal{A} \cup \mathcal{B}$  we need to fix a certain decomposition. First we will assign to certain ordinals  $\mu$  a fixed sequence  $(\mu_i)_i$  of ordinals. If  $\mu = 1$ , put  $\mu_i = 0$  for each  $i \in \mathbb{N}$ . If  $\mu = \tau \cdot \omega$  for some  $\tau$ , put  $\mu_i = \tau \cdot i$  for each  $i \in \mathbb{N}$ , and if  $\mu$  is a  $\sigma$ -ordinal not of the form  $\tau \cdot \omega$ , let  $(\mu_i)_i$  be a strictly increasing sequence of ordinals such that  $\mu_i \rightarrow \mu$  and  $1 \le \mu_i < \mu$  for each  $i \in \mathbb{N}$ . We now define the desired decompositions:

If  $X = [1, \phi] \oplus [1, \omega^{\tau}) \in \mathcal{A}$ , then  $X = [1, \phi] \oplus [1, \omega^{\tau_1}] \oplus [1, \omega^{\tau_2}] \oplus \cdots$  (this is true because for every *i*,  $\omega^{\tau_i}$  is a prime component).

If  $X = [1, \phi] \oplus [1, \beta \cdot \eta) \in \mathcal{A}$ , then  $X = [1, \phi] \oplus [1, \beta \cdot \eta_1] \oplus [1, \beta \cdot \eta_2] \oplus \cdots$ 

If  $X = [1, \omega^{\tau}) \in \mathcal{B}$ , then  $X = [1, \omega^{\tau_1}] \oplus [1, \omega^{\tau_2}] \oplus \cdots$ 

If  $X = [1, \beta \cdot \eta) \in \mathcal{B}$ , then  $X = [1, \beta \cdot \eta_1] \oplus [1, \beta \cdot \eta_2] \oplus \cdots$ 

If for  $X \in \mathcal{A} \cup \mathcal{B}$  we write  $X = \bigoplus_{i=1}^{\infty} X_i$ , then we implicitly mean that the  $X_i$  are as above.

Now we are going to prove that for every  $\sigma$ -ordinal  $\phi$  there is a space  $Y \in \mathcal{A} \cup \mathcal{B}$  and

a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of  $[1, \phi)$  such that  $C_p(X_i) \sim C_p(Y_i)$ . We first need the following

**2.6.5 LEMMA:** Let  $\beta \ge \omega$  be an initial ordinal and  $\tau$  a successor or a  $\sigma$ -ordinal, such that  $\overline{\omega}^{\tau} = \overline{\beta}$ .

- (a) If  $\beta = \omega$ ,  $\beta$  is singular, or  $\omega^{\tau'} \ge \beta^2$  and if  $\tau$  is not a prime component, then there is a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of  $[1, \omega^{\tau})$ , such that for every *i*,  $C_p(X_i) \sim C_p([1, \omega^{\tau' \cdot i}])$ . In particular  $C_p([1, \omega^{\tau})) \sim C_p([1, \omega^{\tau' \cdot \omega}))$ .
- (b) If  $\beta$  is uncountable regular and  $\omega^{\tau'} \in [\beta, \beta^2)$  then one of the following holds: (i)  $\omega^{\tau} = \beta \cdot \eta$  with  $\eta \sigma$ -initial and  $\omega \le \eta \le \beta$ , or
  - (ii) there is an initial ordinal  $\eta$  such that  $\omega \le \eta \le \beta$  and there is a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of  $[1, \omega^{\tau})$ , such that for every i,  $C_p(X_i) \sim C_p([1, \beta \cdot \eta \cdot i])$ . In particular  $C_p([1, \omega^{\tau})) \sim C_p([1, \beta \cdot \eta \cdot \omega))$ . Furthermore  $\beta \cdot \eta \cdot \omega \le \omega^{\tau}$ .

**PROOF:** First notice that  $[1, \omega^{\tau}) = \bigoplus_{i=1}^{\infty} [1, \omega^{\tau_i}]$ , where  $\tau_i = \nu$  if  $\tau = \nu + 1$  and  $(\tau_i)_i$  is a strictly increasing sequence (not necessary equal to the fixed sequence associated with  $\tau$ ) with limit  $\tau$  if  $\tau$  is a  $\sigma$ -ordinal. Both in (a) and (b)(ii) we will get  $X_i = [1, \omega^{\tau_i}]$ .

We first prove (a). Since  $\tau$  is not a prime component, we have  $\tau' < \tau < \tau' \cdot \omega$  (lemma 2.1.19), and we can assume  $\tau' \leq \tau_i$  for each *i*. Now  $\omega^{\tau'} \leq \omega^{\tau_i} < \omega^{\tau} < (\omega^{\tau'})^{\omega}$ . With the help of proposition 2.1.13 it now easily follows that  $\overline{\omega^{\tau_i}} = \overline{\omega^{\tau'}} = \overline{\beta}$ . Since  $\beta = \omega$ ,  $\beta$  is singular, or  $\omega^{\tau'} \geq \beta^2$ , we can apply theorem 2.5.13 (b) to obtain

 $C_p([1, \omega^{\tau_i}]) \sim C_p([1, \omega^{\tau'}])$ 

By a similar argument

 $C_p([1, \omega^{\tau'}]) \sim C_p([1, \omega^{\tau' \cdot i}]),$ 

whence by lemma 2.3.7,

$$C_p([1, \omega^{\tau})) = C_p(\bigoplus_{i=1}^{\infty} [1, \omega^{\tau_i}]) \sim C_p(\bigoplus_{i=1}^{\infty} [1, \omega^{\tau' \cdot i}]) = C_p([1, \omega^{\tau' \cdot \omega})).$$

For (b) we distinguish two cases.

Case 1:  $\omega^{\tau} > \beta^2$ .

Since  $\beta$  is initial, it is a prime component, so by theorem 2.1.21  $\beta = \omega^{\rho}$  for some ordinal  $\rho$ . Then  $\rho \cdot 2 < \tau$  and we can assume that  $\rho \cdot 2 \leq \tau_i$  for each *i*. We conclude that  $\omega^{\tau_i} \geq \beta^2$ . Since  $\omega^{\tau'} < \beta^2$  we have  $\tau' < \rho \cdot 2 < \tau$ , so  $\tau < \rho \cdot \omega$  by lemma 2.1.19. Thus  $\beta^2 \leq \omega^{\tau_i} < (\beta^2)^{\omega}$ . By proposition 2.1.13 it now easily follows that  $\overline{\omega^{\tau_i}} = \overline{\beta}$ , so by theorem 2.5.13 (b)

$$C_p([1, \omega^{\tau_i}]) \sim C_p([1, \beta^2])$$

and similarly

$$C_p([1, \beta^2]) \sim C_p([1, \beta^2 \cdot i]).$$

Consequently,

$$C_p([1, \omega^{\tau})) \sim C_p([1, \beta^2 \cdot \omega)).$$

Since  $\beta^2 \cdot \omega$  is the smallest prime component larger than  $\beta^2$  (lemma 2.1.19),  $\omega^{\tau} \ge \beta^2 \cdot \omega$ , and so we have established (ii) for  $\eta = \beta$ .

Case 2:  $\omega^{\tau} \leq \beta^2$ .

If  $\omega^{\tau} = \beta^2$ ,  $\omega^{\tau}$  satisfies (i), so we may assume  $\omega^{\tau} < \beta^2$ . There are ordinals  $\delta, \eta^* < \beta$ such that  $\omega^{\tau} = \beta \cdot \eta^* + \delta$ . Since  $\delta < \omega^{\tau}$  and  $\omega^{\tau}$  is a prime component,  $\delta = 0$ . In addition, since  $\beta$  is regular and  $cf(\omega^{\tau}) = \omega, \eta^* \ge \omega$ . If  $\eta^*$  is initial we are done, so suppose  $\eta^*$  is not initial. Let  $\eta$  be the initial ordinal of the same power as  $\eta^*$ . Then  $\omega \le \eta < \eta^*$ , hence  $\beta \cdot \eta < \beta \cdot \eta^* = \omega^{\tau}$ , so we can assume  $\beta \cdot \eta \le \omega^{\tau_i}$  for each *i*. Write  $\omega^{\tau_i} = \beta \cdot \eta_i + \delta_i$  with  $\eta_i, \delta_i < \beta$ . Then  $\delta_i = 0$  since  $\omega^{\tau_i}$  is a prime component, and  $\beta \cdot \eta \le \beta \cdot \eta_i < \beta \cdot \eta^*$ , whence  $\overline{\eta} = \overline{\eta_i}$ . Since  $\omega^{\tau_i} = \beta \cdot \eta_i, \beta \cdot \eta \in [\beta, \beta^2]$ , it follows by theorem 2.5.13 (c) that  $C_p([1, \omega^{\tau_i}]) \sim C_p([1, \beta \cdot \eta])$ . Since  $C_p([1, \beta \cdot \eta]) \sim C_p([1, \beta \cdot \eta \cdot i])$ , (ii) can be established as in case 1.  $\Box$ 

**2.6.6 LEMMA:** Let  $\phi$  be a  $\sigma$ -ordinal. Then there is a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of [1,  $\phi$ ) and a space  $Y \in \mathcal{A} \cup \mathcal{B}$  such that  $C_p(X_i) \sim C_p(Y_i)$  (where  $Y_i$  is the *i*<sup>th</sup> term in the fixed decomposition of Y). In particular  $C_p([1, \phi]) \sim C_p(Y)$  and  $C_0([1, \phi]) \sim C_0(Y)$ .

**PROOF:** By theorems 2.1.16 and 2.1.21 there are ordinals  $\psi$  and  $\tau$  such that  $\phi = \psi + \omega^{\tau}$ , with  $\tau > 0$  and  $\psi = 0$  or  $\psi \ge \omega^{\tau}$ . Notice that  $\tau$  is a successor or a  $\sigma$ -ordinal. Let  $\alpha$  and  $\beta$  be initial such that  $\overline{\alpha} = \overline{\psi}$  and  $\overline{\beta} = \overline{\omega^{\tau}}$ . Notice that  $\omega^{\tau'} \ne \beta^2$ , because if not then  $\omega^{\tau'} = \beta^2 = \omega^{\rho \cdot 2}$  for some prime component  $\rho$  which implies that  $\tau' = \rho \cdot 2$  is not a prime component.

Case 1:  $\psi = 0$ ,  $\tau' = \tau$ .

If  $\beta$  is singular,  $\beta = \omega$  or  $\omega^{\tau} > \beta^2$ , we have  $[1, \phi] = [1, \omega^{\tau}] \in \mathcal{B}$ .

If  $\beta$  is uncountable regular and  $\omega^{\tau} \in [\beta, \beta^2)$ , then by lemma 2.6.5 we have either  $[1, \omega^{\tau}) \in \mathcal{B}$  or there is  $Y \in \mathcal{B}$  such that  $C_p([1, \phi]) = C_p([1, \omega^{\tau}]) \sim C_p(Y)$  with the desired decomposition and such that if  $Y = [1, \delta)$  then  $\delta \le \omega^{\tau}$ . *Case 2:*  $\psi = 0, \tau' \ne \tau$ .

If  $\beta$  is singular,  $\beta = \omega$  or  $\omega^{\tau'} > \beta^2$ , we have by lemma 2.6.5 (a) that there exists a space  $Y \in \mathcal{B}$  such that  $C_p([1, \phi]) = C_p([1, \omega^{\tau}]) \sim C_p(Y)$  with the desired decomposition

86

 $(Y \in \mathcal{B} \text{ because } Y = [1, \omega^{\tau' \cdot \omega}) \text{ and } \tau' \cdot \omega \text{ is a } \sigma \text{-prime component}).$ 

If  $\beta$  is uncountable regular and  $\omega^{\tau'} < \beta^2$ , by lemma 2.6.5 (b) there is a space  $Y \in \mathcal{B}$  such that  $C_p([1, \phi]) \sim C_p(Y)$  and which has the desired decomposition. Case  $\beta: \psi \ge \omega^{\tau}, \tau' = \tau$ .

There is an ordinal  $\mu \ge \tau$  such that  $\psi' = \omega^{\mu}$ . Notice that  $\tau' \le \mu$ . By case 1, there is a space  $Y' = [1, \delta] \in \mathcal{B}$  with  $\delta \le \omega^{\tau}$ , such that  $C_p([1, \omega^{\tau'})) \sim C_p(Y')$  and which has the desired decomposition.

If  $\alpha$  is singular,  $\alpha = \omega$  or  $\omega^{\mu'} > \alpha^2$ , by theorem 2.5.13 (b)  $C_p([1, \omega^{\mu'}]) \sim C_p([1, \psi])$ (because  $\omega^{\mu'} \le \omega^{\mu} \le \psi < \omega^{\mu+1} < (\omega^{\mu'})^{\omega}$ ). Since  $\omega^{\mu'} \ge \omega^{\tau'} = \omega^{\tau} \ge \delta$ ,

 $[1, \omega^{\mu'}] \oplus [1, \delta) = [1, \omega^{\mu'}] \oplus Y' \in \mathcal{A}.$ 

If  $\alpha$  is uncountable regular and  $\omega^{\mu'} < \alpha^2$  we have to consider two subcases *Subcase 3.1*:  $\psi \ge \alpha^2$ .

Then  $\alpha^2 \le \psi < (\omega^{\mu'})^{\omega} \le (\alpha^2)^{\omega}$ , so by theorem 2.5.13 (b),  $C_p([1, \psi]) \sim C_p([1, \alpha^2])$ . Since  $\alpha^2 > \omega^{\mu'} \ge \omega^{\tau'} \ge \delta$ , the space  $Y = [1, \alpha^2] \oplus Y' \in \mathcal{A}$  and  $C_p([1, \phi]) \sim C_p(Y)$  and moreover has the desired decomposition.

#### Subcase 3.2: $\psi < \alpha^2$ .

Then  $\psi = \alpha \cdot \xi^* + \delta$  with  $1 \le \xi^* < \alpha$  and  $\delta < \alpha$ . If we let  $\xi$  the initial ordinal with the same power as  $\xi^*$ , then by theorem 2.5.13 (c),  $C_p([1, \psi]) \sim C_p([1, \alpha \cdot \xi])$ . Now let  $Y = [1, \alpha \cdot \xi] \oplus Y'$ . It is easily seen that if  $\alpha \cdot \xi < \delta$ , then  $Y \approx Y' \in \mathcal{B}$  and otherwise  $Y \in \mathcal{A}$ . *Case* 4:  $\psi \ge \omega^{\tau}$ ,  $\tau' \neq \tau$ .

This is a combination of the cases 2 and 3.

Notice that the last remark in the lemma easily follows from lemma 2.3.7 and corollary 1.2.21.

Now we are going to prove that for every X,  $Y \in \mathcal{A} \cup \mathcal{B}$  we have  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$  if and only if X = Y. For that we first have to do some prepatory work.

#### 2.6.7 LEMMA:

(a) If  $\theta: C_0([1, \omega^{\mu}]) \to C_0([1, \omega^{\nu}])$  is a linear embedding with  $\mu, \nu \ge 1$ , then (i)  $\mu < \nu \cdot \omega$ , hence  $\omega^{\mu} < (\omega^{\nu})^{\omega}$ , and

(*ii*) if  $\mu$  is a prime component, then  $\mu \leq \nu$ , hence  $\omega^{\mu} \leq \omega^{\nu}$ .

(b) Let  $\alpha$  be an uncountable regular ordinal and  $\xi, \eta \in [1, \alpha]$ . If  $\theta: C_0([1, \alpha \cdot \xi]) \to C_0([1, \alpha \cdot \eta])$  is a linear embedding, then  $\overline{\xi} \leq \overline{\eta}$ .

**PROOF:** We first prove (a). Suppose  $\mu \ge v \cdot \omega$ . Then  $\omega^{\mu} \ge (\omega^{\nu})^{\omega}$ , so by corollary 2.5.3

 $C_0([1, \omega^v])$  has smaller linear dimension than  $C_0([1, \omega^\mu])$ , which contradicts the fact that  $\theta: C_0([1, \omega^\mu]) \rightarrow C_0([1, \omega^v])$  is a linear embedding. This proves (i). For (ii) let  $\mu$  be a prime component. Since  $\mu < v \cdot \omega$ , by lemma 2.1.19,  $\mu \le v' \le v$ .

For (b) suppose  $\overline{\eta} < \overline{\xi}$ . Then  $\eta < \xi$ , so there is a linear embedding  $\phi: C_0([1, \alpha \cdot \eta]) \rightarrow C_0([1, \alpha \cdot \xi])$ . This gives that  $C_0([1, \alpha \cdot \eta])$  and  $C_0([1, \alpha \cdot \xi])$  have the same linear dimension, so by theorem 2.5.1 (c)  $\overline{\eta} = \overline{\xi}$ . Contradiction.  $\Box$ 

#### 2.6.8 LEMMA:

- (a) Let  $X = Z \oplus [1, \omega)$  with Z a compact space, and  $Y = \bigoplus_{i=1}^{\infty} Z_i$  where each  $Z_i$  is an infinite compact space. Then  $C_0(X)$  and  $C_0(Y)$  are not linearly homeomorphic.
- (b) Let  $X = Z_1 \oplus Z_2$  with  $Z_1$  an infinite compact space. Then  $C_0(X)$  is not linearly homeomorphic to  $C_0([1, \omega))$ .

**PROOF:** For (a) suppose that  $C_0(X)$  is linearly homeomorphic to  $C_0(Y)$ . Then by corollary 1.2.15 (b) there is  $n \in \mathbb{N}$  such that  $\sup Z \subset \bigoplus_{i=1}^{n} Z_i$ . Again by corollary 1.2.15 (b) there is  $m \in \mathbb{N}$  such that  $\sup Z_{n+1} \subset Z \oplus [1, m]$ . By lemma 2.6.3, there is a linear embedding  $\theta: C_0(Z_{n+1}) \to C_0([1, m]) = \mathbb{R}^m$ . Since  $Z_{n+1}$  is infinite we have a contradiction, because the algebraic dimension of  $C_0(Z_{n+1})$  is infinite.

For (b) suppose that  $C_0(X)$  is linearly homeomorphic to  $C_0([1, \omega))$ . Then by corollary 1.2.15 (b), there is  $m \in \mathbb{N}$  such that  $\operatorname{supp} Z_1 \subset [1, m]$ . By corollary 2.6.4, there is a linear embedding  $\theta: C_0(Z_1) \to C_0[1, m] = \mathbb{R}^m$ . Again we have a contradiction.  $\Box$ 

**2.6.9 LEMMA:** Let  $X = [1, \xi_1] \oplus [1, \xi_2)$  and  $Y = [1, \eta_1] \oplus [1, \eta_2)$  where  $\xi_2$  and  $\eta_2$  are  $\sigma$ -prime components,  $\xi_1 \ge \xi_2$  and  $\eta_1 \ge \eta_2$ . Then  $C_0(X) \sim C_0(Y)$  implies  $\overline{\xi_1} = \overline{\eta_1}$  and  $\overline{\xi_2} = \overline{\eta_2}$ .

**PROOF:** Suppose  $\overline{\xi_1} < \overline{\eta_1}$  and  $C_0(X) \sim C_0(Y)$ . By corollary 1.2.15 (b) there is  $\delta < \xi_2$  such that supp  $[1, \eta_1] \subset [1, \xi_1 + \delta]$ , which implies by corollary 2.6.4 that there is a linear embedding  $\theta$ :  $C_0([1, \eta_1]) \rightarrow C_0([1, \xi_1 + \delta])$ . Since  $\overline{\xi_1 + \delta} = \overline{\xi_1}, \overline{\xi_1 + \delta} < \overline{\eta_1}$  and so  $\xi_1 + \delta < \eta_1$ . But then there is also a linear embedding  $\phi$ :  $C_0([1, \xi_1 + \delta]) \rightarrow C_0([1, \eta_1])$  and we conclude that  $C_0([1, \xi_1 + \delta])$  and  $C_0([1, \eta_1])$  have the same linear dimension. This contradicts theorem 2.5.1 (a). By symmetry we conclude that  $\overline{\xi_1} = \overline{\eta_1}$ .

Now suppose  $\overline{\xi_2} < \overline{\eta_2}$ . By corollary 1.2.15 (b) there is  $\delta < \eta_2$  such that supp  $[1, \xi_1] \subset [1, \eta_1 + \delta]$ . Since  $\eta_2$  is a prime component,  $Y \approx [1, \eta_1 + \delta] \oplus [1, \eta_2)$ .

CLAIM: There is an ordinal  $\tau < \eta_2$ , such that  $\overline{\tau} > \overline{\xi_2}$ .

Indeed, choose  $\omega_{\alpha}$  and  $\omega_{\beta}$  such that  $\overline{\xi_2} = \overline{\omega_{\alpha}}$  and  $\overline{\eta_2} = \overline{\omega_{\beta}}$ . Notice that  $\alpha + 1 \le \beta$  (theorem 2.1.6). If  $\alpha + 1 < \beta$ , then  $\tau = \omega_{\alpha+1}$  satisfies the claim. If  $\alpha + 1 = \beta$ , then  $\omega_{\beta}$  is regular (theorem 2.1.10). Since  $\eta_2$  is a  $\sigma$ -ordinal, it follows that  $\omega_{\beta} < \eta_2$ , so let  $\tau = \omega_{\beta}$ .

Now choose  $\sigma < \xi_2$  such that supp  $[1, \tau] \subset [1, \xi_1] \oplus [1, \sigma]$ . By lemma 2.6.3, there is a linear embedding  $\theta: C_0([1, \tau]) \to C_0([1, \sigma])$ . Since  $\overline{\sigma} \le \overline{\xi_2} < \overline{\tau}, \sigma < \tau$ . But then there is also a linear embedding  $\phi: C_0([1, \sigma]) \to C_0([1, \tau])$  and we may conclude that  $C_0([1, \sigma])$  and  $C_0([1, \tau])$  have the same linear dimension. This contradicts theorem 2.5.1 (a). By symmetry we conclude that  $\overline{\xi_2} = \overline{\eta_2}$ .

#### **2.6.10 LEMMA:** Let $\alpha$ be an initial ordinal.

- (a) Let  $X = [1, \omega^{\mu}] \oplus [1, \beta)$  and  $Y = [1, \omega^{\sigma}] \oplus [1, \gamma)$ , where  $1 \le \mu \le \sigma$  are prime components,  $\omega^{\mu} \ge \beta$  and  $\omega^{\sigma} \ge \gamma$ . Then  $C_0(X) \sim C_0(Y)$ , implies  $\mu = \sigma$ .
- (b) Let  $X = [1, \alpha \xi] \oplus [1, \beta)$  and  $Y = [1, \alpha \eta] \oplus [1, \gamma)$  where  $\alpha$  is uncountable regular,  $1 \le \xi \le \eta < \alpha$ ,  $\xi$  and  $\eta$  are initial,  $\alpha \xi \ge \beta$ , and  $\alpha \eta \ge \gamma$ . Then  $C_0(X) \sim C_0(Y)$  implies  $\xi = \eta$ .

**PROOF:** For (a), by corollary 1.2.15 (b) there is  $\delta < \beta$  such that supp  $[1, \omega^{\sigma}] \subset [1, \omega^{\mu}] \oplus [1, \delta]$ . Since  $\omega^{\mu}$  is a prime component, we have  $[1, \omega^{\mu}] \oplus [1, \delta] \approx [1, \omega^{\mu}]$ . Hence by corollary 2.6.4, there is a linear embedding  $\theta: C_0([1, \omega^{\sigma}]) \rightarrow C_0([1, \omega^{\mu}])$ . Then by lemma 2.6.7 (a) we have  $\sigma \leq \mu$ . Since by assumption  $\sigma \geq \mu$ ,  $\sigma = \mu$ .

For the proof of (b), let  $\delta < \beta$  be such that  $\sup[1, \alpha \cdot \eta] \subset [1, \alpha \cdot \xi] \oplus [1, \delta]$ . Since  $\delta < \alpha \cdot \xi$  we have  $\delta + \alpha \cdot \xi = \alpha \cdot \xi$  and so  $[1, \alpha \cdot \xi] \oplus [1, \delta] \approx [1, \alpha \cdot \xi]$ . But then by corollary 2.6.4 there is a linear embedding  $\theta: C_0([1, \alpha \cdot \eta]) \rightarrow [1, \alpha \cdot \xi])$ . Hence by lemma 2.6.7 (b),  $\overline{\eta} \leq \overline{\xi}$ . Since  $\eta$  and  $\xi$  are initial,  $\eta \leq \xi$  and so  $\eta = \xi$  (because by assumption  $\eta \geq \xi$ ).  $\Box$ 

#### **2.6.11 LEMMA:** Let $\alpha$ be an initial ordinal.

- (a) Let  $X = Z_1 \oplus [1, \omega^{\delta})$  and  $Y = Z_2 \oplus [1, \omega^{\tau \cdot \omega})$ , where  $Z_1$  and  $Z_2$  are compact spaces,  $\delta$ ,  $\tau$  are prime components,  $\delta = 1$  or  $\delta$  is a  $\sigma$ -ordinal,  $1 \le \delta \le \tau$ , and  $\overline{\omega^{\delta}} = \overline{\omega^{\tau}} = \overline{\alpha}$ . If  $\alpha$  is singular,  $\alpha = \omega$  or  $\omega^{\delta} \ge \alpha^2$ , then  $C_0(X)$  and  $C_0(Y)$  are not linearly homeomorphic.
- (b) Let  $X = Z_1 \oplus [1, \alpha \cdot \xi)$  and  $Y = Z_2 \oplus [1, \alpha \cdot \eta \cdot \omega)$ , where  $Z_1$  and  $Z_2$  are compact spaces,  $\xi$  is  $\sigma$ -initial,  $\eta$  is initial, and  $\omega \le \xi \le \eta \le \alpha$ . If  $\alpha$  is uncountable regular, then  $C_0(X)$  and  $C_0(Y)$  are not linearly homeomorphic.

**PROOF:** For (a) suppose the  $C_0(X) \sim C_0(Y)$ . Then by lemma 2.6.8 (a),  $\delta > 1$ . By

corollary 1.2.15 (b) there is  $n \in \mathbb{N}$  such that  $\sup Z_1 \subset Z_2 \oplus [1, \omega^{\tau \cdot n}]$ . Again by corollary 1.2.15 (b) there is  $\delta_i$  in the fixed sequence associated with  $\delta$  such that  $\sup [1, \omega^{\tau \cdot (n+1)}] \subset Z_1 \oplus [1, \omega^{\delta_i}]$ . By lemma 2.6.3 and the fact that  $C_0([1, \omega^{\tau \cdot (n+1)}]) \sim C_0([1, \omega^{\tau}])$  (theorem 2.5.13 (b)), there is a linear embedding  $\Theta: C_0([1, \omega^{\tau}]) \to C_0([1, \omega^{\delta_i}])$ . Then by lemma 2.6.7 (a),  $\tau \leq \delta_i < \delta$ , which is a contradiction.

For (b) suppose  $C_0(X) \sim C_0(Y)$ . There is  $n \in \mathbb{N}$  such that  $\operatorname{supp} Z_1 \subset Z_2 \oplus [1, \alpha \cdot \eta \cdot n]$ and ξi in the fixed sequence associated with ξ such that supp  $[1, \alpha \cdot \eta \cdot (n+1)] \subset Z_1 \oplus [1, \alpha \cdot \xi_i]$ . By lemma 2.6.3 and the fact that  $C_0([1, \alpha \cdot \eta \cdot (n+1)]) \sim C_0([1, \alpha \cdot \eta])$  (theorem 2.5.13 (c)), there is a linear embedding  $\theta: C_0([1, \alpha \cdot \eta]) \to C_0([1, \alpha \cdot \xi_i])$ , so by lemma 2.6.7 (b),  $\overline{\eta} \leq \overline{\xi_i} < \overline{\xi}$ , which is a contradiction.

**2.6.12 LEMMA:** Let  $\alpha$  be an initial ordinal.

- (a) Let  $X = Z_1 \oplus [1, \omega^{\delta})$  and  $Y = Z_2 \oplus [1, \omega^{\tau})$ , where  $Z_1$  and  $Z_2$  are compact spaces,  $\delta$ ,  $\tau$  are  $\sigma$ -prime components or 1,  $1 \le \delta \le \tau$ , and  $\overline{\omega^{\delta}} = \overline{\omega^{\tau}} = \overline{\alpha}$ . If  $\alpha$  is singular,  $\alpha = \omega$  or  $\omega^{\delta} \ge \alpha^2$ , then  $C_0(X) \sim C_0(Y)$  implies  $\delta = \tau$ .
- (b) Let  $X = Z_1 \oplus [1, \alpha \cdot \xi)$  and  $Y = Z_2 \oplus [1, \alpha \cdot \eta)$ , where  $Z_1$  and  $Z_2$  are compact spaces,  $\xi$  and  $\eta$  are  $\sigma$ -initial or of the form  $\tau \cdot \omega$  with  $\tau$  initial, and  $\omega \le \xi \le \eta \le \alpha$ . If  $\alpha$  is uncountable regular, then  $C_0(X) \sim C_0(Y)$  implies  $\xi = \eta$ .

**PROOF:** For (a) suppose  $\delta < \tau$ . By lemma 2.6.8 (a),  $\delta > 1$ . By corollary 1.2.15 (b), there is  $\tau_i$  in the fixed sequence associated with  $\tau$  such that  $\operatorname{supp} Z_1 \subset Z_2 \oplus [1, \omega^{\tau_i}]$ . Now let j > i. Again by corollary 1.2.15 (b), there is  $\delta_k$  in the fixed sequence associated with  $\delta$  such that  $\operatorname{supp} [1, \omega^{\tau_j}] \subset Z_1 \oplus [1, \omega^{\delta_k}]$ . By lemma 2.6.3 there is a linear embedding  $\theta: C_0([1, \omega^{\tau_j}]) \to C_0([1, \omega^{\delta_k}])$ . So by lemma 2.6.7 (a), we have  $\tau_j < \delta_k \cdot \omega \le \delta \cdot \omega$ , which implies  $\delta < \tau \le \delta \cdot \omega$ . So since  $\delta$  and  $\tau$  are prime components, we have  $\tau = \delta \cdot \omega$ . But this contradicts lemma 2.6.11.

For (b) suppose  $\xi < \eta$ . There is  $\eta_i \ge 1$  in the fixed sequence associated with  $\eta$  such that supp  $Z_1 \subset Z_2 \oplus [1, \alpha \cdot \eta_i]$ . For j > i, there is  $\xi_k \ge 1$  in the fixed sequence associated with  $\xi$  such that supp  $[1, \alpha \cdot \eta_j] \subset Z_2 \oplus [1, \alpha \cdot \xi_k]$ . By lemma 2.6.3, there is a linear embedding  $\theta: C_0([1, \alpha \cdot \eta_j]) \to C_0([1, \alpha \cdot \xi_k])$ . By lemma 2.6.7 (b),  $\overline{\eta_j} \le \overline{\xi_k} \le \overline{\xi}$ . So  $\overline{\eta} \le \overline{\xi} \le \overline{\eta}$  and hence  $\overline{\eta} = \overline{\xi}$ . Now we have four cases: *Case 1*: $\xi$ ,  $\eta$  are initial.

Since  $\xi < \eta$ , we then have  $\overline{\xi} < \overline{\eta}$ . Contradiction. Case 2:  $\xi$  is initial,  $\eta = \tau \cdot \omega$  with  $\tau$  initial. Then  $\overline{\xi} = \overline{\eta} = \overline{\tau}$ , so  $\tau = \xi$ . But then we have a contradiction with lemma 2.6.11 (b). *Case 3*:  $\xi = \tau \cdot \omega$  with  $\tau$  initial and  $\eta$  is initial.

By the same arguments as in case 3 we can derive a contradiction.

*Case 4*:  $\xi = \tau \cdot \omega$  and  $\eta = \delta \cdot \omega$  with  $\tau$  and  $\delta$  initial.

Then  $\overline{\tau} = \overline{\delta}$  and so  $\tau = \delta$ , so  $\xi = \eta$ , which is a contradiction.

**2.6.13 LEMMA:** Let  $\alpha$  be an uncountable regular ordinal, and  $X = [1, \omega^{\mu}] \oplus [1, \beta)$ , where  $\mu$  is a prime component,  $\beta \le \omega^{\mu}$ ,  $\beta$  a  $\sigma$ -ordinal,  $\omega^{\mu} > \alpha^2$  and  $\overline{\omega^{\mu}} = \overline{\alpha}$ . Let  $Y = [1, \alpha \cdot \xi] \oplus [1, \gamma)$ , where  $\xi$  is initial,  $1 \le \xi \le \alpha$ ,  $\gamma$  a  $\sigma$ -ordinal and  $\gamma \le \alpha \cdot \xi$ . Then  $C_0(X)$ and  $C_0(Y)$  are not linearly homeomorphic.

**PROOF:** To the contrary suppose  $C_0(X) \sim C_0(Y)$ . There is  $\delta < \gamma$  such that supp  $[1, \omega^{\mu}] \subset [1, \alpha \cdot \xi] \oplus [1, \delta] \approx [1, \alpha \cdot \xi]$ . By corollary 2.6.4, there is a linear embedding  $\theta: C_0([1, \omega^{\mu}]) \rightarrow C_0([1, \alpha \cdot \xi])$ . But then by lemma 2.6.7 (a),  $\omega^{\mu} \le \alpha \cdot \xi \le \alpha^2$ , which contradicts the fact that  $\omega^{\mu} > \alpha^2$ .

**2.6.14 LEMMA:** Let  $\alpha$  be an uncountable regular ordinal, and  $X = Z_1 \oplus [1, \alpha \cdot \xi)$ and  $Y = Z_2 \oplus [1, \omega^{\mu})$ , where  $Z_1$  and  $Z_2$  are compact spaces,  $\xi \le \alpha$  is  $\sigma$ -initial or of the form  $\tau \cdot \omega$  with  $\tau$  initial,  $\mu$  is a  $\sigma$ -prime component,  $\omega^{\mu} > \alpha^2$  and  $\overline{\omega}^{\mu} = \overline{\alpha}$ . Then  $C_0(X)$ and  $C_0(Y)$  are not linearly homeomorphic.

**PROOF:** To the contrary suppose  $C_0(X) \sim C_0(Y)$ . By corollary 1.2.15 (b) there is  $\mu_i$  in the fixed sequence associated with  $\mu$  such that  $\sup Z_1 \subset Z_2 \oplus [1, \omega^{\mu_i}]$ . Let j > i such that  $\omega^{\mu_j} > \alpha^2$ . By corollary 1.2.15 (b) there is  $k \in \mathbb{N}$  such that  $\sup [1, \omega^{\mu_j}] \subset Z_1 \oplus [1, \alpha \cdot \xi_k]$ . Notice that  $\xi_k < \xi \le \alpha$ . By lemma 2.6.3 there is a linear embedding from  $C_0([1, \omega^{\mu_j}])$  into  $C_0([1, \alpha \cdot \xi_k])$ . Since  $\alpha^2 < \omega^{\mu_j}$ , there is also a linear embedding from  $C_0([1, \alpha^2])$  into  $C_0([1, \omega^{\mu_j}])$ , thus there is a linear embedding  $\theta: C_0([1, \alpha^2]) \to C_0([1, \alpha \cdot \xi_k])$ . So by lemma 2.6.7 (b),  $\overline{\alpha} \le \overline{\xi_k}$  and hence  $\alpha \le \xi_k$ . Contradiction.  $\Box$ 

We now come to the announced

#### 2.6.15 COROLLARY:

- (a)Let  $X \in \mathcal{A}$  and  $Y \in \mathcal{A}$ . Then  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$  if and only if X = Y.
- (b)Let  $X \in \mathcal{B}$  and  $Y \in \mathcal{B}$ . Then  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$  if and only if X = Y.

**PROOF:** If  $C_p(X) \sim C_p(Y)$ , then by remark 2.6.2,  $C_0(X) \sim C_0(Y)$ . Now suppose  $C_0(X) \sim C_0(Y)$ . Let  $\alpha$  and  $\beta$  be ordinals such that  $X \in \mathscr{A}^{(\alpha, \beta)}$ . But then by lemma 2.6.9,  $Y \in \mathscr{A}^{(\alpha, \beta)}$ . By lemmas 2.6.10, 2.6.12, 2.6.13 and 2.6.14 it then follows that X = Y. If X = Y, then evidently  $C_p(X) \sim C_p(Y)$  and (a) is proved.

Similarly (b) follows from lemmas 2.6.9, 2,6,12 and 2.6.14.

**2.6.16 LEMMA:** Let  $X \in \mathcal{A}$  and  $Y \in \mathcal{B}$ . Then  $C_0(X)$  is not linearly homeomorphic to  $C_0(Y)$ .

**PROOF:**  $X = [1, \phi] \oplus [1, \psi)$  and  $Y = [1, \xi)$  with  $\phi$  a prime component,  $\psi$  and  $\xi \sigma$ ordinals and  $\phi \ge \psi$ . Suppose  $C_0(X) \sim C_0(Y)$ . By theorem 2.5.1 (a),  $\overline{X} = \overline{Y}$  and therefore  $\overline{\psi} \le \overline{\xi}$ . As in lemma 2.6.9 we can derive  $\overline{\psi} = \overline{\xi}$ . Let  $\alpha$  be the initial ordinal such that  $\overline{\psi} = \overline{\xi} = \overline{\alpha}$ . By lemma 2.6.14 we have to consider two cases:

*Case 1*:  $\alpha$  is singular,  $\alpha = \omega$  or  $\psi$ ,  $\xi > \alpha^2$ .

Then  $\psi = \omega^{\tau}$  and  $\xi = \omega^{\mu}$  with  $\mu$  and  $\tau \sigma$ -prime components or 1. By lemma 2.6.12 (a),  $\mu = \tau$  and by lemma 2.6.8 (b)  $\tau > 1$ . There is  $\tau_i < \tau$  such that supp  $[1, \phi] \subset [1, \omega^{\tau_i}]$ . So there is a linear embedding  $\theta: C_0([1, \phi]) \rightarrow C_0([1, \omega^{\tau_i}])$  (corollary 2.6.4). By lemma 2.6.7,  $\phi < \omega^{\tau_i \cdot \omega} \le \omega^{\tau} = \psi$ . Contradiction.

Case 2:  $\alpha$  is uncountable and regular.

Then  $\psi = \alpha \cdot \eta$  and  $\xi = \alpha \cdot \tau$  with  $\eta$  and  $\tau \sigma$ -initial or of the form initial  $\omega, \omega \leq \eta, \tau \leq \alpha$ . By lemma 2.6.12 (b),  $\eta = \tau$ . There is  $i \in \mathbb{N}$  such that supp  $[1, \phi] \subset [1, \alpha \cdot \eta_i]$ . So by corollary 2.6.4, there is a linear embedding  $\theta: C_0([1, \phi]) \rightarrow C_0([1, \alpha \cdot \eta_i])$ . Since  $\alpha \cdot \eta_i < \alpha \cdot \eta = \psi \leq \phi$ , there is a linear embedding  $\theta': C_0([1, \alpha \cdot \eta_i]) \rightarrow C_0([1, \phi])$ . This means that  $C_0([1, \alpha \cdot \eta_i])$  and  $C_0([1, \phi])$  have the same linear dimension. So by theorem 2.5.1 (c)  $\phi = \alpha \cdot \gamma + \delta$  for some  $\gamma \leq \alpha$  and  $\delta < \alpha$  with  $\overline{\gamma} = \overline{\eta_i}$ . But then  $\overline{\gamma} < \overline{\eta}$ , so  $\gamma < \eta$ , which implies  $\phi < \alpha \cdot \eta = \psi$ . Contradiction.  $\Box$ 

The following theorem gives the classification announced in the introduction of this section.

**2.6.17 THEOREM:** Let  $\alpha$  and  $\beta$  be  $\sigma$ -ordinals Then the following statements are equivalent:

(1)  $C_p([1, \alpha)) \sim C_p([1, \beta))$ 

(2)  $C_0([1, \alpha)) \sim C_0([1, \beta))$ 

(3) There are compacta  $X_i$  and  $Y_i$   $(i \in \mathbb{N})$  such that  $[1, \alpha) = \bigoplus_{i=1}^{\infty} X_i$ ,  $[1, \beta) = \bigoplus_{i=1}^{\infty} Y_i$  and for every  $i \in \mathbb{N}$ ,  $C_p(X_i) \sim C_p(Y_i)$ . (4) There are compacta  $X_i$  and  $Y_i$   $(i \in \mathbb{N})$  such that  $[1, \alpha) = \bigoplus_{i=1}^{\infty} X_i$ ,  $[1, \beta) = \bigoplus_{i=1}^{\infty} Y_i$  and for every  $i \in \mathbb{N}$ ,  $C_0(X_i) \sim C_0(Y_i)$ .

(In fact the  $X_i$  and the  $Y_i$  are compact ordinal spaces.)

**PROOF:** For (1)  $\Rightarrow$  (2) apply corollary 1.2.21. Furthermore (2)  $\Rightarrow$  (4) easily follows from lemma 2.6.6, corollary 2.6.15 and lemma 2.6.16. For (4)  $\Rightarrow$  (3) notice that for compact ordinals we have the same isomorphical classification for the topology of pointwise convergence and the compact-open topology (section 2.5), so  $C_0(X_i) \sim C_0(Y_i)$  implies  $C_p(X_i) \sim C_p(Y_i)$ . Finally (3)  $\Rightarrow$  (1) follows from lemma 2.3.7, and the theorem is proved.  $\Box$ 

**2.6.18 EXAMPLE:** Notice that  $[1, \omega^{\omega}) = \bigoplus_{i=1}^{\infty} [1, \omega^n]$  and  $[1, \omega^2) = \bigoplus_{i=1}^{\infty} [1, \omega]$ . By theorem 2.4.7,  $C_p([1, \omega^n]) \sim C_p([1, \omega])$  for each  $n \in \mathbb{N}$  (because  $\omega \le \omega^n < \omega^{\omega}$ ). So by theorem 2.6.17,  $C_p([1, \omega^2)) \sim C_p([1, \omega^{\omega}))$ .

With the next lemma and theorems 2.6.17, 2.4.1 and 2.4.7, we have obtained a complete isomorphical classification for the spaces  $C_p(X)$  and  $C_0(X)$  for  $\sigma$ -compact ordinal spaces X. Notice that from the classification it follows that for these spaces  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$ .

**2.6.19 LEMMA:** Let  $\alpha$  and  $\beta$  be ordinals such that  $C_0([1, \alpha)) \sim C_0([1, \beta))$ . Then (a)  $\alpha$  is a successor if and only if  $\beta$  is a successor, and (b)  $\alpha$  is a  $\sigma$ -ordinal if and only if  $\beta$  is a  $\sigma$ -ordinal.

**PROOF:** For (a), if  $\alpha$  is a successor, then  $[1, \alpha)$  is compact. So by theorem 1.5.7  $[1, \beta)$  is compact and thus  $\beta$  is a successor.

For (b), let  $\alpha$  be a  $\sigma$ -ordinal. By remark 2.6.2 [1,  $\alpha$ ) is a non-compact non-pseudocompact space, so by theorem 1.5.7 [1,  $\beta$ ) is a non-compact non-pseudocompact space. But then by remark 2.6.2,  $\beta$  is a  $\sigma$ -ordinal.

By the obtained classification theorems we conclude that for locally compact spaces X and Y and their respective one-point compactifications  $\omega X$  and  $\omega Y$ , the fact that  $C_p(X) \sim C_p(Y)$  does not necessarily imply that  $C_p(\omega X) \sim C_p(\omega Y)$ , and vica versa. For example,  $C_p([1, \omega^{\omega}))$  is linearly homeomorphic to  $C_p([1, \omega^2])$  (example 2.6.18), however  $C_p([1, \omega^{\omega}])$  is not linearly homeomorphic to  $C_p([1, \omega^2])$ . Furthermore,  $C_p([1, \omega])$  is linearly homeomorphic to  $C_p([1, \omega))$  is not linearly homeomorphic to  $C_p([1, \omega])$ .

The same remark applies to the compact-open topology.

The question now arises whether we can derive a similar classification for the spaces of bounded continuous functions. This seems impossible by the methods of this section. Simply observe that corollary 1.2.15 (b) plays a fundamental role, and that it does not hold for spaces of bounded continuous functions (example 1.2.12). In section 4.6 we will come back to this and we will show there that for  $\sigma$ -ordinals  $\alpha$ , theorem 2.6.17 does not hold for the spaces  $C_p^*([1, \alpha))$ .

Another question is whether a similar classification can be derived for arbitrary ordinal spaces. Again it seems that this is impossible by the methods of this section, because we essentially used that every closed and bounded subset of  $[1, \alpha)$  is compact (with  $\alpha$  a  $\sigma$ -ordinal), and by remark 2.6.2 this is not true for the spaces  $[1, \alpha)$  if  $\alpha$  is an ordinal with  $cf(\alpha) > \omega$ .

Finally we remark that the results in this section are new. They are extensions of the results in [3] for the countable case.

# **§2.7.** Separable metric zero-dimensional locally compact spaces

In this section we will give a complete isomorphical classification of the function spaces  $C_p(X)$  and  $C_0(X)$  with X a separable metric zero-dimensional locally compact space. Notice that for separable metric zero-dimensional compact spaces X we already have a complete classification of the spaces  $C_0(X)$  (cf. theorem 2.4.1) and  $C_p(X)$  (cf. theorem 2.4.7). This classification is such that for two spaces X and Y it follows that  $C_p(X)$  is linearly homeomorphic to  $C_p(Y)$  if and only if  $C_0(X)$  is linearly homeomorphic to  $C_0(Y)$  (cf. remark 2.4.8). By theorems 1.5.1 and 1.5.4 it remains to present a complete classification of the spaces  $C_0(X)$  and  $C_p(X)$  with X separable metric zero-dimensional locally compact but not compact. For convenience *in this section every space is separable metric*.

**2.7.1 LEMMA:** Let X be a countable space which is locally compact but not compact. Then there is a  $\sigma$ -limit ordinal  $\alpha$  such that  $X \approx [1, \alpha)$ .

**PROOF:** Let  $\omega X$  be the Alexandroff one-point compactification of X. By proposition

94

2.2.7 and theorem 2.2.8, there is a limit ordinal  $\lambda$  such that  $\omega X \approx [1, \lambda]$ . So X is a dense subset of  $[1, \lambda]$  such that  $[1, \lambda] \setminus X$  contains only one point, say  $\mu$ . Since X is dense in  $[1, \lambda]$ ,  $\mu$  is a limit ordinal. So

$$X = [1, \lambda] \setminus \{\mu\} = [1, \mu] \oplus [\mu + 1, \lambda] = [\mu + 1, \lambda] \oplus [1, \mu] = [\mu + 1, \lambda + \mu] \approx [1, \alpha)$$

for some limit ordinal  $\alpha$ . Since  $\alpha$  is countable,  $\alpha$  is a  $\sigma$ -limit ordinal.

By lemma 2.7.1 countable spaces which are locally compact but not compact are homeomorphic to ordinal spaces. Since these ordinals are  $\sigma$ -ordinals we already have a complete classification for their function spaces  $C_p(X)$  and  $C_0(X)$  (cf. theorem 2.6.17). We shall now consider the case of uncountable locally compact spaces which are not compact. The proof of their classification is similar to the one in section 2.6. We define a class of spaces such that for every uncountable zero-dimensional space X which is locally compact but not compact,  $C_p(X)$  is linearly homeomorphic to a space in this class. After that, we prove that two different spaces in this class are not linearly homeomorphic, which gives the classification.

**2.7.2 LEMMA:** Let X be an uncountable zero-dimensional space which is locally compact but not compact. Then there is a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of X consisting of compacta such that either every  $X_i$  is uncountable or  $X_i$  is uncountable iff i = 1.

**PROOF:** Let  $X = \bigoplus_{i=1}^{\infty} Z_i$  be a decomposition of X consisting of compacta (this is possible because X is zero-dimensional).

Case 1: Only finitely many  $Z_i$  are uncountable.

Let  $n = \max\{i : Z_i \text{ is uncountable}\}$ . Let  $X_1 = Z_1 \oplus \cdots \oplus Z_n$  and  $X_i = Z_{n+i-1}$   $(i \ge 2)$ . Case 2: Infinitely many  $Z_i$  are uncountable.

Suppose  $Z_{i_1}, Z_{i_2}, ...$  are uncountable. Let  $X_n = Z_{i_{n-1}+1} \oplus \cdots \oplus Z_{i_n}$   $(i_0 = 0)$ . Since  $X_n$  is compact and uncountable we are done.  $\Box$ 

We now define the class  $\mathcal{C} \cup \mathcal{D}$  of spaces as follows:

 $\mathscr{C} = \{C \oplus [1, \omega^{\tau}) : C \text{ is the Cantor set and } 1 \le \tau < \omega_1 \text{ is a prime component}\},\$  $\mathcal{D} = \{\bigoplus_{i=1}^{\infty} C_i : C_i \text{ is a copy of the Cantor set}\}.$ 

Observe the following:

If  $X \in \mathcal{C}$ , say  $X = C \oplus [1, \omega^{\tau})$ , then  $X = C \oplus [1, \omega^{\tau_1}] \oplus [1, \omega^{\tau_2}] \oplus ....$  where  $(\tau_i)_i$  is the fixed sequence cofinal with  $\tau$  which was chosen on page 90. If for  $X \in \mathcal{C}$  we write  $X = \bigoplus_{i=1}^{\infty} X_i$ , then we implicitly mean that the  $X_i$  are as above. If  $X \in \mathcal{D}$  then we consid-

er the fixed decomposition  $\bigoplus_{i=1}^{\infty} C_i$ .

**2.7.3 LEMMA:** Let X be an uncountable zero-dimensional space which is locally compact but not compact. Then there is a decomposition  $\bigoplus_{i=1}^{\infty} X_i$  of X and a space  $Y \in \mathcal{C} \cup \mathcal{D}$  such that  $C_p(X_i) \sim C_p(Y_i)$  (where  $Y_i$  is the i-th component of the decomposition of Y stated as above). In particular  $C_p(X) \sim C_p(Y)$  and  $C_0(X) \sim C_0(Y)$ .

**PROOF:** By lemma 2.7.2 there is a decomposition  $\bigoplus_{i=1}^{\infty} X'_i$  of X consisting of compacta, such that either every  $X'_i$  is uncountable or  $X'_i$  is uncountable iff i = 1. Case 1:  $X'_i$  is uncountable iff i = 1.

Since  $X' = \bigoplus_{i=2}^{\infty} X'_i$  is a countable space which is locally compact but not compact, by lemma 2.7.1 and lemma 2.6.6 there is a decomposition  $\bigoplus_{i=1}^{\infty} Z_i$  of X' and a space  $Y' \in \mathcal{A} \cup \mathcal{B}$  such that  $C_p(Z_i) \sim C_p(Y'_i)$ . By lemma 2.6.9  $Y' \in \mathcal{A}^{(\omega, \omega)} \cup \mathcal{B}^{(\omega, \omega)}$ , because X' is countable.

If  $Y' \in \mathcal{A}^{(\omega, \omega)}$ , then  $Y' = [1, \omega^{\mu}] \oplus [1, \omega^{\tau})$ , where  $\mu$  and  $\tau$  are prime components such that  $1 \leq \mu, \tau < \omega_1$ . Then  $C_p(Z_1) \sim C_p([1, \omega^{\mu}])$ . Let  $X_1 = X'_1 \oplus Z_1$  and for  $i \geq 2$  let  $X_i = Z_i$ . Since  $X_1$  is zero-dimensional, uncountable and compact, by theorem 2.4.7  $C_p(X_1) \sim C_p(C)$ . So if we let  $Y = C \oplus [1, \omega^{\tau})$  we are done.

If  $Y' \in \mathcal{B}^{(\omega, \omega)}$ , say  $Y' = [1, \omega^{\tau})$  with  $\tau$  a prime component,  $1 \le \tau < \omega_1$ , then let  $Y = C \oplus [1, \omega^{\tau}), X_1 = X'_1$  and for  $i \ge 2, X_i = Z_i$ .

Case 2: Every  $X'_i$  is uncountable.

Define  $Y = \bigoplus_{i=1}^{\infty} C_i$ . By theorem 2.4.7  $C_p(X_i) \sim C_p(C_i)$ , so let  $X_i = X'_i$ .

#### 2.7.4 LEMMA:

- (a) If  $X, Y \in \mathcal{C}$ , then  $C_p(X) \sim C_p(Y)$  if and only if  $C_0(X) \sim C_0(Y)$  if and only if X = Y.
- (b) If  $X \in \mathcal{C}$  and  $Y \in \mathcal{D}$ , then  $C_0(X)$  and  $C_0(Y)$  are not linearly homeomorphic.

#### **PROOF:** Part (a) follows directly from lemma 2.6.12 (a).

For (b), suppose that  $X = C \oplus [1, \omega^{\tau})$  and  $Y = \bigoplus_{i=1}^{\infty} C_i$ . Assume  $C_0(X) \sim C_0(Y)$ . There is  $n \in \mathbb{N}$  such that supp  $C \subset C_1 \oplus \cdots \oplus C_n$  (corollary 1.2.15 (b)). There is  $i \in \mathbb{N}$  such that supp  $C_{n+1} \subset C \oplus [1, \omega^{\tau_i}]$ . So by lemma 2.6.3, there is an embedding  $\phi: C_0(C) \to C_0([1, \omega^{\tau_i}]).$ Since theorem 2.4.1 by (c) we have  $C_0(C) \sim C_0(C \oplus [1, \omega^{\tau \cdot \omega}]),$ a we have linear embedding  $\theta: C_0([1, \omega^{\tau \cdot \omega}]) \to C_0([1, \omega^{\tau_i}])$ . But then by lemma 2.6.7 (a),  $\tau \cdot \omega < \tau_i \cdot \omega$ . This is a contradiction.

§2.7. Separable metric zero-dimensional locally compact spaces

**2.7.5 THEOREM:** Let X and Y be uncountable zero-dimensional spaces which are both locally compact but not compact. Then the following statements are equivalent:

- (1)  $C_p(X) \sim C_p(Y)$
- (2)  $C_0(X) \sim C_0(Y)$
- (3) There are compacta  $X_i$  and  $Y_i$   $(i \in \mathbb{N})$  such that  $X = \bigoplus_{i=1}^{\infty} X_i$ ,  $Y = \bigoplus_{i=1}^{\infty} Y_i$  and  $C_p(X_i) \sim C_p(Y_i)$ .
- (4) There are compacta  $X_i$  and  $Y_i$   $(i \in \mathbb{N})$  such that  $X = \bigoplus_{i=1}^{\infty} X_i$ ,  $Y = \bigoplus_{i=1}^{\infty} Y_i$  and  $C_0(X_i) \sim C_0(Y_i)$ .

**PROOF:** For (1)  $\Rightarrow$  (2) apply corollary 1.2.21. Furthermore (2)  $\Rightarrow$  (4) follows easily from lemmas 2.7.3 and 2.7.4. For (4)  $\Rightarrow$  (3) notice that for compact zero-dimensional spaces we have the same isomorphical classification for the topology of pointwise convergence and the compact-open topology (section 2.5), so  $C_0(X_i) \sim C_0(Y_i)$  implies  $C_p(X_i) \sim C_p(Y_i)$ . Finally (3)  $\Rightarrow$  (1) follows from lemma 2.3.7.  $\Box$ 

**REMARK:** In view of the remark after theorem 2.6.17 we have the following: Let X and Y be spaces such as in theorem 2.7.5 and let  $\omega X$  and  $\omega Y$  be their respective one point compactifications. By theorem 2.4.7,  $C_p(\omega X)$  is linearly homeomorphic to  $C_p(\omega Y)$ , irrespective of whether  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic.

Again, the same remark applies to the compact-open topology.

We almost completed the isomorphical classification of the function spaces  $C_p(X)$ and  $C_0(X)$  of locally compact zero-dimensional spaces X. It remains to distinguish between "countable" and "uncountable". For the pointwise topology, if  $C_p(X)$  and  $C_p(Y)$  are linearly homeomorphic, we have that X is countable if and only if Y is countable (by theorem 1.5.9). The same holds for the compact open topology as is shown by the following

**2.7.6 PROPOSITION:** Let X and Y be locally compact zero-dimensional spaces such that  $C_0(X)$  and  $C_0(Y)$  are linearly homeomorphic. Then X is countable if and only if Y is countable.

**PROOF:** Suppose that X is countable and Y is uncountable. By theorem 1.5.4b and theorem 2.4.1 we may assume that X and Y are not compact. By lemma 2.6.6 and lemma 2.7.3 we may assume that  $X \in \mathcal{A} \cup \mathcal{B}$  and  $Y \in \mathcal{C} \cup \mathcal{D}$ . There is a clopen copy of C in Y. Then supp C is contained in a clopen copy of  $[1, \alpha]$  in X for some countable ordinal  $\alpha$ . So by corollary 2.6.4 there is a linear embedding from  $C_0(C)$  into  $C_0([1, \alpha])$ . Since

 $C_0(C \oplus [1, \alpha^{\omega}]) \sim C_0(C)$  (theorem 2.4.1) we then have a linear embedding  $\theta: C_0([1, \alpha^{\omega}]) \rightarrow C_0([1, \alpha])$ . But this is impossible by corollary 2.5.3.

Notice that by lemma 1.4.1, proposition 1.4.3, theorems 1.5.1, 1.5.4, 2.4.1, 2.4.7, 2.6.17, 2.7.5 and proposition 2.7.6 we have as announced in the introduction of this chapter that for locally compact zero-dimensional spaces X and Y,  $C_p(X)$  is linearly homeomorphic to  $C_p(Y)$  if and only if  $C_0(X)$  is linearly homeomorphic to  $C_0(Y)$ .

Finally we remark that the main results of this section were published in [3].

## CHAPTER 3

# On topological equivalence of function spaces

All spaces considered in this chapter are separable and metrizable.

Let h be a homeomorphism between  $\mathbb{R}$  and (-1, 1). Then for each space X and each bounded  $f: X \to \mathbb{R}$ , there is  $m \in \mathbb{N}$  such that  $(h \circ f)(X) \subset [-1 + 1/m, 1 - 1/m]$ . This allows us to identify  $C_p(X)$  and the subspace

 $\{f: X \to (-1, 1): f \text{ is continuous}\}$ 

of  $(-1, 1)^X$ ; similarly we can identify  $C_p^*(X)$  and

$$\{f \in C_p(X) : \text{there is } m \in \mathbb{N} \text{ such that } f(X) \subset [-1 + \frac{1}{m}, 1 - \frac{1}{m}]\}.$$

In particular if X is countable,  $C_p(X)$  and  $C_p^*(X)$  can be regarded as subspaces of the Hilbert cube.

Let  $X = \{x_0, x_1, x_2, ...\}$  be a countable space,  $C_{p,0}(X) = \{f \in C_p(X) : f(x_0) = 0\}$  and  $C_{p,0}^*(X) = \{f \in C_p^*(X) : f(x_0) = 0\}$ . In this chapter we mainly consider non-locally compact countable spaces. For  $X = \{x_0, x_1, x_2, ...\}$  not locally compact, we assume that X is not locally compact at  $x_0$ .

In [38], van Mill showed that for a non-locally compact countable space X,  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ , where

$$\sigma_{\omega} = (l_f^2)^{\infty}$$
 and  $l_f^2 = \{x \in l^2 : x_i = 0 \text{ for all but finitely many } i\}$ 

 $(l^2$  denotes separable Hilbert space).

One of our main results in this chapter is that for a non-locally compact countable space X,  $C_p(X)$  is homeomorphic to  $\sigma_{\omega}$ . We will give two proofs. The first proof in section 3.2 is quite technical: Among other things we prove that whenever Y is any other non-locally compact countable space, then there exists a homeomorphism from the Hilbert cube onto itself arbitrary close to the identity which maps  $C_{p,0}(X)$  onto  $C_{p,0}(Y)$ . The second proof in section 3.3 is in the spirit of van Mill's proof that  $C_p^*(X)$  and  $\sigma_{\omega}$  are homeomorphic [38]. The strategy followed depends strongly on results of Toruńczyk [50], [51]. It is less technical than the proof in section 3.2, but it gives only that  $C_p(X)$  and  $\sigma_{\omega}$  are homeomorphic.

Section 3.1 contains preliminaries from infinite-dimensional topology which will be used to prove our main results. In that section we also present some results on Q-matrices

The question remains whether for non-locally compact countable spaces X and Y, there is a homeomorphism from the Hilbert cube onto itself arbitrary close to the identity which maps  $C_{p,0}^*(X)$  onto  $C_{p,0}^*(Y)$ . In section 3.2 we give a positive answer to this question.

In section 3.4 we give some final remarks. We state recent theorems of Dobrowolski, Gulko and Mogilski [20] and Cauty [16] from which can be concluded that for a non-discrete countable space X,  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ . Since for any countable discrete space X,  $C_p(X)$  is homeomorphic to  $\mathbb{R}^n$ , where  $n \in \mathbb{N} \cup \{\infty\}$  is the cardinality of X, we obtain a complete topological classification of the spaces  $C_p(X)$ , for countable spaces X. Furthermore in that section we state the uniform classification derived by Gulko [27] of the uniform spaces  $C_p(X)$ , for countable infinite compact spaces X.

## §3.1. Preliminaries and *Q*-matrices

In this chapter we consider products of spaces at several places. It will be convenient to explicitly define an admissible metric on such a product. For every  $i \in \mathbb{N}$ , let  $P_i$  be a space with an admissible metric  $d_i$  such that each  $d_i$  is bounded by c for a fixed  $c \in \mathbb{R}$ . If we have a finite product of spaces  $P = \prod_{i=1}^{n} P_i$  then the specific admissible metric d on P is defined by  $d = \max\{d_1, \ldots, d_n\}$ , and if we have a countable infinite product of spaces  $P = \prod_{i=1}^{\infty} P_i$  then the specific admissible metric d on P is defined by  $d = \max\{d_1, \ldots, d_n\}$ , and if we have a countable infinite product of spaces  $P = \prod_{i=1}^{\infty} P_i$  then the specific admissible metric d on P is defined by

$$d(x, y) = \sum_{i=1}^{\infty} 2^{-i} d_i(x_i, y_i),$$

where  $x = (x_i)_{i \in \mathbb{N}}$ ,  $y = (y_i)_{i \in \mathbb{N}} \in P$ . Whenever for each  $i \in \mathbb{N}$ ,  $P_i = X$  for some space X, we denote P by  $X^{\infty}$ .

Consider *the Hilbert cube*  $Q = \prod_{i=1}^{\infty} [-1, 1]_i$ , where  $[-1, 1]_i = [-1, 1]$  for every  $i \in \mathbb{N}$ . Then the topology of Q is given by the metric

$$d(x, y) = \sum_{i=1}^{\infty} 2^{-i} |x_i - y_i|,$$

where  $x = (x_i)_{i \in \mathbb{N}}$ ,  $y = (y_i)_{i \in \mathbb{N}} \in Q$ . The subset  $s = \prod_{i=1}^{\infty} (-1, 1)_i$  of Q, where  $(-1, 1)_i = (-1, 1)$  for every  $i \in \mathbb{N}$ , is called *the pseudo-interior* of Q.  $B(Q) = Q \setminus s$  is called *the pseudo-boundary* of Q. A space which is homeomorphic to Q is called *a Hilbert cube*.

A subspace A of separable Hilbert space  $\ell^2$  will be called a *Keller space* whenever it is compact, convex and infinite dimensional. In [33] it is proved that a Keller space is a Hilbert cube (see also [39]). Since there is an affine embedding from Q into  $\ell^2$ , we obtain the following

#### **3.1.1 THEOREM:** A Keller space in Q is a Hilbert cube.

For spaces X and Y let

 $C(X, Y) = \{ f : X \rightarrow Y : f \text{ is continuous} \}$ 

and

 $\mathcal{H}(X, Y) = \{ f : X \to Y : f \text{ is a homeomorphism} \}.$ 

Whenever X = Y we write  $\mathcal{H}(X)$  for  $\mathcal{H}(X, X)$ . For  $f, g \in C(X, Y)$  we define

 $\hat{d}(f, g) = \sup \{ d(f(x), g(x)) : x \in X \} \in [0, \infty],$ 

where d is an admissible metric on Y. As is easily seen we have the following

**3.1.2 LEMMA:** Let X, Y and Z be spaces. Let  $f, g \in C(Y, Z)$  and  $h \in C(X, Y)$ . Then  $\hat{d}(f \circ h, g \circ h) \leq \hat{d}(f, g)$ . If moreover h is surjective, then  $\hat{d}(f \circ h, g \circ h) = \hat{d}(f, g)$ .

Let X be a compact space and let A be a closed subspace of X. Then A is a Z-set in X if and only if for every  $f \in C(Q, X)$  and for every  $\varepsilon > 0$ , there is a  $g \in C(Q, X)$  such that

(a) 
$$d(f, g) < \varepsilon$$
, and  
(b)  $g(O) \cap A = \emptyset$ .

The definition of a Z-set is independent from the chosen metric on X. By  $\mathcal{Z}(X)$  we denote the family of all Z-sets in X. A countable union of Z-sets is called a  $\sigma Z$ -set. The family of all  $\sigma Z$ -sets in X is denoted by  $\mathcal{Z}_{\sigma}(X)$ . An embedding  $f: X \to Y$ , where Y is another compact space, is called a  $\mathcal{Z}$ -embedding whenever  $f(X) \in \mathcal{Z}(Y)$ .

**3.1.3 LEMMA** ([39, Lemma 6.2.2]): Let X be a space. Then (a) If  $A \in \mathcal{J}(X)$  and  $B \subset A$  is closed, then  $B \in \mathcal{J}(X)$ . (b) If  $A \in \mathcal{J}(X)$ , then A has empty interior in X.

**PROOF:** (a) follows directly from the definition of a Z-set. For (b) suppose that Int  $A \neq \emptyset$ . Let  $x \in \text{Int } A$  and put  $\varepsilon = d(x, X \setminus \text{Int } A)$ . Let  $f \in C(Q, X)$  be the constant function with value x. If  $g \in C(Q, X)$  satisfies  $\hat{d}(f, g) < \varepsilon/2$ , then obviously  $g(Q) \cap A \neq \emptyset$ . Hence A is not a Z-set in X.  $\Box$ 

**3.1.4 LEMMA** ([39, Lemma 6.2.3]): Let  $P = \prod_{i=1}^{\infty} P_i$  be a countable infinite product of compact spaces. Let  $A \subset P$  be closed such that  $\pi_j(A) \neq P_j$  for infinitely many j. Then  $A \in \mathcal{J}(P)$ .

**PROOF:** Let  $f = (f_1, f_2, ...) \in C(Q, P)$  and  $\varepsilon > 0$ . For each  $i \in \mathbb{N}$ , let  $d_i$  be an admissible metric on  $P_i$  bounded by 1. Find  $j \in \mathbb{N}$  such that  $2^{-j} < \varepsilon$ . By assumption there are k > j and  $t \in P_k \setminus \pi_k(A)$ . Define  $g \in C(Q, P)$  by

 $g(x) = (f_1(x), \ldots, f_{k-1}(x), t, f_{k+1}(x), \ldots)$ 

Then  $g(Q) \cap A = \emptyset$  and

 $\hat{d}(f, g) = \sup \{ d(f(q), g(q)) : q \in Q \}$ = sup {  $\Sigma_{i=1}^{\infty} 2^{-i} d_i (f_i(q), g_i(q)) : q \in Q \}$ = sup {  $2^{-k} d_k (f_k(q), t) : q \in Q \}$  $\leq 2^{-k} < 2^{-j} < \varepsilon. \square$ 

**3.1.5 THEOREM** ([39, Th. 6.4.6]): Let  $E, F \in \mathcal{I}(Q)$  and let  $f: E \to F$  be a homeomorphism such that  $\hat{d}(f, 1_E) < \varepsilon$ . Then f can be extended to a homeomorphism  $\tilde{f}: Q \to Q$  such that  $\hat{d}(\tilde{f}, 1) < \varepsilon$ .

**3.1.6 THEOREM** ([39, Th. 6.4.8]): Let X be a compact space, let  $A \subset X$  be closed and let  $f: X \to Q$  be continuous such that  $f \mid A$  is a  $\mathcal{I}$ -embedding. Then for every  $\varepsilon > 0$ there is a  $\mathcal{I}$ -embedding  $g: X \to Q$  such that  $\hat{d}(f, g) < \varepsilon$  and  $g \mid A = f \mid A$ .

Let  $\{A_n\}_{n \in \mathbb{N}}$  be an increasing family of Z-sets in a compact space X. Then  $\{A_n\}_{n \in \mathbb{N}}$  is a *skeleton* in X whenever for every  $\varepsilon > 0$ ,  $n \in \mathbb{N}$  and  $Z \in \mathcal{I}(X)$ , there are  $h \in \mathcal{H}(X)$  and  $m \in \mathbb{N}$  such that

(a)  $\hat{d}(h, 1) < \varepsilon$ , (b)  $h \mid A_n = 1$ , and (c)  $h(Z) \subset A_m$ .

102

The definition of a skeleton is independent from the chosen metric on X. A subset A of X is called a *skeletoid* in X if there is a skeleton  $\{A_n\}_{n \in \mathbb{N}}$  in X such that  $A = \bigcup_{n=1}^{\infty} A_n$ . Note that if A is a skeletoid in X and  $h \in \mathcal{H}(X, Y)$ , then h(A) is a skeletoid in Y. In the following theorem sufficient conditions are given for an increasing sequence of Z-sets in a Keller space to be a skeleton.

**3.1.7 THEOREM:** If  $\{A_i\}_{i \in \mathbb{N}}$  is an increasing family of Z-sets in a Keller space P such that,

(a) for every  $i \in \mathbb{N}$ ,  $A_i \in \mathcal{Z}(A_{i+1})$ ,

(b) for every  $i \in \mathbb{N}$ ,  $A_i$  is convex and infinite-dimensional, and

(c)  $\bigcup_{i=1}^{\infty} A_i$  is dense in P, then  $\{A_i\}_{i \in \mathbb{N}}$  is a skeleton in P.

**PROOF:** Let  $Z \in \mathcal{J}(P)$ ,  $n \in \mathbb{N}$  and  $\varepsilon > 0$ . Since P is a Hilbert cube, there is by theorem 3.1.5,  $\delta > 0$  such that if  $E, F \in \mathcal{J}(P)$  and if  $f: E \to F$  is a homeomorphism with  $\hat{d}(f, 1_E) < \delta$ , then f can be extended to a homeomorphism  $\tilde{f}: P \to P$  such that  $\hat{d}(\tilde{f}, 1) < \varepsilon$  (we use that a homeomorphism between P and Q is uniformly continuous).

Find  $\{x_1, \ldots, x_k\} \subset P$  such that  $P = \bigcup_{j=1}^k B(x_j, \delta/4)$ . There is  $m \ge n$  such that for each  $j \le k$ ,  $B(x_j, \delta/4) \cap A_m \ne \emptyset$ . By [39, Cor. 8.2.2] there is a retraction  $r: P \to A_m$  such that for each  $x \in P$ ,  $d(x, r(x)) = d(x, A_m)$ . We claim that  $\hat{d}(r, 1) < \delta/2$ . Indeed let  $x \in P$ . Let  $j \le k$  be such that  $x \in B(x_j, \delta/4)$ . Find  $y \in B(x_j, \delta/4) \cap A_m$ . Then

$$d(x, r(x)) = d(x, A_m) \le d(x, y) \le d(x, x_j) + d(x_j, y) < \frac{\delta}{2}.$$

Let  $r'=r \mid (Z \cup A_n): Z \cup A_n \to A_m$ . Then  $r' \mid A_n$  is a  $\mathcal{Z}$ -embedding. Note that  $A_m$  is a Keller space, hence a Hilbert cube. So by theorem 3.1.6 there is a  $\mathcal{Z}$ -embedding  $s: Z \cup A_n \to A_m$  such that  $s \mid A_n = r' \mid A_n = 1_{A_n}$  and  $\hat{d}(s, r') < \delta/2$ . Hence  $\hat{d}(s, 1) < \delta$ . Note that  $Z \cup A_n \in \mathcal{Z}(P)$  and  $s(Z \cup A_n) \in \mathcal{Z}(P)$  and  $s: Z \cup A_n \to s(Z \cup A_n)$  is a homeomorphism. Hence there is  $h \in \mathcal{H}(P)$  with  $\hat{d}(h, 1) < \varepsilon$  and  $h \mid Z \cup A_n = s$ . This implies that  $h(Z) \subset A_m$  and  $h \mid A_n = 1$ .  $\Box$ 

**3.1.8 EXAMPLE:** For every  $n \in \mathbb{N}$ , let  $\Sigma_n = [-1+1/n, 1-1/n]^{\infty}$ , and let  $\Sigma = \bigcup_{n=1}^{\infty} \Sigma_n$ . By lemma 3.1.4 we have for  $i \in \mathbb{N}$ ,  $\Sigma_i \in \mathcal{I}(Q)$  and  $\Sigma_i \in \mathcal{I}(\Sigma_{i+1})$ . Since each  $\Sigma_i$  is convex and infinite-dimensional and  $\Sigma$  is dense in Q it follows from theorem 3.1.7 that  $\{\Sigma_n\}_{n \in \mathbb{N}}$  is a skeleton in Q, so that  $\Sigma$  is a skeletoid in Q.

Another well-known example of a skeletoid in Q is B(Q).

There are interesting theorems on skeletoids. We mention a few which we will use in the sequel.

**3.1.9 THEOREM** ([17, Lemma 4.3]): Let A and B be skeletoids in a Hilbert cube P. Let  $Z \in \mathcal{Z}(P)$  such that  $Z \cap (A \cup B) = \emptyset$ . Then for every  $\varepsilon > 0$ , there is  $h \in \mathcal{H}(P)$  such that

(a) h(A) = B, (b) h | Z = 1, and (c)  $\hat{d}(h, 1) < \varepsilon$ .

**3.1.10 THEOREM** ([17, Th. 6.7; 39, Th. 6.5.3 (2)]): Let A be a skeletoid in a Hilbert cube P,  $B \in \mathcal{I}(P)$  and  $C \in \mathcal{I}_{\sigma}(P)$ . Then  $A \setminus B$  and  $A \cup C$  are skeletoids in P.

**3.1.11 COROLLARY:** Let A be a skeletoid in Q,  $B \in \mathcal{I}(Q)$  and  $C \in \mathcal{I}_{\sigma}(Q)$  such that  $C \subset B$ . Then for every  $\varepsilon > 0$  there is  $h \in \mathcal{H}(Q)$  such that

(a)  $\hat{d}(h, 1) < \varepsilon$ , and (b)  $h(B) \cap A = h(C)$ .

**PROOF:** By theorem 3.1.10,  $(A \setminus B) \cup C$  is a skeletoid. By theorem 3.1.9 there is  $h \in \mathcal{H}(Q)$  such that

(1)  $\hat{d}(h, 1) < \varepsilon$ , and (2)  $h((A \setminus B) \cup C) = A$ .

Then we have

$$h(B) \cap A = h(B) \cap h((A \setminus B) \cup C)$$
$$= h(B \cap ((A \setminus B) \cup C))$$
$$= h(B \cap C)$$
$$= h(C). \Box$$

We now present the notions of a  $\mathcal{I}$ -matrix and a Q-matrix. These notions were introduced by van Mill in [38].

A  $\mathcal{I}$ -matrix in a compact space X is a collection  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  of Z-sets in X such that for every  $m, n \in \mathbb{N}$ ,

(a)  $A_1^n = \emptyset$ , (b)  $A_m^n \subset A_{m+1}^n$ , and (c)  $A_m^{n+1} \subset A_m^n$ .

104

Define the *kernel of*  $\mathcal{A}$  by ker  $\mathcal{A} = \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} A_m^n$ . Then clearly ker  $\mathcal{A}$  is an  $F_{\sigma\delta}$ -subset of X.

Let  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  be a  $\mathcal{I}$ -matrix in a compact space X. Then  $\mathcal{A}$  is a *Q*-matrix if and only if  $\mathcal{A}$  has the following properties:

(a) For every  $n \in \mathbb{N}$ ,  $\{A_m^n\}_{m>1}$  is a skeleton in X,

and for every  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$ ,

(b)  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Hilbert cube, (c) for every  $p \in \mathbb{N}$ ,  $\{\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i>1}$  is a skeleton in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$ , and (d) for every  $s, t \in \mathbb{N}$  such that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \notin A_i^s$  we have  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^s \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k}).$ 

Note that if  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  is a *Q*-matrix in *X* and  $h \in \mathcal{H}(X, Y)$ , then  $h(\mathcal{A}) = \{h(A_m^n) : n, m \in \mathbb{N}\}$  is a *Q*-matrix in *Y* (we use that *h* is uniformly continuous).

Let  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  be a  $\mathcal{P}$ -matrix and let  $A_{m_1}^{n_1}$  and  $A_{m_2}^{n_2}$  be in  $\mathcal{A}$  such that  $n_1 < n_2$  and  $m_1 \ge m_2$ . Then  $A_{m_2}^{n_2} \subset A_{m_1}^{n_1}$  so  $A_{m_1}^{n_1} \cap A_{m_2}^{n_2} = A_{m_2}^{n_2}$ . So for  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$  we may assume  $i_1 < \cdots < i_m$  if we are interested in  $\bigcap_{k=1}^m A_{i_k}^{n_k}$ .

**3.1.12 THEOREM** ([38]): If  $\mathcal{A}$  and  $\mathcal{B}$  are Q-matrices in Q, then

(a) ker  $\mathcal{A}$  is homeomorphic to  $\sigma_{\omega}$ , and

(b) for every  $\varepsilon > 0$  there is  $h \in \mathcal{H}(Q)$  such that  $d(h, 1) < \varepsilon$  and  $h(\ker \mathcal{A}) = \ker \mathcal{B}$ .

**3.1.13 COROLLARY:** Let  $P_1$  and  $P_2$  be Hilbert cubes and let A and B be Q-matrices in  $P_1$  resp.  $P_2$ . Then

(a) ker  $\mathcal{A}$  is homeomorphic to  $\sigma_{\omega}$ , and

(b) for every  $h \in \mathcal{H}(P_1, P_2)$  and  $\varepsilon > 0$ , there is  $g \in \mathcal{H}(P_1, P_2)$  such that  $\hat{d}(h, g) < \varepsilon$  and  $g (\ker \mathcal{A}) = \ker \mathcal{B}$ .

**PROOF:** Observe that (a) is a triviality. For (b), let  $h_1: Q \to P_1$  be a homeomorphism. Then  $\mathcal{C} = h_1^{-1}(\mathcal{A})$  and  $\mathcal{D} = (h \circ h_1)^{-1}(\mathcal{B})$  are Q-matrices in Q. Since  $h \circ h_1$  is uniformly continuous, there is  $\delta > 0$  such that if  $d(x, y) < \delta$ , then  $d((h \circ h_1)(x), (h \circ h_1)(y)) < \varepsilon/2$ . By theorem 3.1.12 (b) there is  $\alpha \in \mathcal{H}(Q)$  such that  $\hat{d}(\alpha, 1) < \delta$  and  $\alpha(\ker \mathcal{C}) = \ker \mathcal{D}$ . Let  $g = h \circ h_1 \circ \alpha \circ h_1^{-1}$ . Then  $g: P_1 \to P_2$  is a homeomorphism and  $g(\ker \mathcal{A}) = \ker \mathcal{B}$ . Furthermore by lemma 3.1.2 and the choice of  $\delta$ ,

$$\hat{\mathbf{d}}(h, g) = \hat{\mathbf{d}}(h, h \circ h_1 \circ \alpha \circ h_1^{-1}) = \hat{\mathbf{d}}(h \circ h_1, h \circ h_1 \circ \alpha) < \varepsilon. \square$$

Van Mill used theorem 3.1.12 to prove that if X is a non-locally compact countable space, then  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ . The strategy of the proof is the following: First a test space T of X is constructed and a Q-matrix  $\mathcal{B}$  is found such that ker  $\mathcal{B} = C_p^*(T)$ . So by theorem 3.1.12 (a) it follows that  $C_p^*(T)$  is homeomorphic to  $\sigma_{\omega}$ . Then by applying strong results of Toruńczyk [50], [51] he derives that  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ . In section 3.3 we will use the same strategy to prove that  $C_p(X)$ is homeomorphic to  $\sigma_{\omega}$ .

**3.1.14 EXAMPLE:** Let  $\Sigma_n$  and  $\Sigma$  be as in example 3.1.8. Let  $P = \prod_{i=1}^{\infty} Q_i$ , where  $Q_i = Q$  for every  $i \in \mathbb{N}$ . Clearly P is a Hilbert cube. For every  $n, m \in \mathbb{N}$  define  $A_m^n \subset P$  as follows

(1)  $A_1^n = \emptyset$  for every  $n \in \mathbb{N}$  and (2)  $A_m^n = (\Sigma_m)^n \times Q \times Q \times Q \times \cdots$  for every  $n \in \mathbb{N}$  and  $m \ge 2$ .

We claim that  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  is a *Q*-matrix in *P*. By lemma 3.1.4 for each  $n, m \in \mathbb{N}, A_m^n \in \mathcal{I}(P)$  and  $A_m^n \in \mathcal{I}(A_{m+1}^n)$ . For each  $n \in \mathbb{N}, A_m^n$  is convex and infinitedimensional, and  $\bigcup_{m=1}^{\infty} A_m^n$  is dense in *P*, so by theorem 3.1.7 we have that  $\{A_m^n\}_{m>1}$  is a skeleton in *P* for every  $n \in \mathbb{N}$ . Now let  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$ . By the observation made above we may assume  $i_1 < \cdots < i_m$ . Then

$$\bigcap_{k=1}^{m} A_{i_{k}}^{n_{k}} = (\Sigma_{i_{1}})^{n_{1}} \times (\Sigma_{i_{2}})^{n_{2}-n_{1}} \times \cdots \times (\Sigma_{i_{m}})^{n_{m}-n_{m-1}} \times Q \times Q \times Q \times \cdots$$

is a product of Hilbert cubes and hence a Hilbert cube itself.

For  $p \in \mathbb{N}$  and  $i \ge i_m$ ,

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} = (\Sigma_{i_1})^{n_1} \times \cdots \times (\Sigma_{i_m})^{n_m-n_{m-1}} \times (\Sigma_i)^p \times Q \times Q \times Q \times \cdots$$

and for  $p \in \mathbb{N}$  and  $i < i_m$ ,  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p} = A_i^{n_m+p}$ . By the above formulas and by lemma 3.1.4, for each  $i \in \mathbb{N}$ ,

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k}) \text{ and } \bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p})$$

Furthermore we have that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is convex and infinite-dimensional, and

106
$\bigcup_{i=1}^{\infty} (\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}) \text{ is dense in } \bigcap_{k=1}^{m} A_{i_k}^{n_k}. \text{ So again by theorem 3.1.7, for every } p \in \mathbb{N}, \{\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i>1} \text{ is a skeleton in } \bigcap_{k=1}^{m} A_{i_k}^{n_k}.$ 

Finally let s,  $t \in \mathbb{N}$  be such that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \not\subset A_t^s$ . Then by the above formulas we can write for t > 1

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} = \prod_{i=1}^{\infty} E_i \text{ and } \bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_t^s = \prod_{i=1}^{\infty} F_i,$$

where all but finitely many  $E_i$ 's and  $F_i$ 's are equal to Q and the remaining finitely many  $E_i$ 's and  $F_i$ 's are elements of the family  $\{\Sigma_n\}_{n>1}$ . Since  $\prod_{i=1}^{\infty} F_i$  is a proper subset of  $\prod_{i=1}^{\infty} E_i$ , there is  $i \in \mathbb{N}$  such that  $F_i$  is a proper subset of  $E_i$ . Since  $F_i = \Sigma_n$  for some  $n \in \mathbb{N}$  or Q and  $F_i$  also, it follows that each factor space of  $F_i$  is a proper subset of the corresponding factor space of  $E_i$ . Hence by lemma 3.1.4 it follows that  $\prod_{i=1}^{\infty} F_i \in \mathcal{J}(\prod_{i=1}^{\infty} E_i)$ .

It is easily seen that ker  $\mathcal{A} = \Sigma^{\infty}$ , so that by theorem 3.1.12 (a)  $\Sigma^{\infty}$  is homeomorphic to  $\sigma_{\omega}$ .

**COROLLARY 3.1.15:** Let  $\{P_i : i \in \mathbb{N}\}$  be a family of Hilbert cubes, and  $P = \prod_{i=1}^{\infty} P_i$ . Then

- (a) if  $\mathcal{B}$  and  $\mathcal{C}$  are Q-matrices in  $P_1$  resp.  $P_2$ , then there is a Q-matrix  $\mathcal{A}$  in  $P_1 \times P_2$  such that ker  $\mathcal{A} = \ker \mathcal{B} \times \ker \mathcal{C}$ ,
- (b) if for each  $i \in \mathbb{N}$ ,  $\mathcal{A}_i$  is a Q-matrix in  $P_i$ , then there is a Q-matrix  $\mathcal{A}$  in P such that ker  $\mathcal{A} = \prod_{i=1}^{\infty} \ker \mathcal{A}_i$ ,
- (c) if for each  $i \in \mathbb{N}$ ,  $A_i$  is a skeletoid in  $P_i$ , then there is a Q-matrix  $\mathcal{A}$  in P such that ker  $\mathcal{A} = \prod_{i=1}^{\infty} A_i$ ,
- (d) if  $\mathcal{B}$  is a Q-matrix in  $P_1$  and A is a skeletoid in  $P_2$ , then there is a Q-matrix A in  $P_1 \times P_2$  such that ker  $\mathcal{A} = \ker \mathcal{B} \times A$ , and
- (e) if  $\mathcal{B}$  is a Q-matrix in  $P_1$ , then there is a Q-matrix  $\mathcal{A}$  in  $[-1, 1] \times P_1$  such that  $\ker \mathcal{A} = (-1, 1) \times \ker \mathcal{B}$ .

**PROOF:** If  $\mathcal{A}$  is a Q-matrix in a Hilbert cube  $Q_1$ , there is by corollary 3.1.13 (b) and example 3.1.14, a homeomorphism  $h: Q^{\infty} \to Q_1$  such that  $h(\Sigma^{\infty}) = \ker \mathcal{A}$ . Moreover by theorem 3.1.9 and example 3.1.8, there is for each skeletoid A in a Hilbert cube  $Q_1$  a homeomorphism  $g: Q \to Q_1$  such that  $g(\Sigma) = A$ . It is easily seen that

- (1) there is a homeomorphism  $h_1: Q^{\infty} \to (Q^{\infty})^2$  such that  $h_1(\Sigma^{\infty}) = (\Sigma^{\infty})^2$ ,
- (2) there is a homeomorphism  $h_2: Q^{\infty} \to (Q^{\infty})^{\infty}$  such that  $h_2(\Sigma^{\infty}) = (\Sigma^{\infty})^{\infty}$ ,
- (3) there is a homeomorphism  $h_3: Q^{\infty} \to Q \times Q^{\infty}$  such that  $h_3(\Sigma^{\infty}) = \Sigma \times \Sigma^{\infty}$ ,

and

Taking the right combinations of the obtained homeomorphisms one can prove (a) through (e). We outline (e). The other proofs are similar.

Let  $h: Q^{\infty} \to P_1$  be a homeomorphism such that  $h(\Sigma^{\infty}) = \ker \mathcal{B}$ . Define  $\alpha: P_1 \to [-1, 1] \times P_1$  by  $\alpha = (1 \times h) \circ h_4 \circ h^{-1}$ . Then  $\alpha$  is clearly a well-defined homeomorphism. Furthermore

$$\alpha(\ker \mathcal{B}) = ((1 \times h) \circ h_4 \circ h^{-1})(\ker \mathcal{B})$$
$$= ((1 \times h) \circ h_4)(\Sigma^{\infty})$$
$$= (1 \times h)((-1, 1) \times \Sigma^{\infty})$$
$$= (-1, 1) \times \ker \mathcal{B}.$$

Then  $\mathcal{A} = \alpha(\mathcal{B})$  is a *Q*-matrix in  $[-1, 1] \times P_1$  such that ker  $(\mathcal{A}) = (-1, 1) \times \text{ker } \mathcal{B}$ .

In contrast to the theory of skeletoids, the theory of Q-matrices is hardly developed. In view of theorem 3.1.10, a first question to ask, is whether for a Q-matrix  $\mathcal{A}$  in a Hilbert cube P and  $F \in \mathcal{Z}(P)$ , there is a Q-matrix  $\mathcal{B}$  such that ker  $\mathcal{B} = \ker \mathcal{A} \setminus F$ . We were not able to prove this straight from the definition of a Q-matrix. However we can prove the weaker statement that ker  $\mathcal{A}$  and ker  $\mathcal{A} \setminus F$  are homeomorphic (theorem 3.1.21). As will be clear in the sequel, the proof of this statement unfortunately has nothing to do with Q-matrices. Before the proof can be given we have to present some more definitions and known theorems.

A space X is said to be a  $\sigma_{\omega}$ -manifold if there is an open cover of X consisting of sets homeomorphic to open subsets of  $\sigma_{\omega}$ . Two spaces X and Y have the same homo-topy type whenever there are  $f \in C(X, Y)$  and  $g \in C(Y, X)$  such that  $f \circ g$  is homotopic to  $1_X$  and  $g \circ f$  is homotopic to  $1_X$ . We have the following theorem of Henderson.

**3.1.16 THEOREM** ([30]): If X and Y are  $\sigma_{\omega}$ -manifolds, then X is homeomorphic to Y if and only if X and Y have the same homotopy type.

A space X is an *absolute retract* (abbreviated AR), resp. an *absolute neigborhood retract* (abbreviated ANR), whenever for every space Y and for every closed subspace A of Y, every continuous function  $f: A \to X$  has an extension  $\tilde{f}: Y \to X$ , resp. an extension  $\tilde{f}: U \to X$  over a neigborhood U of A in Y. By the Dugundji Extension Theorem,  $\sigma_{\omega}$  is an AR. A space X which admits an open cover consisting of ANR's, is itself an

108

ANR ([39, Th. 5.4.5]). A space X is *contractible* whenever the identity  $1_X$  is homotopic to a constant mapping. Note that two contractible spaces have the same homotopy type. A space X is *homotopically trivial* whenever for every  $n \in \mathbb{N}$  and every continuous  $f: S^n \to X$ , there is a continuous extension  $g: B^{n+1} \to X$ . The above notions are related by the following

3.1.17 THEOREM ([39, Th. 5.2.15]): For a space X are equivalent

- (1) X is an AR,
- (2) X is an ANR and contractible, and
- (3) X is an ANR and homotopically trivial.

We proceed by proving the announced statement from the previous page (cf. Theorem 3.1.21). We start with three lemmas.

**3.1.18 LEMMA:** Let  $K \subset \Sigma$  be compact. Then for every  $\varepsilon > 0$ , there is an embedding  $f: Q \to \Sigma$  such that (a)  $f \mid K = 1$ , and

 $(b) \hat{d}(f, 1) < \varepsilon$ 

**PROOF:** By lemma 3.1.4,  $K \in \mathcal{J}(Q)$  and by example 3.1.8,  $\{\Sigma_n\}_{n \in \mathbb{N}}$  is a skeleton in Q, so there are  $n \in \mathbb{N}$  and  $h \in \mathcal{H}(Q)$  such that

(1)  $h(K) \subset \Sigma_n$ , and (2)  $\hat{d}(h, 1) < \varepsilon/6$ .

By lemma 3.1.4,  $h(K) \in \mathcal{Z}(Q)$ . By theorem 3.1.10,  $\Sigma \setminus h(K)$  is a skeletoid in Q. Furthermore  $h(\Sigma \setminus K)$  is a skeletoid in Q and h(K) misses  $\Sigma \setminus h(K)$  and  $h(\Sigma \setminus K)$  so that by theorem 3.1.9, there is  $\alpha \in \mathcal{H}(Q)$  such that

(3)  $\alpha(h(\Sigma \setminus K)) = \Sigma \setminus h(K),$ (4)  $\alpha \mid h(K) = 1,$  and (5)  $\hat{d}(\alpha, 1) < \varepsilon/6.$ 

Since  $h(K) \subset \Sigma$ , we have by (3) and (4) that  $\alpha$  has the additional property that  $\alpha(h(\Sigma)) = \Sigma$ . Let  $\beta = \alpha \circ h$ . Find  $m \in \mathbb{N}$  and a homeomorphism  $\xi: Q \to \Sigma_m$  such that

(6)  $\xi | \Sigma_n = 1$ , and (7)  $\hat{d}(\xi, 1) < \varepsilon/3$ .

Let  $f = \beta^{-1} \circ \xi \circ \beta$ :  $Q \to \Sigma$ . Then f is clearly a well-defined embedding. Furthermore for

and by (2), (5), (7), and lemma 3.1.2,

$$\hat{d}(f, 1) \leq \hat{d}(\beta^{-1} \circ \xi \circ \beta, \xi \circ \beta) + \hat{d}(\xi \circ \beta, \beta) + \hat{d}(\beta, 1)$$
  
$$\leq \hat{d}(\beta^{-1}, 1) + \hat{d}(\xi, 1) + \hat{d}(\beta, 1)$$
  
$$= 2\hat{d}(\beta, 1) + \hat{d}(\xi, 1)$$
  
$$\leq 2\hat{d}(\alpha, 1) + 2\hat{d}(h, 1) + \hat{d}(\xi, 1) < \varepsilon. \square$$

**3.1.19 LEMMA:** Let  $K \subset \Sigma^{\infty}$  be compact. Then for every  $\varepsilon > 0$ , there is an embedding  $f: Q^{\infty} \to \Sigma^{\infty}$  such that (a)  $f \mid K = 1$ , and (b)  $\widehat{V}(\varepsilon, 1) \to \infty$ 

 $(b) \hat{d}(f, 1) < \varepsilon.$ 

**PROOF:** For every  $n \in \mathbb{N}$ ,  $\pi_n(K) \subset \Sigma$ , so by lemma 3.1.18, there is an embedding  $f_n: Q \to \Sigma$  such that  $f_n | \pi_n(K) = 1$  and  $\hat{d}(f_n, 1) < \varepsilon$ . Define  $f: Q^{\infty} \to \Sigma^{\infty}$  by  $f = (f_1, f_2, ...)$  Then f is easily seen to be as required.  $\Box$ 

**3.1.20 LEMMA:** Let  $K \subset \Sigma^{\infty}$  be compact and  $Z \in \mathcal{J}(Q^{\infty})$  such that  $K \cap Z = \emptyset$ . Then for every  $\varepsilon > 0$  there is an embedding  $f : Q^{\infty} \to \Sigma^{\infty}$  such that

(a) f | K = 1, (b)  $f (Q^{\infty}) \cap Z = \emptyset$ , and (c)  $\hat{d}(f, 1) < \varepsilon$ .

**PROOF:** There is a continuous  $h_1: Q^{\infty} \to Q^{\infty}$  such that  $h_1(Q^{\infty}) \cap Z = \emptyset$  and  $\hat{d}(h_1, 1) < \varepsilon/8$ . Let  $\eta = d(h_1(Q^{\infty}), Z)$ . By theorem 3.1.6 there is a Z-embedding  $h_2: Q^{\infty} \to Q^{\infty}$  such that  $\hat{d}(h_2, h_1) < \min\{\eta, \varepsilon/8\}$ . Then  $h_2(Q^{\infty}) \in \mathcal{I}(Q^{\infty})$  and  $h_2(Q^{\infty}) \cap Z = \emptyset$ .

Define  $g: h_2(K) \cup Z \to K \cup Z$  by

$$g(x) = \begin{cases} h_2^{-1}(x) & \text{if } x \in h_2(K) \\ x & \text{if } x \in Z \end{cases}$$

110

Then g is a homeomorphism between Z-sets in  $Q^{\infty}$  such that  $\hat{d}(g, 1_{h_2(K)\cup Z}) < \varepsilon/4$ , so by theorem 3.1.5, g extends to a homeomorphism  $\overline{g}: Q^{\infty} \to Q^{\infty}$  such that  $\hat{d}(\overline{g}, 1) < \varepsilon/4$ . Let  $h = \overline{g} \circ h_2$ . Then h satisfies

(1)  $h \mid K = 1$ , (2)  $h(Q^{\infty}) \cap Z = \emptyset$ , and (3)  $\hat{d}(h, 1) < \varepsilon/2$ .

Let  $\xi = d(h(Q^{\infty}), Z)$ . By lemma 3.1.19, there is an embedding  $\alpha: Q^{\infty} \to \Sigma^{\infty}$  such that  $\alpha \mid K = 1$  and  $\hat{d}(\alpha, 1) < \min\{\epsilon/2, \xi\}$ . Let  $f = \alpha \circ h$ . Then f is easily seen to be as required.

We are now in a position to prove the announced

**3.1.21 THEOREM:** Let  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  be a *Q*-matrix in a Hilbert cube *P* and  $Z \in \mathcal{I}(P)$ . Then ker  $\mathcal{A}$  and ker  $\mathcal{A} \setminus Z$  are homeomorphic.

**PROOF:** By corollary 3.1.13 (b) and example 3.1.14, there is a homeomorphism  $h: P \to Q^{\infty}$  such that  $h(\ker \mathcal{A}) = \Sigma^{\infty}$ . It suffices to prove that for  $Z \in \mathcal{A}(Q^{\infty})$ ,  $\Sigma^{\infty} \setminus Z$  is homeomorphic to  $\Sigma^{\infty}$ .

By corollary 3.1.13 (a),  $\Sigma^{\infty}$  is homeomorphic to  $\sigma_{\omega}$ , so  $\Sigma^{\infty}$  is an AR. Obviously  $\Sigma^{\infty} \setminus Z$  is a  $\sigma_{\omega}$ -manifold and hence an ANR.

CLAIM:  $\Sigma^{\infty} \setminus Z$  is homotopically trivial.

Let  $f: S^n \to \Sigma^{\infty} \setminus Z$  be a continuous function. Then since  $\Sigma^{\infty}$  is an AR, f extends to a continuous function  $g_1: B^{n+1} \to \Sigma^{\infty}$ . Then  $f(S^n)$  is a compact subset of  $\Sigma^{\infty}$  such that  $f(S^n) \cap Z = \emptyset$ . By lemma 3.1.20 we then have an embedding  $g_2: Q^{\infty} \to \Sigma^{\infty}$  such that  $g_2 | f(S^n) = 1$  and  $g_2(Q^{\infty}) \cap Z = \emptyset$ . Let  $g = g_2 \circ g_1: B^{n+1} \to \Sigma^{\infty} \setminus Z$ . Then g is easily seen to be an extension of f.

By theorem 3.1.17 and the claim we now have that  $\Sigma^{\infty}$  and  $\Sigma^{\infty} \setminus Z$  are both contractible and hence they have the same homotopy type. By theorem 3.1.16 we then have that  $\Sigma^{\infty}$  is homeomorphic to  $\Sigma^{\infty} \setminus Z$ .

We finish this section with the remark that we did not prove all results in this section. Their proofs are beyond the scope of this monograph. For more information on infinite-dimensional topology we refer to [11], [17], [18] and [39]. For more information on AR theory we refer to [14], [31] and [39].

## §3.2. Homeomorphic function spaces part 1

Let  $X = \{x_0, x_1, x_2, ...\}$  be a countable space which fails to be locally compact at  $x_0$ . We shall prove that  $C_{p,0}(X)$  can be written as the kernel of a *Q*-matrix in some Hilbert cube.

Let  $\psi: \mathbb{N} \to X \times \mathbb{N}$  be a bijection and define  $\phi: \mathbb{N} \to X$  by  $\phi = \pi_1 \circ \psi$ , where  $\pi_1: X \times \mathbb{N} \to X$  is the projection. The following lemma is of fundamental importance in the process of describing  $C_{p,0}(X)$  as the kernel of a *Q*-matrix.

**3.2.1 LEMMA:** There exists a decreasing clopen base  $\{U_i^{x_0}\}_{i \in \mathbb{N}}$  at  $x_0$  and for each  $x \neq x_0$  there exists a clopen neighborhood  $U^x$  of x such that for every  $n \in \mathbb{N}$ ,

- (a) if  $\phi(n) \neq x_0$ , then  $U^{\phi(n)} \cap U_n^{x_0} = \emptyset$ , and
- (b) for s,  $n \in \mathbb{N}$ , we have  $U_s^{x_0} \setminus [U_{s+1}^{x_0} \cup \bigcup \{U^{\phi(j)} : j \le n, \phi(j) \ne x_0\}]$  is infinite.

**PROOF:** Since no neighborhood of  $x_0$  is compact, there exists a decreasing clopen base  $\{V_i^{x_0}\}_{i \in \mathbb{N}}$  at  $x_0$  such that for every  $i \in \mathbb{N}$ ,  $V_i^{x_0} \setminus V_{i+1}^{x_0}$  contains an infinite closed discrete subset  $D_i$ .

We construct inductively a strictly increasing sequence  $(i_n)_{n \in \mathbb{N}}$  of natural numbers and for each  $n \in \mathbb{N}$  such that  $\phi(n) \neq x_0$  a clopen neighborhood  $V_n$  of  $\phi(n)$  satisfying

- (1)  $V_n \cap V_{i_n}^{\chi_0} = \emptyset$ , and
- (2)  $V_n \cap \bigcup_{s \le i_s} D_s$  contains at most one point.

Suppose we found for  $n \in \mathbb{N}$ ,  $i_1, \ldots, i_{n-1}$  and  $V_j$  for j < n such that  $\phi(j) \neq x_0$ . If  $\phi(n) = x_0$  let  $i_n > i_{n-1}$  be arbitrary. If  $\phi(n) = \phi(j) \neq x_0$  for some j < n, let  $V_n = V_j$  and  $i_n > i_{n-1}$  arbitrary. Since we deal with a decreasing base at  $x_0$  we have  $V_n \cap V_{i_n}^{x_0} = \emptyset$  and  $V_n \cap \bigcup_{s \ge i_j} D_s = \emptyset$ , hence also  $V_n \cap \bigcup_{s < i_n} D_s$  contains at most one point. If  $\phi(n) \notin \{\phi(j) : j < n, \phi(j) \neq x_0\} \cup \{x_0\}$ , we can find a clopen neighborhood U of  $\phi(n)$  and  $i_n > i_{n-1}$  such that  $U \cap V_{i_n}^{x_0} = \emptyset$ . Since  $\bigcup_{s < i_n} D_s$  is closed and discrete we can find a clopen neighborhood  $V_n$  of  $\phi(n)$  contained in U such that  $V_n \cap \bigcup_{s < i_n} D_s$  contains at most one point.

For  $n \in \mathbb{N}$ , let  $U_n^{x_0} = V_{i_n}^{x_0}$ . For  $x \neq x_0$ , let  $k(x) = \min \phi^{-1}(x)$ , and let  $U^x = V_{k(x)}$ . Let  $n \in \mathbb{N}$  with  $\phi(n) \neq x_0$ . Then  $k(\phi(n)) \le n$ , so by (1),  $U^{\phi(n)} \cap U_n^{x_0} = \emptyset$ . In addition we have that for  $s \ge i_n$ ,  $U^{\phi(n)} \cap D_s = \emptyset$ . Hence by (2) we have for  $s, n \in \mathbb{N}$  that

 $U_s^{x_0} \setminus [U_{s+1}^{x_0} \cup \bigcup \{U^{\phi(j)} : j \le n, \phi(j) \ne x_0\}]$  is infinite. This proves the lemma.

As mentioned in the introduction of this chapter,  $C_p(X) = \{f : X \to (-1, 1) : f \text{ is continuous}\}$ . Recall that  $C_{p,0}(X) = \{f \in C_p(X) : f(x_0) = 0\}$ . We look at these spaces as subspaces of  $[-1, 1]^X$  endowed with the product topology.  $[-1, 1]^X$  is obviously a Hilbert cube. Recall that on  $[-1, 1]^X$  we use the metric

$$d(f, g) = \sum_{i=0}^{\infty} 2^{-i} |f(x_i) - g(x_i)|,$$

for  $f, g \in [-1, 1]^X$ .

We will now give another description of the space  $C_{p,0}(X)$ , in terms of the kernel of a Q-matrix. Let  $Y = X \setminus \{x_0\}$  and  $P = \{0\} \times [-1, 1]^Y$ . Evidently there is a convexity preserving homeomorphism between P and Q. Hence by theorem 3.1.1 each Keller space in P is a Hilbert cube. Let  $\{U_n^{x_0}\}_{n \in \mathbb{N}}$  and  $\{U^x : x \neq x_0\}$  be as in lemma 3.2.1. For  $x \neq x_0$  let  $\{U_n^x\}_{n \in \mathbb{N}}$  be a clopen decreasing base at x such that  $U_1^x = U^x$ . For every  $x \in X$  and n,  $m \in \mathbb{N}$ , we define

(A) 
$$B_m^{(x,n)} = \{ g \in P : g(U_m^x) \subset [g(x) - \frac{1}{n}, g(x) + \frac{1}{n}] \}.$$

Furthermore for every  $n, m \in \mathbb{N}$  we define

(B) 
$$C_m^n = B_m^{(x_0, n)} \cap \bigcap_{j=1}^n B_m^{(\phi(j), n)},$$
  
(C)  $L_m^n = \{0\} \times [-1 + \frac{1}{m}, 1 - \frac{1}{m}]^{\{x_1, \dots, x_n\}} \times [-1, 1] \times [-1, 1] \times \cdots,$  and  
(D)  $A_1^n = \emptyset$ , and  $A_m^n = C_m^n \cap L_m^n$  for  $m \ge 2$ .

It will turn out that the family  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}, n > 1\}$  is a *Q*-matrix in *P* such that ker  $\mathcal{A} = C_{p,0}(X)$ .

**3.2.2 LEMMA:** For every  $n, m \in \mathbb{N}$ , we have

(a)  $A_m^n$  is closed in P, (b)  $A_m^n \subset A_{m+1}^n$ , and (c)  $A_m^{n+1} \subset A_m^n$ .

**PROOF:** It is easily seen that for every  $x \in X$  and for every  $n, m \in \mathbb{N}$  we have,

(1)  $B_m^{(x, n)}$  and  $L_m^n$  are closed in  $[-1, 1]^X$ ,

(2) 
$$B_m^{(x, n)} \subset B_{m+1}^{(x, n)}, B_m^{(x, n+1)} \subset B_m^{(x, n)},$$
  
(3)  $L_m^n \subset L_{m+1}^n$  and  $L_m^{n+1} \subset L_m^n.$ 

Now the lemma follows from (1), (2) and (3).  $\Box$ 

We first prove that ker  $\mathcal{A} = C_{p,0}(X)$  (lemma 3.2.4). Define  $F = \{f \in [-1, 1]^X : f \text{ is continuous and } f(x_0) = 0\}$ . Then  $F \subset P$ . Observe that by the definition of continuity,  $F = \bigcap_{x \in X} \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} B_m^{(x, n)}$  (see [19]).

**3.2.3 LEMMA:** 
$$F = \bigcap_{n=2}^{\infty} \bigcup_{m=1}^{\infty} C_m^n$$
.

**PROOF:** First suppose that  $f \in F$  and  $n \ge 2$ . Then  $f \in P$ . Because f is continuous, there exists  $m \in \mathbb{N}$  such that for each  $j \le n$ ,

$$f(U_m^{\phi(j)}) \subset [f(\phi(j)) - \frac{1}{n}, f(\phi(j)) + \frac{1}{n}],$$

and

$$f(U_m^{x_0}) \subset [-\frac{1}{n}, \frac{1}{n}].$$

This implies  $f \in B_m^{(x_0,n)} \cap \bigcap_{j=1}^n B_m^{(\phi(j),n)} = C_m^n$ . We conclude that  $f \in \bigcap_{n=2}^\infty \bigcup_{m=1}^\infty C_m^n$ .

Secondly suppose  $f \in \bigcap_{n=2}^{\infty} \bigcup_{m=1}^{\infty} C_m^n$ . Let  $x \in X$  and  $n \in \mathbb{N}$ . Because  $\psi$  is a bijection, there exists  $n_x \in \mathbb{N}$  such that  $k = \psi^{-1}(x, n_x) > n$ . Then  $\phi(k) = x$ . There exists  $m \in \mathbb{N}$  such that  $f \in C_m^k$ , hence

$$f \in C_m^k \subset B_m^{(x, k)} \subset B_m^{(x, n)}.$$

So  $f \in \bigcap_{x \in X} \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} B_m^{(x, n)} = F$ . This completes the proof of the lemma.

**3.2.4 LEMMA:** 
$$C_{p,0}(X) = \bigcap_{n=2}^{\infty} \bigcup_{m=1}^{\infty} A_m^n$$
.

**PROOF:** First suppose  $f \in C_{p,0}(X)$  and  $n \ge 2$ . There exists  $k \ge 2$  such that

$$\{f(x_1),\ldots,f(x_n)\} \subset [-1+\frac{1}{k},1-\frac{1}{k}],\$$

hence  $f \in L_k^n$ . Since  $f \in F$ , there exists  $m \ge k$  such that  $f \in C_m^n$ . So  $f \in C_m^n \cap L_m^n = A_m^n$ . We conclude that  $f \in \bigcap_{m=2}^{\infty} \bigcup_{m=1}^{\infty} A_m^n$ .

114

Secondly suppose  $f \in \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} A_m^n$ . Since for every  $n, m \in \mathbb{N}$ ,  $A_m^n \subset C_m^n$ , we have by lemma 3.2.3,  $f \in F$ . Since for every  $n \in \mathbb{N}$  there is  $m \in \mathbb{N}$  such that  $f \in A_m^n \subset L_m^n$ , we have that  $\{f(x_1), ..., f(x_n)\} \subset (-1, 1)$ . So  $f \in C_{p,0}(X)$ . This proves the lemma.  $\Box$ 

**3.2.5 LEMMA:** For every  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$  we have that  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is a Keller space in P, hence a Hilbert cube.

**PROOF:** To prove that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Keller space, we have to verify that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is compact, convex and infinite-dimensional.

By lemma 3.2.2 (a),  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is closed in *P*, hence it is compact. To prove that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is convex we first claim that for every  $x \in X$  and for every  $n, m \in \mathbb{N}, B_m^{(x, n)}$  is convex. Indeed, let  $f, g \in B_m^{(x, n)}, \lambda \in [0, 1]$  and  $h = \lambda f + (1 - \lambda)g$ . Since *P* is convex, we have  $h \in P$ . Furthermore if  $y \in U_m^x$ , then

$$|h(x) - h(y)| \le \lambda |f(x) - f(y)| + (1 - \lambda) |g(x) - g(y)|$$
  
$$\le \lambda \frac{1}{n} + (1 - \lambda) \frac{1}{n} = \frac{1}{n}.$$

So  $h \in B_m^{(x,n)}$ , so  $B_m^{(x,n)}$  is convex. It is easily seen that for every  $n, m \in \mathbb{N}, L_m^n$  is convex. Since the intersection of convex sets is again convex,  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is convex.

Finally, to see that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is infinite-dimensional notice that for every  $x \in X$  and for every  $n, m \in \mathbb{N}$ ,

$$\{0\} \times \left[-\frac{1}{2n}, \frac{1}{2n}\right]^Y \subset B_m^{(x, n)}$$

and for every  $n \in \mathbb{N}$  and  $m \in \mathbb{N} \setminus \{1\}, \{0\} \times [-\frac{1}{2n}, \frac{1}{2n}]^Y \subset L_m^n$ , so

$$\{0\}\times \left[-\frac{1}{2n_m}, \frac{1}{2n_m}\right]^Y \subset \bigcap_{k=1}^m A_{i_k}^{n_k}$$

We conclude that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Keller space in *P*, hence by theorem 3.1.1  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Hilbert cube.  $\Box$ 

For every  $i, k \in \mathbb{N}$  we define  $h(i, k): P \to P$  by

$$h(i, k)(f)(x) = \begin{cases} f(x) & \text{if } x \neq x_i \\ \frac{1}{k} & \text{if } x = x_i \end{cases}$$

Notice that since  $x_i \neq x_0$ , h(i, k) is well-defined.

**3.2.6 LEMMA:** Let  $i, k \in \mathbb{N}$ . Then (a) h(i, k) is continuous and  $\hat{d}(h(i, k), 1) \le 2^{-i+1}$ , and (b) if  $i > 1, k \le p < i$  and  $s \in \mathbb{N}$  are such that  $x_i \in U_s^{x_0} \setminus [U_{s+1}^{x_0} \cup \bigcup \{U_1^{\varphi(j)} : j \le p, \varphi(j) \ne x_0\}]$ , then (i) for  $n \le k$  and  $m \in \mathbb{N}$ ,  $h(i, k)(A_m^n) \subset A_m^n$ , (ii) for  $n \le p$  and m > s,  $h(i, k)(A_m^n) \subset A_m^n$ , and (iii) for n > k,  $h(i, k)(P) \cap A_s^n = \emptyset$ .

**PROOF:** For every  $x \in X$ , let  $\pi_x: P \to [-1, 1]$  be the projection onto the *x*-th coordinate. Then for  $x \neq x_i$ , we have  $\pi_x \circ h(i, k) = \pi_x$  and for  $x = x_i$ , we have  $\pi_x \circ h(i, k) \equiv 1/k$ , so that for each  $x \in X$ ,  $\pi_x \circ h(i, k)$  is continuous. So we have that h(i, k) is continuous. Furthermore we have

$$\begin{aligned} \hat{d}(h(i, k), 1) &= \sup \left\{ d(h(i, k)(f), f) : f \in P \right\} \\ &= \sup \left\{ \sum_{j=1}^{\infty} 2^{-j} |h(i, k)(f)(x_j) - (f)(x_j)| : f \in P \right\} \\ &= \sup \left\{ 2^{-i} |1/k - f(x_i)| : f \in P \right\} \\ &\leq 2^{-i} \cdot 2 \neq 2^{-i+1} \end{aligned}$$

We prove (b)(i) and (b)(ii) simultaneously. Let  $n, m \in \mathbb{N}$  be such that  $n \le k$  and  $m \in \mathbb{N}$  or such that  $n \le p$  and m > s. If m = 1 there is nothing to prove, so let m > 1 and  $f \in A_m^n$ . We have to prove that  $h(i, k)(f) \in A_m^n = L_m^n \cap C_m^n$ .

Since  $f \in L_m^n$  and  $n \le p < i$ , we have for  $j \le n$ 

$$h(i, k)(f)(x_j) = f(x_j) \in [-1 + \frac{1}{m}, 1 - \frac{1}{m}].$$

Hence  $h(i, k)(f) \in L_m^n$ . To prove that  $h(i, k)(f) \in C_m^n$ , we take  $y \in U_m^x$ , where  $x \in \{x_0\} \cup \{\phi(j) : j \le n\}$ . Notice that  $x \ne x_i$  because

$$x_i \notin U_{s+1}^{x_0} \cup \bigcup \{ U_{\uparrow}^{\phi(j)} : j \leq p, \ \phi(j) \neq x_0 \},\$$

and  $n \leq p$ .

If  $y \neq x_i$ , then since  $f \in B_m^{(x, n)}$ ,

$$|h(i, k)(f)(y) - h(i, k)(f)(x)| = |f(y) - f(x)| \le \frac{1}{n}.$$

Now assume that  $y = x_i$ . If  $x \neq x_0$ , then  $x = \phi(j)$  for some  $j \le n$  with  $\phi(j) \neq x_0$ . Then

 $x_i = y \in U_m^{\phi(j)} \subset U_1^{\phi(j)},$ 

116

which is a contradiction. Hence  $x = x_0$ . If m > s, then  $U_m^{x_0} \subset U_{s+1}^{x_0}$ , so  $x_i \notin U_m^{x_0}$ . But this means that  $y \notin U_m^x$ , which gives a contradiction. So  $n \le k$  and we get

$$|h(i, k)(f)(y) - h(i, k)(f)(x)| = |\frac{1}{k} - 0| = \frac{1}{k} \le \frac{1}{n}.$$

We conclude that  $h(i, k)(f) \in B_m^{(x, n)}$ , hence

$$h(i, k)(f) \in B_m^{(x_0, n)} \cap \bigcap_{j=1}^n B_m^{(\phi(j), n)} = C_m^n.$$

So  $h(i, k)(f) \in C_m^n \cap L_m^n = A_m^n$ . This proves (b)(i) and (b)(ii). For (b)(iii), let n > k. Since  $x_i \in U_x^{x_0}$ , we have for  $f \in P$ ,

$$|h(i, k)(f)(x_i) - h(i, k)(f)(x_0)| = \frac{1}{k} > \frac{1}{n}.$$

We conclude that  $h(i, k)(f) \notin B_s^{(x_0, n)}$ , hence  $h(i, k)(f) \notin A_s^n$ . This proves the lemma.

**3.2.7 COROLLARY:** For every  $n, m \in \mathbb{N}$  with n > 1 we have that  $A_m^n \in \mathcal{J}(P)$  and  $A_m^n \in \mathcal{J}(A_{m+1}^n)$ .

**PROOF:** By lemma 3.2.2, we have that  $A_m^n$  is closed in P and  $A_m^n \subset A_{m+1}^n$ , hence  $A_m^n$  is closed in  $A_{m+1}^n$ .

Let  $f \in C(Q, P)$  or  $f \in C(Q, A_{m+1}^n)$  and let  $\varepsilon > 0$ . By lemma 3.2.1 (b),

 $U_m^{x_0} \setminus [U_{m+1}^{x_0} \cup \bigcup \{U_1^{\phi(j)} : j \le n, \phi(j) \ne x_0\}]$ 

is infinite so we can choose  $x_i$  in this set such that i > n, and  $2^{-i} < \varepsilon/2$ . Define  $g: Q \to P$  by  $g = h(i, 1) \circ f$ . Then g is well-defined. By lemma 3.2.6 (a), g is continuous and

 $\hat{d}(g, f) \le \hat{d}(h(i, 1), 1) \le 2^{-i+1} < \varepsilon.$ 

If  $f \in C(Q, A_{m+1}^n)$ , we have by lemma 3.2.6 (b)(ii),

 $g(Q) \subset h(i, 1)(A_{m+1}^n) \subset A_{m+1}^n$ .

Furthermore because n > 1, we have by lemma 3.2.6 (b)(ii),  $g(Q) \cap A_m^n = \emptyset$ . We conclude that  $A_m^n \in \mathcal{Z}(P)$  and  $A_m^n \in \mathcal{Z}(A_{m+1}^n)$ .

**3.2.8 COROLLARY:** For every  $n_1 < \cdots < n_m \in \mathbb{N}$ ,  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$ ,  $p \in \mathbb{N}$  and  $i \in \mathbb{N}$  we have that

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{Z}(\bigcap_{k=1}^{m} A_{i_k}^{n_k})$$

and

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{Z}(\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p}).$$

**PROOF:** By lemma 3.2.2, we have that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is closed in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  and  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is closed in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p}$ .

Let  $f: Q \to \bigcap_{k=1}^{m} A_{i_k}^{n_k}$  or  $f: Q \to \bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p}$  be continuous and let  $\varepsilon > 0$ . By lemma 3.2.1 (b),

$$U_{i}^{x_{0}} \setminus [U_{i+1}^{x_{0}} \cup \bigcup \{U_{1}^{\phi(j)} : j \leq n_{m} + p, \phi(j) \neq x_{0}\}]$$

is infinite, so we can choose  $x_{i_0}$  in this set, such that  $i_0 > n_m + p$ , and  $2^{-i_0} < \varepsilon/2$ . Define  $g: Q \to P$  by  $g = h(i_0, n_m) \circ f$ . Then g is a well-defined continuous function and  $\hat{d}(g, f) < \varepsilon$ 

Since  $f(Q) \subset \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ , we have by lemma 3.2.6 (b)(i),  $g(Q) \subset \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ . Furthermore if  $f(Q) \subset A_{i+1}^{n_m+p}$ , then by lemma 3.2.6 (b)(ii),  $g(Q) \subset A_{i+1}^{n_m+p}$ . By lemma 3.2.6 (b)(iii),  $g(Q) \cap A_{i+1}^{n_m+p} = \emptyset$ . We conclude that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k})$  and  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p})$ .

**3.2.9 LEMMA:** For every n > 1:  $\{A_m^n\}_{m>1}$  is a skeleton in *P*.

**PROOF:** Observe that *P* is a Keller space. To prove the lemma we shall verify the conditions in theorem 3.1.7. By corollary 3.2.7 and lemma 3.2.2,  $\{A_m^n\}_{m>1}$  is an increasing family of *Z*-sets in *P*. Again by corollary 3.2.7, we have for every  $m \in \mathbb{N}$ ,  $A_m^n \in \mathcal{I}(A_{m+1}^n)$ . By lemma 3.2.5,  $A_m^n$  is convex and infinite-dimensional. So we only have to verify that  $\bigcup_{m=2}^{\infty} A_m^n$  is dense in *P*. Notice that  $C_{p,0}(X)$  is dense in  $\{0\} \times (-1, 1)^Y$  which is dense in *P*, hence  $C_{p,0}(X)$  is dense in *P*. Since  $\bigcup_{m=2}^{\infty} A_m^n \supset C_{p,0}(X), \bigcup_{m=2}^{\infty} A_m^n$  is dense in *P*. We obtain that  $\{A_m^n\}_{m>1}$  is a skeleton in *P*.  $\Box$ 

**3.2.10 LEMMA:** For every  $n_1 < \cdots < n_m \in \mathbb{N}$ ,  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$  and  $p \in \mathbb{N}$  we have that  $\{\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i>1}$  is a skeleton in  $\bigcap_{k=1}^m A_{i_k}^{n_k}$ .

**PROOF:** By lemma 3.2.5,  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Keller space. To prove the lemma we again shall verify the conditions in theorem 3.1.7. As mentioned in section 3.1 we may assume  $i_1 < \cdots < i_m$ . By corollary 3.2.8, we have for every  $i \in \mathbb{N}$ ,

§3.2. Homeomorphic function spaces part 1

and

$$\bigcap_{k=1}^{m} A_{i_{k}}^{n_{k}} \cap A_{i}^{n_{m}+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_{k}}^{n_{k}}),$$

$$\bigcap_{k=1}^{m} A_{i_{k}}^{n_{k}} \cap A_{i}^{n_{m}+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_{k}}^{n_{k}} \cap A_{i+1}^{n_{m}+p}).$$

By lemma 3.2.5,  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is convex and infinite dimensional, so we only have to verify that  $\bigcup_{i=1}^{m} A_i^{n_m+p} \cap \bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is dense in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$ . To this end we have to prove that for arbitrary  $g \in \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ ,  $y_1, \ldots, y_n \in X$  and  $\varepsilon > 0$ , we have for

$$U = \{ f \in P : |f(y_l) - g(y_l)| < \varepsilon \text{ for every } l \le n \}$$

that  $(U \cap \bigcap_{k=1}^{m} A_{i_k}^{n_k}) \cap \bigcup_{i=1}^{\infty} A_i^{n_m+p} \neq \emptyset$ . Since we deal with decreasing clopen bases it is possible to find  $i_0 > n_m + p$  such that  $1/i_0 < \varepsilon$  and

(1) 
$$U_{i_0}^{\phi(j_1)} \cap U_{i_0}^{\phi(j_2)} = \emptyset$$
 if  $\phi(j_1) \neq \phi(j_2)$  and  $j_1, j_2 \le n_m + p$ ,  
(2)  $U_{i_0}^{\phi(j_0)} \cap U_{i_k}^{\phi(j)} = \emptyset$  if  $\phi(j_0) \notin U_{i_k}^{\phi(j)}$ ,  $k \le m$ ,  $j \le n_k$  and  $j_0 \le n_m + p$ ,  
(3)  $U_{i_0}^{\phi(j_0)} \cap U_{i_k}^{x_0} = \emptyset$  if  $\phi(j_0) \notin U_{i_k}^{x_0}$ ,  $k \le m$  and  $j_0 \le n_m + p$ ,  
(4) for  $a \in \{\phi(j) : j \le n_m + p\} \cup \{x_0\}$  and  $y_l \ne a$  we have  $y_l \notin U_{i_0}^a$   $(l \le n)$ , and  
(5) for  $\phi(j) \notin \{x_0, ..., x_{n_m}\}$  we have  $U_{i_0}^{\phi(j)} \cap \{x_0, ..., x_{n_m}\} = \emptyset$   $(j \le n_m + p)$ .

Now define  $f: X \rightarrow [-1, 1]$  by

J

$$f(x) = \begin{cases} (1 - \frac{1}{i_0})g(\phi(j)) & \text{if } x \in \bigcup_{j \le n_m + p} U_{i_0}^{\phi(j)}, \\ 0 & \text{if } x \in U_{i_0}^{x_0}, \\ (1 - \frac{1}{i_0})g(x) & \text{elsewhere.} \end{cases}$$

By lemma 3.2.1 (a) and (1) there is for every  $x \in X$  at most one  $a \in \{\phi(j) : j \le n_m + p\} \cup \{x_0\}$  with  $x \in U_{i_0}^a$ , and since  $g(X) \subset [-1, 1]$ ,  $f(X) \subset [-1, 1]$ , so we conclude that f is a well-defined mapping. Furthermore  $f(x_0) = 0$ , hence  $f \in P$ . To prove the claim we will show that  $f \in (U \cap \bigcap_{k=1}^m A_{i_k}^{n_k}) \cap \bigcup_{i=1}^\infty A_i^{n_m + p}$ . For that we first prove  $f \in U$ . To this end let  $l \le n$ . If  $y_l \notin \{\phi(j) : j \le n_m + p\} \cup \{x_0\}$  then by (4) and by definition of f,  $f(y_l) = (1 - 1/i_0)g(y_l)$  so

$$|f(y_l) - g(y_l)| = \frac{1}{i_0} |g(y_l)| \le \frac{1}{i_0} < \varepsilon.$$

If  $y_l \in \{\phi(j) : j \le n_m + p\} \cup \{x_0\}$  then by the definition of f,

$$|f(y_l) - g(y_l)| \leq \frac{1}{i_0} < \varepsilon.$$

So we indeed have  $f \in U$ .

To prove that  $f \in \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ , we first show that  $f \in \bigcap_{k=1}^{m} C_{i_k}^{n_k}$ . To this end let  $k \le m$ and  $b \in \{\phi(j) : j \le n_k\} \cup \{x_0\}$ . Notice that  $b \in U_{i_0}^b$ , hence  $f(b) = (1-1/i_0)g(b)$ . Let  $x \in U_{i_k}^b$ . First suppose there is  $a \in \{\phi(l) : l \le n_m + p\} \cup \{x_0\}$  such that  $x \in U_{i_0}^a$ . In this case  $f(x) = (1 - 1/i_0)g(a)$ . If  $a = x_0$  then by lemma 3.2.1 (a),  $a = b \in U_{i_k}^b$ , and if  $a \ne x_0$ then by (2) and (3),  $a \in U_{i_k}^b$ . Since

$$g \in C_{i_k}^{n_k} \subset B_{i_k}^{(b, n_k)},$$

we now have

$$|f(x) - f(b)| = (1 - \frac{1}{i_0})|g(a) - g(b)| \le (1 - \frac{1}{i_0})\frac{1}{n_k} < \frac{1}{n_k}.$$

Secondly suppose for every  $a \in \{ \phi(l) : l \le n_m + p \} \cup \{ x_0 \}$  we have  $x \notin U_{i_0}^a$ . Then

$$|f(x) - f(b)| = (1 - \frac{1}{i_0})|g(x) - g(b)| \le (1 - \frac{1}{i_0})\frac{1}{n_k} < \frac{1}{n_k}$$

We conclude that  $f \in \bigcap_{k=1}^{m} C_{i_k}^{n_k}$ . Now let  $k \le m$  and  $j \le n_k$ . By (5) we have

$$|f(x_j)| = |(1 - \frac{1}{i_0})g(x_j)| \le |g(x_j)| \le 1 - \frac{1}{i_k}.$$

So  $f \in \bigcap_{k=1}^{m} C_{i_k}^{n_k} \cap \bigcap_{k=1}^{m} L_{i_k}^{n_k} = \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ .

Finally we have to prove that  $f \in \bigcup_{i=1}^{\infty} A_i^{n_m+p}$ . In fact we show that  $f \in A_{i_0}^{n_m+p}$ . Let  $a \in \{\phi(j) : j \le n_m + p\} \cup \{x_0\}$ , and let  $x \in U_{i_0}^a$ . Then  $f(x) = (1 - 1/i_0)g(a) = f(a)$ , so

$$|f(x) - f(a)| = 0 \le \frac{1}{n_m + p}$$

So  $f \in C_{i_0}^{n_m+p}$ . Since for every  $x \in X$ ,  $|f(x)| \le 1 - 1/i_0$  we have  $f \in A_{i_0}^{n_m+p}$ . This proves the lemma.  $\Box$ 

120

§3.2. Homeomorphic function spaces part 1

**3.2.11 LEMMA:** For every  $n_1 < \cdots < n_m \in \mathbb{N}$ ,  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$  and  $s, t \in \mathbb{N}$  such that s > 1 and  $\bigcap_{k=1}^m A_{i_k}^{n_k} \not\subset A_i^s$ , we have  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^s \in \mathcal{I}(\bigcap_{k=1}^m A_{i_k}^{n_k})$ .

**PROOF:** Let  $n_1 < \cdots < n_m \in \mathbb{N}$ ,  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$  and suppose  $\bigcap_{k=1}^m A_{i_k}^{n_k} \not\subset A_i^s$ . As mentioned in section 3.1 we may assume  $i_1 < \cdots < i_m$ .

If  $s > n_m$  we have by corollary 3.2.8,

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^s \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k}).$$

So from now on we assume that  $s \le n_m$ . If there exists  $l \le m$  such that  $s \le n_l$  and  $t \ge i_l$ , then by lemma 3.2.2,

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \subset A_{i_l}^{n_l} \subset A_t^s$$

and we have a contradiction. So for every  $k \le m$ ,  $s > n_k$  or  $t < i_k$ . There exists  $r \le m$  such that  $n_{r-1} < s \le n_r$ . (Let  $n_0 = 1$ ).

Let  $f: Q \to \bigcap_{k=1}^{m} A_{i_k}^{n_k}$  and  $\varepsilon > 0$ . By lemma 3.2.1 (b),

$$U_t^{x_0} \setminus [U_{t+1}^{x_0} \cup \bigcup \{U_1^{\phi(j)} : j \leq n_m, \phi(j) \neq x_0\}]$$

is infinite. Choose  $x_i$  in this set, such that  $i > n_m$ ,  $2^{-i} < \varepsilon/2$ . Notice that  $x_i \notin \{\phi(j) : j \le n_m\}$ . Define  $g : Q \to P$  by  $g = h(i, n_{r-1}) \circ f$ . Then g is well-defined and  $\hat{d}(f, g) < \varepsilon$ . We claim that  $g(Q) \subset \bigcap_{k=1}^m A_{i_k}^{n_k}$ . To this end let  $k \le m$ . If  $n_k \le n_{r-1}$ , then by lemma 3.2.6 (b)(i),

$$h(i, n_{r-1})(A_{i_k}^{n_k}) \subset A_{i_k}^{n_k}$$

If  $n_{r-1} < n_k \le n_m$ , then  $s \le n_r \le n_k$ . So  $t < i_k$ . Then by lemma 3.2.6 (b)(ii),

 $h(i, n_{r-1})(A_{i_k}^{n_k}) \subset A_{i_k}^{n_k}.$ 

So  $g(Q) \in \bigcap_{k=1}^{m} A_{i_k}^{n_k}$ .

To finish the proof of this lemma, notice that by lemma 3.2.6 (b)(iii),  $g(Q) \cap A_t^s = \emptyset$ (because  $s > n_{r-1}$ ), so that  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_t^s \in \mathcal{Z}(\bigcap_{k=1}^m A_{i_k}^{n_k})$ .

### 3.2.12 THEOREM: A is a Q-matrix in P.

**PROOF:** By lemma 3.2.2 and corollary 3.2.7,  $\mathcal{A}$  is a  $\mathcal{Z}$ -matrix. The theorem now follows directly from the lemmas 3.2.5, 3.2.9, 3.2.10 and 3.2.11.

We now come to the main result of this section.

#### 3.2.13 THEOREM: Let X be a non-locally compact countable space. Then

- (a)  $C_p(X)$  is homeomorphic to  $\sigma_{\omega}$ , and
- (b) If Y is another non-locally compact countable space, then for every  $\varepsilon > 0$ there is  $h \in \mathcal{H}(Q)$  such that  $h(C_{p,0}(X)) = C_{p,0}(Y)$  and  $\hat{d}(h, 1) < \varepsilon$ .

**PROOF:** By lemma 3.2.2 and lemma 3.2.4 it follows that  $C_{p,0}(X) = \ker \mathcal{A}$ , so by corollary 3.1.13 (a),  $C_{p,0}(X)$  is homeomorphic to  $\sigma_{\omega}$ . By proposition 2.3.2,  $C_p(X)$  is homeomorphic to  $C_{p,0}(X) \times \mathbb{R}$ , hence

 $C_p(X) \approx C_{p,0}(X) \times \mathbb{R} \approx \sigma_{\omega} \times \mathbb{R} \approx \sigma_{\omega}.$ 

Now let *Y* be another non-locally compact countable space and let  $\varepsilon > 0$ . Then by corollary 3.1.13 (b), there is  $h \in \mathcal{H}(Q)$  such that  $h(C_{p,0}(X)) = C_{p,0}(Y)$  and  $\hat{d}(h, 1) < \varepsilon$ .  $\Box$ 

In [38] it was proved that for a non-locally compact countable space X,  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ . He did not prove theorem 3.2.13 (b) for  $C_{p,0}^*(X)$ . To show that in this case theorem 3.2.13 (b) is also valid we will give, for a non-locally compact countable space X, a Q-matrix  $\mathcal{A}$  such that  $C_{p,0}^*(X) = \ker \mathcal{A}$ . Since the calculations are more or less the same as in the case of  $C_{p,0}(X)$  we will be brief.

Let X be a countable space which fails to be locally compact at some point  $x_0 \in X$ . Again let  $X = \{x_0, x_1, x_2, ...\}, Y = \{x_1, x_2, ...\}, P = \{0\} \times [-1, 1]^Y$  and let  $\phi$  be the map  $\pi_1 \circ \psi$ :  $\mathbb{N} \to X$ . We consider the same clopen bases as above. As mentioned in the introduction,  $C_{p,0}^*(X) = \{f \in C_{p,0}(X) : f(X) \subset [-1+1/m, 1-1/m] \text{ for some } m \in \mathbb{N}\}.$ 

For every  $x \in X$  and  $n, m \in \mathbb{N}$ , define

$$B_m^{(x, n)} = \{ g \in \{0\} \times [-1 + \frac{1}{m}, 1 - \frac{1}{m}]^Y : g(U_m^x) \subset [g(x) - \frac{1}{n}, g(x) + \frac{1}{n}] \}.$$

For every  $n, m \in \mathbb{N}$ , define  $A_1^n = \emptyset$ , and for m > 1,

$$A_m^n = B_m^{(x_0, n)} \cap \bigcap_{j=1}^n B_m^{(\phi(j), n)}.$$

As in lemma 3.2.4 we have the following:

**3.2.14 LEMMA:** 
$$C_{p,0}^*(X) = \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} A_m^n$$
.

One can prove that  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  is a *Q*-matrix in *P*. The proof is more or less the same as the proof of theorem 3.2.12 and we will only give some remarks.

It is not necessary to copy the proof of corollary 3.2.7. We have  $A_m^n \in \mathcal{Z}(P)$  because of lemma 3.1.4. Furthermore we can simplify the proof of the claim in lemma 3.2.10: The condition in (5) can be skipped but we need  $i_0 > i_m$  to prove  $f \in A_{i_0}^{n_m+p}$ . The function f can be defined as follows:

$$f(x) = \begin{cases} g(\phi(j)) & \text{if } x \in U_{i_0}^{\phi(j)}(j \le n_m + p), \\ 0 & \text{if } x \in U_{i_0}^{x_0}, \\ g(x) & \text{elsewhere.} \end{cases}$$

As in theorem 3.2.13, we have

٢

### 3.2.15 THEOREM: Let X be a non-locally compact countable space. Then

- (a)  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ , and
- (b) If Y is another non-locally compact countable space, then for every  $\varepsilon > 0$ there is  $h \in \mathcal{H}(Q)$  such that  $h(C_{p,0}^*(X)) = C_{p,0}^*(Y)$  and  $\hat{d}(h, 1) < \varepsilon$ .

The question remains whether for non-locally compact countable spaces X and Y there is a homeomorphism from the Hilbert cube onto itself arbitrary close to the identity which maps  $C_p(X)$  onto  $C_p(Y)$  resp.  $C_p^*(X)$  onto  $C_p^*(Y)$ . By theorem 3.2.13, we have for  $\varepsilon > 0$ ,  $h \in \mathcal{H}([-1, 1] \times P)$  such that  $\hat{d}(h, 1) < \varepsilon$ , and

 $h((-1, 1) \times C_{p,0}(X)) = (-1, 1) \times C_{p,0}(Y).$ 

By proposition 2.3.2,  $(-1, 1) \times C_{p,0}(X)$  is homeomorphic to  $C_p(X)$ . This is not what we need to solve the above question. We actually need a homeomorphism from  $[-1, 1] \times Q$  to Q which maps  $(-1, 1) \times C_{p,0}(X)$  onto  $C_p(X)$ . Whether such a homeomorphism exists remains unsolved.

## §3.3. Homeomorphic function spaces part 2

In this section we give another proof of the statement that for a non-locally compact countable space X the function space  $C_p(X)$  is homeomorphic to  $\sigma_{00}$ . We first compare the strategies followed in this section and section 3.2. In section 3.2 we found a Q-matrix  $\mathcal{A}$  such that ker  $\mathcal{A} = C_{p,0}(X)$ . Using corollary 3.1.13, it was then easily deduced

that  $C_p(X)$  and  $\sigma_{\omega}$  are homeomorphic. The *Q*-matrix involved asked for a lot of technical calculations. However this was not a waste of time, since this strategy also gives the result stated in theorem 3.2.13 (b). The strategy in this section starts with a test space *T*. One could say that *T* is the "simplest" non-locally compact countable space: moreover *T* is a closed subspace of any non-locally compact countable space (lemma 3.3.1). Next a *Q*-matrix  $\mathcal{B}$  will be given such that ker  $\mathcal{B} = C_{p,0}(T)$ . This *Q*-matrix is much easier to deal with than the one in section 3.2. It follows that  $C_p(T)$  is homeomorphic to  $\sigma_{\omega}$ . To get this also for arbitrary non-locally compact spaces *X* we use strong results of Toruńczyk [50], [51], which gives the necessary connection between  $C_p(X)$  and  $C_p(T)$ . The method of this section was used by van Mill in [38], where he proved that  $C_p^*(X)$  is homeomorphic to  $\sigma_{\omega}$ .

We first define the test space T. The underlying set of T is  $\mathbb{N}^2 \cup \{\infty\}$ . Each point of  $\mathbb{N}^2$  is isolated and  $\{(\{n, n+1, \cdots\} \times \mathbb{N}) \cup \{\infty\}\}_{n \in \mathbb{N}}$  is a local open base at  $\infty$ . Then T is obviously a countable space which is not locally compact at  $\infty$ . Among the non-locally compact countable spaces, T is a special one as is shown in the following

**3.3.1 LEMMA:** Let X be a non-locally compact space. Then X contains a closed copy of T

**PROOF:** Let  $x_0$  be a point where X fails to be locally compact. Let  $\{U_n : n \in \mathbb{N}\}$  be a decreasing open base at  $x_0$ . Since no  $\overline{U}_n$  is compact we may assume that for each  $n \in \mathbb{N}, \overline{U}_n \setminus U_{n+1}$  contains an infinite closed discrete subset  $D_n$ . Let  $S = \{x\} \cup \bigcup_{n=1}^{\infty} D_n$ . Then S is obviously closed in X and homeomorphic to T.  $\Box$ 

Recall from the introduction that  $C_p(T) = \{ f : T \to (-1, 1) : f \text{ is continuous} \}$  and  $C_{p,0}(T) = \{ f \in C_p(T) : f(\infty) = 0 \}.$ 

For convenience let I = [-1, 1],  $I_m = [-1 + 1/m, 1 - 1/m]$  for every  $m \in \mathbb{N}$  and

 $B(\varepsilon) = \prod_{i=1}^{\infty} [-\varepsilon, \varepsilon]_i$  for every  $\varepsilon > 0$ .

For every  $i \in \mathbb{N}$ , let  $Q_i = \prod_{j=1}^{\infty} I_{ij}$ , where  $I_{ij} = I$  for every  $j \in \mathbb{N}$ . Let  $P = \prod_{i=1}^{\infty} Q_i$ , where  $Q_i = Q$  for every  $i \in \mathbb{N}$ . Observe that there is a convexity preserving homeomorphism between P and Q, hence by theorem 3.1.1 each Keller space in P is a Hilbert cube. Define  $\phi: C_{p,0}(T) \to P$  by

$$\phi(f)_i = (f((i,j)))_{i \in \mathbb{N}}, \text{ for every } i \in \mathbb{N}.$$

Then  $\phi$  is easily seen to be an embedding. For every  $n, m \in \mathbb{N}$  define  $A_m^n \subset P$  by

(1) 
$$A_1^n = \emptyset$$
 for every  $n \in \mathbb{N}$  and  
(2)  $A_m^n = \prod_{i=1}^m ((I_m)^n \times I \times I \times \cdots)_i \times \prod_{i=m+1}^\infty B_i(2^{-n})$  for every  $n \in \mathbb{N}$  and  $m \ge 2$ .

It will turn out that  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$  is a *Q*-matrix in *P* such that its kernel is equal to  $\phi(C_{p,0}(T))$ . As will be clear in the sequel the calculations involved are not so comprehensive as the ones in section 3.2.

```
3.3.2 LEMMA: A is a I-matrix in P.
```

**PROOF:** By lemma 3.1.4 we have for every  $n, m \in \mathbb{N}$ , that  $A_m^n \in \mathcal{Q}(P)$ . It is clear that for every  $n, m \in \mathbb{N}, A_m^n \subset A_{m+1}^n$  and  $A_m^{n+1} \subset A_m^n$ .

```
3.3.3 LEMMA: ker \mathcal{A} = \phi(C_{p,0}(T)).
```

**PROOF:** Let  $f = (f_{ij})_{(i,j) \in \mathbb{N}^2} \in \ker \mathcal{A}$  and  $(i, j) \in \mathbb{N}^2$ . Define  $\tilde{f}: T \to (-1, 1)$  by  $\tilde{f}((i, j)) = f_{ij}$  for  $(i, j) \in \mathbb{N}^2$  and  $\tilde{f}(\infty) = 0$ . Since  $f \in \bigcup_{m=1}^{\infty} A_m^j$ , there is  $m \in \mathbb{N}$  with  $f \in A_m^j$ . If  $i \le m$  then  $f_{ij} \in I_m \subset (-1, 1)$  and if i > m then  $f_{ij} \in [-2^{-j}, 2^{-j}] \subset (-1, 1)$ , hence  $\tilde{f}$  is well-defined. We will prove that  $\tilde{f}$  is continuous. To this end we only have to prove that  $\tilde{f}$  is continuous at  $\infty$ . Let  $\varepsilon > 0$  and  $n \in \mathbb{N}$  such that  $2^{-n} < \varepsilon$ . Let  $m \in \mathbb{N}$  be such that  $f \in A_m^n$ . Then  $|f_{ij}| \le 2^{-n} < \varepsilon$  for i > m and  $j \in \mathbb{N}$ . So  $\tilde{f}((\{m+1, m+2, \cdots\} \times \mathbb{N}) \cup \{\infty\}) \subset (-\varepsilon, \varepsilon)$ , hence  $\tilde{f} \in C_{p,0}(T)$ . Obviously  $\phi(\tilde{f}) = f$ , so  $f \in \phi(C_{p,0}(T))$ .

Conversely let  $f \in C_{p,0}(T)$  and  $n \in \mathbb{N}$ . Since f is continuous at  $\infty$ , there is  $m_1 \in \mathbb{N}$  with  $|f(i, j)| < 2^{-n}$  for  $i > m_1$  and  $j \in \mathbb{N}$ . There is  $m \ge m_1$  such that for every  $i \le m$  and  $j \le n$  we have  $|f(i, j)| \le 1 - 1/m$ . Then  $\phi(f) \in A_m^n$  and we conclude that  $\phi(f) \in \ker \mathcal{A}$ .

3.3.4 LEMMA: A is a Q-matrix in P.

**PROOF:** By lemma 3.3.2 *A* is a  $\mathcal{I}$ -matrix in *P*. Notice that by lemma 3.1.4 we have for every  $\varepsilon > 0$  and  $\delta < \varepsilon$  that  $B(\delta) \in \mathcal{I}(B(\varepsilon))$ .

CLAIM 1: For every  $n \in \mathbb{N}$ ,  $\{A_m^n\}_{m>1}$  is a skeleton in P.

Notice that P is a Keller space. To prove this claim we verify the conditions in

theorem 3.1.7. By lemma 3.1.4 we have for every  $n, m \in \mathbb{N}$  that  $A_m^n \in \mathcal{I}(P)$  and  $A_m^n \in \mathcal{I}(A_{m+1}^n)$ . Because each  $A_m^n$  (m > 1) is a product of non-degenerate intervals, it is convex and infinite-dimensional. To prove that for every  $n \in \mathbb{N}$ ,  $\bigcup_{m=1}^{\infty} A_m^n$  is dense in P, let  $f \in P$  and  $U = \prod_{(i, j) \in \mathbb{N}^2} U_{ij}$  a standard neighborhood of f in P. There is  $m_1 \in \mathbb{N}$  such that  $i > m_1$  implies  $U_{ij} = I$  for every  $j \in \mathbb{N}$ . There is  $m \ge m_1$  such that  $U_{ij} \cap I_m \neq \emptyset$  for every  $i, j \in \mathbb{N}$ . We claim that  $U \cap A_m^n \neq \emptyset$ . Indeed, let  $(i, j) \in \mathbb{N}^2$ . If  $i \le m$  then  $\pi_{ij}(U \cap A_m^n) \supset U_{ij} \cap I_m \neq \emptyset$ . If i > m then  $\pi_{ij}(U \cap A_m^n) = I \cap B(2^{-n}) \neq \emptyset$ . We conclude that  $\{A_m^n\}_{m \ge 1}$  is a skeleton in P.

Now let  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$ . As mentioned in section 3.1 we may assume  $i_1 < \cdots < i_m$ 

CLAIM 2:  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is is a Keller space in *P*, hence a Hilbert cube.

Since each  $A_{i_k}^{n_k}$  is a product of closed intervals,  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is a product of closed intervals. Hence  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is compact and convex. Since  $A_{i_1}^{n_m} \subset \bigcap_{k=1}^m A_{i_k}^{n_k}$  and  $i_1 \neq 1$ ,  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is infinite-dimensional. We conclude that  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is a Keller space in *P*. Hence by theorem 3.1.1,  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is a Hilbert cube.

CLAIM 3: For every  $p \in \mathbb{N}$ ,  $\{\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i>1}$  is a skeleton in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$ .

By claim 2,  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is a Keller space in *P*. We prove this claim by verifying the conditions in theorem 3.1.7. Let  $p \in \mathbb{N}$  and  $i \in \mathbb{N} \setminus \{1\}$ . Let *j* be greater than max $(i, i_m)$ . The *j*-th factor space of  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$  is  $B(2^{-n_m})$  and the *j*-th factor space of  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^{m} A_{i_k}^{n_k})$  (by lemma 3.1.4).

If  $i \ge i_m$ , then the (i+1)-th factor space of  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is  $B(2^{-n_m-p})$  and the (i+1)-th factor space of  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p}$  is  $B(2^{-n_m})$ . Hence by lemma 3.1.4 we have for every  $i \ge i_m$ ,  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{I}(\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p})$ . Since  $\bigcup_{i\ge i_m} (\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p})$  is dense in  $\bigcap_{k=1}^m A_{i_k}^{n_k}$ , we have by theorem 3.1.7 that  $\{\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i\ge i_m}$  is a skeleton in  $\bigcap_{k=1}^m A_{i_k}^{n_k}$ . Hence by the definition of skeletons,  $\{\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^{n_m+p}\}_{i>1}$  is a skeleton in  $\bigcap_{k=1}^m A_{i_k}^{n_k}$ .

CLAIM 4: For every  $s \in \mathbb{N}$  and for every  $t \in \mathbb{N}$  such that  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \not\subset A_t^s$  we have  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_t^s \in \mathcal{Z}(\bigcap_{k=1}^{m} A_{i_k}^{n_k})$ .

If  $s > n_m$ , then by claim 3,  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^s \in \mathcal{I}(\bigcap_{k=1}^m A_{i_k}^{n_k})$ . If  $s \le n_m$ , there is  $p \le m$ such that  $n_{p-1} < s \le n_p$  (let  $n_0 = 0$ ). This implies  $t < i_p$ , otherwise  $\bigcap_{k=1}^m A_{i_k}^{n_k} \subset A_{i_l}^{n_l} \subset A_i^s$ . So there is  $r \le p$  such that  $i_{r-1} < t+1 \le i_r$  (let  $i_0 = 0$ ). The (t+1)-th factor space of  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^s$  is  $B(2^{-n_{r-1}})$  and the (t+1)-th factor space of  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^s$  is  $B(2^{-s})$ . Because  $s > n_{p-1} \ge n_{r-1}$ , by lemma 3.1.4  $\bigcap_{k=1}^m A_{i_k}^{n_k} \cap A_i^s \in \mathcal{I}(\bigcap_{k=1}^m A_{i_k}^{n_k})$ .

By claims 1-4 we have that  $\mathcal{A}$  is a Q-matrix in P.

**3.3.5 COROLLARY:**  $C_p(T)$  is homeomorphic to  $\sigma_{\omega}$ .

**PROOF:** By lemmas 3.3.3, 3.3.4 and corollary 3.1.13 (a), we have that  $C_{p,0}(T)$  is homeomorphic to  $\sigma_{\omega}$ . As in theorem 3.2.13 we can prove that  $C_p(T)$  is homeomorphic to  $\sigma_{\omega}$ .  $\Box$ 

In [38] van Mill constructed a Q-matrix  $\mathcal{A}$  such that  $C_{p,0}^*(T) = \ker \mathcal{A}$ , hence  $C_p^*(T)$ and  $\sigma_{\omega}$  are homeomorphic. From this result he derived for an arbitrary non-locally compact countable space X, that  $C_p^*(X)$  and  $\sigma_{\omega}$  are homeomorphic. We proceed in the same way to derive that  $C_p(X)$  and  $\sigma_{\omega}$  are homeomorphic. We first need results of Toruńczyk (cf. theorem 3.3.6).

For a linear space E we define  $\Sigma E = \{x \in E^{\infty} : x_i = 0 \text{ for all but finitely many } i\}$ .

**3.3.6 THEOREM:** Let *E* be a locally convex linear space. Then (a) ([50]) for a closed AR X in *E* we have  $X \times \Sigma E \approx \Sigma E$ , and (b) ([51])  $\Sigma(E^{\infty}) \approx \Sigma \mathbb{R} \times E^{\infty}$ .

This theorem will be used in the proof of theorem 3.3.9, which formulates in a sense the connection between  $C_p(T)$  and  $C_p(X)$  for an arbitrary non-locally compact space X. Before we come to this theorem we have to prove some lemmas. Recall from examples 3.1.8 and 3.1.14 that  $\Sigma = \{x \in Q : \exists n \in \mathbb{N} \forall i \in \mathbb{N}, |x_i| \le 1 - 1/n\}$  is a skeletoid in Q and that  $\Sigma^{\infty}$  is homeomorphic to  $\sigma_{\omega}$ .

**3.3.7 LEMMA:** Let X be a  $\sigma$ -compact space. Then  $\Sigma$  contains a closed copy of X.

**PROOF:** Since every space admits an embedding in the Hilbert cube, and  $[-1/2, 1/2]^{\infty}$  is a Hilbert cube, X has a compactification  $\alpha X \subset Q$  such that  $\alpha X \in \mathcal{J}(Q)$  (cf. lemma 3.1.4). Since X is  $\sigma$ -compact we then have by lemma 3.1.3 (a),  $X \in \mathcal{I}_{\sigma}(Q)$ . Then by corollary 3.1.11, there is a homeomorphism  $h: Q \to Q$  such that

 $h(\alpha X) \cap \Sigma = h(X)$ . Consequently h(X) is a closed copy of X in  $\Sigma$ .

A space X is called an *absolute*  $F_{\sigma\delta}$  if X is an  $F_{\sigma\delta}$  in Y, for every space Y in which X is embedded. It is well-known that a space X is an absolute  $F_{\sigma\delta}$  whenever it is an  $F_{\sigma\delta}$  in some completely metrizable space.

**3.3.8 COROLLARY:** Let X be an absolute  $F_{\sigma\delta}$ . Then  $\sigma_{\omega}$  contains a closed copy of X.

**PROOF:** We may assume that X is a subspace of Q. Let  $X = \bigcap_{i=1}^{\infty} F_i$ , where each  $F_i$  is a  $F_{\sigma}$ -subspace of Q. Then each  $F_i$  is  $\sigma$ -compact. So by lemma 3.3.7 there exists for every  $i \in \mathbb{N}$  a closed embedding  $f_i: F_i \to \Sigma$ . Now define  $f: X \to \Sigma^{\infty}$  by  $f(x) = (f_1(x), f_2(x), \cdots)$ . It is easily seen that f is a continuous injection.

Define  $\phi: X \to \prod_{i=1}^{\infty} F_i$  by  $\phi(x) = (x, x, \cdots)$  and  $g: \prod_{i=1}^{\infty} F_i \to \Sigma^{\infty}$  by  $g((x_i)_{i \in \mathbb{N}}) = (f_i(x_i))_{i \in \mathbb{N}}$ . Then  $f = g \circ \phi$  and g is easily seen to be a closed embedding. Hence to prove that f is a closed embedding it suffices to prove that  $\phi(X)$  is closed in  $\prod_{i=1}^{\infty} F_i$ . Let  $y = (y_i)_{i \in \mathbb{N}} \in \prod_{i=1}^{\infty} F_i \setminus \phi(X)$ . Then there are  $i, j \in \mathbb{N}$  with  $y_i \neq y_j$ . There are U open in  $F_i$  and V open in  $F_j$  such that  $y_i \in U$  and  $y_j \in V$  and  $U \cap V = \emptyset$ . Let

 $O = F_1 \times \cdots F_{i-1} \times U \times F_{i+1} \cdots \times F_{j-1} \times V \times F_{j+1} \times \cdots$ 

Then  $y \in O$  and  $O \cap \phi(X) = \emptyset$ . We conclude that f is a closed embedding. Since  $\Sigma^{\infty}$  and  $\sigma_{\omega}$  are homeomorphic, we are done.  $\Box$ 

**3.3.9 THEOREM:** Let X be an absolute  $F_{\sigma\delta}$  which moreover is an AR. Then  $X \times \sigma_{\omega}$  is homeomorphic to  $\sigma_{\omega}$ .

**PROOF:** By corollary 3.3.8, we may assume that X is closed in  $\sigma_{\omega}$ . Then by theorem 3.3.6 (a),  $X \times \Sigma \sigma_{\omega}$  and  $\Sigma \sigma_{\omega}$  are homeomorphic. By theorem 3.3.6 (b) we have

 $\Sigma \sigma_{\omega} \approx \Sigma (\sigma_{\omega})^{\infty} \approx \Sigma \mathbb{R} \times (\sigma_{\omega})^{\infty} \approx l_{f}^{2} \times (\sigma_{\omega})^{\infty} \approx \sigma_{\omega}.$ 

So we conclude  $X \times \sigma_{\omega}$  is homeomorphic to  $\sigma_{\omega}$ .

In the proof of theorem 3.3.11 it will be clear how this theorem connects  $C_p(T)$  with  $C_p(X)$ , for arbitrary non-locally compact countable spaces X. We need one more lemma.

**3.3.10 LEMMA** ([19]): If X is a countable space, then  $C_p(X)$  is an absolute  $F_{\sigma\delta}$ .

§3.4. Remarks

**PROOF:** For every  $x \in X$ , let  $\{U_n^x\}_{n \in \mathbb{N}}$  be a decreasing clopen base at x. Then by the definition of continuity,

$$C_p(X) = \bigcap_{x \in X} \bigcap_{n=1}^{\infty} \bigcup_{m=1}^{\infty} \{ g \in \mathbb{R}^X : g(U_m^x) \subset [g(x) - \frac{1}{n}, g(x) + \frac{1}{n}] \}.$$

Since each set  $\{g \in \mathbb{R}^X : g(U_m^x) \subset [g(x) - \frac{1}{n}, g(x) + \frac{1}{n}]\}$  is closed in  $\mathbb{R}^X$ ,  $C_p(X)$  is an  $F_{\sigma\delta}$  in  $\mathbb{R}^X$  and hence an absolute  $F_{\sigma\delta}$ .  $\Box$ 

**3.3.11 THEOREM:** Let X be a non-locally compact countable space. Then  $C_p(X)$  is homeomorphic to  $\sigma_{\omega}$ .

**PROOF:** By lemma 3.3.1 we may assume that T is a closed subspace of X. Then by proposition 2.3.2,  $C_p(X)$  and  $C_{p,T}(X) \times C_p(T)$  are homeomorphic. Since  $C_{p,T}(X)$  is a linear subspace of the locally convex space  $C_p(X)$  it is locally convex as well. By The Dugundji Extension Theorem [39, Th. 1.4.13],  $C_{p,T}(X)$  is an AR. It is easily seen that  $C_{p,T}(X)$  is closed in  $C_p(X)$ , hence by lemma 3.3.10,  $C_{p,T}(X)$  is an absolute  $F_{\sigma\delta}$ . So by theorem 3.3.9,  $C_{p,T}(X) \times \sigma_{\omega}$  and  $\sigma_{\omega}$  are homeomorphic. We conclude that by corollary 3.3.5

 $C_p(X) \approx C_{p, T}(X) \times \sigma_{\omega} \approx \sigma_{\omega}. \Box$ 

Theorem 3.3.11 can be found in [6].

## §3.4. Remarks

Van Mill conjectured in [38] that for a non-discrete countable space X,  $C_p^*(X)$  and  $\sigma_{\omega}$  are homeomorphic. In the preceding sections, it became clear that *Q*-matrices were a handy tool to prove for non-locally compact countable spaces X, that  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ . The question remains whether this also holds for arbitrary non-discrete countable spaces. Recently Dobrowolski, Gulko and Mogilski in [20] and Cauty in [16] independently answered this question in the affirmative. In this section we shortly discuss both papers.

In section 3.3 the test space T plays an important role. It is the "simplest" nonlocally compact countable space, which is a closed subspace of any non-locally compact countable space. In the class of non-discrete countable spaces, the role of T is played by  $[1, \omega]$  because any non-discrete countable space contains a closed copy of  $[1, \omega]$ . Following the strategy in section 3.3, we have to prove that  $C_p([1, \omega])$  is homeomorphic to  $\sigma_{\omega}$ , to obtain for arbitrary non-discrete countable spaces X, that  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ . Both in [16] and [20] it is proved that  $C_{p,0}([1, \omega]) = \{f \in C_p([1, \omega]) : f(\omega) = 0\}$  is homeomorphic to  $\sigma_{\omega}$  (hence  $C_p([1, \omega])$  is homeomorphic to  $\sigma_{\omega}$ ). The approaches in both papers are in a sense the same: they both rely on theorems of Bestvina and Mogilski [13].

First we discuss the proof in [20]. For a space X and  $x \in X$ , let

 $W(X, x) = \{x \in X^{\infty} : x_n = x \text{ for all but finitely many } n\}.$ 

We have the following characterization of  $\sigma_{\omega}$ ,

**3.4.1 THEOREM** ([20]): An AR X is homeomorphic to  $\sigma_{\omega}$  iff the following conditions are satisfied,

(a)  $X = \bigcup_{i=1}^{\infty} X_i$ , where each  $X_i$  is an absolute  $F_{\sigma\delta}$  and a Z-set in X,

(b) there is  $x \in X$  and there is a copy Y of X such that  $W(X, x) \subset Y \subset X^{\infty}$  and,

(c) X contains a closed copy of  $X^{\infty}$ .

The proof of this theorem depends strongly on results derived by Bestvina and Mogilski in [13]. In [20] it is proved that  $C_{p,0}([1, \omega])$  satisfies the conditions in theorem 3.4.1, so that  $C_{p,0}([1, \omega])$  is homeomorphic to  $\sigma_{\omega}$ . Hence following the strategy of section 3.3, for non-discrete countable spaces X,  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ .

The proof in [16] depends on a theorem derived by Bestvina and Mogilski in [13]. Before we can formulate this theorem we have to give some definitions.

Let X and Y be spaces and let  $\mathcal{U}$  be an open cover of Y. Two functions  $f, g \in C(X, Y)$  are said to be  $\mathcal{U}$ -close if for every  $x \in X$ , there is  $U \in \mathcal{U}$  such that  $\{f(x), g(x)\} \subset U$ . We have to extend the definition of a Z-set to arbitrary spaces. A closed subspace A of X is called a Z-set in X, whenever for every open cover  $\mathcal{U}$  of X and for every  $f \in C(Q, X)$ , there is  $g \in C(Q, X)$   $\mathcal{U}$ -close to f and  $g(Q) \cap A = \emptyset$ . For compact spaces, this definition coincides with the one given in section 3.1. For an ANR X, we have by [39, Th. 7.2.5]:

A closed subset A of X is a Z-set in X iff for every open cover  $\mathcal{U}$  of X there exists  $f \in C(X, X)$  such that f and  $1_X$  are  $\mathcal{U}$ -close and  $f(X) \cap A = \emptyset$ .

#### §3.4. Remarks

For an ANR X, a closed subset A of X is said to be a *strong Z-set* if for every open cover  $\mathcal{U}$  of X there exists a continuous function  $f: X \to X$  such that f and  $1_X$  are  $\mathcal{U}$ -close and  $\overline{f(X)} \cap A = \emptyset$ . A Z-set need not to be a strong Z-set (an example is given in [12]).

A space X is strongly  $\mathcal{F}_{\sigma\delta}$ -universal if for every  $f \in C(A, X)$ , where A is an absolute  $\mathcal{F}_{\sigma\delta}$ , for every  $B \subset A$  closed such that  $f \mid B : B \to X$  is a Z-embedding, and for every open cover  $\mathcal{U}$  of X, there exists a Z-embedding  $h: A \to X$  such that  $h \mid B = f \mid B$  and f and h are  $\mathcal{U}$ -close.

We can now state the announced theorem of Bestvina and Mogilski,

**3.4.2 THEOREM** ([13]): An AR X which is an absolute  $F_{\sigma\delta}$  is homeomorphic to  $\sigma_{\omega}$  iff the following conditions are satisfied

(a) X is strongly  $F_{\sigma\delta}$ -universal, and

(b)  $X = \bigcup_{n=1}^{\infty} X_n$ , where each  $X_n$  is a strong Z-set in X.

In [16] it is proved that  $C_{p,0}([1, \omega])$  satisfies the conditions of theorem 3.4.2 and hence  $C_{p,0}([1, \omega])$  is homeomorphic to  $\sigma_{\omega}$ . So again we have that for non-discrete countable spaces X,  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ .

Let  $X = \{x_0, x_1, x_2, \dots\}$  be a countable space which is not discrete at  $x_0$ . Now that we have that  $C_p(X)$  and  $C_p^*(X)$  are homeomorphic to  $\sigma_{\omega}$ , the question remains whether theorem 3.2.13 (b) also holds for this X. That is if Y is another non-discrete countable space, is it true then that there is  $h \in \mathcal{H}(Q)$  arbitrary close to the identity which maps  $C_{p,0}(X)$  onto  $C_{p,0}(Y)$ . For this purpose we actually would like to write  $C_{p,0}(X)$  as the kernel of a Q-matrix. The Q-matrix in section 3.2 essentially uses the non-locally compactness of X, and as far as we see it cannot be used for non-discrete countable spaces. A weaker question is whether  $C_{p,0}([1, \omega])$  can be written as the kernel of a Q-matrix. As with the test space T, there is a natural candidate. However this candidate unfortunately is not a Q-matrix. We shall present it and prove that it is not a Q-matrix.

We identify  $C_{p,0}([1, \omega])$  with the following subspace of Q,

$$\{(x_n)_{n\in\mathbb{N}}\in S: \lim_{n\to\infty}x_n=0\}.$$

For convenience for every  $n, m \in \mathbb{N}$ , let

$$I_m = [-1 + \frac{1}{m}, 1 - \frac{1}{m}]$$
 and  $I_{mn} = [-(1 - \frac{1}{m})2^{-n}, (1 - \frac{1}{m})2^{-n}].$ 

For every  $n, m \in \mathbb{N}$ , let

(1) 
$$A_1^n = \emptyset$$
, and

(2) 
$$A_m^n = (\mathbf{I}_m)^m \times \mathbf{I}_{mn} \times \mathbf{I}_{mn} \times \mathbf{I}_{mn} \times \cdots$$
, if  $m \ge 2$ .

Let  $\mathcal{A} = \{A_m^n : n, m \in \mathbb{N}\}$ . By lemma 3.1.4 it is easily seen that  $\mathcal{A}$  is a  $\mathcal{I}$ -matrix in Q. Again by lemma 3.1.4 we have for every  $n, m \in \mathbb{N}$  that  $A_m^n \in \mathcal{I}(A_{m+1}^n)$ . Since each  $A_m^n$  is convex and infinite-dimensional and since  $\bigcup_{m \in \mathbb{N}} A_m^n$  is dense in Q, we have by theorem 3.1.7, that  $\{A_m^n\}_{m>1}$  is a skeleton in Q.

Now fix  $n_1 < \cdots < n_m \in \mathbb{N}$  and  $i_1, \ldots, i_m \in \mathbb{N} \setminus \{1\}$ . By the observation in section 3.1 we assume  $i_1 < \cdots < i_m$ . Then it is easily seen that

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} = B_1 \times \cdots \times B_{i_m} \times I_{i_m n_m} \times I_{i_m n_m} \times I_{i_m n_m} \times \cdots$$

where each  $B_i$  is a non-degenerate closed subinterval of  $I_m$ . So  $\bigcap_{k=1}^m A_{i_k}^{n_k}$  is a Keller space, hence a Hilbert cube. Furthermore for  $p \in \mathbb{N}$  and  $i \ge i_m$ ,

$$\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} = B_1 \times \cdots \times B_{i_m} \times (I_{i_m n_m})^{i_{-i_m}} \times I_{i_{n_m+p}} \times I_{i_{n_m+p}} \times I_{i_{n_m+p}} \times \cdots$$

By lemma 3.1.4 we have  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{Z}(\bigcap_{k=1}^{m} A_{i_k}^{n_k})$  and  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p} \in \mathcal{Z}(\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_{i+1}^{n_m+p})$ . Since  $\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}$  is convex and infinite-dimensional, and  $\bigcup_{i=1}^{\infty} (\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p})$  is dense in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$ , we have by theorem 3.1.7 that  $\{\bigcap_{k=1}^{m} A_{i_k}^{n_k} \cap A_i^{n_m+p}\}$  is a skeleton in  $\bigcap_{k=1}^{m} A_{i_k}^{n_k}$ .

It seems that we are on the right way to prove that  $\mathcal{A}$  is a *Q*-matrix. Unfortunately condition (d) in the definition of a *Q*-matrix is not satisfied. Indeed for

$$A_{2}^{1} = \left[-\frac{1}{2}, \frac{1}{2}\right] \times \left[-\frac{1}{2}, \frac{1}{2}\right] \times \left[-\frac{1}{4}, \frac{1}{4}\right] \times \left[-\frac{1}{4}, \frac{1}{4}\right] \times \left[-\frac{1}{4}, \frac{1}{4}\right] \times \cdots$$

and

$$A_3^2 = \left[-\frac{1}{3}, \frac{1}{3}\right] \times \left[-\frac{1}{3}, \frac{1}{3}\right] \times \left[-\frac{1}{3}, \frac{1}{3}\right] \times \left[-\frac{1}{6}, \frac{1}{6}\right] \times \left[-\frac{1}{6}, \frac{1}{6}\right] \times \cdots$$

we have

$$A_2^1 \cap A_3^2 = [-\frac{1}{3}, \frac{1}{3}] \times [-\frac{1}{3}, \frac{1}{3}] \times [-\frac{1}{4}, \frac{1}{4}] \times [-\frac{1}{6}, \frac{1}{6}] \times [-\frac{1}{6}, \frac{1}{6}] \times \cdots$$

Furthermore  $A_3^2 \not\subset A_2^1$ . However by lemma 3.1.3 (b),  $A_2^1 \cap A_3^2 \notin \mathcal{Z}(A_3^2)$ . So this natural description of  $C_{p,0}([1, \omega])$  fails to be a *Q*-matrix, hence the question remains open whether  $C_{p,0}([1, \omega])$  can be described as the kernel of a *Q*-matrix.

Our last remark in this section concerns uniform spaces. At this moment there ex-

#### §3.4. Remarks

ists a topological classification of function spaces  $C_p(X)$ , for countable non-discrete spaces X and an isomorphical classification of function spaces  $C_p(X)$ , for countable infinite compact spaces X (cf. chapter 2). We can also consider  $C_p(X)$  as a uniform space. The uniformity on  $C_p(X)$  is given by the subbase  $\{U(P, \varepsilon): P \subset X \text{ finite}, \varepsilon > 0\}$ , where

$$U(P, \varepsilon) = \{ (f, g) \in C_p(X) \times C_p(X) : |f(x) - g(x)| < \varepsilon \text{ for every } x \in P \}.$$

Since every linear homeomorphism is a uniform homeomorphism and every uniform homeomorphism is a homeomorphism, it is interesting to find a uniform classification of the function spaces  $C_p(X)$ , for countable infinite compact spaces X. In [27], Gulko derived the following

**3.4.3 THEOREM** ([27]): Let X be a countable infinite compact space. Then  $C_p(X)$  is uniformly homeomorphic to  $C_p([1, \omega])$ .

So for countable infinite compact spaces the topological and uniform classification coincide. As a corollary we also have that there are spaces X and Y such that  $C_p(X)$  and  $C_p(Y)$  are uniformly homeomorphic but not linearly homeomorphic. In [27] Gulko announces a complete uniform classification of  $C_p(X)$  for all countable metric spaces.

·

## **CHAPTER 4**

# On the $l_p$ -equivalence of metric spaces

#### All spaces considered in this chapter are Tychonov.

In chapter 3 we stated a topological classification result for the spaces  $C_p(X)$ , where X is any countable metric space. In the light of this result the question naturally arises which of these function spaces are in fact linearly homeomorphic, i.e., isomorphic as linear spaces. In chapter 2, we already obtained an isomorphical classification of the spaces  $C_p(X)$ , where X is any locally compact zero-dimensional separable metric space. In this chapter we also consider non-locally compact zero-dimensional separable metric spaces.

In section 4.1, we introduce the notion of  $l_p$ -equivalent pairs, which is a useful tool in deriving topological properties of metric spaces which are preserved by  $l_p$ equivalence. In section 4.2 we show that the topological properties preserved by  $l_p$ equivalence, found in section 4.1, are sufficient to give an isomorphical classification of the function spaces  $C_p(X)$ , where X is any countable metric space with scattered height less than or equal to  $\omega$ . Unfortunately these properties are not sufficient to give a complete isomorphical classification for the class of all countable metric spaces X. The results in sections 4.1 and 4.2 can be found in [4]. In section 4.3, we present other topological properties preserved by  $l_p$ -equivalence. In section 4.4, we state a conjecture on a complete isomorphical classification for the function spaces considered. Some remarks are made concerning the difficulties one encounters when one attempts to prove the conjecture. Finally, some partial results are given on  $l_0$ -equivalence (section 4.5) and  $l_p^*$ -equivalence (section 4.6).

## §4.1. $l_p$ -equivalent properties of metric spaces

In this section we present  $l_p$ -equivalent properties of metric spaces. The notion of

 $l_p$ -equivalent pairs provides us with these properties. Before we give the definition of  $l_p$ -equivalent pairs, we first need the following:

Let X be a space and  $X_0 \subset X$ . For every ordinal we define the set  $X^{\{\alpha\}}$  with respect to the pair  $(X, X_0)$  by transfinite induction as follows:

- (1)  $X^{\{0\}} = X_0$ .
- (2) If  $\alpha$  is a successor, say  $\alpha = \beta + 1$ , then  $x \in X^{\{\alpha\}}$  if and only if for every neighborhood U of x,  $\overline{U} \cap X^{\{\beta\}}$  is not compact.
- (3) If  $\alpha$  is a limit ordinal, then  $X^{\{\alpha\}} = \bigcap_{\beta < \alpha} X^{\{\beta\}}$ .

The construction of the sets  $X^{\{\alpha\}}$  is a special case of a construction in [49]. Note that, whereas taking the derivative of a space means "throwing away all isolated points", the above procedure throws away all points with a compact neighborhood. There are also some similarities between both operations which are formulated in the following two lemmas. They will be used frequently but will not always be mentioned. For a subset U of X, we define as above for every ordinal  $\alpha$ , the set  $U^{\{\alpha\}}$  with respect to the pair  $(U, U_0)$ , where  $U_0 = U \cap X_0$ . Compare the following two lemmas with proposition 2.2.2, corollary 2.2.3 and proposition 2.2.4.

**4.1.1 LEMMA:** Let X be a space and  $X_0$  a closed subspace of X. Then for every ordinal  $\alpha$ ,

(a)  $X^{\{\alpha\}}$  is closed in X, (b)  $X^{\{\alpha\}} \subset X^{(\alpha)}$ , (c) for  $\beta < \alpha, X^{\{\alpha\}} \subset X^{\{\beta\}}$ , and (d)  $X^{\{\alpha+1\}} = (X^{\{\alpha\}})^{\{1\}}$ .

**PROOF:** We prove each case by transfinite induction on  $\alpha$ .

For (a), the case  $\alpha = 0$  is a triviality. First suppose that  $\alpha > 0$  is a successor, say  $\alpha = \beta + 1$ . Let  $x \in X \setminus X^{\{\alpha\}}$ . Then there is an open neighborhood U of x such that  $\overline{U} \cap X^{\{\beta\}}$  is compact. So  $U \cap X^{\{\alpha\}} = \emptyset$ , hence  $X^{\{\alpha\}}$  is closed. Secondly, if  $\alpha$  is a limit ordinal, then  $X^{\{\alpha\}} = \bigcap_{\beta < \alpha} X^{\{\beta\}}$ , so by our inductive hypothesis,  $X^{\{\alpha\}}$  is closed in X. For (b), the case  $\alpha = 0$  is a triviality. If  $\alpha > 0$  is a successor, say  $\alpha = \beta + 1$ , then for no neighborhood U of a point  $x \in X^{\{\alpha\}}$ ,  $\overline{U} \cap X^{\{\beta\}}$  is compact. By the inductive hypothesis we have  $\overline{U} \cap X^{\{\beta\}}$  is not compact, hence  $\overline{U} \cap X^{\{\beta\}} \setminus \{x\} \neq \emptyset$ . We conclude that  $x \in X^{\{\alpha\}}$ . For  $\alpha$  a limit ordinal, part (b) is clear.

For (c), first let  $\alpha = 1$ . If  $x \notin X_0$ , there is a neighborhood U of x such that  $\overline{U} \cap X_0 = \emptyset$ , hence  $x \notin X^{\{1\}}$ . So  $X^{\{1\}} \subset X^{\{0\}}$ . If  $\alpha > 1$  is a successor, say  $\alpha = \gamma + 1$ , then for every

 $x \in X^{\{\alpha\}}$  and for every neighborhood U of x,  $\overline{U} \cap X^{\{\gamma\}}$  is not compact. Hence  $\overline{U} \cap X^{\{\beta\}}$  is not compact and so  $x \in X^{\{\beta\}}$ . For  $\alpha$  a limit ordinal part (c) is a triviality.

For (d), let  $x \in X^{\{\alpha+1\}}$ . Then for each neighborhood U of x,  $\overline{U} \cap X^{\{\alpha\}}$  is not compact if and only if  $\overline{U} \cap X^{\{\alpha\}} \cap X_0$  is not compact if and only if  $x \in (X^{\{\alpha\}})^{\{1\}}$ .

One can now easily see that if  $X_0$  and  $Y_0$  are closed in X and  $Y_0 \subset X_0$ , then for each ordinal  $\alpha$  the set  $X^{\{\alpha\}}$  with respect to the pair  $(X, Y_0)$  is a subset of the set  $X^{\{\alpha\}}$  with respect to the pair  $(X, X_0)$ .

**4.1.2 LEMMA:** Let X be a space and U a subset of X, and  $X_0$  a closed subset of X. Then for each ordinal  $\alpha$ ,

(a) if U is closed, then  $U^{\{\alpha\}} \subset U \cap X^{\{\alpha\}}$ , and (b) if U is open, then  $U \cap X^{\{\alpha\}} \subset U^{\{\alpha\}}$ .

**PROOF:** We prove this proposition by transfinite induction on  $\alpha$ . If  $\alpha = 0$ , the lemma is obviously true, so suppose that  $\alpha > 0$  and that for each  $\beta < \alpha$  the lemma has been proved. First suppose that  $\alpha$  is a successor, say  $\alpha = \beta + 1$ .

For (a), suppose U is closed, let  $x \in U^{\{\alpha\}}$  and let V be a neighborhood of x. Then by the inductive hypothesis

$$\overline{V} \cap U^{\{\beta\}} \subset \overline{V} \cap (U \cap X^{\{\beta\}}) \subset \overline{V} \cap X^{\{\beta\}}.$$

By lemma 4.1.1,  $\overline{V} \cap U^{\{\beta\}}$  is a closed subset of U and because U is closed in X,  $\overline{V} \cap U^{\{\beta\}}$  is a closed subset of  $\overline{V} \cap X^{\{\beta\}}$ . Since  $\overline{V} \cap U^{\{\beta\}}$  is not compact, we then have that  $\overline{V} \cap X^{\{\beta\}}$  is not compact. So  $x \in X^{\{\alpha\}} \cap U$ .

For (b), suppose U is open and let  $x \in U \cap X^{\{\alpha\}}$ . Let V be a neighborhood of x in X such that  $\overline{V} \subset U$ . Then  $\overline{V} \cap X^{\{\beta\}}$  is not compact. So by the inductive hypothesis

$$\overline{V} \cap X^{\{\beta\}} = \overline{V} \cap U \cap X^{\{\beta\}} \subset \overline{V} \cap U^{\{\beta\}}$$

hence  $\overline{V} \cap U^{\{\beta\}}$  is not compact. We conclude that  $x \in U^{\{\alpha\}}$ .

If  $\alpha$  is a limit ordinal, then by the inductive hypothesis we have for closed U that

$$U^{\{\alpha\}} = \bigcap_{\beta < \alpha} U^{\{\beta\}} \subset \bigcap_{\beta < \alpha} (U \cap X^{\{\beta\}}) = U \cap \bigcap_{\beta < \alpha} X^{\{\beta\}} = U \cap X^{\{\alpha\}},$$

and for open U

$$U \cap X^{\{\alpha\}} = U \cap \bigcap_{\beta < \alpha} X^{\{\beta\}} = \bigcap_{\beta < \alpha} (U \cap X^{\{\beta\}}) \subset \bigcap_{\beta < \alpha} U^{\{\beta\}} = U^{\{\alpha\}}.$$

This completes the proof of the lemma.

This lemma implies that whenever U is a clopen subset of a space X, then for each ordinal  $\alpha$ ,  $U \cap X^{\{\alpha\}} = U^{\{\alpha\}}$ ; furthermore by proposition 2.2.4,  $U^{(\alpha)} = U \cap X^{(\alpha)}$ . We will use this frequently without explicit reference.

**4.1.3 LEMMA:** Let X be a paracompact space,  $X_0$  closed in X and  $\alpha \ge 1$  an ordinal. Let  $V \subset X$  be open such that  $\overline{V} \cap X^{\{\alpha\}} = \emptyset$ . Then there is a locally finite family  $\{V_s : s \in S\}$  consisting of open sets such that  $V = \bigcup_{s \in S} V_s$  and for every  $s \in S$ , there is  $\beta < \alpha$  with  $\overline{V}_s \cap X^{\{\beta\}}$  compact.

**PROOF:** Case 1:  $\alpha$  is a successor, say  $\alpha = \beta + 1$ .

Since  $\overline{V} \cap X^{\{\alpha\}} = \emptyset$ , for every  $x \in \overline{V}$ , there is a neighborhood  $U_x$  of x such that  $\overline{U}_x \cap X^{\{\beta\}}$  is compact. Since  $\{U_x : x \in \overline{V}\} \cup \{X \setminus \overline{V}\}$  is an open cover of X, there is a locally finite open refinement  $\{O_s : s \in S\}$  of it. For every  $s \in S$ , let  $V_s = O_s \cap V$ . Then  $\{V_s : s \in S\}$  is a locally finite family consisting of open sets such that  $V = \bigcup_{s \in S} V_s$ . In addition, if  $s \in S$  and  $V_s \neq \emptyset$  there is  $x \in \overline{V}$  with  $V_s \subset U_x$ . Then  $\overline{V}_s \cap X^{\{\beta\}} \subset \overline{U}_x \cap X^{\{\beta\}}$ . So  $\overline{V}_s \cap X^{\{\beta\}}$  is compact.

Case 2:  $\alpha$  is a limit ordinal.

Then  $\mathcal{U} = \{X \setminus X^{\{\beta\}} : \beta < \alpha\} \cup \{X \setminus \overline{V}\}$  is an open cover of X, so there is a locally finite open covering  $\{O_s : s \in S\}$  of X such that  $\{\overline{O}_s : s \in S\}$  refines  $\mathcal{U}$ . For every  $s \in S$ put  $V_s = V \cap O_s$ . Then  $\{V_s : s \in S\}$  is a locally finite family of open sets such that  $V = \bigcup_{s \in S} V_s$ . Now fix  $s \in S$  and suppose  $V_s \neq \emptyset$ . Then there is  $\beta < \alpha$  such that  $\overline{V}_s \subset X \setminus X^{\{\beta\}}$ , which implies  $\overline{V}_s \cap X^{\{\beta\}} = \emptyset$ .  $\Box$ 

Let X be a space. There are several possibilities to combine the two operations  $X^{(\alpha)}$ and  $X^{\{\alpha\}}$ . The one that is important for our purposes is the case where  $X_0 = X^{(\alpha)}$  for some ordinal  $\alpha$ . In the sequel the set  $X^{\{\beta\}}$  with respect to  $X_0 = X^{(\alpha)}$  will be denoted by  $X^{(\alpha,\beta)}$ . Another subset of X we need in section 4.3 is  $X^{\ll\alpha,\beta\gg}$  defined for limit ordinals  $\alpha$  by

$$X^{\boldsymbol{\ll}\alpha,\ \boldsymbol{\beta}\boldsymbol{\gg}} = \bigcap_{\boldsymbol{\gamma}<\boldsymbol{\alpha}} X^{(\boldsymbol{\gamma},\ \boldsymbol{\beta})}.$$

Note that the sets  $X^{(\alpha,\beta)}$  and  $X^{\ll\alpha,\beta\gg}$  are closed in X, and if  $\beta=0$ , then  $X^{\ll\alpha,\beta\gg}=X^{(\alpha)}$ . As in lemma 4.1.2 one can prove that for U clopen in X,  $U \cap X^{(\alpha,\beta)}=U^{(\alpha,\beta)}$ , and  $U \cap X^{\ll\alpha,\beta\gg}=U^{\ll\alpha,\beta\gg}$ .

In this chapter it will be made clear that the isomorphical classification of  $C_p(X)$  for countable metric X depends upon the behaviour of X with respect to the above operations and that neither of the operations is redundant. We need the following lemma in

§4.1. lp-equivalent properties of metric spaces

this section and also in section 4.3.

**4.1.4 LEMMA:** Let X be a zero-dimensional separable metric space. Let  $\alpha \ge 1$  and  $\beta$  be ordinals and let V be an open subset of X. Suppose that

(a)  $\overline{V} \cap X^{(\alpha)} = \emptyset$ , or

(b)  $\overline{V} \cap X^{(\beta,\alpha)} = \emptyset$ , or

(c)  $\alpha$  is a limit ordinal and  $\overline{V} \cap X^{\ll \alpha, \beta \gg} = \emptyset$ .

Then there is a discrete clopen family  $\{A_i : i \in \mathbb{N}\}$  such that  $V \subset \bigcup_{i \in \mathbb{N}} A_i$  and for each

 $i \in \mathbb{N}$ , there is  $\gamma < \alpha$  such that

if (a) holds, then  $A_i^{(\gamma)}$  is finite and if moreover  $\alpha$  is a limit, then  $A_i^{(\gamma)} = \emptyset$ ; if (b) holds, then  $A_i^{(\beta,\gamma)}$  is compact and if moreover  $\alpha$  is a limit, then  $A_i^{(\beta,\gamma)} = \emptyset$ ; if (c) holds, then  $A_i^{(\gamma,\beta)} = \emptyset$ .

**PROOF:** The proof is almost the same as the proof of lemma 4.1.3, hence we will be brief and present a proof of case (a) only.

If  $\alpha = \gamma + 1$  is a successor, there is for each  $x \in \overline{V}$  a clopen neighborhood  $U_x$  of x such that  $U_x^{(\gamma)}$  is finite. The open cover  $\{U_x : x \in \overline{V}\} \cup X \setminus \overline{V}$  of X has a clopen disjoint refinement  $\{A_i : i \in \mathbb{N}\}$ . Put  $I = \{i \in \mathbb{N} : A_i \cap V \neq \emptyset\}$ . Then  $\{A_i : i \in I\}$  is a discrete clopen family which is as required.

If  $\alpha$  is a limit ordinal, then  $\mathcal{U} = \{X \setminus X^{(\beta)} : \beta < \alpha\} \cup \{X \setminus \overline{V}\}$  is an open cover of X. Let  $\{A_i : i \in \mathbb{N}\}$  be a disjoint clopen refinement of  $\mathcal{U}$ , and put  $I = \{i \in \mathbb{N} : A_i \cap V \neq \emptyset\}$ . Then  $\{A_i : i \in I\}$  is a discrete clopen family which satisfies the desired conditions.  $\Box$ 

**4.1.5 COROLLARY:** Let X be a zero-dimensional separable metric space. Let  $\alpha \ge 1$  and  $\beta$  be ordinals and let V be an clopen subset of X. Suppose that

(a)  $V^{(\alpha)} = \emptyset$ , or

(b)  $V^{(\beta,\alpha)} = \emptyset$ , or

(c)  $\alpha$  is a limit ordinal and  $V^{\ll \alpha, \beta \gg} = \emptyset$ .

Then there is a discrete clopen family  $\{A_i : i \in \mathbb{N}\}$  such that  $V = \bigcup_{i \in \mathbb{N}} A_i$  and for each  $i \in \mathbb{N}$ , there is  $\gamma < \alpha$  such that

if (a) holds, then  $A_i^{(\gamma)}$  is finite and if moreover  $\alpha$  is a limit, then  $A_i^{(\gamma)} = \emptyset$ ;

if (b) holds, then  $A_i^{(\beta,\gamma)}$  is compact and if moreover  $\alpha$  is a limit, then  $A_i^{(\beta,\gamma)} = \emptyset$ ; if (c) holds, then  $A_i^{(\gamma,\beta)} = \emptyset$ .

We now define some additional notions. Let X and Y be spaces. Let  $X_0$  be closed in X and  $Y_0$  be closed in Y. Let  $\phi: C_p(X) \to C_p(Y)$  be a linear bijection and  $\alpha$  an ordinal. We define the pair  $(X, X_0)$  to be  $(\phi, \alpha)$ -relative to the pair  $(Y, Y_0)$  if the following

holds:

If U and V are open in X and W is open in Y such that  $(\sup D) \cap W = \emptyset$  and  $\sup D W \subset U \cup V$ , then  $W \cap Y^{\{\alpha\}} \neq \emptyset$  implies  $\overline{V} \cap X^{\{\alpha\}} \neq \emptyset$ .

We define  $(X, X_0)$  and  $(Y, Y_0)$  to be  $l_p$ -equivalent pairs if there is a linear homeomorphism  $\phi: C_p(X) \to C_p(Y)$  such that  $(X, X_0)$  is  $(\phi, 0)$ -relative to  $(Y, Y_0)$  and  $(Y, Y_0)$  is  $(\phi^{-1}, 0)$ -relative to  $(X, X_0)$ . Note that two spaces X and Y are  $l_p$ -equivalent if and only if  $(X, \emptyset)$  and  $(Y, \emptyset)$  are  $l_p$ -equivalent pairs.

The importance of  $l_p$ -equivalent pairs will become clear in proposition 4.1.9 and proposition 4.1.12.

**4.1.6 LEMMA:** Let X and Y be metric spaces,  $X_0$  closed in X and  $Y_0$  closed in Y. Let  $\phi: C_p(X) \rightarrow C_p(Y)$  be a continuous linear bijection such that  $(X, X_0)$  is  $(\phi, 0)$ -relative to  $(Y, Y_0)$ . Then for every ordinal  $\alpha$ ,  $(X, X_0)$  is  $(\phi, \alpha)$ -relative to  $(Y, Y_0)$ .

**PROOF:** We prove the lemma by transfinite induction on  $\alpha$ . Since  $(X, X_0)$  is  $(\phi, 0)$ -relative to  $(Y, Y_0)$ , the case  $\alpha = 0$  is clear. So assume the lemma to be true for every ordinal  $\beta < \alpha$  with  $\alpha \ge 1$ . Suppose that the lemma is false for  $\alpha$ . Then there are U and V open in X and W open in Y such that  $(\operatorname{supp} U) \cap W = \emptyset$ ,  $\operatorname{supp} W \subset U \cup V$ ,  $W \cap Y^{\{\alpha\}} \neq \emptyset$  and  $\overline{V} \cap X^{\{\alpha\}} = \emptyset$ . By lemma 4.1.3, there is a locally finite family  $\{V_s : s \in S\}$  consisting of open sets such that  $V = \bigcup_{s \in S} V_s$  and for every  $s \in S$  there is  $\beta < \alpha$  such that  $\overline{V}_s \cap X^{\{\beta\}}$  is compact. Choose  $y \in W \cap Y^{\{\alpha\}}$  and a neighborhoodbase  $\{W_m : m \in \mathbb{N}\}$  at y in W such that for every  $m \in \mathbb{N}$ ,  $\overline{W}_{m+1} \subset W_m$ . By corollary 1.2.15 (a) and lemma 1.2.10, there are  $m \in \mathbb{N}$  and  $s_1, \ldots, s_m \in S$  with

$$\operatorname{supp} W_m \cap \bigcup_{s \notin \{s_1, \dots, s_m\}} V_s = \emptyset.$$
<sup>(1)</sup>

Now let  $A = \bigcup_{i=1}^{m} V_{s_i}$ . Fix  $\beta < \alpha$  such that  $\overline{A} \cap X^{\{\beta\}}$  is compact. Also, notice the following: A and U are open in X,  $W_m$  is open in Y,  $(\sup p U) \cap W_m = \emptyset$  (because  $W_m \subset W$  and  $(\sup p U) \cap W = \emptyset$ ) and  $\sup p W_m \subset U \cup A$  (by (1) and the fact that  $\sup p W \subset U \cup V$ ). Since  $y \in W_m \cap Y^{\{\beta\}}$ , our inductive hypothesis implies that  $\overline{A} \cap X^{\{\beta\}} \neq \emptyset$ . We have that X is a metric space, so there is an open neighborhoodbase  $\{A_s : s \in \mathbb{N}\}$  at  $\overline{A} \cap X^{\{\beta\}}$  in X such that  $\overline{A}_{s+1} \subset A_s$  for every  $s \in \mathbb{N}$ . Since  $y \in Y^{\{\alpha\}}$  and  $\overline{W}_{m+1}$  is a neighborhood of y,  $\overline{W}_{m+1} \cap Y^{\{\beta\}}$  is not compact, so in Y there is a closed discrete subset  $\{y_s : s \in \mathbb{N}\}$  contained in  $W_m \cap Y^{\{\beta\}}$ . Let  $\{O_s : s \in \mathbb{N}\}$  be an open discrete family in  $W_m$  such that  $y_s \in O_s$ . Then by corollary 1.2.15 (a) and lemma 1.2.10,

§4.1. lp-equivalent properties of metric spaces

there is  $s \in \mathbb{N}$  with

$$\operatorname{supp} A_s \cap \bigcup_{i \ge 0} O_i = \emptyset. \tag{2}$$

Now put  $U' = U \cup A_s$ ,  $V' = A \setminus \overline{A}_{s+1}$  and  $W' = O_s$ . Then U' and V' are open in X and W' is open in Y. We also have

$$(\operatorname{supp} U') \cap W' = (\operatorname{supp} U \cup \operatorname{supp} A_s) \cap O_s = \emptyset$$
 (by 2)

and

$$\operatorname{supp} W' \subset \operatorname{supp} W_m \subset U \cup A \subset U' \cup V'.$$

Furthermore,  $y_s \in W' \cap Y^{\{\beta\}}$  and

$$\overline{V'} \cap X^{\{\beta\}} = (\overline{A \setminus \overline{A_{s+1}}}) \cap X^{\{\beta\}} \subset (\overline{A} \setminus A_{s+1}) \cap X^{\{\beta\}} = \emptyset.$$

This contradicts our inductive assumption.

**4.1.7 THEOREM:** Let X and Y be metric spaces,  $X_0$  closed in X and  $Y_0$  closed in Y. Suppose that  $(X, X_0)$  and  $(Y, Y_0)$  are  $l_p$ -equivalent pairs. Then for every ordinal  $\alpha$  we have

- (a)  $X^{\{\alpha\}} = \emptyset$  if and only if  $Y^{\{\alpha\}} = \emptyset$ , (b)  $X^{\{\alpha\}}$  is compact if and only if  $Y^{\{\alpha\}}$  is compact, and
- (c)  $X^{\{\alpha\}}$  is locally compact if and only if  $Y^{\{\alpha\}}$  is locally compact.

**PROOF:** Let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism such that  $(X, X_0)$  is  $(\phi, 0)$ -relative to  $(Y, Y_0)$  and  $(Y, Y_0)$  is  $(\phi^{-1}, 0)$ -relative to  $(X, X_0)$ . For (a), by applying lemma 4.1.6 and the definition of  $(\phi, \alpha)$ -relativeness to  $U = \emptyset$ , V = X and W = Y, we get  $X^{\{\alpha\}} = \emptyset$  if  $Y^{\{\alpha\}} = \emptyset$ .

For (b) suppose that  $Y^{\{\alpha\}}$  is compact and  $X^{\{\alpha\}}$  is not. Since  $X^{\{\alpha\}} \neq \emptyset$ , by (a) we have  $Y^{\{\alpha\}} \neq \emptyset$ . Let  $\{W_m : m \in \mathbb{N}\}$  be an open decreasing base in Y at  $Y^{\{\alpha\}}$  such that for every  $m \in \mathbb{N}$ ,  $\overline{W}_{m+1} \subset W_m$ . Furthermore, let  $\{x_m : m \in \mathbb{N}\}$  be closed and discrete in  $X^{\{\alpha\}}$ . Let  $\{O_m : m \in \mathbb{N}\}$  be an open discrete family in X such that  $x_m \in O_m$ . Then by corollary 1.2.15 (a) and lemma 1.2.10, there is  $m \in \mathbb{N}$  such that

 $\operatorname{supp} W_m \cap \bigcup_{i \ge m} O_i = \emptyset.$ 

Now let  $U = W_m$ ,  $V = Y \setminus \overline{W}_{m+1}$  and  $W = O_m$ . Then U and V are open, W is open, (supp U)  $\cap W = \emptyset$  and supp  $W \subset Y = U \cup V$ . In addition

$$\overline{V} \cap Y^{\{\alpha\}} = Y \setminus \overline{W}_{m+1} \cap Y^{\{\alpha\}} = \emptyset$$

On the  $l_p$ -equivalence of metric spaces

and

$$W \cap X^{\{\alpha\}} = O_m \cap X^{\{\alpha\}} \neq \emptyset.$$

This contradicts lemma 4.1.6.

For (c) notice that  $X^{\{\alpha\}}$  is locally compact if and only if  $X^{\{\alpha+1\}} = \emptyset$ . So (c) follows directly from (a).

Theorem 4.1.7 is a useful theorem. In the remaining part of this section we give some applications of it. We will prove for  $l_p$ -equivalent spaces X and Y that if  $\alpha$  is a finite prime component, then  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are  $l_p$ -equivalent pairs (proposition 4.1.9), and if  $\alpha$  is an infinite countable prime component and X and Y are zerodimensional separable metric spaces, then  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are  $l_p$ -equivalent pairs (proposition 4.1.12). We will distinguish between the cases of finite and infinite prime components. Although the result for finite prime components is much stronger than the result for countable infinite prime components, the latter case requires most of the work. We first need the following

**4.1.8 LEMMA:** Let X be a first countable space and  $\alpha < \omega_1$  an ordinal such that  $X^{(\alpha)} \neq \emptyset$ . Then there is  $K \subset X$  such that  $K \approx [1, \omega^{\alpha}]$ .

**PROOF:** We prove the lemma by transfinite induction on  $\alpha$ . For  $\alpha = 0$ , it is a triviality. Now suppose the lemma is true for every ordinal  $\beta < \alpha$ , with  $\alpha \ge 1$ . Fix  $x \in X^{(\alpha)}$ . *Case 1*:  $\alpha$  is a successor, say  $\alpha = \beta + 1$ .

Choose a sequence  $(x_n)_{n \in \mathbb{N}}$  in  $X^{(\beta)}$  such that  $x_n \to x$ , and a decreasing open base  $\{U_n : n \in \mathbb{N}\}$  at x such that for each  $n \in \mathbb{N}$ ,  $x_n \in V_n = U_n \setminus \overline{U}_{n+1}$ . Notice that  $V_n$  is open, so  $V_n^{(\beta)} = V_n \cap X^{(\beta)}$ . Hence,  $x_n \in V_n^{(\beta)}$ . So by the inductive hypothesis, there are  $K_n \subset V_n$  such that  $K_n \approx [1, \omega^{\beta}]$ . Notice that for every  $n \neq m$ ,  $K_n \cap K_m = \emptyset$ . Let  $K = \bigcup_{n=1}^{\infty} K_n \cup \{x\}$ . Then by theorem 2.2.8,  $K \approx [1, \omega^{\alpha}]$ . *Case 2*:  $\alpha$  is a limit ordinal.

Let  $(\beta_n)_n$  be an increasing sequence converging to  $\alpha$ . Since  $x \in X^{(\alpha)}$ , there is a decreasing open base  $\{U_n : n \in \mathbb{N}\}$  at x such that if  $V_n = U_n \setminus \overline{U}_{n+1}$ , then  $V_n^{(\beta_n)} \neq \emptyset$ . By the inductive hypothesis there are  $K_n \subset V_n$  such that  $K_n \approx [1, \omega^{\beta_n}]$ . Then by theorem 2.2.8,  $K = \bigcup_{n=1}^{\infty} K_n \cup \{x\}$  is as required.  $\Box$ 

In [9], J.W. Baker gives conditions for a space to have an ordinal interval as a subspace.

142
§4.1.  $l_p$ -equivalent properties of metric spaces

We first deal with finite prime components, i.e., the numbers 0 and 1.

**4.1.9 PROPOSITION:** Let  $\alpha \in \{0, 1\}$  and let X and Y be  $l_p$ -equivalent metric spaces. Then  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are  $l_p$ -equivalent pairs.

**PROOF:** By proposition 2.2.2 (a),  $X^{(\alpha)}$  is closed in X. Let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism. It suffices to prove that  $(X, X^{(\alpha)})$  is  $(\phi, 0)$ -relative to  $(Y, Y^{(\alpha)})$ . To this end let U and V be open disjoint in X and W open in Y such that

```
(\operatorname{supp} U) \cap W = \emptyset, and
\operatorname{supp} W \subset U \cup V.
```

Suppose that  $W \cap Y^{(\alpha)} \neq \emptyset$  and  $\overline{V} \cap X^{(\alpha)} = \emptyset$ .

Case 1:  $\alpha = 0$ .

Since  $\overline{V} = \emptyset$ , we have supp  $W \subset U$ . So by proposition 1.4.3,

 $W \subset \text{supp supp } W \subset \text{supp } U$ .

Since  $(\sup D) \cap W = \emptyset$  this gives  $W = \emptyset$ , hence we arrived at a contradiction. Case 2:  $\alpha = 1$ .

Since  $\overline{V} \cap X^{(1)} = \emptyset$ ,  $V = \overline{V}$  consists of isolated points, say  $V = \{x_s : s \in S\}$ . Choose  $y \in W \cap Y^{(1)}$  and let  $\{W_m : m \in \mathbb{N}\}$  be a decreasing open base at y in W. By corollary 1.2.15 (a) and lemma 1.2.10, there is  $m \in \mathbb{N}$  and  $s_1, \ldots, s_m \in S$  such that

 $\operatorname{supp} W_m \cap \{x_s : s \notin \{s_1, \ldots, s_m\}\} = \emptyset.$ 

Now let  $V' = \{x_{s_1}, \ldots, x_{s_m}\}$ . Since supp  $W_m \subset U \cup V'$ , it follows that

 $W_m \subset \operatorname{supp} \operatorname{supp} W_m \subset \operatorname{supp} (U \cup V') = \operatorname{supp} U \cup \operatorname{supp} V'.$ 

Since  $W_m \cap \text{supp } U = \emptyset$ , we have  $W_m \subset \text{supp } V'$ . Because V' is finite, we have by lemma 1.4.1 that  $W_m$  is finite. This contradicts the fact that  $y \in W_m^{(1)}$ .

**4.1.10 THEOREM:** Let X and Y be  $l_p$ -equivalent metric spaces, let  $\alpha \in \{0, 1\}$ , and let  $\beta$  be an ordinal. Then

(a)  $X^{(\alpha, \beta)} = \emptyset$  if and only if  $Y^{(\alpha, \beta)} = \emptyset$ ,

(b)  $X^{(\alpha, \beta)}$  is compact if and only if  $Y^{(\alpha, \beta)}$  is compact, and

(c)  $X^{(\alpha,\beta)}$  is locally compact if and only if  $Y^{(\alpha,\beta)}$  is locally compact.

**PROOF:** This follows directly from proposition 4.1.9 and theorem 4.1.7.

In section 4.2, we prove that the conditions (a) and (b) are sufficient to obtain an isomorphical classification for countable metric spaces with scattered height less than or equal to  $\omega$ , i.e., if two such spaces satisfy conditions (a) and (b) in theorem 4.1.10 for all ordinals  $\alpha \in \{0, 1\}$  and  $\beta$ , then they are  $\ell_p$ -equivalent.

In our search for  $l_p$ -equivalent pairs we now consider pairs  $(X, X^{(\alpha)})$  for infinite countable prime components  $\alpha$ . We start with the following

**4.1.11 LEMMA:** Let X be a metric space and A a closed subspace of X. Let O be an open neighborhood of A in X. Then there is a continuous linear function  $\phi: C_0(A) \rightarrow C_0(X)$  such that for each  $f \in C(A)$ ,

 $\phi(f) \mid A = f, \, \phi(f)(X) \subset \operatorname{conv} (f(A) \cup \{0\}), \, and \, \phi(f)(X \setminus O) = \{0\}.$ 

**PROOF:** We will construct a continuous linear function  $\phi: C_p(A) \to C_p(X)$  with the required properties. Then by proposition 1.2.19,  $\phi$  considered as a function from  $C_0(A) \to C_0(X)$  is also continuous, and hence is as required.

Since  $A \cup (X \setminus O)$  is a closed subset of X, there is by theorem 2.3.1, a continuous linear function  $\psi: C_p(A \cup (X \setminus O)) \to C_p(X)$  such that for each  $f \in C(A \cup (X \setminus O))$  we have  $\psi(f) \mid (A \cup (X \setminus O)) = f$  and  $\psi(f)(X) \subset \operatorname{conv} f(A \cup (X \setminus O))$ .

For each  $f \in C(A)$ , define  $f^*: A \cup (X \setminus O) \to \mathbb{R}$  by

$$f^{*}(x) = \begin{cases} f(x) & \text{if } x \in A \\ 0 & \text{if } x \in X \setminus O \end{cases}$$

Then  $f^*$  is a well-defined continuous function. Define  $\theta: C_p(A) \to C_p(A \cup (X \setminus O))$  by  $\theta(f) = f^*$ . Then  $\theta$  is a well-defined continuous linear function. Finally define  $\phi: C_p(A) \to C_p(X)$  by  $\phi = \psi \circ \theta$ . Then  $\phi$  is a continuous linear function, and we claim that it is as required. Let  $f \in C(A)$ . Then

 $\phi(f) | A = \psi(\theta(f)) | A = \theta(f) | A = f,$   $\phi(f)(X) = \psi(\theta(f))(X) \subset \operatorname{conv} (\theta(f)(A \cup X \setminus O)) = \operatorname{conv} (f (A) \cup \{0\}), \text{ and }$  $\phi(f)(X \setminus O) = \psi(\theta(f))(X \setminus O) = \theta(f)(X \setminus O) = \{0\}.$ 

This proves the lemma.

**4.1.12 PROPOSITION:** Let  $\alpha < \omega_1$  be a prime component and let X and Y be  $l_p$ -equivalent zero-dimensional separable metric spaces. Then  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are  $l_p$ -equivalent pairs.

**PROOF:** By proposition 2.2.2 (a),  $X^{(\alpha)}$  is closed in X. Let  $\phi: C_p(X) \to C_p(Y)$  be a

linear homeomorphism. Then by corollary 1.2.21,  $\phi$  considered as a function from  $C_0(X)$  to  $C_0(Y)$  is also a linear homeomorphism. It suffices to prove that  $(X, X^{(\alpha)})$  is  $(\phi, 0)$ -relative to  $(Y, Y^{(\alpha)})$ . To this end let U and V be open in X and W open in Y such that

 $(\operatorname{supp} U) \cap W = \emptyset$ , and  $\operatorname{supp} W \subset U \cup V$ .

Suppose that  $W \cap Y^{(\alpha)} \neq \emptyset$  and  $\overline{V} \cap X^{(\alpha)} = \emptyset$ . By proposition 4.1.9 we must have  $\alpha \ge \omega$ . Let  $y \in W \cap Y^{(\alpha)}$  and let  $\{W_m : m \in \mathbb{N}\}$  be a decreasing clopen base at y in W. By lemma 4.1.4 (a), there is a discrete clopen family  $\{V_m : m \in \mathbb{N}\}$  such that  $V \subset \bigcup_{m \in \mathbb{N}} V_m$  and for each  $m \in \mathbb{N}$ , there is  $\beta < \alpha$  such that  $(V_m)^{(\beta)} = \emptyset$ . By corollary 1.2.15 (a) and lemma 1.2.10, there is  $m \in \mathbb{N}$  such that

$$\operatorname{supp} W_m \cap \bigcup_{i>m} V_m = \emptyset.$$

Let  $V' = \bigcup_{i=1}^{m} V_{s_i}$  Notice that V' is clopen and  $\sup W_m \subset U \cup V'$ . Fix  $\beta < \alpha$  such that  $(V')^{(\beta)} = \emptyset$ . Since  $W_m^{(\alpha)} = W_m \cap Y^{(\alpha)} \neq \emptyset$ , by lemma 4.1.8 there is a set  $K \subset W_m$  such that  $K \approx [1, \omega^{\alpha}]$ . Let  $L = \overline{\operatorname{supp} K} \cap V'$ . Then by corollary 1.2.15 (b), L is compact. Furthermore  $L \subset V'$ . We also have that L is non-empty. Indeed, if  $(\operatorname{supp} K) \cap V' = \emptyset$ , then  $\operatorname{supp} K \subset U$ , and so by proposition 1.4.3,

 $K \subset \text{supp supp } K \subset \text{supp } U$ .

Since  $(\sup U) \cap K = \emptyset$ , we then have  $K = \emptyset$ . Contradiction.

By lemma 4.1.11, there is a continuous linear function  $\psi_1: C_0(K) \to C_0(Y)$  such that for each  $f \in C(K)$ ,

 $\psi_1(f) \mid K = f \text{ and } \psi_1(f)(Y \setminus W_m) = \{0\}.$ 

Again by lemma 4.1.11, there is a continuous linear function  $\psi_2: C_0(L) \to C_0(X)$  such that for each  $f \in C(L)$ ,

 $\psi_2(f) \mid L = f \text{ and } \psi_2(f)(X \setminus V') = \{0\}.$ 

Define

$$\psi: C_0(K) \to C_0(L) \text{ by } \psi(f) = \phi^{-1}(\psi_1(f)) \mid L, \text{ and}$$
  
$$\theta: C_0(L) \to C_0(K) \text{ by } \theta(f) = \phi(\psi_2(f)) \mid K.$$

Observe that  $\psi$  and  $\theta$  are linear.

CLAIM: For each  $f \in C(K)$ ,  $\theta(\psi(f)) = f$ .

Suppose there is  $f \in C(K)$  such that  $\theta(\psi(f)) \neq f$ . Then

On the  $l_p$ -equivalence of metric spaces

 $\phi(\psi_2(\psi(f))) \mid K \neq f = \psi_1(f) \mid K.$ 

So by corollary 1.4.2 (b),

 $\psi_2(\psi(f)) \mid \operatorname{supp} K \neq \phi^{-1}(\psi_1(f)) \mid \operatorname{supp} K.$ 

Since  $\psi_1(f)(Y \setminus W_m) = \{0\}$  and  $\sup D \subset Y \setminus W_m$ , it follows from corollary 1.4.2 (b), that  $\phi^{-1}(\psi_1(f))(U) = \{0\}$ . Since  $U \setminus V' \subset X \setminus V'$ ,  $\psi_2(\psi(f))((\sup K \setminus V') = \{0\}$ . Hence

$$\Psi(f) = \Psi_2(\Psi(f)) | L \neq \phi^{-1}(\Psi_1(f)) | L = \Psi(f).$$

Contradiction and the claim is proved.

From the claim we conclude that  $\psi$  is a linear embedding. Since  $L \subset V'$ , we have  $L^{(\beta)} = \emptyset$ . Since L is separable metric, it is countable by the Cantor-Bendixson theorem and so by theorem 2.2.8, there is  $\gamma < \beta$  and  $n \in \mathbb{N}$  such that  $L \approx [1, \omega^{\gamma} \cdot n]$ . Since by theorem 2.4.1  $C_0([1, \omega^{\gamma} \cdot n]) \sim C_0([1, \omega^{\gamma}])$ , we have a linear embedding  $\psi: C_0([1, \omega^{\alpha}]) \rightarrow C_0([1, \omega^{\gamma}])$ . By lemma 2.6.7 and the fact that  $\alpha$  is a prime component it follows that  $\alpha \leq \gamma$ . This gives a contradiction since  $\gamma < \beta < \alpha$ .

**REMARK:** a) For  $\alpha < \omega_1$  not a prime component, there are  $l_p$ -equivalent countable metric spaces X and Y such that  $X^{(\alpha)} = \emptyset$ ,  $Y^{(\alpha)} \neq \emptyset$ . So  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are not  $l_p$ -equivalent pairs. For example let  $X = [1, \omega^{\alpha'}]$  and let  $Y = [1, \omega^{\alpha}]$ . Then by theorem 2.4.7, X and Y are  $l_p$ -equivalent. Since  $\alpha$  is not a prime component,  $\alpha' < \alpha$  hence  $X^{(\alpha)} = \emptyset$ . However  $Y^{(\alpha)} \neq \emptyset$ .

b) The question arises whether "being  $l_p$ -equivalent pairs" is independent from the choice of the linear homeomorphism. From the proof of proposition 4.1.12 it follows that for any linear homeomorphism  $\phi$  between  $C_p(X)$  and  $C_p(Y)$  we have that  $(X, X^{(\alpha)})$  and  $(Y, Y^{(\alpha)})$  are  $l_p$ -equivalent pairs.

**4.1.13 THEOREM:** Let X and Y be  $l_p$ -equivalent zero-dimensional separable metric spaces and let  $\alpha$ ,  $\beta$  be ordinals with  $\alpha < \omega_1$  a prime component. Then

(a)  $X^{(\alpha, \beta)} = \emptyset$  if and only if  $Y^{(\alpha, \beta)} = \emptyset$ ,

(b)  $X^{(\alpha, \beta)}$  is compact if and only if  $Y^{(\alpha, \beta)}$  is compact, and

(c)  $X^{(\alpha, \beta)}$  is locally compact if and only if  $Y^{(\alpha, \beta)}$  is locally compact.

**PROOF:** This follows directly from proposition 4.1.12 and theorem 4.1.7.

**4.1.14 COROLLARY:** Let X and Y be  $l_p$ -equivalent zero-dimensional separable metric spaces and let  $\alpha < \omega_1$  be a prime component. Then

146

§4.1. lp-equivalent properties of metric spaces

(a) X<sup>(α)</sup> =Ø if and only if Y<sup>(α)</sup> =Ø,
(b) X<sup>(α)</sup> is compact if and only if Y<sup>(α)</sup> is compact, and
(c) X<sup>(α)</sup> is locally compact if and only if Y<sup>(α)</sup> is locally compact.

**PROOF:** This is an application of theorem 4.1.13: take  $\beta = 0$ .

The strength of theorem 4.1.7 has now become clear. Once we have  $l_p$ -equivalent pairs such as in propositions 4.1.9 and 4.1.12, we immediately get  $\omega_1 l_p$ -equivalent properties.

Although we were not able to prove proposition 4.1.12 for arbitrary metric spaces, we can give for this class of spaces a direct proof of corollary 4.1.14 (a).

**4.1.15 THEOREM:** Let X and Y be  $l_p$ -equivalent metric spaces and let  $\alpha < \omega_1$  be a prime component. Then  $X^{(\alpha)} = \emptyset$  if and only if  $Y^{(\alpha)} = \emptyset$ .

**PROOF:** By theorem 4.1.10 we may assume that  $\alpha \ge \omega$ . Suppose  $X^{(\alpha)} = \emptyset$  and  $Y^{(\alpha)} \neq \emptyset$ . Choose  $y \in Y^{(\alpha)}$  and let  $\{W_n : n \in \mathbb{N}\}$  be an open decreasing base at y.

CLAIM 1: There is a locally finite open covering  $\{V_s : s \in S\}$  of X such that for each  $s \in S$ , there is  $\beta < \alpha$  such that  $\overline{V}_s \cap X^{(\beta)} = \emptyset$ .

Since  $X^{(\alpha)} = \emptyset$ ,  $\mathcal{U} = \{X \setminus X^{(\beta)} : \beta < \alpha\}$  is an open covering of X. Let  $\{V_s : s \in S\}$  be a locally finite open covering of X such that  $\{\overline{V}_s : s \in S\}$  refines  $\mathcal{U}$ . Then for  $s \in S$  there is  $\beta < \alpha$  such that  $\overline{V}_s \subset X \setminus X^{(\beta)}$ . Hence  $\overline{V}_s \cap X^{(\beta)} = \emptyset$ .

Fix  $\beta$  as in the claim. Let  $\{F_s : s \in S\}$  be a closed covering of X such that for each  $s \in S$ ,  $F_s \subset V_s$ . By corollary 1.2.15 (a) and lemma 1.2.10, there are  $m \in \mathbb{N}$  and  $\{s_1, \ldots, s_m\} \subset S$  such that

 $\operatorname{supp} W_m \cap \bigcup_{s \notin \{s_1, \ldots, s_m\}} V_s = \emptyset.$ 

Let  $V' = \bigcup_{i=1}^{n} V_{s_i}$  and  $F' = \bigcup_{i=1}^{n} F_{s_i}$ . Then V' is open, F' is closed and  $F' \subset V'$ . Find a copy K of  $[1, \omega^{\alpha}]$  in  $W_m$  (lemma 4.1.8). Let  $L = \overline{\operatorname{supp} K}$ . Note that  $\operatorname{supp} K \subset F'$ , hence  $L \subset V'$ . Furthermore L is compact. If  $L = \emptyset$ , then  $\operatorname{supp} K = \emptyset$ , hence  $K = \emptyset$ . This gives a contradiction, so  $L \neq \emptyset$ . Let  $\psi_1 : C_0(K) \to C_0(Y)$ ,  $\psi_2 : C_0(L) \to C_0(X)$ ,  $\psi : C_0(K) \to C_0(L)$  and  $\theta : C_0(L) \to C_0(K)$  be continuous linear functions such as in the proof of proposition 4.1.12.

CLAIM 2: For each  $f \in C(K)$ ,  $\theta(\psi(f)) = f$ .

Suppose there is  $f \in C(K)$  such that  $\theta(\psi(f)) \neq f$ . Then

 $\phi(\psi_2(\psi(f))) \mid K \neq f = \psi_1(f) \mid K,$ 

so by corollary 1.4.2 (b),

$$\psi_2(\psi(f)) | \operatorname{supp} K \neq \phi^{-1}(\psi_1(f)) | \operatorname{supp} K.$$

This implies that

$$\Psi(f) = \Psi_2(\Psi(f)) | L \neq \phi^{-1}(\Psi_1(f)) | L = \Psi(f).$$

Contradiction, and the claim is proved.

From the claim we conclude that  $\psi$  is a linear embedding. Since  $L \subset \overline{V'}$ , we have by proposition 2.2.4,

$$L^{(\beta)} \subset L \cap \overline{V'}^{(\beta)} \subset L \cap \overline{V'} \cap X^{(\beta)} = \emptyset.$$

As in proposition 4.1.12, we arrive at a contradiction.  $\Box$ 

By the Cantor-Bendixson theorem, each scattered separable metric space has scattered height less than  $\omega_1$ . Hence we have the following

**4.1.16 COROLLARY:** Let X and Y be  $l_p$ -equivalent separable metric spaces. Then X is scattered if and only if Y is scattered.

The question arises whether corollary 4.1.16 holds in the class of metric spaces, or whether theorem 4.1.15 holds for all prime components. In the proof of theorem 4.1.15 we used the isomorphical classification of function spaces of countable compact spaces (cf. section 2.4). For prime components larger than or equal to  $\omega_1$  we cannot use this result. Moreover, we are not able to use larger compact ordinal intervals (cf. section 2.6) because they are not metric. There is one case in which we can overcome these technical problems.

# **4.1.17 THEOREM:** Let X and Y be $l_p$ -equivalent metric spaces. Then $X^{(\omega_1)} = \emptyset$ if and only if $Y^{(\omega_1)} = \emptyset$ .

**PROOF**: The proof is almost the same as the proof of theorem 4.1.15. Replace  $\alpha$  in the proof of theorem 4.1.15 by  $\omega_1$ . Copy this proof until the set *K* is introduced. We have  $\overline{V'} \cap X^{(\beta)} = \emptyset$  for some  $\beta < \omega_1$ . Find a prime component  $\alpha < \omega_1$ , such that  $\beta < \alpha$ . Find a copy *K* of  $[1, \omega^{\alpha}]$  in  $W_m$  and let  $L = \overline{\text{supp } K}$ . As in theorem 4.1.15 we obtain a

§4.2. An isomorphical classification

contradiction.  $\square$ 

We finish this section by posing the following

**Question:** Let X and Y be  $l_p$ -equivalent metric spaces. For which ordinals  $\alpha$  is theorem 4.1.15 true? Is it true for prime components? Is it only true for prime components? Is it true for  $\omega_1 \cdot 2$ ?

## §4.2. An isomorphical classification

In this section we present an isomorphical classification of function spaces of countable metric spaces which have scattered height less than or equal to  $\omega$ . In chapter 2 we have considered finite spaces. We will assume in this section that all spaces are infinite.

Let X be a space. For ordinals  $\alpha$  and  $\beta$ , we define the following:

 $X(\alpha, \beta) = 0$  if and only if  $X^{(\alpha, \beta)} = \emptyset$ ,

- $X(\alpha, \beta) = 1$  if and only if  $X^{(\alpha, \beta)}$  is non-empty and compact, and
- $X(\alpha, \beta) = 2$  if and only if  $X^{(\alpha, \beta)}$  is not compact.

With this notation, part of the results in section 4.1 can be reformulated as follows:

(I) Let X and Y be  $l_p$ -equivalent spaces zero-dimensional separable metric spaces. Then for every pair of ordinals  $\alpha$ ,  $\beta$  with  $\alpha < \omega_1$  a prime component, we have  $X(\alpha, \beta) = Y(\alpha, \beta)$ .

As mentioned above we restrict ourselves in this section to countable metric spaces X which have scattered height less than or equal to  $\omega$ . In this class of spaces, (I) takes the following form (note that for such X, we have  $X^{(\omega)} = \emptyset$ , so the ordinals we have to consider here are the finite ordinals):

(II) Let X and Y be  $l_p$ -equivalent countable metric spaces which have scattered height less than or equal to  $\omega$ . Then for every  $n \in \mathbb{N} \cup \{0\}, X(0, n) = Y(0, n)$  and X(1, n) = Y(1, n).

In this section we will show that the necessary conditions in (II) are also sufficient, i.e., if for two infinite countable metric spaces X and Y which have scattered height less than

or equal to  $\omega$ , X(0, n) = Y(0, n) and X(1, n) = Y(1, n) for every  $n \in \mathbb{N} \cup \{0\}$ , then X and Y are  $l_p$ -equivalent.

Before we consider function spaces of countable metric spaces, we first deal with the countable metric spaces itself. Some of the next lemmas are formulated in a more general form in case their proofs do not use special properties of countable metric spaces.

**4.2.1 LEMMA:** Let X be a space. Then for every  $n \in \mathbb{N}$ ,  $X^{(0,n)} \subset X^{(1,n-1)} \subset X^{(0,n-1)}$ .

**PROOF:** It is easily seen that

 $X^{(0,1)} \subset X^{(1)} \subset X = X^{(0)},$ 

from which it follows that for every  $n \in \mathbb{N}$ ,

$$(X^{(0,1)})^{(0,n-1)} \subset (X^{(1)})^{(0,n-1)} \subset X^{(0,n-1)}$$

This completes the proof of this lemma.

**4.2.2 COROLLARY:** Let X be a space, such that there is  $n \in \mathbb{N}$  with X(0, n) = 0. Let  $n_0 = \min\{n : X(0, n) = 0\}$ . Then  $n_1 = \min\{n : X(1, n) = 0\}$  is well-defined and  $n_0 = n_1$  or  $n_0 = n_1 + 1$ .

**PROOF:** By lemma 4.2.1,  $X^{(1, n_0)} \subset X^{(0, n_0)}$ , so that  $n_1 \le n_0$ . Again by lemma 4.2.1,  $X^{(0, n_1+1)} \subset X^{(1, n_1)}$ , so that  $n_0 \le n_1 + 1$ .

We can distinguish the spaces X with scattered height less than or equal to  $\omega$  into two types. The first type consists of those X such that for each  $n \in \mathbb{N}$ , X(0, n) = 2; then also X(1, n) = 2 for each  $n \in \mathbb{N}$ , by lemma 4.2.1. For the other spaces, there is  $n \in \mathbb{N}$ such that X(0, n) = 1 or X(0, n) = 0; if X(0, n) = 1, then X(0, n + 1) = 0, so in both cases X(0, n) = 0 for some  $n \in \mathbb{N}$ . For spaces of the second type we have

**4.2.3 LEMMA:** Let X be a space such that there is  $m \in \mathbb{N}$  with X(0, m) = 0. Then there is  $n \in \mathbb{N}$  such that X satisfies one of the following conditions:

 $(a)_n X(0, n) = 0, X(0, n-1) = 1, and X(1, n-1) = 1,$   $(b)_n X(0, n) = 0, X(0, n-1) = 2, and X(1, n-1) = 1,$   $(c)_n X(0, n) = 0, X(0, n-1) = 2, and X(1, n-1) = 2,$  $(d)_n X(1, n) = 0, X(1, n-1) = 1, and X(0, n) = 1,$ 

150

 $(e)_n X(1, n) = 0, X(1, n-1) = 2, and X(0, n) = 1,$   $(f)_n X(1, n) = 0, X(1, n-1) = 2, and X(0, n) = 2, or$  (g) X(0, 1) = 0, X(0, 0) = 2, and X(1, 0) = 0,(i.e., X is an infinite discrete space).

#### **PROOF:** As in corollary 4.2.2, let

 $n_0 = \min\{n : X(0, n) = 0\}, \text{ and } n_1 = \min\{n : X(1, n) = 0\}.$ 

Then  $n_0 = n_1$  or  $n_0 = n_1 + 1$ . Since  $X \neq \emptyset$ ,  $n_0 \in \mathbb{N}$ . case 1:  $n_0 = n_1$ .

In this case  $X(0, n_0) = X(1, n_0) = 0$ ,  $X(0, n_0 - 1) \neq 0$  and  $X(1, n_0 - 1) \neq 0$ . If  $X(0, n_0 - 1) = 1$ , then by lemma 4.2.1,  $X(1, n_0 - 1) = 1$ , so X satisfies  $(a)_{n_0}$ . If  $X(0, n_0 - 1) = 2$ , then X satisfies  $(b)_{n_0}$  or  $(c)_{n_0}$ .

case 2:  $n_0 = n_1 + 1$ , where  $n_1 \in \mathbb{N}$ .

In this case  $X(0, n_1+1) = X(1, n_1) = 0$ ,  $X(0, n_1) \neq 0$  and  $X(1, n_1-1) \neq 0$ . If  $X(1, n_1-1) = 1$ , then by lemma 4.2.1,  $X(0, n_1) = 1$ , so X satisfies  $(d)_{n_1}$ . If  $X(1, n_1-1) = 2$ , then X satisfies  $(e)_{n_1}$  or  $(f)_{n_1}$ .

*case 3*:  $n_1 = 0$ , and  $n_0 = 1$ .

Since X(1, 0) = 0, X has no accumulation points, so in this case, X is an infinite discrete space.  $\Box$ 

At the end of this section we will show that for each case, there exist countable metric spaces with scattered height less than or equal to  $\omega$  satisfying the corresponding conditions. We will now present a special decomposition of the spaces of interest in this section. (cf. corollaries 4.2.5, 4.2.7 and 4.2.9). We restrict ourselves to countable metric spaces.

**4.2.4 LEMMA:** Let X be a countable metric space. Let A and B be closed in X with  $A \subset B$  and suppose that A(0, 1) = B(0, 1) = 1. Then there is a decreasing clopen base  $\{U_n : n \in \mathbb{N}\}$  at  $B^{(0, 1)}$  in X such that  $U_1 = X$  and  $(U_n \setminus U_{n+1}) \cap A$  is not compact for every  $n \in \mathbb{N}$ .

**PROOF:** Since  $B^{(0,1)}$  is compact, there is a decreasing clopen base  $\{V_n : n \in \mathbb{N}\}$  at  $B^{(0,1)}$  in X. We now inductively find the  $U_n$ . Let  $U_1 = X$  and suppose we have chosen  $U_1, \ldots, U_n$  for some  $n \in \mathbb{N}$ . Since  $A^{(0,1)} \subset B^{(0,1)}$ ,  $U_n$  is a neighborhood of  $A^{(0,1)}$ . But then  $U_n \cap A$  is not compact, from which it follows that there is an infinite closed discrete set E in  $U_n \cap A$ . Since  $B^{(0,1)}$  is compact, without loss of generality we may as-

sume that  $E \cap B^{(0,1)} = \emptyset$ , so there is i > n such that  $V_i \subset X \setminus E$ . If we now let  $U_{n+1} = V_i$ , then  $E \subset (U_n \setminus U_{n+1}) \cap A$ .

**4.2.5 COROLLARY:** Let X be a countable metric space, and let  $m \in \mathbb{N}$ .

- (a) If X(0,m) = X(1,m) = 1, then there is a clopen decreasing base  $\{U_n : n \in \mathbb{N}\}$  at  $X^{(0,m)}$  in X, such that  $U_1 = X$  and  $(U_n \setminus U_{n+1})(1, m-1) = 2$  for every  $n \in \mathbb{N}$ .
- (b) If X(1, m) = X(0, m+1) = 1, then there is a clopen decreasing base  $\{U_n : n \in \mathbb{N}\}$  at  $X^{(1, m)}$  in X, such that  $U_1 = X$  and  $(U_n \setminus U_{n+1})(0, m) = 2$  for every  $n \in \mathbb{N}$ .

**PROOF:** This is a direct consequence of lemmas 4.2.1 and 4.2.4.  $\Box$ 

**4.2.6 LEMMA:** Let X be a countable metric space. Let A and B be closed in X with  $A \subset B$ . If A and B are locally compact but not compact, then X can be written as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} X_i$  such that for each i,  $X_i \cap A$  and  $X_i \cap B$  are compact and non-empty.

**PROOF:** Since *B* is locally compact but not compact and *X* is zero-dimensional, we can write *X* as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} K_i$  such that for every  $i \in \mathbb{N}$ ,  $K_i \cap B$  is compact. Since  $A \subset B$ , for every  $i \in \mathbb{N}$ ,  $A \cap K_i$  is compact as well. Since *A* is not compact we can find a strictly increasing sequence  $(i_n)_{n \in \mathbb{N}}$  such that for each  $n \in \mathbb{N}$ ,  $A \cap K_{i_n}$  is not empty. Taking  $X_n = \bigcup_{i=i_{n-1}}^{i_n} K_i$  (where  $i_0 = 1$ ) we obtain the desired decomposition.  $\Box$ 

**4.2.7 COROLLARY:** Let X be a countable metric space and let  $m \in \mathbb{N}$ .

- (a) If X(0, m) = 0 and X(1, m-1) = 2, then X can be written as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} A_i$  such that for every  $i \in \mathbb{N}$ ,  $A_i(0, m-1) = A_i(1, m-1) = 1$ .
- (b) If X(1, m) = 0 and X(0, m) = 2, then X can be written as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} A_i$  such that for every  $i \in \mathbb{N}$ ,  $A_i(0, m) = A_i(1, m-1) = 1$ .

**PROOF:** This a direct consequence of lemmas 4.2.1 and 4.2.6.

**4.2.8 LEMMA:** Let X be a countable metric space. Let A and B be closed in X with  $A \subset B$ . If A is compact and non-empty and B is locally compact but not compact, then X can be written as a clopen disjoint union  $X = X_1 \cup X_2$  such that

(1)  $X_1 \cap B$  is compact and non-empty and (2)  $X_2 \cap A = \emptyset$ .

**PROOF:** As in the proof of lemma 4.2.6, X can be written as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} K_i$  with for each  $i \in \mathbb{N}$ ,  $K_i \cap B$  is compact and non-empty. Since A is compact, there is  $i_0$  such that  $A \cap \bigcup_{i>i_0} K_i = \emptyset$ . Now let  $X_1 = \bigcup_{i=1}^{i_0} K_i$  and  $X_2 = \bigcup_{i>i_0} K_i$ .

**4.2.9 COROLLARY:** Let X be a countable metric space and let  $m \in \mathbb{N} \cup \{0\}$ .

- (a) If X(0, m) = 2, X(0, m+1) = 0 and X(1, m) = 1, then X can be written as a clopen disjoint union  $X = X_1 \cup X_2$  such that  $X_1(0, m) = 1$  and  $X_2(1, m) = 0$ .
- (b) If X(1, m) = 2, X(1, m+1) = 0 and X(0, m+1) = 1, then X can be written as a clopen disjoint union  $X = X_1 \cup X_2$  such that  $X_1(1, m) = 1$  and  $X_2(0, m+1) = 0$ .

**PROOF:** This is a direct consequence of lemmas 4.2.1 and 4.2.8.  $\Box$ 

**REMARK:** Notice that in corollary 4.2.9 (a) we also have that  $X_1(1, m) = 1$  because  $X^{(1, m)} = X_1^{(1, m)} \cup X_2^{(1, m)}$ . Similarly we have  $X_2(0, m+1) = 0, X_2(0, m) = 2$  and if  $m \neq 0, X_2(1, m-1) = 2$ . In addition, in corollary 4.2.9 (b) we have  $X_1(0, m+1) = 1, X_2(1, m+1) = 0, X_2(1, m) = 2$  and  $X_2(0, m) = 2$ .

We return to the subject of function spaces. The following lemma together with the "decomposition" lemmas above will play a fundamental role in proving the announced isomorphical classification of function spaces of infinite countable metric spaces which have scattered height less than or equal to  $\omega$ .

**4.2.10 LEMMA:** Let X be a countable metric space, A a non-empty compact subspace of X, and  $\{U_n : n \in \mathbb{N}\}$  a clopen decreasing base at A in X such that  $U_1 = X$ . Let Y be a countable metric space, B a non-empty compact subspace of Y, and  $\{V_n : n \in \mathbb{N}\}$  a clopen decreasing base at B in Y such that  $V_1 = X$ . Let  $k \in \mathbb{N}$ . Suppose that for every  $n \in \mathbb{N}$ ,  $\phi_n : C_p(U_n \setminus U_{n+1}) \to C_p(V_n \setminus V_{n+1})$  is a linear k-mapping. Define  $\phi: C_{p,A}(X) \to C_{p,B}(Y)$  by

 $\phi(f) | V_n \setminus V_{n+1} = \phi_n(f | U_n \setminus U_{n+1}) \text{ and } \phi(f) | B = 0.$ 

Then  $\phi$  is a well-defined linear k-mapping. If moreover each  $\phi_n$  is a linear k-homeomorphism, then  $\phi$  is a linear k-homeomorphism.

**PROOF:** To prove that  $\phi$  is well-defined it suffices to prove that  $\phi(f)$  is continuous at points of *B*. Let  $\varepsilon > 0$ . Since f(A) = 0, there is an open neighborhood *W* of *A* with  $f(W) \subset \left(-\frac{\varepsilon}{k}, \frac{\varepsilon}{k}\right)$ . There is  $n_0 \in \mathbb{N}$  such that  $A \subset U_{n_0} \subset W$ , so

$$f(U_{n_0})\!\subset\!(-\frac{\varepsilon}{k},\frac{\varepsilon}{k}).$$

Then it easily follows by k-linearity of  $\phi_n$  for every n, that  $\phi(f)(V_{n_0}) \subset (-\varepsilon, \varepsilon)$ , so that  $\phi(f)$  is continuous at points of B.

To prove continuity of  $\phi$ , it suffices to prove that  $\phi$  is continuous at 0. Let  $P \subset Y$  be finite and  $\varepsilon > 0$ . For each  $n \in \mathbb{N}$ , let  $P_n = P \cap (V_n \setminus V_{n+1})$ . Since P is finite, there is  $n_0 \in \mathbb{N}$  such that for each  $n > n_0$ ,  $P_n = \emptyset$ . Let  $n \le n_0$ . Since  $\phi_n$  is continuous, there are a finite  $Q_n \subset U_n \setminus U_{n+1}$  and  $\delta_n > 0$  such that

$$\phi_n(<0, Q_n, \delta_n >) \subset <0, P_n, \varepsilon >.$$

Let  $Q = \bigcup_{n=1}^{n_0} Q_n$  and  $\delta = \min\{\delta_i : 1 \le i \le n_0\}$ . Then it is easily seen that

 $\phi(\langle 0, Q, \delta \rangle) \subset \langle 0, P, \varepsilon \rangle.$ 

The k-linearity of  $\phi$  is an easy exercise.

Now suppose each  $\phi_n$  is a linear k-homeomorphism. Define  $\psi: C_{p, B}(Y) \to C_{p, A}(X)$  by

$$\Psi(f) | U_n \setminus U_{n+1} = \phi_n^{-1}(f | V_n \setminus V_{n+1}) \text{ and } \Psi(f) | A = 0.$$

Then  $\psi$  is a well-defined linear k-mapping which is easily seen to be equal to  $\phi^{-1}$ , hence  $\phi$  is a linear k-homeomorphism.  $\Box$ 

We are now in a position to prove an isomorphical classification of function spaces of countable metric spaces which have scattered height less than or equal to  $\omega$ . First we consider the case of countable metric spaces which have scattered height strictly less than  $\omega$ . The proof will be an inductive one. In the following two lemmas we deal with spaces at a "low level". The space *T* in lemma 4.2.12 is the one defined in section 3.3.

**4.2.11 LEMMA:** Let  $q \in \mathbb{N}$ . There is  $k_q \in \mathbb{N}$  such that if X and Y are infinite countable compact spaces with  $\kappa(X), \kappa(Y) \leq q$ , then  $C_p(X) \stackrel{k_q}{\to} C_p(Y)$ .

**PROOF:** Let X be an infinite countable compact metric space with  $\kappa(X) \le q$ . By theorem 2.2.8, there are  $1 \le m \le q$  and  $n \in \mathbb{N}$  such that  $X \approx [1, \omega^m \cdot n]$ . Let  $a = \omega^m$  and

 $A = X^{(m)}$ . Notice that  $A = \{\omega^m \cdot i : 1 \le i \le n\}$ . Then

$$C_{p}(X) \stackrel{2}{\sim} C_{p, a}([1, \omega^{m}]) \times C_{p}(A) \quad \text{(corollaries 2.2.10 and 2.3.4)}$$

$$\stackrel{1}{\sim} C_{p, a}(A \oplus [1, \omega^{m}]) \quad \text{(lemma 2.3.6)}$$

$$\stackrel{1}{\sim} C_{p, a}([1, \omega^{m}]) \quad \text{(lemma 2.3.5)}$$

$$\stackrel{2}{\sim} C_{p}([1, \omega^{m}]) \quad \text{(lemma 2.3.9)}.$$

So that  $C_p(X) \stackrel{4}{\sim} C_p([1, \omega^m])$ .

To finish the lemma it suffices to prove the following

CLAIM: There is  $l \in \mathbb{N}$  such that for every  $1 \le r \le q$  we have

 $C_p([1,\omega^r]) \stackrel{l}{\sim} C_p([1,\omega]).$ 

Let  $1 \le r \le q$ . By theorem 2.4.7, there is a linear homeomorphism  $\phi: C_p([1, \omega^r]) \to C_p([1, \omega])$ . Then by corollary 1.2.21,  $\phi: C_0([1, \omega^r]) \to C_0([1, \omega])$  is also a linear homeomorphism. Since these two function spaces are Banach spaces, there is  $l(r) \in \mathbb{N}$  such that for every  $f \in C_0([1, \omega^r])$  we have

 $\frac{1}{l(r)} \|f\| < \|\phi(f)\| < l(r)\|f\|.$ 

Then  $l = \max\{l(r) : r \le q\}$  is as required.  $\Box$ 

From now on we fix for each  $q \in \mathbb{N}$ ,  $k_q$  as in lemma 4.2.11.

**4.2.12 LEMMA:** Let  $q \in \mathbb{N}$ . There is  $l_q \ge k_q$  such that if X and Y are countable metric spaces with  $\kappa(X)$ ,  $\kappa(Y) \le q$ , X(0, 1) = Y(0, 1) = 1 and X(1, 0) = Y(1, 0) = 1, then

$$C_p(X) \stackrel{i_q}{\sim} C_p(Y).$$

**PROOF:** Let X be a countable metric space with  $\kappa(X) \le q$  and X(0, 1) = X(1, 0) = 1. Let  $A = X^{(1)}$ . Then by assumption A is compact. Since X/A is a perfect image of X, and X is not locally compact, we have that X / A is a non-locally compact countable metric space with exactly one non-isolated point. It then easily follows that X/A is homeomorphic to the space T. Then by corollary 2.3.4,

 $C_p(X) \stackrel{2}{\sim} C_{p,\infty}(T) \times C_p(A).$ 

If A is finite, then  $T \oplus A$  is homeomorphic to T, so  $C_p(X) \stackrel{2}{\sim} C_{p,\infty}(T)$ . Now suppose that A is infinite. We have by lemma 4.2.11,  $C_p(A) \stackrel{k_q}{\sim} C_p([1, \omega])$ . Note that by the above argument  $C_p(T) \stackrel{2}{\sim} C_{p,\infty}(T)$ , so that  $C_p(X) \stackrel{4\cdot k_q}{\sim} C_p(T \oplus [1, \omega])$ . Since  $(T \oplus [1, \omega])^{(1)}$  is finite, the same argument gives  $C_p(T \oplus [1, \omega]) \stackrel{2}{\sim} C_{p,\infty}(T)$ . We conclude that  $C_p(X) \stackrel{8 \cdot k_q}{\sim} C_p(T)$ . Then  $l_q = (8 \cdot k_q)^2$  is as required.  $\Box$ 

From now on we fix for each  $q \in \mathbb{N}$ ,  $l_q$  as in lemma 4.2.12.

The isomorphical classification for countable metric spaces with scattered height less than  $\omega$  will be given after the following lemma, in which all the tools developed in this section are used.

**4.2.13 LEMMA:** For every  $q \in \mathbb{N}$ , there is  $r_q \in \mathbb{N}$ , such that if X and Y are infinite countable metric spaces with  $\kappa(X) \leq q$ ,  $\kappa(Y) \leq q$  and for every  $n \in \mathbb{N}$ , X(0, n) = Y(0, n), and X(1, n) = Y(1, n), then  $C_p(X)^{\frac{r_q}{2}}C_p(Y)$ .

**PROOF:** Let  $q \in \mathbb{N}$  and for every  $m \leq q$ , let  $s_m = 4^m \cdot l_q$ . Let X and Y be infinite countable metric spaces with  $\kappa(X) \leq q$ ,  $\kappa(Y) \leq q$  and for every  $n \in \mathbb{N}$ , X(0, n) = Y(0, n), and X(1, n) = Y(1, n). Since  $X^{(q)} = Y^{(q)} = \emptyset$ , X(0, q) = Y(0, q) = 0, hence X and Y both satisfy the condition in lemma 4.2.3. If X and Y do not satisfy (g), find  $m \in \mathbb{N}$  such that X and Y both satisfy one of the cases  $(a)_m$  through  $(f)_m$ . Notice that  $1 \leq m \leq q$ . We will

prove that  $C_p(X) \stackrel{s_m}{\sim} C_p(Y)$  by induction on m. Then  $r_q = s_q$  is as required. The inductive proof is organized as follows: We prove case  $(a)_m$  for m = 1 and for m > 1 we use  $(c)_{m-1}$ . The proof of case  $(b)_m$  makes use of case  $(a)_m$  and for m > 1 it makes use of case  $(f)_{m-1}$ . We prove case  $(c)_m$  using case  $(a)_m$ . We prove case  $(d)_m$  for m = 1 and for m > 1 and for m > 1 we use  $(f)_{m-1}$ . For case  $(e)_m$  we use  $(d)_m$  and if m > 1 we use  $(c)_m$ . Finally in case  $(f)_m$  we use  $(d)_m$ .

case  $(a)_m$ : X(0, m) = Y(0, m) = 0, X(0, m-1) = Y(0, m-1) = 1, and X(1, m-1) = Y(1, m-1) = 1.

Notice that in this case we have X(1, m) = Y(1, m) = 0. For m = 1 we have that X

and Y are infinite countable compact spaces, so by lemma 4.2.11,  $C_p(X) \stackrel{k_q}{\sim} C_p(Y)$ . Note that  $k_q \le l_q \le s_1$ . For m > 1, let  $A = X^{(0, m-1)}$  and  $B = Y^{(0, m-1)}$ . By proposition 2.3.2,

$$C_p(X) \stackrel{2}{\sim} C_{p,A}(X) \times C_p(A) \text{ and } C_p(Y) \stackrel{2}{\sim} C_{p,B}(Y) \times C_p(B).$$
(1)

Define  $Z_1 = X \oplus A$  and  $Z_2 = Y \oplus B$ . Notice that since m > 1,

$$Z_1^{(0, m-1)} = X^{(0, m-1)}, Z_1^{(1, m-1)} = X^{(1, m-1)},$$
  

$$Z_2^{(0, m-1)} = Y^{(0, m-1)}, \text{ and } Z_2^{(1, m-1)} = Y^{(1, m-1)}.$$

Let  $C = Z_1^{(0, m-1)}$  and  $D = Z_2^{(0, m-1)}$ . Then by (1) and by lemma 2.3.6,

 $C_p(X) \stackrel{2}{\sim} C_{p, C}(Z_1)$  and  $C_p(Y) \stackrel{2}{\sim} C_{p, D}(Z_2)$ .

By corollary 4.2.5 (a), there are clopen decreasing bases  $\{U_n : n \in \mathbb{N}\}$  and  $\{V_n : n \in \mathbb{N}\}$  at C and D, respectively such that  $U_1 = Z_1$ ,  $V_1 = Z_2$ ,

$$(U_n \setminus U_{n+1})(1, m-2) = 2$$
 and  $(V_n \setminus V_{n+1})(1, m-2) = 2$ .

Notice that then also

$$(U_n \setminus U_{n+1})(0, m-2) = (V_n \setminus V_{n+1})(0, m-2) = 2.$$

It is easily seen that

$$(U_n \setminus U_{n+1})(0, m-1) = 0$$
 and  $(V_n \setminus V_{n+1})(0, m-1) = 0$ .

Then  $(c)_{m-1}$  gives  $C_p(U_n \setminus U_{n+1}) \stackrel{s_{m-1}}{\sim} C_p(V_n \setminus V_{n+1})$  for every  $n \in \mathbb{N}$ , whence by lemma 4.2.10,  $C_{p, C}(Z_1) \stackrel{s_{m-1}}{\sim} C_{p, D}(Z_2)$ . In conclusion we have  $C_p(X) \stackrel{s_m}{\sim} C_p(Y)$ . This completes the proof of case  $(a)_m$ .

case  $(b)_m$ : X(0, m) = Y(0, m) = 0, X(0, m-1) = Y(0, m-1) = 2, and X(1, m-1) = Y(1, m-1) = 1.

Again in this case we have X(1, m) = Y(1, m) = 0. By corollary 4.2.9 (a), X and Y can be written as clopen disjoint unions,  $X = A \cup B$  and  $Y = C \cup D$  such that A(0, m-1) = C(0, m-1) = 1 and B(1, m-1) = D(1, m-1) = 0. By the remark following corollary 4.2.9 we now have by case  $(a)_m$ ,  $C_p(A) \stackrel{s_m}{\sim} C_p(C)$  and for m > 1, by  $(f)_{m-1}$ ,  $C_p(B) \stackrel{s_{m-1}}{\sim} C_p(D)$ . If m = 1 then B and D are infinite discrete and so  $C_p(B) \stackrel{1}{\sim} C_p(D)$ . With lemma 2.3.7 it now follows that  $C_p(X) \stackrel{s_m}{\sim} C_p(Y)$ . This completes the proof of case  $(b)_m$ .

case  $(c)_m$ : X(0, m) = Y(0, m) = 0, X(0, m-1) = Y(0, m-1) = 2, and X(1, m-1) = Y(1, m-1) = 2.

Again X(1, m) = Y(1, m) = 0. We have by corollary 4.2.7 (a), X and Y can be written as clopen disjoint unions,  $X = \bigcup_{i=1}^{\infty} A_i$  and  $Y = \bigcup_{i=1}^{\infty} B_i$  such that for each  $i \in \mathbb{N}$ ,  $A_i(0, m-1) = B_i(0, m-1) = A_i(1, m-1) = B_i(1, m-1) = 1$ . By case  $(a)_m$ , we then have  $C_p(A_i) \stackrel{s_m}{\sim} C_p(B_i)$ , so that by lemma 2.3.7,  $C_p(X) \stackrel{s_m}{\sim} C_p(Y)$ . This completes the proof of case  $(c)_m$ .

case 
$$(d)_m$$
:  $X(1, m) = Y(1, m) = 0$ ,  $X(1, m-1) = Y(1, m-1) = 1$ , and  
 $X(0, m) = Y(0, m) = 1$ .

The proof of this case is almost the same as the proof of case  $(a)_m$ . Instead of  $(c)_{m-1}$  we use  $(f)_{m-1}$ , instead of lemma 4.2.11, we use lemma 4.2.12 and instead of corollary 4.2.5 (a), we use corollary 4.2.5 (b).

case 
$$(e)_m$$
:  $X(1, m) = Y(1, m) = 0$ ,  $X(1, m-1) = Y(1, m-1) = 2$ , and  
 $X(0, m) = Y(0, m) = 1$ .

The proof is almost the same as the proof of case  $(b)_m$ . Instead of corollary 4.2.9 (a), we use corollary 4.2.9 (b), instead of  $(a)_m$  we use  $(d)_m$  and instead of  $(f)_{m-1}$  we use  $(c)_m$ .

case 
$$(f)_m$$
:  $X(1, m) = Y(1, m) = 0$ ,  $X(1, m-1) = Y(1, m-1) = 2$ , and  
 $X(0, m) = Y(0, m) = 2$ .

The proof is almost the same as the proof of case  $(c)_m$ . Instead of corollary 4.2.7 (a), we use corollary 4.2.7 (b), and instead of  $(a)_m$  we use  $(d)_m$ .

case (g): X and Y are infinite discrete spaces.

Since X and Y are countable we have that X and Y are both homeomomorphic to  $\mathbb{N}$ . Now apply lemma 2.3.5.

This completes the proof of this lemma.

**4.2.14 THEOREM:** Let X and Y be infinite countable metric spaces with  $\kappa(X), \kappa(Y) < \omega$  such that for every  $n \in \mathbb{N}, X(0, n) = Y(0, n)$  and X(1, n) = Y(1, n). Then  $C_p(X) \sim C_p(Y)$ .

**PROOF:** Let  $q = \max(\kappa(X), \kappa(Y))$  and apply lemma 4.2.13.

We have completed the case of countable metric spaces with scattered height less than  $\omega$ , so from now on we consider spaces with scattered height equal to  $\omega$ . So let X be a countable metric space with  $\kappa(X) = \omega$ . As mentioned in the beginning of this section, there are two cases:

(a) there is *n* ∈ **N**, such that *X*(0, *n*)=0,
(b) for every *n* ∈ **N**, *X*(0, *n*)=2.

Note that if a space X satisfies condition (b), then  $\kappa(X) \ge \omega$ .

We will first deal with spaces for which case (a) holds. We need another decomposition lemma.

**4.2.15 LEMMA:** Let X and Y be countable metric spaces such that  $\kappa(X) = \omega$ ,  $\kappa(Y) \le \omega$ , X(0, n) = Y(0, n) and X(1, n) = Y(1, n) for every  $n \in \mathbb{N} \cup \{0\}$  and such that case (a) holds for X. Then X and Y can be written as clopen disjoint unions,  $X = \bigcup_{i=1}^{\infty} X_i$  and  $Y = \bigcup_{i=1}^{\infty} Y_i$  such that  $\kappa(X_i)$ ,  $\kappa(Y_i) < \omega$  and for every  $i, n \in \mathbb{N}$ ,  $X_i(0, n) = Y_i(0, n)$  and  $X_i(1, n) = Y_i(1, n)$ .

**PROOF:** Since X satisfies (a), and  $\kappa(X) = \omega$ , there is by lemma 4.2.3,  $k \in \mathbb{N}$  such that X and Y both satisfy one of the cases  $(a)_k$  through  $(f)_k$ . We prove the lemma by induction on k and the inductive proof is organized as the inductive proof in lemma 4.2.13.

case 
$$(a)_m$$
:  $X(0, m) = Y(0, m) = 0$ ,  $X(0, m-1) = Y(0, m-1) = 1$ , and  
 $X(1, m-1) = Y(1, m-1) = 1$ .

Since  $\kappa(X) = \omega$ , X is not compact. This implies m > 1. By corollary 4.2.5 (a), there are clopen decreasing bases  $\{U_n : n \in \mathbb{N}\}$  and  $\{V_n : n \in \mathbb{N}\}$  at  $X^{(0, m-1)}$  and  $Y^{(0, m-1)}$  respectively, such that  $U_1 = X$  and  $V_1 = Y$ ,

$$(U_n \setminus U_{n+1})(1, m-2) = 2$$
 and  $(V_n \setminus V_{n+1})(1, m-2) = 2$ .

CLAIM: There is  $l \in \mathbb{N}$  such that  $\kappa(U_l) < \omega$ .

Since  $\kappa(X) = \omega$ ,  $\mathcal{U} = \{X \setminus X^{(n)} : n \in \mathbb{N}\}$  is an open cover of X without finite subcover. Since X is zero-dimensional, there is a disjoint clopen refinement  $\{A_i : i \in \mathbb{N}\}$  of  $\mathcal{U}$ . Since  $X^{(0, m-1)}$  is compact, there is n such that  $X^{(0, m-1)} \subset \bigcup_{i=1}^n A_i$ . There is  $l \in \mathbb{N}$  such that  $U_l \subset \bigcup_{i=1}^n A_i$ , and this l satisfies the claim.

Without loss of generality we may assume that also  $\kappa(V_l) < \omega$ . Now let  $X_1 = U_l$  and  $Y_1 = V_l$ . Notice that

 $X_1(0, n) = Y_1(0, n)$  and  $X_1(1, n) = Y_1(1, n)$  for every  $n \in \mathbb{N}$ ,  $(X \setminus U_l)(1, m-2) = (Y \setminus V_l)(1, m-2) = 2$  and  $(X \setminus U_l)(0, m-1) = (Y \setminus V_l)(0, m-1) = 0.$ 

So by  $(c)_{m-1}$  we have that X and Y can be written as clopen disjoint unions,  $X \setminus U_l = \bigcup_{i=2}^{\infty} X_i$  and  $Y \setminus V_k = \bigcup_{i=2}^{\infty} Y_i$  such that for every  $i \ge 2$  and  $n \in \mathbb{N}$ ,  $X_i(0, n) = Y_i(0, n)$  and  $X_i(1, n) = Y_i(1, n)$  and the lemma has been proved in this case.

case 
$$(b)_m$$
:  $X(0, m) = Y(0, m) = 0$ ,  $X(0, m-1) = Y(0, m-1) = 2$ , and

On the  $l_p$ -equivalence of metric spaces

$$X(1, m-1) = Y(1, m-1) = 1.$$

By corollary 4.2.9 (a), X and Y can be written as clopen disjoint unions  $X = A \cup B$ and  $Y = C \cup D$  such that A(0, m-1) = C(0, m-1) = 1 and B(1, m-1) = D(1, m-1) = 0. By the remark following corollary 4.2.9 we now have in cases of scattered height  $\omega$  by case  $(a)_m$  or by case  $(f)_{m-1}$ , the desired decomposition of X and Y.

case  $(c)_m$ : X(0, m) = Y(0, m) = 0, X(0, m-1) = Y(0, m-1) = 2, and X(1, m-1) = Y(1, m-1) = 2.

We have by corollary 4.2.7 (a), X and Y are clopen disjoint unions  $X = \bigcup_{i=1}^{\infty} A_i$  and  $Y = \bigcup_{i=1}^{\infty} B_i$  such that  $A_i(0, m-1) = B_i(0, m-1) = A_i(1, m-1) = B_i(1, m-1) = 1$ . By case (a)<sub>m</sub> (applied in cases where  $A_i$  or  $B_i$  has scattered height  $\omega$ ), we have the desired decomposition of X and Y.

case 
$$(d)_m$$
:  $X(1, m) = Y(1, m) = 0$ ,  $X(1, m-1) = Y(1, m-1) = 1$ , and  
 $X(0, m) = Y(0, m) = 1$ .

The proof of this case is almost the same as the proof of case  $(a)_m$ . Instead of corollary 4.2.5 (a), we use corollary 4.2.5 (b), and instead of  $(c)_{m-1}$  we use  $(f)_{m-1}$ .

case  $(e)_m$ : X(1, m) = Y(1, m) = 0, X(1, m-1) = Y(1, m-1) = 2, and X(0, m) = Y(0, m) = 1.

The proof is almost the same as the proof of case  $(b)_m$ . Instead of corollary 4.2.9 (a), we use corollary 4.2.9 (b), instead of  $(a)_m$  we use  $(d)_m$  and instead of  $(f)_{m-1}$  we use  $(c)_m$ .

case 
$$(f)_m$$
:  $X(1, m) = Y(1, m) = 0$ ,  $X(1, m-1) = Y(1, m-1) = 2$ , and  
 $X(0, m) = Y(0, m) = 2$ .

The proof is almost the same as the proof of case  $(c)_m$ . Instead of corollary 4.2.7 (a), we use corollary 4.2.7 (b) and instead of  $(a)_m$  we use  $(d)_m$ .

This completes the proof of this lemma.

**4.2.16 THEOREM:** Let X and Y be countable metric spaces such that  $\kappa(X) = \omega$ ,  $\kappa(Y) \le \omega$  and for every  $n \in \mathbb{N} \cup \{0\}$ , X(0, n) = Y(0, n) and X(1, n) = Y(1, n). If X is a space satisfying (a), then X and Y are  $l_p$ -equivalent.

**PROOF:** This follows directly from theorem 4.2.14, lemmas 2.3.7 and 4.2.15.

160

This theorem completes the case for spaces satisfying (a). So now we only have to consider spaces satisfying (b). Again a decomposition of these spaces is needed. This will be given in a claim in the following:

**4.2.17 THEOREM:** Let X and Y be countable metric spaces such that  $\kappa(X) = \kappa(Y) = \omega$  and both satisfying (b). Then X and Y are  $l_p$ -equivalent.

#### **PROOF:** We begin with the following:

CLAIM: We can write  $X = \bigcup_{i=1}^{\infty} X_i$  and  $Y = \bigcup_{i=1}^{\infty} Y_i$  as clopen disjoint unions, such that there are sequences  $(n_i)_{i \in \mathbb{N}}$  and  $(m_i)_{i \in \mathbb{N}}$  such that  $n_i + 1 < m_i$ ,  $m_i + 1 < n_{i+1}$ ,  $X_i(1, n_i) \neq 0$ ,  $X_i(1, n_{i+1}) = 0$ ,  $Y_i(1, m_i) \neq 0$  and  $Y_i(1, m_{i+1}) = 0$ .

It is easily seen that  $\{X \setminus X^{(1,n)} : n \in \mathbb{N}\}$  is an open cover of X without finite subcover. er. Since X is countable, there is a clopen disjoint refinement  $\{A_i : i \in \mathbb{N}\}$  of this cover. We may assume that there exists a strictly increasing sequence  $(k_i)_{i \in \mathbb{N}}$  of natural numbers such that for each  $i \in \mathbb{N}$ ,  $A_i(1, k_i) \neq 0$  and  $A_i(1, k_i + 1) = 0$  (note that X satisfies condition (b), so take unions of the  $A_i$ 's). In the same way Y can be written as a clopen disjoint union  $Y = \bigcup_{i=1}^{\infty} B_i$  such that there are  $l_1 < l_2 \cdots$  with  $B_i(1, l_i) \neq 0$  and  $B_i(1, l_i + 1) = 0$  for each  $i \in \mathbb{N}$ . Now let  $(n_i)_{i \in \mathbb{N}}$  and  $(m_i)_{i \in \mathbb{N}}$  be subsequences of  $(k_i)_{i \in \mathbb{N}}$  and  $(l_i)_{i \in \mathbb{N}}$ , respectively, such that  $n_i + 1 < m_i, m_i + 1 < n_{i+1}$ . By letting  $X_i$  be a appropriate finite union of the  $A_i$ 's and the same for the  $Y_i$ 's, we are done.

Let  $Z = X_1 \oplus Y_1 \oplus X_2 \oplus Y_2 \oplus \cdots$ .

Because  $n_i + 1 < m_i$ ,  $(X_i \oplus Y_i)(0, n) = Y_i(0, n)$  and  $(X_i \oplus Y_i)(1, n) = Y_i(1, n)$  for every  $n \in \mathbb{N} \cup \{0\}$ . Both  $X_i \oplus Y_i$  and  $Y_i$  satisfy (a), so by theorem 4.2.14 or theorem 4.2.16,  $C_p(X_i \oplus Y_i) \sim C_p(Y_i)$ , so that  $C_p(Z) \sim C_p(Y)$ . By interchanging the role of X and Y we also have  $C_p(Z) \sim C_p(X)$ . We conclude that  $C_p(X) \sim C_p(Y)$ .  $\Box$ 

Since we have considered all possible cases we can now formally state the result announced at the beginning of this section.

**4.2.18 THEOREM:** Let X and Y be infinite countable metric spaces, such that  $\kappa(X)$ ,  $\kappa(Y) \le \omega$ . Then X and Y are  $l_p$ -equivalent if and only if for every  $n \in \mathbb{N} \cup \{0\}$ , X(0, n) = Y(0, n) and X(1, n) = Y(1, n).

**PROOF:** This follows immediately from theorems 4.2.14, 4.2.16, 4.2.17 and 4.1.13.

The question naturally arises whether theorem 4.2.18 can be generalized to all countable metric spaces. One is tempted to conjecture the following:

Let X and Y be countable metric spaces. Then X and Y are  $l_p$ -equivalent if and only if for every prime component  $\alpha$  and ordinal  $\beta$  we have  $X(\alpha, \beta) = Y(\alpha, \beta)$ .

In section 4.3 we will show that this conjecture is false.

In this section we saw that an infinite countable metric space X, with scattered height less than or equal to  $\omega$ , satisfies one of the conditions in lemma 4.2.3 or for every  $n \in \mathbb{N}$ , X(0, n) = X(1, n) = 2. The question remains whether each of the considered classes is non-empty. We will prove that in each case there are  $\omega$ -many spaces satisfying the given conditions (except for case (g) in lemma 4.2.3 of course).

For convenience, for every  $n \in \mathbb{N}$  define  $S_n = [1, \omega^n]$ . Let X be a space. We define T(X) to be the space obtained from T by replacing each isolated point of T by a copy of X. Each copy of X will then be clopen in T(X). Inductively we define  $T^k(X) = T(T^{k-1}(X))$  for k > 1, and let  $T^0(X) = X$ . Similarly  $S_k(X)$  will be the space obtained from  $S_k$  by replacing each isolated point of  $S_k$  by a copy of X.

**4.2.19 LEMMA:** Let X be a non-discrete space. Then for every  $n \in \mathbb{N}$  and  $m \in \mathbb{N} \cup \{0\}, S_n(X)(0, m) = X(0, m)$  and  $S_n(X)(1, m) = X(1, m)$ .

**PROOF:** Since X is non-empty we have for every  $n \in \mathbb{N}$ ,  $(S_n(X))^{(1)} = S_n(X^{(1)})$ . Since X is non-discrete, we have

 $S_n(X)(0, 0) = X(0, 0)$  and  $S_n(X)(1, 0) = X(1, 0)$ .

Since  $S_n$  is compact, we also have for every  $m \in \mathbb{N}$ ,

 $S_n(X)(0, m) = X(0, m)$  and  $S_n(X)(1, m) = X(1, m)$ .

If X is a scattered space we have for  $n \neq m$  that  $S_n(X)$  and  $S_m(X)$  are not homeomorphic. So this lemma implies that we only have to give one countable metric space for each of the cases mentioned above, since the lemma then immediately gives  $\omega$ -many. For every  $n \in \mathbb{N}$  define  $X_n = T^{n-1}(T)$ , and  $Y_n = T^{n-1}(S_1)$ . Let  $X_0$  be any onepoint space.

**4.2.20 PROPOSITION:** For every  $n \in \mathbb{N}$ , we have

(a)  $Y_n$  satisfies the conditions in lemma 4.2.3 (a)<sub>n</sub>,

- (b)  $Y_n \bigoplus (X_{n-1} \times \mathbb{N})$  satisfies the conditions in lemma 4.2.3 (b)<sub>n</sub>,
- (c)  $Y_n \times \mathbb{N}$  satisfies the conditions in lemma 4.2.3 (c)<sub>n</sub>,
- (d)  $X_n$  satisfies the conditions in lemma 4.2.3 (d)<sub>n</sub>,
- (e)  $X_n \bigoplus (Y_n \times \mathbb{N})$  satisfies the conditions in lemma 4.2.3 (e)<sub>n</sub>,
- (f)  $X_n \times \mathbb{N}$  satisfies the conditions in lemma 4.2.3 (f)<sub>n</sub>, and
- (g)  $Y = \bigoplus_{n=1}^{\infty} Y_n$  satisfies Y(0, m) = Y(1, m) = 2 for each  $m \in \mathbb{N}$ .

#### **PROOF:** We start this proof with the following

CLAIM: Let X be a non-discrete space, and let  $m \in \mathbb{N} \cup \{0\}$ . If X(0, m) = 1, then T(X)(0, m + 1) = 1, and if X(1, m) = 1, then T(X)(1, m + 1) = 1.

Notice that since X is non-empty,  $T(X)^{(0,m)} = T(X^{(0,m)})$ . Furthermore because  $X^{(0,m)}$  is compact and non-empty,  $T(X)^{(0,m+1)}$  contains only one point, hence T(X)(0, m+1) = 1. Since X is a non-discrete space,  $T(X)^{(1)} = T(X^{(1)})$ , so the second part follows from the fist part. This proves the claim.

Since  $Y_1 = S_1$  satisfies the conditions in lemma 4.2.3  $(a)_1$ , we have by the claim that  $Y_n$  satisfies the conditions in lemma 4.2.3  $(a)_n$ . We can prove (d) similarly. Case (c) follows easily from (a) and case (f) easily follows from (d). It is easily seen that  $Y_1 \bigoplus (X_0 \times \mathbb{N})$  satisfies the conditions in lemma 4.2.3  $(b)_1$ . For n > 1, case (b) follows from (a) and (f). Case (e) is a combination of (d) and (c). Finally (g) follows from (d) since for each n > 2,  $Y_n(0, n-2) = Y_n(1, n-2) = 2$ .

Of course the spaces constructed above are not the only possible ones. In fact one can replace isolated points in T by other countable metric spaces to obtain more examples.

## §4.3. More $l_p$ -equivalent properties of metric spaces

The  $l_p$ -equivalent pairs found in section 4.1 do not provide a complete isomorphical classification for the function spaces  $C_p(X)$ , with X countable and metric. In this section we present two different types of  $l_p$ -equivalent properties which show that an isomorphical classification for these function spaces must be more complicated than the one derived in section 4.2, where we dealt with countable metric spaces of scattered height less than or equal to  $\omega$ .

Let  $\alpha \ge \omega$  be a countable prime component and let  $\beta$  be a countable ordinal. Recall from section 4.1 that for any space *X*,

$$X^{\ll\alpha, \beta \gg} = \bigcap_{\gamma < \alpha} X^{(\gamma, \beta)}.$$

Let  $\alpha \ge \omega$  be a countable prime component and let  $\gamma$ ,  $\beta$  be countable ordinals. By  $X^{\ll \alpha, \beta, \gamma \gg}$  we denote the set  $X^{\{\gamma\}}$  with respect to the pair  $(X, X^{\ll \alpha, \beta \gg})$ . Notice that if  $\gamma$  is a successor, say  $\gamma = \delta + 1$ , then we have  $X^{\ll \alpha, \beta, \gamma \gg} = (X^{\ll \alpha, \beta, \delta \gg})^{(0, 1)}$ .

The numbers  $X \ll \alpha$ ,  $\beta$ ,  $\gamma \gg$  are defined for ordinals  $\alpha$ ,  $\beta$  and  $\gamma$  similarly to the numbers  $X(\alpha, \beta)$  as follows:

 $X \ll \alpha, \beta, \gamma \gg = 0$  if and only if  $X^{\ll \alpha, \beta, \gamma \gg} = \emptyset$ ,

 $X \ll \alpha$ ,  $\beta$ ,  $\gamma \gg = 1$  if and only if  $X^{\ll \alpha, \beta, \gamma \gg}$  is non-empty and compact, and

 $X \ll \alpha, \beta, \gamma \gg = 2$  if and only if  $X^{\ll \alpha, \beta, \gamma \gg}$  is not compact.

For a subset A of a space X we will denote by  $C_{0,A}(X)$  the subspace  $\{f \in C(X) : f(A) = \{0\}\}$  of  $C_0(X)$ .

**4.3.1 PROPOSITION:** Let X and Y be zero-dimensional separable metric  $l_p$ -equivalent spaces and  $\alpha \ge \omega$  a countable prime component. Then  $(X, X^{\ll \alpha, 1\gg})$  and  $(Y, Y^{\ll \alpha, 1\gg})$  are  $l_p$ -equivalent pairs.

**PROOF:** Let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism. It suffices to prove that  $(X, X^{\ll \alpha, 1\gg})$  is  $(\phi, 0)$ -relative to  $(Y, Y^{\ll \alpha, 1\gg})$ . Let U and V be open subsets of X and W an open subset of Y, such that

 $(\operatorname{supp} U) \cap W = \emptyset$ , and  $\operatorname{supp} W \subset U \cup V$ ,

Suppose  $W \cap Y^{\ll \alpha, 1 \gg} \neq \emptyset$  and  $\overline{V} \cap X^{\ll \alpha, 1 \gg} = \emptyset$ .

Let  $(\gamma_n)_{n \in \mathbb{N}}$  be a strictly increasing sequence of ordinals such that  $\gamma_n \to \alpha$   $(n \to \infty)$ . Let  $y \in W \cap Y^{\ll \alpha, 1\gg}$  and let  $\{W_n \mid n \in \mathbb{N}\}$  be a clopen decreasing base at y in W. For each  $n \in \mathbb{N}$ ,  $y \in W_n^{(\gamma_n, 1)}$ , hence we may assume that for each  $n \in \mathbb{N}$  we can find a closed copy of  $[1, \omega^{\gamma_n}] \times \mathbb{N}$  in  $W_n \setminus W_{n+1}$ . By lemma 4.1.4, there is a discrete clopen family  $\{A_i : i \in \mathbb{N}\}$  such that  $V \subset \bigcup_{i=1}^{\infty} A_i$  and such that for every  $i \in \mathbb{N}$ , there is  $\gamma_i < \alpha$  with  $A_i^{(\gamma_i, 1)} = \emptyset$ . By corollary 1.2.15 (a) and lemma 1.2.10, there is  $p \in \mathbb{N}$  such that

$$\operatorname{supp} W_p \cap \bigcup_{i>p} A_i = \emptyset.$$

Let  $A = \bigcup_{i=1}^{p} A_i$ . Then there is  $\gamma < \alpha$  such that  $A^{(\gamma,1)} = \emptyset$ . By corollary 4.1.5, there is a

discrete clopen family  $\{B_i : i \in \mathbb{N}\}$  such that  $A = \bigcup_{i=1}^{\infty} B_i$  and such that for every  $i \in \mathbb{N}$ ,  $B_i^{(\gamma)}$  is compact. By corollary 1.2.15 (a) and lemma 1.2.10 there is k > p such that

$$\operatorname{supp} W_k \cap \bigcup_{i>k} B_i = \emptyset.$$

Let  $B = \bigcup_{i=1}^{k} B_i$ . Then  $B^{(\gamma)}$  is compact. We now have

$$(\operatorname{supp} U) \cap W_k = \emptyset$$
, and  
 $\operatorname{supp} W_k \subset U \cup B$ ,

Since  $W_k^{(\alpha)} \neq \emptyset$ , we have by proposition 4.1.12, that  $B^{(\alpha)} \neq \emptyset$ , so that  $B^{(\gamma)} \neq \emptyset$ . This implies that supp  $B^{(\gamma)}$  is a non-empty compactum in Y.

Since  $\overline{\operatorname{supp} B^{(\gamma)}}$  is compact, this implies that there exists a closed copy  $L_n$  of  $[1, \omega^{\gamma_n}]$ in  $W_n$  such that  $\overline{\operatorname{supp} B^{(\gamma)}} \cap L_n = \emptyset$ . If  $\operatorname{supp} L_n \cap B = \emptyset$ , then  $\operatorname{supp} L_n \subset U$  hence by proposition 1.4.3,  $L_n \subset \operatorname{supp} \operatorname{supp} L_n \subset \operatorname{supp} U$ . This implies  $L_n = \emptyset$ , contradiction. Hence  $\operatorname{supp} L_n \cap B \neq \emptyset$ . Let

$$M = (\operatorname{supp} B^{(\gamma)} \cap W_k) \cup \{y\},\$$
  

$$L_{\alpha} = \bigcup_{n > k} L_n \cup \{y\},\$$
  

$$L = L_{\alpha} \cup M, \text{ and}$$
  

$$K = (\operatorname{supp} L \cap B) \cup B^{(\gamma)}.$$

By lemma 4.1.11, there is a continuous linear function  $\eta_1: C_0(L) \to C_0(Y)$  such that for each  $f \in C(L)$ ,

$$\eta_1(f) \mid L = f \text{ and } \eta_1(f)(Y \setminus W_k) = \{0\}.$$

Again by lemma 4.1.11, there is a continuous linear function  $\eta_2: C_0(K) \to C_0(X)$  such that for each  $f \in C(K)$ ,

$$\eta_2(f) \mid K = f \text{ and } \eta_2(f)(X \setminus B) = \{0\}.$$

Define

$$\theta: C_{0,M}(L) \to C_{0,B}(M) \text{ by } \theta(f) = \phi^{-1}(\eta_1(f)) \mid K, \text{ and}$$
  
$$\psi: C_0(K) \to C_0(L) \text{ by } \psi(f) = \phi(\eta_2(f)) \mid L.$$

Let  $f \in C_{0,M}(L)$ . Then  $\eta_1(f)(Y \setminus W_k) = \{0\}$ . Since  $\overline{\operatorname{supp} B^{(\gamma)}} \cap W_k \subset M$ , we have that  $\eta_1(f)(\operatorname{supp} B^{(\gamma)}) = \{0\}$ . So by corollary 1.4.2,  $\phi^{-1}(\eta_1(f))(B^{(\gamma)}) = \{0\}$ , hence  $\theta(f)(B^{(\gamma)}) = \{0\}$ . Now it is easily seen that both  $\theta$  and  $\psi$  are well-defined continuous linear functions.

CLAIM:  $\theta$  is a linear embedding.

It suffices to prove that for  $f \in C_{0,M}(L)$ , we have  $\psi(\theta(f)) = f$ . Suppose to the contrary that for some  $f \in C_{0,M}(L)$ , we have  $\psi(\theta(f)) \neq f$ . This implies that

 $\phi(\eta_2(\theta(f))) \mid L \neq \eta_1(f) \mid L.$ 

Then by corollary 1.4.2,

 $\eta_2(\theta(f)) | \operatorname{supp} L \neq \phi^{-1}(\eta_1(f)) | \operatorname{supp} L.$ 

Now  $\eta_2(\theta(f)) \equiv 0$  on  $X \setminus B$ , and  $\eta_1(f) \equiv 0$  on  $Y \setminus W_k$ . Since supp  $U \subset Y \setminus W_k$ , we have by corollary 1.4.2 that  $\phi^{-1}(\eta_1(f)) \equiv 0$  on U. Since supp  $L \subset U \cup B$ , we have

 $\eta_2(\boldsymbol{\theta}(f)) \mid K \neq \phi^{-1}(\eta_1(f)) \mid K.$ 

Hence  $\theta(f) \neq \theta(f)$ . Contradiction. This proves the claim.

Since  $L_{\alpha} \cap M = \{y\}$  it follows that  $C_{0,M}(L)$  is linearly homeomorphic to  $C_{0,\{y\}}(L_{\alpha})$ . Note that  $L_{\alpha} \approx [1, \omega^{\alpha}]$ . By lemma 2.3.9 and remark 2.3.10,  $C_{0,\{y\}}(L_{\alpha})$  is linearly homeomorphic to  $C_{0}(L_{\alpha})$ . By lemma 2.2.9, the space  $B/B^{(\gamma)}$  obtained from B by identifying  $B^{(\gamma)}$  to a single point a has scattered height  $\gamma+1$ . Let  $Z = K/B^{(\gamma)}$ . Then  $\kappa(Z) \leq \gamma+1$ , so there are  $\beta \leq \gamma$  and  $m \in \mathbb{N}$  such that  $Z \approx [1, \omega^{\beta} \cdot m]$ . By lemma 2.3.3 and remark 2.3.10,  $C_{0,\{a\}}(Z)$  is linearly homeomorphic to  $C_{0,B}(\gamma)(K)$ . Furthermore by lemma 2.3.9 and remark 2.3.10,  $C_{0,\{a\}}(Z)$  is linearly homeomorphic to  $C_{0}(L_{\alpha})$  to  $C_{0}(Z)$ . Now the claim implies that we have a linear embedding from  $C_{0}(L_{\alpha})$  to  $C_{0}(Z)$ , or equivalently a linear embedding from  $C_{0}[1, \omega^{\alpha}]$  to  $C_{0}[1, \omega^{\gamma}]$ . Since  $\gamma < \alpha$  we have a contradiction by lemma 2.6.7.  $\Box$ 

**4.3.2 COROLLARY:** Let X and Y be  $l_p$ -equivalent zero-dimensional separable metric spaces, let  $\alpha \ge \omega$  be a countable prime component and let  $\gamma$  be a countable ordinal. Then

(a)  $X^{\ll\alpha, 1, \gamma \gg} = \emptyset$  if and only if  $Y^{\ll\alpha, 1, \gamma \gg} = \emptyset$ ,

(b)  $X^{\ll\alpha, 1, \gamma \gg}$  is compact if and only if  $Y^{\ll\alpha, 1, \gamma \gg}$  is compact, and

(c)  $X^{\ll\alpha, 1, \gamma \gg}$  is locally compact if and only if  $Y^{\ll\alpha, 1, \gamma \gg}$  is locally compact.

**PROOF:** This follows directly from proposition 4.3.1 and theorem 4.1.7.

We will now give an example of two countable metric metric spaces X and Y such that for every pair of ordinals  $\alpha$ ,  $\beta$  with  $\alpha$  a countable prime component,  $X(\alpha, \beta) = Y(\alpha, \beta)$ , and such that X and Y are not  $l_p$ -equivalent.

Let X be the space obtained from T by replacing each  $(i, j) \in T$  by  $[1, \omega^i]$   $(i, j \in \mathbb{N})$ and let  $Y = T([1, \omega]) \oplus [1, \omega^{\omega}]$  (for definitions see section 4.2). Let p be any point. Then

 $X^{(1)} \approx X \text{ and } Y^{(1)} \approx T \bigoplus [1, \omega^{\omega}],$   $X^{(0, 1)} \approx \{p\} \text{ and } Y^{(0, 1)} \approx \{p\},$   $X^{(1, 1)} \approx \{p\} \text{ and } Y^{(1, 1)} \approx \{p\}, \text{ and }$  $X^{(\omega)} \approx \{p\} \text{ and } Y^{(\omega)} \approx \{p\}.$ 

So for every pair of ordinals  $\alpha$ ,  $\beta$  with  $\alpha$  a countable prime component, we have  $X(\alpha, \beta) = Y(\alpha, \beta)$ . However by corollary 4.3.2, X and Y are not  $\ell_p$ -equivalent since

 $X^{\ll\omega, 1\gg} \approx \{p\}$  and  $Y^{\ll\omega, 1\gg} = \emptyset$ .

Note that the scattered height of X and Y is  $\omega + 1$ , so a classification such as in section 4.2 does not hold when we consider countable metric spaces with scattered height one higher than  $\omega$ . In this example only  $\alpha = 0$ ,  $\alpha = 1$ ,  $\alpha = \omega$ ,  $\beta = 0$  and  $\beta = 1$  are necessary. It is possible to build more complex examples in which higher ordinals are involved.

Question 1: Let  $\alpha \ge \omega$  be a countable prime component, and let X and Y be  $l_p$ -equivalent zero-dimensional separable metric spaces. For which ordinals  $\beta$  do we have that  $(X, X^{\ll \alpha, \beta \gg})$  and  $(Y, Y^{\ll \alpha, \beta \gg})$  are  $l_p$ -equivalent pairs?

By propositions 4.1.12 and 4.3.1 we have a positive answer to this question for  $\beta = 0$  and  $\beta = 1$ . We conjecture that this question has a positive answer for all ordinals  $\beta$ .

By corollary 4.3.2 and question 1 one could think that for two countable metric spaces X and Y which satisfy  $X(\alpha, \beta) = Y(\alpha, \beta)$  and  $X \ll \alpha, \beta, \gamma \gg = Y \ll \alpha, \beta, \gamma \gg$ , for all ordinals  $\alpha$ ,  $\beta$  and  $\gamma$  with  $\alpha$  a prime component, we have that X and Y are  $l_p$ -equivalent. In the sequel we will give an example which shows that this is not the case. We first need a new notion which will be used in proposition 4.3.7. We present it in a general setting since it seems to be interesting in itself.

Let X and Y be spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a linear function. Let A be a non-empty closed subset of X and U a neighborhood of A in X. Let B be a non-empty closed subset of Y and V a neighborhood of B in Y. Finally let  $m \in \mathbb{N}$ . We say that the triple (U, V, m) is **relatively bounded** with respect to the triple  $(A, B, \phi)$  whenever for each  $g \in C(X)$  satisfying

(a)  $g((X \setminus U) \cup A) = \{0\}$ , (b)  $g(U) \subset (-1/m, 1/m)$ , and (c)  $\phi(g)(B) = \{0\}$ ,

we have

(d)  $\phi(g)(V) \subset (-1, 1)$ .

**4.3.3 LEMMA:** Let X and Y be spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a continuous linear function. Let A be a non-empty closed subset of X and U a neighborhood of A in X. Let B be a non-empty closed subset of Y and V a neighborhood of B in Y. Let  $m \in \mathbb{N}$ . Suppose (U, V, m) is relatively bounded with respect to  $(A, B, \phi)$ . If  $U_1$  is a neighborhood of A such that  $U_1 \subset U$ ,  $V_1$  is a neighborhood of B such that  $V_1 \subset V$  and  $k \ge m$ , then  $(U_1, V_1, k)$  is also relatively bounded with respect to  $(A, B, \phi)$ .

**PROOF:** Let  $g \in C(X)$  be such that  $g((X \setminus U_1) \cup A) = \{0\}$ ,  $g(U_1) \subset (-1/k, 1/k)$ , and  $\phi(g)(B) = \{0\}$ . Then obviously  $g((X \setminus U) \cup A) = \{0\}$ . For  $z \in U \setminus U_1$ , g(z) = 0, hence  $g(U) \subset (-1/m, 1/m)$ . So  $\phi(g)(V) \subset (-1, 1)$ , hence  $\phi(g)(V_1) \subset (-1, 1)$ .

**4.3.4 LEMMA:** Let X and Y be metric spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a continuous linear function. Let A be a non-empty compact subset of X and let B be a non-empty compact subset of Y. Then there are a neighborhood U of A in X, a neighborhood V of B in Y and  $m \in \mathbb{N}$  such that (U, V, m) is relatively bounded with respect to  $(A, B, \phi)$ .

**PROOF:** Suppose the lemma is false. Let  $\{U_n\}_{n \in \mathbb{N}}$  be an open base at A in X such that for each  $n \in \mathbb{N}$ ,  $\overline{U}_{n+1} \subset U_n$ . Let  $\{V_n\}_{n \in \mathbb{N}}$  be an open decreasing base at B in Y. By induction we construct  $\{k_i : i \in \mathbb{N}\} \subset \mathbb{N}$ ,  $\{g_i : i \in \mathbb{N}\} \subset C(X)$  and  $\{y_i : i \in \mathbb{N}\} \subset Y$  such that

(1)  $1 = k_1 < k_2 < k_3 < \cdots$ ,

and for every  $i \in \mathbb{N}$ ,

(2)  $g_i((X \setminus U_{k_i}) \cup A) = \{0\},$ (3) for every  $j \le i$ ,  $g_j(U_{k_i}) \subset (-\frac{1}{i^2}, \frac{1}{i^2}),$ (4)  $\phi(g_i)(B) = \{0\},$ (5)  $y_i \in V_{k_i}$  and  $|\phi(g_i)(y_i)| \ge 1,$ (6) for every j < i,  $\phi(g_j)(V_{k_i}) \subset (-\frac{1}{2(i-1)}, \frac{1}{2(i-1)}),$  and (7) for every j < i,  $\phi(g_i)(y_j) = 0.$ 

Let  $k_1 = 1$ . By assumption there is  $g_1 \in C(X)$  such that  $g_1((X \setminus U_{k_1}) \cup A) = \{0\}$ ,  $g_1(U_{k_1}) \subset (-1, 1), \ \phi(g_1)(B) = \{0\}$  and  $\phi(g_1)(V_{k_1}) \not\subset (-1, 1)$ . Let  $y_1 \in V_{k_1}$  be such that  $|\phi(g_1)(y_1)| \ge 1.$ 

Let  $m \ge 1$  and suppose we found  $k_1, \ldots, k_m, g_1, \ldots, g_m$  and  $y_1, \ldots, y_m$ . For each  $j \le m$  we have  $g_j(A) = 0$  and  $\phi(g_1)(B) = \{0\}$ . By continuity of  $g_1, \ldots, g_m$ , there is  $k_{m+1} > k_m$  such that

(8) for each  $j \le m$ ,  $g_j(U_{k_{m+1}}) \subset (-\frac{1}{(m+1)^2}, \frac{1}{(m+1)^2})$ , (9)  $U_{k_{m+1}} \cap \text{supp} \{y_1, \dots, y_m\} \subset A$ , and (10) for each  $j \le m$ ,  $\phi(g_j)(V_{k_{m+1}}) \subset (-\frac{1}{2m}, \frac{1}{2m})$ .

Again by assumption there is  $g_{m+1} \in C(X)$  such that  $g_{m+1}((X \setminus U_{k_{m+1}}) \cup A) = \{0\}$ ,

$$g_{m+1}(U_{k_{m+1}}) \subset (-\frac{1}{(m+1)^2}, \frac{1}{(m+1)^2})$$

 $\phi(g_{m+1})(B) = \{0\}$  and  $\phi(g_{m+1})(V_{k_{m+1}}) \not\subset (-1, 1)$ . Let  $y_{m+1} \in V_{k_{m+1}}$  be such that  $|\phi(g_{m+1})(y_{m+1})| \ge 1$ . To complete the inductive construction we have to verify (7) for i = m+1. Since  $g_{m+1}(\operatorname{supp}(y_j)) = \{0\}$  for  $j \le m$ , we have by corollary 1.4.2,  $\phi(g_{m+1})(y_j) = 0$ . This completes the inductive construction.

Now let  $g = \sum_{i=1}^{\infty} g_i$ . We will show that  $g \in C(X)$ . For  $i \in \mathbb{N}$  and  $z \notin \overline{U}_{k_{i+1}}$ , we have by (2),  $g(z) = \sum_{j=1}^{i} g_j(z)$ . So  $g \mid X \setminus A$  is well-defined and continuous. It remains to prove that g is continuous at points of A. Since g(A) = 0 this follows from the fact that for every  $i \in \mathbb{N}$ ,  $g(U_{k_i}) \subset (-1/i, 1/i)$ . Indeed let  $z \in U_{k_i} \setminus A$ . Then there is  $j \ge i$  such that  $z \in U_{k_i} \setminus U_{k_{i+1}}$ . Then by (2),  $g(z) = \sum_{k=1}^{i} g_j(z)$ , and hence by (3),

$$|g(z)| \le \sum_{k=1}^{j} |g_k(z)| < j \cdot \frac{1}{j^2} \le \frac{1}{i}.$$

We conclude that  $g \in C(X)$ . So  $\phi(g) = \sum_{i=1}^{\infty} \phi(g_i) \in C(Y)$ . Since B is compact we may assume that  $y_n \to b$   $(n \to \infty)$  for some  $b \in B$ . By (4),  $\phi(g)(b) = 0$ , hence  $\phi(g)(y_n) \to 0$   $(n \to \infty)$ . However for every  $i \in \mathbb{N}$ , we have

$$\begin{aligned} |\phi(g)(y_i)| &= |\sum_{j=1}^{\infty} \phi(g_j)(y_i)| \\ &= |\sum_{j=1}^{i-1} \phi(g_j)(y_i) + \phi(g_i)(y_i) + \sum_{j=i+1}^{\infty} \phi(g_j)(y_i)| \\ &= |\sum_{j=1}^{i-1} \phi(g_j)(y_i) + \phi(g_i)(y_i)| \qquad \text{by (7)} \\ &\ge |\phi(g_i)(y_i)| - \sum_{j=1}^{i-1} |\phi(g_j)(y_j)| \\ &\ge 1 - (i-1) \cdot \frac{1}{2(i-1)} = \frac{1}{2} \qquad \text{by (5) and (6)} \end{aligned}$$

This contradiction proves the lemma.

**4.3.5 COROLLARY:** Let X and Y be metric spaces and let  $\phi: C_p(X) \to C_p(Y)$  be a linear homeomorphism. Let A be a non-empty compact subset of X and let B be a non-empty compact subset of Y. Then there are a neighborhood U of A in X, a neighborhood V of B in Y and  $m \in \mathbb{N}$  such that (U, V, m) is relatively bounded with respect to  $(A, B, \phi)$  and (V, U, m) is relatively bounded with respect to  $(B, A, \phi^{-1})$ .

**PROOF:** By lemma 4.3.4, there are neighborhoods  $U_1$  and  $U_2$  of A in X, neighborhoods  $V_1$  and  $V_2$  of B in Y, and  $m_1, m_2 \in \mathbb{N}$  such that  $(U_1, V_1, m)$  is relatively bounded with respect to  $(A, B, \phi)$  and  $(V_1, U_1, m)$  is relatively bounded with respect to  $(B, A, \phi^{-1})$ . Let  $m = \max(m_1, m_2)$ ,  $U = U_1 \cap U_2$  and  $V = V_1 \cap V_2$ . Then by lemma 4.3.3, this U, V and m satisfy the conditions in the lemma.  $\square$ 

Let X and Y be spaces. Let E and F be linear subspaces of  $C_0(X)$  resp.  $C_0(Y)$ . Let  $m \in \mathbb{N}$ . A linear function  $\phi: E \to F$  is said to be a *linear m-embedding* whenever  $\phi$  is an embedding and

(1) if  $f \in E$  satisfies  $f(X) \subset (-1/m, 1/m)$ , then  $\phi(f)(Y) \subset (-1, 1)$ , and (2) if  $f \in E$  satisfies  $\phi(f)(Y) \subset (-1/m, 1/m)$ , then  $f(X) \subset (-1, 1)$ .

This definition is comparable with the notion of a linear k-mapping introduced in section 2.3. Since we need linear m-embeddings only in a very specific situation our definition is not in the most general form as was the case in section 2.3.

**4.3.6 LEMMA:** Let X and Y be compact spaces. Let  $x \in X$  and let  $\{U_n : n \in \mathbb{N}\}$  be a clopen decreasing base at  $x \in X$  such that  $U_1 = X$ . Let  $y \in Y$  and let  $\{V_n : n \in \mathbb{N}\}$  be a clopen decreasing base at  $y \in Y$  such that  $V_1 = Y$ . Let  $k \in \mathbb{N}$ . Suppose that for every  $n \in \mathbb{N}$ ,

$$\phi_n: C_0(U_n \setminus U_{n+1}) \to C_0(V_n \setminus V_{n+1})$$

is a linear k-embedding. Define  $\phi: C_{0,\{x\}}(X) \rightarrow C_{0,\{y\}}(Y)$  by

 $\phi(f) \mid V_n \setminus V_{n+1} = \phi_n(f \mid U_n \setminus U_{n+1}) \text{ and } \phi(f)(y) = 0.$ 

Then  $\phi$  is a well-defined linear k-embedding.

**PROOF:** As in lemma 4.2.10,  $\phi$  is well-defined and linear. To prove continuity of  $\phi$  it suffices to prove that  $\phi$  is continuous at 0. To this end let  $P \subset Y$  be compact, let  $\varepsilon > 0$  and observe that by linearity of  $\phi$ ,

§4.3. More  $l_p$ -equivalent properties of metric spaces

 $\phi(\langle 0, X, \varepsilon/k \rangle) \subset \langle 0, P, \varepsilon \rangle.$ 

To prove that  $\phi$  is an embedding, let  $P \subset X$  be compact, let  $\varepsilon > 0$  and observe that

 $<0, Y, \varepsilon/k > \cap \phi(C_{0, \{x\}}(X)) \subset \phi(<0, P, \varepsilon). \Box$ 

We will now give an example of two countable metric metric spaces X and Y such that for all ordinals  $\alpha$ ,  $\beta$  and  $\gamma$  with  $\alpha$  a countable prime component, we have  $X(\alpha, \beta) = Y(\alpha, \beta), X \ll \alpha, \beta, \gamma \gg = Y \ll \alpha, \beta, \gamma \gg$ , and X and Y are not  $\ell_p$ -equivalent.

Let X be the space obtained from T by replacing each  $(i, j) \in T$  by  $[1, \omega^i]$   $(i, j \in \mathbb{N})$ . Let  $Y = S([1, \omega^{\omega}))$ . Let x be the unique point in  $X^{(\omega)}$  and y the unique point in  $Y^{(\omega)}$ . Then

 $X^{(1)} \approx X \text{ and } Y^{(1)} \approx Y$   $X^{(0, 1)} = \{x\} \text{ and } Y^{(0, 1)} = \{y\},$   $X^{(1, 1)} = \{x\} \text{ and } Y^{(1, 1)} = \{y\},$   $X^{(\omega)} = \{x\} \text{ and } Y^{(\omega)} = \{y\}, \text{ and }$  $X^{<\omega, 1>} = \{x\} \text{ and } Y^{<\omega, 1>} = \{y\}.$ 

However we have

### **4.3.7 PROPOSITION:** X and Y are not $l_p$ -equivalent.

**PROOF:** Suppose  $\phi: C_p(X) \to C_p(Y)$  is a linear homeomorphism. Let  $\{W_n : n \in \mathbb{N}\}$  be a clopen decreasing base at y in Y such that for each  $n \in \mathbb{N}$ ,  $W_n \setminus W_{n+1}$  contains a clopen copy of  $[1, \omega^{\omega})$  and let  $\{V_n : n \in \mathbb{N}\}$  be a clopen decreasing base at  $\{x\} \cup \text{supp}(y)$  in X. By corollary 4.3.5, there is  $m \in \mathbb{N}$  such that

 $(V_m, W_m, m)$  is relatively bounded with respect to  $(\{x\} \cup \text{supp}(y), \{y\}, \phi)$ , and

 $(W_m, V_m, m)$  is relatively bounded with respect to  $(\{y\}, \{x\} \cup \text{supp}(y), \phi^{-1}),$ 

Notice that  $X \setminus V_m$  is locally compact. So  $X \setminus V_m = \bigcup_{i=1}^{\infty} A_i$  a clopen disjoint union such that for each  $i \in \mathbb{N}$ ,  $A_i$  is compact. By corollary 1.2.15 (a) and lemma 1.2.10, there is  $k \ge m$  such that

 $\operatorname{supp} W_k \cap \bigcup_{i>k} A_i = \emptyset.$ 

Let  $A = \bigcup_{i=1}^{k} A_i$ . Then A is compact. Let K be a clopen copy of  $[1, \omega^{\omega})$  in  $W_k \setminus W_{k+1}$ . Write  $K = \bigcup_{i=1}^{\infty} K_i$  as a clopen disjoint union such that for each  $i \in \mathbb{N}$ ,  $K_i = [1, \omega^i]$ . Since  $A \cup \text{supp}(y)$  is compact, there is by corollary 1.2.15 (a) and lemma 1.2.10, p > k

On the  $l_p$ -equivalence of metric spaces

such that

 $\operatorname{supp}(V_p \cup A \cup \operatorname{supp}(y)) \cap \bigcup_{i>p} K_i = \emptyset.$ 

Note that the scattered height of  $X \setminus V_p$  is less than  $\omega$ . Let  $s \in \mathbb{N}$  be the scattered height of  $X \setminus V_p$ .

Fix i > p. We have

supp  $(A \cup V_p) \cap K_i = \emptyset$ , and supp  $K_i \subset (A \cup V_p) \cup (V_m \setminus V_p)$ .

Let  $L_i = \overline{\operatorname{supp} K_i} \cap (V_m \setminus V_p)$ . Then  $L_i$  is compact. If  $L_i = \emptyset$ , then  $\operatorname{supp} K_i \subset A \cup V_p$ , so by proposition 1.4.3,

$$K_i \subset \operatorname{supp} \operatorname{supp} K_i \subset \operatorname{supp} (A \cup V_p),$$

hence  $K_i = \emptyset$ . Contradiction. So  $L_i$  is a non-empty compactum.

By lemma 4.1.11, there is a continuous linear embedding  $\eta_1: C_0(K_i) \to C_0(Y)$  such that for each  $f \in C(K_i)$  we have

 $\eta_1(f) | K_i = f,$   $\eta_1(f)(Y) \subset \text{conv} (f'(K_i) \cup \{0\}), \text{ and }$  $\eta_1(f)(Y \setminus K_i) = \{0\}.$ 

Again by lemma 4.1.11 there is a continuous linear function  $\eta_2: C_0(K_i) \to C_0(X)$  such that for each  $f \in C(L_i)$  we have

 $\eta_2(f) | L_i = f,$   $\eta_2(f)(X) \subset \text{conv} (f (L_i) \cup \{0\}), \text{ and }$  $\eta_2(f)((X \setminus V_m) \cup V_p) = \{0\}.$ 

Define

 $\theta$ :  $C_0(K_i) \rightarrow C_0(L_i)$  by  $\theta(f) = \phi^{-1}(\eta_1(f)) \mid L_i$ , and  $\psi$ :  $C_0(L_i) \rightarrow C_0(K_i)$  by  $\psi(f) = \phi(\eta_2(f)) \mid K_i$ .

Then  $\theta$  and  $\psi$  are clearly well-defined continuous linear functions.

CLAIM:  $\theta$  is a linear *m*-embedding.

We first prove that  $\theta$  is an embedding. It suffices to prove that for each  $f \in C(K_i)$  we have  $\psi(\theta(f)) = f$ . Suppose there is  $f \in C(K_i)$  such that  $\psi(\theta(f)) \neq f$ . Then

 $\phi(\eta_2(\theta(f))) \mid K_i \neq \eta_1(f) \mid K_i.$ 

By corollary 1.4.2, we then have

172

§4.3. More lp-equivalent properties of metric spaces

 $\eta_2(\theta_{(j)}) | \operatorname{supp} K_i \neq \phi^{-1}(\eta_1(f)) | \operatorname{supp} K_i.$ 

Now  $\eta_2(\theta(f)) \equiv 0$  on  $(X \setminus V_m) \cup V_p$  and  $\eta_1(f) \equiv 0$  on  $X \setminus K_i$ . Since  $\operatorname{supp}(A \cup V_p) \subset X \setminus K_i$ , it follows from corollary 1.4.2, that  $\phi^{-1}(\eta_1(f)) \equiv 0$  on  $A \cup V_p$ . So

 $\eta_2(\theta(f)) | L_i \neq \phi^{-1}(\eta_1(f)) | L_i,$ 

which implies that  $\theta(f) \neq \theta(f)$ , contradiction. We conclude that  $\theta$  is a linear embedding.

To prove that  $\theta$  is a linear *m*-embedding, first let  $f \in C(K_i)$  be such that  $f(K_i) \subset (-1/m, 1/m)$ . Then

$$\eta_1(f)(Y) \subset \operatorname{conv} (f(K_i) \cup \{0\}) \subset (-1/m, 1/m), \text{ and}$$
  
 $\eta_1(f)((Y \setminus W_m) \cup \{y\}) = \{0\}.$ 

Since supp  $(\{x\} \cup \text{supp}(y)) \cap K_i = \emptyset$ , we have

 $\eta_1(f)(\text{supp}(\{x\} \cup \text{supp}(y))) = \{0\},\$ 

so by corollary 1.4.2,  $\phi^{-1}(\eta_1(f))(\{x\} \cup \operatorname{supp}(y)) = \{0\}$ . Since  $(W_m, V_m, m)$  is relatively bounded with respect to  $(\{y\}, \{x\} \cup \operatorname{supp}(y), \phi^{-1})$ , we have  $\phi^{-1}(\eta_1(f))(V_m) \subset (-1, 1)$ . Since  $L_i \subset V_m$ , we have  $\theta(f)(L_i) \subset (-1, 1)$ .

Secondly let  $f \in C(K_i)$  be such that  $\theta(f)(L_i) \subset (-1, 1)$ . Then

 $\eta_2(\theta(f))(X) \subset \operatorname{conv} (\theta(f)(L_i) \cup \{0\}) \subset (-1/m, 1/m), \text{ and} \\ \eta_2(\theta(f))((X \setminus V_m) \cup \{x\} \cup \operatorname{supp} (y)) = \{0\}.$ 

Since  $\eta_2(\theta(f))(\sup p(y)) = \{0\}$ , we have by corollary 1.4.2,  $\phi(\eta_2(\theta(f)))(y) = 0$ . Since  $(V_m, W_m, m)$  is relatively bounded with respect to  $(\{x\} \cup \sup p(y), \{y\}, \phi)$ , we have  $\phi(\eta_2(\theta(f)))(W_m) \subset (-1, 1)$ . Since  $K_i \subset W_m$ , we have  $\psi(\theta(f))(K_i) \subset (-1, 1)$ . By the above  $\psi(\theta(f)) = f$ , so  $f(K_i) \subset (-1, 1)$ . We conclude that  $\theta$  is a linear *m*-embedding.

Since  $L_i \subset X \setminus V_p$  and  $\kappa(X \setminus V_p) = s$ , there is a linear 1-embedding from  $C_0(L_i)$  into  $C_0([1, \omega^{s+1}])$ . By the claim we have for each  $i \in \mathbb{N}$  a linear *m*-embedding from  $C_0([1, \omega^i])$  into  $C_0([1, \omega^{s+1}])$ . then by lemma 4.3.6 we have a linear embedding from  $C_0([1, \omega^{\omega}])$  into  $C_0([1, \omega^{s+2}])$ . This is a contradiction with lemma 2.6.7. We conclude that X and Y are not  $l_p$ -equivalent.  $\square$ 

If we look at the spaces X and Y we see that each neighborhood of  $y \in Y$  contains a closed copy of  $[1, \omega^{\omega})$  and that no neighborhood of  $x \in X$  contains a closed copy of  $[1, \omega^{\omega})$ .

For a space X, let

 $X^{<\alpha>} = \{x \in X : \text{ each neighborhood of } x \text{ contains a closed copy of } [1, \omega^{\alpha})\}.$ 

Obviously  $X^{<\alpha>}$  is a closed subspace of X. If  $X^{<\alpha>} = \emptyset$ , then  $X^{(\alpha)}$  is locally compact. Indeed if  $x \in X^{(\alpha)}$ , then each neighborhood contains a closed copy of  $[1, \omega^{\alpha}] \times \mathbb{N}$  (cf. lemma 4.1.8). But then each neighborhood also contains a closed copy of  $[1, \omega^{\alpha})$ .

Let  $\alpha \ge \omega$  be a countable prime component and let  $\beta$  be a countable ordinal. By  $X^{<\alpha,\beta>}$  we denote the set  $X^{\{\beta\}}$  with respect to the pair  $(X, X^{<\alpha>})$ . Notice that if  $\beta$  is a successor, say  $\beta = \gamma + 1$ , then we have  $X^{<\alpha,\beta>} = (X^{<\alpha,\gamma>})^{(0,1)}$ .

The numbers  $X < \alpha, \beta >$  are defined for ordinals  $\alpha$  and  $\beta$  similarly to the numbers  $X(\alpha, \beta)$  as follows:

 $X < \alpha, \beta > = 0$  if and only if  $X^{<\alpha, \beta >} = \emptyset$ ,

 $X < \alpha, \beta > = 1$  if and only if  $X^{<\alpha, \beta>}$  is non-empty and compact, and

 $X < \alpha$ ,  $\beta > = 2$  if and only if  $X^{<\alpha, \beta>}$  is not compact.

Question 2: Let  $\alpha \ge \omega$  be a countable prime component and let X and Y be  $l_p$ -equivalent separable metric zero-dimensional spaces. Are  $(X, X^{<\alpha>})$  and  $(Y, Y^{<\alpha>}) l_p$ -equivalent pairs?

We conjecture that this question has a positive answer.

## §4.4. Remarks on a conjecture

In sections 4.1 and 4.3 we found several  $l_p$ -equivalent pairs. In this section we conjecture that these  $l_p$ -equivalent pairs together with the conjectured  $l_p$ -equivalent pairs in question 1 and 2 in section 4.3 are sufficient to obtain an isomorphical classification for the function spaces  $C_p(X)$ , for countable metric X. In this section we indicate among other things difficulties that one encounters if one tries to prove the conjecture along the lines of the proof of theorem 4.2.18.

**4.4.1 CONJECTURE:** Let X and Y be infinite countable metric spaces. Then X and Y are  $l_p$ -equivalent if and only if for all countable ordinals  $\alpha$ ,  $\beta$  and  $\gamma$ , where  $\alpha$  is a prime component, we have

(a)  $X(\alpha, \beta) = Y(\alpha, \beta)$ ,

174

(b) 
$$X < \alpha, \beta > = Y < \alpha, \beta >$$
, for  $\alpha \ge \omega$ , and  
(c)  $X \ll \alpha, \beta, \gamma \gg = Y \ll \alpha, \beta, \gamma \gg$ , for  $\alpha \ge \omega$ 

Note that conditions (b) and (c) are trivially true when X and Y are countable metric spaces with scattered height less than or equal to  $\omega$ , so the conjecture is in agreement with theorem 4.2.18 when we restrict ourselves to this class of spaces.

Part of the proof of theorem 4.2.18 depends in a rough way on lemma 4.2.10 and on the decomposition theorems at the beginning of section 4.2. If we want to prove for countable metric spaces satisfying conditions (a), (b) and (c) in the above conjecture, that they are  $l_p$ -equivalent, and we want to follow the same strategy as in section 4.2, some problems appear. For example, it is not clear how to decompose each countable metric space such that lemma 4.2.10 becomes applicable. Such a decomposition should have the property that between "building blocks" we have "linear k-homeomorphisms". A second problem is how we should start an inductive proof such as in the proof of lemma 4.2.13. We will try to show why these problems are real obstacles in the process of proving the conjecture. We first present a lemma which gives connections between the operations that are involved in the conjecture. One could say that this lemma is the "replacement" of lemma 4.2.1.

**4.4.2 LEMMA:** Let X be a space. Let  $\alpha$ ,  $\alpha_1$ ,  $\alpha_2$ ,  $\beta_1$  and  $\beta_2$  be ordinals with  $\alpha$ ,  $\alpha_1$  and  $\alpha_2$  prime components. Then

(a)  $X^{(\alpha, 1)} \subset X^{<\alpha>} \subset X^{(\alpha)}$ , (b)  $X^{(\alpha_1, \alpha_2 - \alpha_1)} \subset X^{(\alpha_2)} \subset X^{(\alpha_1)}$   $(\alpha_1 \le \alpha_2)$ , (c)  $X^{\ll \alpha, \beta_1, \beta_2 - \beta_1 \gg} \subset X^{\ll \alpha, \beta_2 \gg} \subset X^{\ll \alpha, \beta_1 \gg}$   $(\beta_1 \le \beta_2, \alpha \ge \omega)$ , and (d)  $X^{<\alpha>} \subset X^{\ll \alpha, 1 \gg}$   $(\alpha \ge \omega)$ .

**PROOF:** For (a) let  $x \in X^{(\alpha, 1)}$ . Then each neighborhood of x in X contains a closed copy of  $[1, \omega^{\alpha}] \times \mathbb{N}$ , hence a closed copy of  $[1, \omega^{\alpha})$ . So  $x \in X^{<\alpha>}$ . For  $x \in X^{<\alpha>}$  each neighborhood of x contains a closed copy of  $[1, \omega^{\alpha})$ , hence  $x \in X^{(\alpha)}$ .

For (b) we have by proposition 2.2.2,  $X^{(\alpha_2)} \subset X^{(\alpha_1)}$ . Furthermore

 $X^{(\alpha_1, \alpha_2 - \alpha_1)} \subset (X^{(\alpha_1)})^{(\alpha_2 - \alpha_1)} = (X^{(\alpha_1 + (\alpha_2 - \alpha_1))}) = X^{(\alpha_2)}.$ 

This completes the proof of (b).

For (c), observe that the second inclusion is a triviality, so we only have to prove that for  $\delta < \alpha$ ,  $X^{\ll \alpha, \beta_1, \beta_2 - \beta_1 \gg} \subset X^{(\delta, \beta_2)}$ . We will prove this by transfinite induction on  $\beta_2$ . If  $\beta_2 = 0$ ,  $\beta_1 = 0$  and we are done. Suppose for all  $\beta < \beta_2$ , the inclusion is true. If  $\beta_1 = \beta_2$  there is nothing to prove, so suppose  $\beta_1 < \beta_2$ . If  $\beta_2$  is a successor, say

$$\beta_2 = \beta + 1$$
, we have  $\beta_2 - \beta_1 = (\beta - \beta_1) + 1$ , so

$$X^{\ll\alpha,\ \beta_1,\ \beta_2-\beta_1\gg} = (X^{\ll\alpha,\ \beta_1,\ \beta-\beta_1\gg})^{(0,\ 1)} \subset (X^{(\delta,\ \beta)})^{(0,\ 1)} \subset X^{(\delta,\ \beta_2)}.$$

If  $\beta$  is a limit ordinal we have

$$X^{\boldsymbol{\ll}\boldsymbol{\alpha},\ \boldsymbol{\beta}_1,\ \boldsymbol{\beta}_2-\boldsymbol{\beta}_1\boldsymbol{\gg}} = \bigcap_{\boldsymbol{\beta}_1<\boldsymbol{\gamma}<\boldsymbol{\beta}_2} X^{\boldsymbol{\ll}\boldsymbol{\alpha},\ \boldsymbol{\beta}_1,\ \boldsymbol{\gamma}-\boldsymbol{\beta}_1\boldsymbol{\gg}} = \bigcap_{\boldsymbol{\beta}_1<\boldsymbol{\gamma}<\boldsymbol{\beta}_2} X^{(\boldsymbol{\delta},\ \boldsymbol{\gamma})} \subset X^{(\boldsymbol{\delta},\ \boldsymbol{\beta}_2)}.$$

This completes the proof of case (c).

For (d) we have to prove for  $\delta < \alpha$  that  $X^{<\alpha>} \subset X^{(\delta, 1)}$ . So let  $x \in X^{<\alpha>}$ . Then each neighborhood  $U_x$  of x contains a closed copy of  $[1, \omega^{\alpha})$ . Hence  $U_x^{(\delta)}$ , cannot be compact, so  $x \in X^{(\delta, 1)}$ . This completes the proof of (e) and hence the proof of this lemma.

**4.4.3 COROLLARY:** Let X be a space. Let  $\alpha$ ,  $\alpha_1$ ,  $\alpha_2$ ,  $\beta$ ,  $\beta_1$  and  $\beta_2$  be ordinals with  $\alpha$ ,  $\alpha_1$  and  $\alpha_2$  prime components. Then

(a)  $X^{(\alpha, 1+\beta)} \subset X^{<\alpha, \beta>} \subset X^{(\alpha, \beta)},$ (b)  $X^{(\alpha_1, \alpha_2-\alpha_1+\beta)} \subset X^{(\alpha_2, \beta)} \subset X^{(\alpha_1, \beta)}$   $(\alpha_1 \le \alpha_2),$ (c)  $X^{\ll\alpha, \beta_1, \beta_2-\beta_1+\beta\gg} \subset X^{\ll\alpha, \beta_2, \beta\gg} \subset X^{\ll\alpha, \beta_1, \beta\gg}$   $(\beta_1 \le \beta_2, \alpha \ge \omega),$  and (d)  $X^{<\alpha, \beta>} \subset X^{\ll\alpha, 1, \beta\gg}$   $(\alpha \ge \omega).$ 

From the lemma it also follows that some of the inclusions are in fact equalities. For example we have the following

**4.4.4 COROLLARY:** Let X be a space and let  $\alpha$ ,  $\beta$  be countable ordinals with  $\alpha$  a prime component and  $\beta \ge \omega$ . Then  $X^{(\alpha, \beta)} = X^{<\alpha, \beta>}$ .

**PROOF:** This follows directly from corollary 4.4.3 (a).  $\Box$ 

Comparing corollary 4.4.3 with lemma 4.2.1, we see that the general situation (countable metric spaces) gives rise to many more cases than the specific situation in section 4.2 (countable metric spaces with scattered height less than or equal to  $\omega$ ). Our next task is to find substitutes for corollary 4.2.2 and lemma 4.2.3. This is almost impossible. To make this clear we will restrict ourselves from now on to a specific class of spaces, which is the most natural one to consider after the results in section 4.2. We will consider countable metric spaces X with  $\omega < \kappa(X) < \omega^2$ . Many of the conditions in our conjecture become empty in this situation. The only ordinals we have to consider are  $\alpha = 0$ ,  $\alpha = 1$ ,  $\alpha = \omega$  and  $\beta$ ,  $\gamma < \omega^2$ . If we reformulate corollary 4.4.3 we get

176

**4.4.5 LEMMA:** Let X be a space. Let  $\beta$ ,  $\beta_1$ ,  $\beta_2 < \omega^2$  be ordinals. Then (a)  $X^{(\omega, 1+\beta)} \subset X^{<\omega, \beta>} \subset X^{(\omega, \beta)}$ , (b) (i)  $X^{(0, 1+\beta)} \subset X^{(1, \beta)} \subset X^{(0, \beta)}$ , (ii)  $X^{(0, \omega+\beta)} \subset X^{(\omega, \beta)} \subset X^{(0, \beta)}$ , (iii)  $X^{(1, \omega+\beta)} \subset X^{(\omega, \beta)} \subset X^{(1, \beta)}$ , (c)  $X^{<\omega, \beta_1, \beta_2-\beta_1+\beta>} \subset X^{<\omega, \beta_2, \beta>} \subset X^{<\omega, \beta_1, \beta>}$  ( $\beta_1 \le \beta_2$ ), and (d)  $X^{<\omega, \beta>} \subset X^{<\omega, 1, \beta>}$ .

In analogy with corollary 4.2.2 we define

 $\pi_{0} = \min\{\pi : X(0, \pi) = 0\},\$   $\pi_{1} = \min\{\pi : X(1, \pi) = 0\},\$   $\gamma_{\omega} = \min\{\gamma : X < \omega, \gamma > = 0\},\$  $\pi_{\omega} = \min\{\pi : X \ll \omega, \pi, 0 \gg = 0\},\$ 

and for every  $\pi \leq \pi_{\omega}$ ,

 $\beta_{\pi} = \min\{\beta: X \ll \omega, \pi, \beta \gg = 0\}.$ 

Of course all these ordinals are well-defined and less than or equal to  $\kappa(X)$ . Note that  $\pi_{\omega} > 0$  because of our choice of X. So  $\beta_1$  is well-defined.

4.4.6 LEMMA: We have the following relations

(a)  $\pi_1 \le \pi_0 \le 1 + \pi_1$ , (b)  $\gamma_{\omega} \le \beta_1 \le \beta_0 \le 1 + \gamma_{\omega} \le \pi_{\omega} \le \pi_1 \le \pi_0 \le \omega + \beta_0$ , (c)  $0 = \beta_{\pi_{\omega}} \le \beta_{\tau} \le \beta_{\sigma} \le \tau - \sigma + \beta_{\tau} \ (\sigma \le \tau \le \pi_{\omega})$ , and (d) for all  $\pi < \pi_{\omega}, \beta_{\pi} \ge 1$ .

**PROOF:** As in corollary 4.2.2, part (a) follows from lemma 4.4.5 (b)(i). For (b), notice that by lemma 4.4.5 (d),

 $X^{<\omega, \beta_1>} \subset X^{\ll\omega, 1, \beta_1>} = \emptyset.$ 

hence  $\gamma_{\omega} \leq \beta_1$ . That  $\beta_1 \leq \beta_0$  follows from (c) and will be proved there. By lemma 4.4.5 (a), we have

 $X^{\ll\omega, 0, 1+\gamma_{\omega}} = X^{(\omega, 1+\gamma_{\omega})} \subset X^{<\omega, \gamma_{\omega}} = \emptyset,$ 

hence  $\beta_0 \le 1 + \gamma_{\omega}$ . Since  $\pi_{\omega} > o$ , we have by lemma 4.4.5 (c) and (d),

 $X^{<\omega, \pi_{\omega}-1>} \subset X^{\ll\omega, 1, \pi_{\omega}-1\gg} \subset X^{\ll\omega, \pi_{\omega}\gg} = \emptyset,$ 

hence  $\gamma_{\omega} \leq \pi_{\omega} - 1$ . This implies  $1 + \gamma_{\omega} \leq \pi_{\omega}$ . Since

$$X^{\ll\omega, \pi_1 \gg} = \bigcap_{n < \omega} X^{(n, \pi_1)} \subset X^{(1, \pi_1)} = \emptyset,$$

we have  $\pi_{\omega} \le \pi_1$ . That  $\pi_1 \le \pi_0$  was proved under (a). Finally by lemma 4.4.5 (b)(ii) we have

$$X^{(0, \omega+\beta_0)} \subset X^{(\omega, \beta_0)} = X^{\ll \omega, 0, \beta_0 \gg} = \emptyset,$$

hence  $\pi_0 \le \omega + \beta_0$ . This completes the proof of (b).

For (c), we have by lemma 4.4.5 (c),

 $X^{\ll\omega, \sigma, \tau-\sigma+\beta_{\tau}\gg} \subset X^{\ll\omega, \tau, \beta_{\tau}\gg} = \emptyset.$ 

hence  $\beta_{\sigma} \leq \tau - \sigma + \beta_{\tau}$ , and

$$X^{\ll\omega, \tau, \beta_{\tau}\gg} \subset X^{\ll\omega, \sigma, \beta_{\sigma}\gg} = \emptyset,$$

hence  $\beta_{\tau} \leq \beta_{\sigma}$ . Obviously  $\beta_{\pi_{\omega}} = 0$ . This proves (c).

Part (d) follows immediately by the definition of  $\pi_{\omega}$ . This completes the proof of this lemma.  $\Box$ .

**4.4.7 COROLLARY:** If  $\sigma < \pi_{\omega}$ , then (a)  $\beta_{\sigma+1} \le \beta_{\sigma} \le 1 + \beta_{\sigma+1}$ , and (b)  $\beta_{\sigma} \le \pi_{\omega} - \sigma$ . Furthermore we have  $\beta_0 \ge 1$ , and for  $\pi \ge \pi_0$ ,  $\beta_{\pi} = 0$ .

**PROOF:** Part (a) and (b) are special cases of lemma 4.4.6 (c). By lemma 4.4.6 (b) we have  $\pi_{\omega} \ge 1 + \gamma_{\omega} \ge 1 > 0$ , hence by lemma 4.4.6 (d),  $\beta_0 \ge 1$ . Since  $\pi_{\omega} \le \pi_0$ , we have for  $\pi \ge \pi_0$ ,  $\beta_{\pi} = 0$ .

Comparing lemma 4.4.6 with corollary 4.2.2 we notice that lemma 4.4.6 covers more cases than corollary 4.2.2. If we want to follow the same strategy as in section 4.2, our next step should be to find a substitute for lemma 4.2.3. In this lemma we had for each natural number exactly six possible cases. This made the situation there suitable for the inductive proof in lemma 4.2.13. The role of that natural number is now played by  $\pi_0$ . However if we increase  $\pi_0$ , the number of possibilities will also increase. This makes it difficult to find a suitable replacement for lemma 4.2.3. To make this clear we will now look at some specific values of  $\pi_0$ . If  $\pi_0$  is fixed,  $\gamma_{\omega}$ ,  $\pi_{\omega}$ ,  $\pi_1$  and  $\beta_{\pi}$ for  $\pi < \pi_0$  are the only defined ordinals that can possibly be non-zero (cf. lemma 4.4.6 and corollary 4.4.7).
First let  $\pi_0 = 1$ . Then we have to consider  $\gamma_{\omega}$ ,  $\pi_{\omega}$ ,  $\pi_1$  and  $\beta_0$ . By lemma 4.4.6 (b) and corollary 4.4.7 we then have  $1 = \beta_0 \le 1 + \gamma_{\omega} \le \pi_{\omega} \le \pi_1 \le \pi_0$ . This gives

$\pi_0 = 1$			
γω	β0	πω	$\pi_1$
0	1	1	1

A space X satisfying this condition can be one of the following types (cf. lemma 4.2.3).

(1)  $X(0, 0) = X(1, 0) = X(\omega, 0) = 1$ , (2) X(0, 0) = 2, and  $X(1, 0) = X(\omega, 0) = 1$ , (3) X(0, 0) = X(1, 0) = 2, and  $X(\omega, 0) = 1$ , and (4)  $X(0, 0) = X(1, 0) = X(\omega, 0) = 2$ ,

Spaces of each of these four types in fact exist. We will give the examples, but leave all calculations to the reader. For case (1) let  $X = [1, \omega^{\omega}]$ , and for case (2) let  $X = [1, \omega^{\omega}] \oplus \mathbb{N}$ . The space  $[1, \omega^{\omega}] \oplus ([1, \omega] \times \mathbb{N})$  satisfies the conditions of (3) and  $[1, \omega^{\omega}] \times \mathbb{N}$  satisfies the conditions of (4).

Let us now consider the case that  $\pi_0 = 2$ . We then have to consider  $\gamma_{\omega}$ ,  $\pi_{\omega}$ ,  $\pi_1$ ,  $\beta_1$  and  $\beta_0$ . Lemma 4.4.6 and corollary 4.4.7 give us the following possibilities

$\pi_0 = 2$					
	γω	$\beta_1$	β <sub>0</sub>	πω	$\pi_1$
1	0	0	1	1	1
2	0	0	1	1	2
3	0	1	1	2	2
4	1	1	1	2	2
5	1	1	2	2	2

Each possibility gives rise to several cases in the same way as for  $\pi_0 = 1$ . The number of these possibilities increases since there is one more variable. We will not try to explicitly describe all these cases, but instead we shall give one example for each of the five cases in the above table, to make clear that the relations in lemma 4.4.6 are at

least in this case sharp.

A space satisfying the conditions in (1) is  $T \oplus [1, \omega^{\omega}]$  and a space satisfying the conditions in (2) is  $T([1, \omega]) \oplus [1, \omega^{\omega}]$ . Let X be the space obtained from T by replacing each  $(i, j) \in T$  by a copy of  $[1, \omega^i]$ . Then X satisfies the conditions in (3). For case (4) we can take  $X = S_1([1, \omega^{\omega}))$  and for case (5) we can take  $X = T([1, \omega^{\omega}])$ . Again we leave all calculations to the reader.

One can see that for  $\pi_0 = 2$ , the situation involves a lot more possibilities then for  $\pi_0 = 1$ . If we consider  $\pi_0 = 3$ , we get another new variable  $\beta_2$ , and by lemma 4.4.6 and corollary 4.4.7 the following table:

$\pi_0 = 3$						
	γω	β2	$\beta_1$	β <sub>0</sub>	πω	$\pi_1$
1	0	0	0	1	1	2
2	0	0	1	1	2	2
3	0	0	0	1	1	3
4	0	0	1	1	2	3
5	0	1	1	1	3	3
6	1	0	1	1	2	2
7	1	0	1	2	2	2
8	1	0	1	1	2	3
9	1	0	1	2	2	3
10	1	1	1	1	3	3
11	1	1	1	2	3	3
12	1	1	2	2	3	3
13	2	1	2	2	3	3
14	2	1	2	3	3	3

Hence  $\pi_0 = 3$  yields 14 possibilities and each of them can be dealt with as in the proof of lemma 4.2.3. It is also possible to describe for each of the above cases an example of a space satisfying the corresponding conditions. It goes too far to present them here.

For  $\pi_0 = 4$ , the number of cases has increased to 32, and things get even worse if  $\pi_0$ 

is infinite. Then there are infinitely many variables to consider, and the number of possibilities will also be infinite. But if we want to follow the same strategy as in section 4.2, we need an explicit description of all possibilities and that is exactly the problem when one wants to find "decomposition lemmas" which can be used in a lemma such as lemma 4.2.13. It is also not obvious that the relations in lemma 4.4.6 are sharp. Can one find for each possibility a space satisfying the corresponding condition?

Another problem in connection with lemma 4.2.13 is to find "linear k-homeomorphisms". In section 4.2 we were in the pleasant situation that for each pair of  $l_p$ -equivalent spaces both with scattered height less than  $\omega$ , we actually had a linear k-homeomorphism for some  $k \in \mathbb{N}$ . That property is lost if we consider spaces X with  $\omega < \kappa(X) < \omega^2$ . This problem already occurs in the case of spaces satisfying  $\pi_0 = 1$ . These spaces should be the "building blocks" for spaces with  $\pi_0 = 2$  (if we want to start a proof by induction on  $\pi_0$ ). If we let  $X = [1, \omega^{\omega}] \oplus ([1, \omega] \times \mathbb{N})$  and  $Y = [1, \omega^{\omega}] \oplus [1, \omega^{\omega})$ , then both spaces have scattered height  $\omega + 1$ , and they both have  $\pi_0 = 1$ . By theorem 2.6.17 they are moreover  $l_p$ -equivalent. However, we cannot use X and Y as building blocks for obtaining linear homeomorphisms between function spaces of spaces with  $\pi_0 = 2$  using lemma 4.2.10 because of the following

**4.4.8 LEMMA:** There is no linear k-homeomorphism between  $C_p(X)$  and  $C_p(Y)$  for any  $k \in \mathbb{N}$ .

**PROOF:** Suppose there is  $k \in \mathbb{N}$  and a linear k-homeomorphism between  $C_p(X)$  and  $C_p(Y)$ . Then by lemma 4.2.10,  $S_1(X)$  and  $S_1(Y)$  are  $l_p$ -equivalent. However  $(S_1(X))^{\ll \omega, 1 \gg} = \emptyset$  and  $(S_1(Y))^{\ll \omega, 1 \gg} \neq \emptyset$ . This contradicts corollary 4.3.2.  $\Box$ 

Comparing the situation here to the one in section 4.2, we see that if we want to "decompose" spaces into spaces of a "lower level", we are forced to avoid situations as above in order to make it possible to apply lemma 4.2.10.

Summarizing we conclude that a proof of the conjecture, even in the case of the relatively simple spaces X with  $\omega < \kappa(X) < \omega^2$ , will be a hard job and will certainly not be as "simple" as the proof given in section 4.2.

We will finish this section by giving a relatively simple result concerning the  $l_p$ -equivalence of non-scattered countable metric spaces.

**4.4.9 PROPOSITION:** Let X be a non-scattered countable metric space. Then X and Q are  $l_p$ -equivalent.

**PROOF:** Since X is a non-scattered countable metric space, X contains a closed copy D of  $\mathbb{Q}$ . Since X is a countable metric space,  $\mathbb{Q}$  contains a closed copy E of  $X \times \mathbb{N}$ . Then

$$C_{p}(X) \sim C_{p,D}(X) \times C_{p}(\mathbb{Q}) \qquad \text{by proposition 2.3.2}$$

$$\sim C_{p,D}(X) \times C_{p}(\mathbb{Q}) \times C_{p}(\mathbb{Q}) \qquad \mathbb{Q} \approx \mathbb{Q} \oplus \mathbb{Q}$$

$$\sim C_{p}(X) \times C_{p,E}(\mathbb{Q}) \times \prod_{i=1}^{\infty} C_{p}(X) \qquad \text{by proposition 2.3.2}$$

$$\sim C_{p,E}(\mathbb{Q}) \times \prod_{i=1}^{\infty} C_{p}(X) \qquad \text{by proposition 2.3.2} \square$$

### §4.5. Partial results on $l_0$ -equivalence

In the previous section we saw that a complete isomorphical classification for the function spaces  $C_p(X)$ , for X countable and metric, seems beyond reach. Only for countable metric spaces with scattered height less than or equal to  $\omega$  the situation is clear. In this section we deal with the compact-open topology instead of the topology of pointwise convergence and we try to make clear that an isomorphical classification for the function spaces  $C_0(X)$ , where X is countable and metric, seems even more beyond reach. First we notice that from theorem 4.2.18 and corollary 1.2.21 we have the following

**4.5.1 THEOREM:** Let X and Y be infinite countable metric spaces, such that  $\kappa(X), \kappa(Y) \leq \omega$ , and for every  $n \in \mathbb{N}, X(0, n) = Y(0, n)$  and X(1, n) = Y(1, n). Then X and Y are  $l_0$ -equivalent.  $\Box$ 

In the proof of the converse implication for  $l_p$ -equivalence, we used the notion of  $l_p$ -equivalent pairs. Of course we can define in a similar way the notion of  $l_0$ -equivalent pairs, and it is then possible to prove a theorem such as theorem 4.1.7. The problem however lies in propositions 4.1.9 and 4.1.12, where we used the precise description of supports obtained in section 1.4 for the topology of pointwise convergence. We were unable to derive such a precise description of supports in the case of the compact-open topology. This complicates the situation quite a bit. We will now derive two theorems that should be compared with corollary 4.1.14.

**4.5.2 THEOREM:** Let X and Y be zero-dimensional separable metric  $l_0$ -equivalent spaces. Then

(a)  $X = \emptyset$  if and only if  $Y = \emptyset$ ,

(b) X is compact if and only if Y is compact, and

(c) X is locally compact if and only if Y is locally compact.

**PROOF:** Part (a) is a triviality. Part (b) follows from theorem 1.5.7. For (c) let  $\phi: C_0(X) \to C_0(Y)$  be a linear homeomorphism and suppose that X is locally compact and Y is not locally compact. Then X can be written as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} A_i$  such that for each  $i \in \mathbb{N}$ ,  $A_i$  is compact. Let  $y \in Y$  be such that y has no compact neighborhood and let  $\{U_n : n \in \mathbb{N}\}$  be a decreasing clopen base at y. By corollary 1.2.15 (a) and lemma 1.2.10, there is  $n \in \mathbb{N}$  such that

 $(\operatorname{supp} U_n) \cap \bigcup_{i > n} A_i = \emptyset.$ 

Let  $A = \bigcup_{i=1}^{n} A_i$ . Then A is compact. So  $\overline{\text{supp}A}$  is compact as well. Since  $U_n$  is not compact, there is a non-empty clopen  $O \subset U_n \setminus \overline{\text{supp}A}$ . Now let  $f \neq 0$  be a Urysohn function such that  $f(Y \setminus O) = \{0\}$ . Since  $Y \setminus O$  is a neighborhood of  $\overline{\text{supp}A}$ , by corollary 1.2.15 (a),  $\phi^{-1}(f)(A) = \{0\}$ . Since A is a neighborhood of supp O, we consequently have  $f(O) = \{0\}$ . But then f = 0, which is a contradiction. This proves the theorem.  $\Box$ 

**4.5.3 THEOREM:** Let X and Y be zero-dimensional separable metric  $l_0$ -equivalent spaces. Then

(a)  $X^{(1)} = \emptyset$  if and only if  $Y^{(1)} = \emptyset$ ,

(b)  $X^{(1)}$  is compact if and only if  $Y^{(1)}$  is compact, and

(c)  $X^{(1)}$  is locally compact if and only if  $Y^{(1)}$  is locally compact.

**PROOF:** Let  $\phi: C_0(X) \to C_0(Y)$  be a linear homeomorphism. For (a) suppose  $X^{(1)} = \emptyset$ and  $Y^{(1)} \neq \emptyset$ . Let K be a copy of  $[1, \omega]$  in Y and let L = supp K. Then L is non-empty and compact and hence is finite. By lemma 4.1.11, there is a continuous linear function  $\eta_1: C_0(K) \to C_0(Y)$  such that for each  $f \in C(K)$ ,  $\eta_1(f) \mid K = f$  and there is a continuous linear function  $\eta_2: C_0(L) \to C_0(X)$  such that for each  $f \in C(L)$ ,  $\eta_2(f) \mid L = f$ . Define

 $\theta: C_0(K) \rightarrow C_0(L)$  by  $\theta(f) = \phi^{-1}(\eta_1(f)) | L$ , and  $\psi: C_0(L) \rightarrow C_0(K)$  by  $\psi(g) = \phi(\eta_2(g)) | K$ .

CLAIM: For every  $f \in C(K)$ ,  $\psi(\theta(f)) = f$ .

To the contrary suppose there is  $f \in C(K)$  such that  $\psi(\theta(f)) \neq f$ . Then

 $\phi(\eta_2(\theta(f))|K \neq \eta_1(f)|K$ . Since *L* is open, *L* is a neighborhood of supp *K*. Hence by corollary 1.2.15 (a),  $\eta_2(\theta(f))|L \neq \phi^{-1}(\eta_1(f))|L$ , which gives  $\theta(f) \neq \theta(f)$ . Contradiction. This proves the claim.

We conclude that  $\theta$  is a linear embedding. So we have a linear embedding from  $C_0([1, \omega])$  into some  $\mathbb{R}^n$  for  $n \in \mathbb{N}$ . This is not possible since the algebraic dimension of  $C_0([1, \omega])$  is infinite. This proves (a).

For (b) suppose  $X^{(1)}$  is compact and  $Y^{(1)}$  is not compact. By (a) we have  $X^{(1)} \neq \emptyset$ . Let  $\{y_n : n \in \mathbb{N}\}$  be a closed discrete subset of Y consisting of non-isolated points. Let  $\{O_n : n \in \mathbb{N}\}$  be a clopen discrete family such that for each  $n \in \mathbb{N}$ ,  $y_n \in O_n$ . Let  $\{U_n : n \in \mathbb{N}\}$  be a clopen decreasing base at  $X^{(1)}$  in X. By corollary 1.2.15 (a) and lemma 1.2.10, there is  $k \in \mathbb{N}$  such that

 $\operatorname{supp} U_k \cap \bigcup_{i \ge k} O_i = \emptyset.$ 

Find a copy K of  $[1, \omega]$  in  $O_k$  containing  $y_k$ . Let  $L = \overline{\operatorname{supp} K} \cap X \setminus U_k$ . Then L is compact, and hence is finite. If  $L = \emptyset$ , then  $U_k$  is a neighborhood of supp K. Furthermore  $Y \setminus O_k$  is a neighborhood of supp  $U_k$ . Let f be a Urysohn function such that  $f(y_k) = 1$  and  $f(Y \setminus O_k) = \{0\}$ . Since  $\phi^{-1}$  is effective,  $\phi^{-1}(f)(U_k) = \{0\}$ . By effectiveness of  $\phi$ , we then have  $f(K) = \{0\}$ . But this gives a contradiction since  $y_k \in K$ . We conclude that  $L \neq \emptyset$ . By lemma 4.1.11 and proposition 1.2.19, there is a continuous linear function  $\eta_1: C_0(K) \to C_0(Y)$  such that for each  $f \in C(K)$ ,  $\eta_1(f) \mid K = f$  and  $\eta_1(f)(Y \setminus O_k) = \{0\}$ , and there is a continuous linear function  $\eta_2: C_0(L) \to C_0(X)$  such that for each  $f \in C(L)$ ,  $\eta_2(f) \mid L = f$  and  $\eta_2(f)(U_k) = \{0\}$ . Define  $\theta: C_0(K) \to C_0(L)$  by  $\theta(f) = \phi^{-1}(\eta_1(f)) \mid L$ , and  $\psi: C_0(L) \to C_0(K)$  by  $\psi(g) = \phi(\eta_2(g)) \mid K$ .

CLAIM: For every  $f \in C(K)$ ,  $\psi(\theta(f)) = f$ .

To the contrary suppose there is  $f \in C_0(K)$  such that  $\psi(\theta(f)) \neq f$ . Then  $\phi(\eta_2(\theta(f)) | K \neq \eta_1(f) | K$ . Since  $U_k \cup L$  is a neighborhood of L, we have by effectiveness of  $\phi$  that  $\eta_2(\theta(f)) | (U_k \cup L) \neq \phi^{-1}(\eta_1(f)) | (U_k \cup L)$ . Now  $\eta_2(\theta(f)) \equiv 0$  on  $U_k$  and  $\eta_1(f) \equiv 0$  on  $Y \setminus O_k$ . Since  $Y \setminus O_k$  is a neighborhood of supp  $U_k$  we have by effectiveness of  $\phi^{-1}$ ,  $\phi^{-1}(\eta_1(f)) \equiv 0$  on  $U_k$ . We conclude that  $\eta_2(\theta(f)) | L \neq \phi^{-1}(\eta_1(f) | L$ , hence  $\theta(f) \neq \theta(f)$ . Contradiction. This proves the claim.

From the claim it follows that  $\theta$  is a linear embedding. Again we have a linear embedding from  $C_0([1, \omega])$  into some  $\mathbb{R}^n$  for  $n \in \mathbb{N}$ , which gives a contradiction. This proves case (b).

For (c) suppose that  $X^{(1)}$  is locally compact and  $Y^{(1)}$  is not locally compact. By (b) we have that  $X^{(1)}$  is not compact. Hence X can be written as a clopen disjoint union

 $X = \bigcup_{i=1}^{\infty} A_i$  such that for each  $i \in \mathbb{N}$ ,  $A_i^{(1)}$  is a non-empty compactum. Let  $y \in Y^{(1)}$  be a point and let  $\{U_n : n \in \mathbb{N}\}$  be a clopen decreasing base at y such that for each  $n \in \mathbb{N}$ ,  $U_n^{(1)}$  is not compact. By corollary 1.2.15 (a) and lemma 1.2.10, there is  $k \in \mathbb{N}$  such that

 $\operatorname{supp} U_k \cap \bigcup_{i>k} A_i = \emptyset.$ 

Let  $A = \bigcup_{i=1}^{k} A_i$ . Then  $A^{(1)}$  is a non-empty compactum. Let  $\{V_n : n \in \mathbb{N}\}$  be a clopen decreasing base at  $A^{(1)}$  in A. Let  $\{y_n : n \in \mathbb{N}\}$  be a closed discrete set of non-isolated points in  $U_k$  and let  $\{O_n : n \in \mathbb{N}\}$  be a clopen discrete family in  $U_k$  such that for each  $n \in \mathbb{N}$ ,  $y_n \in O_n$ . By corollary 1.2.15 (a) and lemma 1.2.10, there is  $p \in \mathbb{N}$  such that

 $\operatorname{supp} V_p \cap \bigcup_{i \ge p} O_i = \emptyset.$ 

Find a copy K of  $[1, \omega]$  in  $O_k$  containing  $y_k$ . Let  $L = \overline{\operatorname{supp} K} \cap X \setminus V_p$ . As under case (b) we can derive a contradiction. This proves the theorem.  $\Box$ 

In the above theorems the proofs are similar to the proofs in section 4.1. The next prime component to consider is  $\omega$ . The specific problems that we encounter when dealing with the compact-open topology now become clear. We are able to prove the following theorem which is much weaker than the result we have for  $l_p$ -equivalence. It is a generalization of a result in [6].

**4.5.4 THEOREM:** Let X and Y be zero-dimensional separable metric spaces. Suppose  $X^{(1)}$  is discrete and  $Y^{(\omega)} \neq \emptyset$ . Then X and Y are not  $l_0$ -equivalent.

**PROOF:** To the contrary suppose there is a linear homeomorphism  $\phi: C_0(X) \to C_0(Y)$ . Write X as a clopen disjoint union  $X = \bigcup_{i=1}^{\infty} X_i$  such that for each  $i \in \mathbb{N}, X_i^{(1)}$  contains at most one point. By theorem 4.5.3, there is at least one  $i \in \mathbb{N}$  with  $X_i^{(1)} \neq \emptyset$ . Let K be a copy of  $[1, \omega^{\omega}]$  in Y and let  $L = \operatorname{supp} K$ . Then L is compact, hence there is  $p \in \mathbb{N}$  such that  $L \subset \bigcup_{i=1}^{p} X_i$ . We may assume that for each  $i \in \mathbb{N}, X_i^{(1)} \neq \emptyset$ , and we let  $x_i$  denote the unique point in  $X_i^{(1)}$ .

CLAIM 1: For every  $f, g \in C_0(Y)$  with  $f | K \neq g | K$  it follows that  $\phi^{-1}(f) | L \neq \phi^{-1}(g) | L$ .

Let  $y \in K$  be such that  $f(y) \neq g(y)$  and let  $W_0$  and  $W_1$  be disjoint open neighborhoods of f(y) and g(y) in  $\mathbb{R}$ , respectively. Then  $A(\{y\}, W_0)$  and  $A(\{y\}, W_1)$  are disjoint open neighborhoods of f and g in  $C_0(Y)$ . So  $\phi^{-1}(A(\{y\}, W_0))$  and  $\phi^{-1}(A(\{y\}, W_1))$  are disjoint open neighborhoods of  $\phi^{-1}(f)$  and  $\phi^{-1}(g)$  in  $C_0(X)$ .

There consequently exist compact subsets  $K_1, \ldots, K_n, L_1, \ldots, L_m$  of X and open subsets  $U_1, \ldots, U_n, V_1, \ldots, V_m$  of  $\mathbb{R}$  such that

$$\phi^{-1}(f) \in \bigcap_{i=1}^{n} A(K_i, U_i) \subset \phi^{-1}(A(\{y\}, W_0))$$

and

$$\phi^{-1}(g) \in \bigcap_{i=1}^{m} A(L_i, V_i) \subset \phi^{-1}(A(\{y\}, W_1)).$$

We claim there is a  $z \in \text{supp}(y) \subset L$  such that  $\phi^{-1}(f)(z) \neq \phi^{-1}(g)(z)$  (and then we are done). Striving for a contradiction, assume the contrary. Let  $M = \{k \leq p \mid x_k \in \text{supp}(y)\}$ . Then by assumption we have that for every  $k \in M$ ,  $\phi^{-1}(f)(x_k) = \phi^{-1}(g)(x_k)$ . For every  $k \in M$  let  $I_k = \{i \leq n \mid x_k \notin K_i\}, J_k = \{i \leq m \mid x_k \notin L_i\}$  and

$$P_k = \bigcup_{i \in I_k} (K_i \cap X_k) \cup \bigcup_{i \in J_k} (L_i \cap X_k).$$

Then  $P_k$  is compact in  $T_k$  and  $x_k \notin P_k$ , so  $P_k$  is finite. Let  $P = \bigcup_{k \in M} P_k$ . Then P is finite and  $P \cap \{x_k : k \in M\} = \emptyset$ . Define  $f': X \to \mathbb{R}$  by

$$f'(x) = \begin{cases} \phi^{-1}(f)(x) & \text{if } x \in P \cup \bigcup_{i \notin M} X_i, \\ \phi^{-1}(f)(x_k) & \text{if } x \in X_k \setminus P_k \text{ for } k \in M \end{cases}$$

and define  $g': X \to \mathbb{R}$  by

$$g'(x) = \begin{cases} \phi^{-1}(g)(x) & \text{if } x \in P \cup \bigcup_{i \notin M} X_i, \\ \phi^{-1}(g)(x_k) & \text{if } x \in X_k \setminus P_k \text{ for } k \in M. \end{cases}$$

Then f' and g' are continuous since for each  $k \in M$ ,  $X_k \setminus P_k$  is a neighborhood of  $x_k$ .

Let  $U = \bigcup_{k \in M} X_k \setminus P_k \cup \text{supp}(y)$ . Then U is a neighborhood of supp(y) on which f'and g' coincide. Since  $\phi$  is effective we have  $\phi(f')(y) = \phi(g')(y)$ .

On the other hand  $f' \in \bigcap_{i=1}^{n} A(K_i, U_i)$ . Indeed let  $i \le n$  and  $x \in K_i$ . If  $x \in P \cup \bigcup_{i \notin M} X_i$ , then  $f'(x) = \phi^{-1}(f)(x) \in U_i$  since  $\phi^{-1}(f) \in \bigcap_{i=1}^{n} A(K_i, U_i)$ . If  $x \notin P \cup \bigcup_{i \notin M} X_i$  we have  $x \in X_k \setminus P_k$  for some  $k \in M$ . Since  $x \in K_i \cap X_k$  and  $x \notin P_k$  we have  $x_k \in K_i$ , so  $f'(x) = \phi^{-1}(f)(x_k) \in U_i$ . Similarly one can prove that

 $g' \in \bigcap_{i=1}^{m} A(L_i, V_i)$ . We then have  $\phi(f') \in A(\{y\}, W_0)$  and  $\phi(g') \in A(\{y\}, W_1)$ . But this means  $\phi(f')(y) \neq \phi(g')(y)$ , which gives a contradiction. This completes the proof of the claim.

By lemma 4.1.11, there is a continuous linear function  $\eta_1: C_0(K) \to C_0(Y)$  such that for each  $f \in C(K)$ ,  $\eta_1(f) | K = f$  and there is a continuous linear function  $\eta_2: C_0(L) \to C_0(X)$  such that for each  $f \in C(L)$ ,  $\eta_2(f) | L = f$ . Define

 $\theta: C_0(K) \to C_0(L) \text{ by } \theta(f) = \phi^{-1}(\eta_1(f)) \mid L, \text{ and}$  $\psi: C_0(L) \to C_0(K) \text{ by } \psi(g) = \phi(\eta_2(g)) \mid K.$ 

CLAIM 2:  $\theta$  is a linear embedding.

It is easy to see that  $\theta$  and  $\theta$  are well-defined continuous functions. We claim that for every  $h \in C_0(K)$  we have  $\psi(\theta(h)) = h$ . To the contrary suppose  $\phi(\eta_2(\theta(h))) | K \neq \eta_1(h) | K$ . By claim 1 we have  $\eta_2(\theta(h)) | L \neq \phi^{-1}(\eta_1(h)) | L$ . But this implies  $\theta(h) \neq \theta(h)$ . Contradiction. Hence  $\theta$  is a linear embedding.

By claim 2, we have a linear embedding from  $C_0([1, \omega^{\omega}])$  into  $C_0([1, \alpha])$ , where  $\alpha < \omega^2$ . However this contradicts theorem 2.4.1. This proves the theorem.  $\Box$ 

# §4.6. Partial results on $l_p^*$ -equivalence

We would like to have classification results for the spaces  $C_p^*(X)$  as we had for the spaces  $C_p(X)$  in the previous sections of this chapter. The theory developed there depends strongly on results derived in chapter 1 (corollary 1.2.15 (b)). Example 1.2.12 shows that the method for  $C_p(X)$  cannot be used for  $C_p^*(X)$ , i.e., we cannot prove a theorem such as theorem 4.1.7 for " $l_p^*$ -equivalent pairs". We have to find another way to prove results for the function spaces  $C_p^*(X)$ .

In this section we will prove for  $l_p^*$ -equivalent metric spaces X and Y, that  $\kappa(X) < \omega$ if and only if  $\kappa(Y) < \omega$ . The proof of this result is a generalization of Pelant's proof that  $C_p^*(T)$  and  $C_p^*(\mathbb{Q})$  are not linearly homeomorphic (cf. [42]). The reader should compare this result with theorem 4.1.15, which states that for  $l_p$ -equivalent metric spaces X and Y,  $\kappa(X) \le \omega$  if and only if  $\kappa(Y) \le \omega$ .

We first need the following definition, which can be found in [23]. A family  $\mathcal{F} \subset C(X)$  is *equicontinuous* if for every  $x \in X$  and  $\varepsilon > 0$ , there is a neighborhood U of x in X such that for each  $f \in \mathcal{F}$  and  $y \in U$ ,  $|f(x) - f(y)| < \varepsilon$ . The following result is well-known.

**4.6.1 PROPOSITION:** If  $\mathcal{F} \subset C_{\mu}(X)$  is compact, then  $\mathcal{F}$  is equicontinuous.

**PROOF:** Let  $x \in X$  and  $\varepsilon > 0$ . The family  $\{\langle f, X, \varepsilon/3 \rangle : f \in \mathcal{F}\}$  is an open cover of  $\mathcal{F}$ . Since  $\mathcal{F}$  is compact, there are  $f_1, \ldots, f_n \in \mathcal{F}$   $(n \in \mathbb{N})$  such that  $\{\langle f_i, X, \varepsilon/3 \rangle : i \leq n\}$  covers  $\mathcal{F}$ . Since each  $f_i$  is continuous, there is a neighborhood U of x such that for all  $y \in U$  and for every  $i \leq n$ ,  $|f_i(y) - f_i(x)| < \varepsilon/3$ . Now let  $f \in \mathcal{F}$  and  $y \in U$ . There is  $i \leq n$  such that  $f \in \langle f_i, X, \varepsilon/3 \rangle$ . This implies  $|f_i(x) - f(x)| < \varepsilon/3$  and  $|f_i(y) - f(y)| < \varepsilon/3$ . Since  $y \in U$ , we now have

$$|f(x) - f(y)| \le |f(x) - f_i(x)| + |f_i(x) - f_i(y)| + |f_i(y) - f(y)| < \varepsilon. \square$$

**4.6.2 THEOREM:** Let X and Y be first countable  $l_p^*$ -equivalent spaces. Then

(a)  $\kappa(X) < 1$  if and only if  $\kappa(Y) < 1$ ,

(b)  $\kappa(X) < 2$  if and only if  $\kappa(Y) < 2$ .

**PROOF:** For (a) observe that  $\kappa(X) < 1$  if and only if  $X = \emptyset$ .

For (b) suppose  $\kappa(X) < 2$  and  $\kappa(Y) \ge 2$ . Then by (a),  $\kappa(X) = 1$ , which gives that X is discrete. Since  $\kappa(Y) \ge 2$  there is  $y \in Y$  which is non-isolated. Let  $\{U_n : n \in \mathbb{N}\}$  be a decreasing open base at y in Y. For every  $n \in \mathbb{N}$  let  $f_n$  be a Urysohn function with  $f_n(y) = 1$  and  $f_n(Y \setminus U_n) = 0$ . Then  $f_n \to \chi_{\{y\}}$  pointwise in  $\mathbb{R}^Y$ . Since  $\chi_{\{y\}} \notin C_p^*(Y)$ ,  $\{f_n : n \in \mathbb{N}\}$  is closed and discrete in  $C_p^*(Y)$ .

Now let  $\phi: C_p^*(X) \to C_p^*(Y)$  be a linear homeomorphism. Then by The Closed Graph Theorem,  $\phi: C_u^*(X) \to C_u^*(Y)$  is also a linear homeomorphism. Since  $C_u^*(X)$  and  $C_u^*(Y)$  are Banach spaces, there is  $k \in \mathbb{N}$  such that for every  $f \in C^*(X)$  we have

 $\frac{1}{k} \|f\| \le \|\phi(f)\| \le k \|f\|.$ 

Let  $g_n = \phi^{-1}(f_n)$ . Then  $||g_n|| \le k ||f_n|| = k$ . Hence  $\{g_n : n \in \mathbb{N}\} \subset [-k, k]^X$ . Since  $[-k, k]^X$  is compact,  $\{g_n : n \in \mathbb{N}\}$  has an accumulation point  $g \in [-k, k]^X$ . Since X is discrete  $[-k, k]^X \subset C_p^*(X)$  and so  $g \in C_p^*(X)$ . However, since  $\{f_n : n \in \mathbb{N}\}$  is closed and discrete in  $C_p^*(Y)$ ,  $\{g_n : n \in \mathbb{N}\}$  is closed and discrete in  $C_p^*(X)$ . Contradiction.  $\Box$ 

Before we prove our announced result we need two fairly simple lemmas. One deals with function spaces and the other one deals with nets.

**4.6.3 LEMMA:** Let X be a metric space with  $\kappa(X) < \omega$ . There is a metric space Y such that  $\kappa(Y) = \kappa(X)$  and  $C_p^*(X) \sim C_{p,A}^*(Y)$  where  $A = Y^{(1)}$ .

**PROOF:** We prove the lemma by induction on  $\kappa(X)$ . If  $\kappa(X) = 1$ , let Y = X. So suppose the lemma has been proved for metric spaces X with  $\kappa(X) < n$  (n > 1). Let X be a metric space with  $\kappa(X) = n$  and let  $B = X^{(1)}$ . Then by proposition 2.3.2, the remark following lemma 2.3.6, and proposition 2.2.2 (a)  $C_p^*(X) \sim C_p^*(B) \times C_{p,B}^*(X)$ . Since  $\kappa(B) = n - 1$  (corollary 2.2.3), there is by the inductive hypothesis a metric space Z such that  $\kappa(Z) = \kappa(B)$  and  $C_p^*(B) \sim C_{p,C}^*(Z)$  where  $C = Z^{(1)}$ . Then  $C_p^*(X) \sim C_{p,C}^*(Z) \times C_{p,B}^*(X) = C_{p,B \cup C}^*(Z \oplus X)$ . Let  $Y = Z \oplus X$ . Then  $Y^{(1)} = B \cup C$  and  $\kappa(Y) = \kappa(X)$ . This finishes the proof of the lemma.  $\Box$ 

**4.6.4 LEMMA:** Let X be a space and B an infinite set. For every  $b \in B$  let  $f_b \in \mathbb{R}^X$  such that for every  $x \in X$ ,  $\{b \in B : f_b(x) \neq 0\}$  is finite. Furthermore let  $\mathcal{G} = \{S \subset B : S \text{ is finite}\}$  and define a relation  $\leq$  on  $\mathcal{G}$  as follows: If  $S_1, S_2 \in \mathcal{G}$  then  $S_1 \leq S_2$  if  $S_1 \subset S_2$ . For every  $S \in \mathcal{G}$  define  $f_S = \Sigma_{b \in S} f_b$ . Then  $\{f_S : S \in \mathcal{G}\}$  is a net in  $\mathbb{R}^X$  and  $\lim_{S \in \mathcal{G}} f_S = \Sigma_{b \in B} f_b$ .

**PROOF:** It is easily seen that  $\mathscr{S}$  is directed by  $\leq$ . Since every  $S \in \mathscr{S}$  is finite,  $f_S \in \mathbb{R}^X$ , hence  $\{f_S : S \in \mathscr{S}\}$  is a net in  $\mathbb{R}^X$ .

Now let  $\varepsilon > 0$  and  $P \subset X$  finite. For every  $p \in P$  let  $S_p = \{b \in B : f_b(p) \neq 0\}$  and  $S_0 = \bigcup_{p \in P} S_p$ . Then  $S_0 \in \mathcal{S}$ . Let  $S \ge S_0$ ,  $p \in P$  and  $f = \sum_{b \in B} f_b$ . Then

 $|f(p) - f_S(p)| = |\Sigma_{b \in B} f_b(p) - \Sigma_{b \in S} f_b(p)| = |\Sigma_{b \in S_n} f_b(p) - \Sigma_{b \in S_n} f_b(p)| = 0 < \varepsilon.$ 

Hence  $\lim_{S \in \mathcal{S}} f_S = f_{\Box}$ 

We now come to the result announced in the introduction of this section.

**4.6.5 THEOREM:** Let X and Y be  $l_p^*$ -equivalent metric spaces. Then  $\kappa(X) < \omega$  if and only if  $\kappa(Y) < \omega$ .

**PROOF:** Suppose  $\kappa(X) < \omega$  and  $\kappa(Y) \ge \omega$ . By lemma 4.6.3 we may assume  $C_{p,A}^*(X) \sim C_p^*(Y)$  where  $A = X^{(1)}$ . Let  $\phi: C_{p,A}^*(X) \to C_p^*(Y)$  be a linear homeomorphism. Then by The Closed Graph Theorem,  $\phi: C_{u,A}^*(X) \to C_u^*(Y)$  is also a linear homeomorphism  $(C_{u,A}^*(X)$  has its obvious meaning). So there is  $k \in \mathbb{N}$  such that for every  $f \in C_{u,A}^*(X)$  we have

On the  $l_p$ -equivalence of metric spaces

$$\frac{1}{k} \|f\| \le \|\phi(f)\| \le k \|f\|.$$

Let  $B = X \setminus A$ . Since every element of *B* is an isolated point in *X*, we have for each  $x \in B$ that  $f_x = \chi_{\{x\}} \in C_{p,A}^*(X)$ , where  $\chi_{\{x\}}$  is the characteristic function of *x*. Notice that for each  $f \in C_{p,A}^*(X)$ ,  $f = \sum_{x \in B} \alpha_x f_x$ , where  $\alpha_x = f(x)$ . For each  $x \in B$ , let  $g_x = \phi(f_x)$ .

For every  $y \in Y$ , let  $C_y = \{x \in B : g_x(y) \neq 0\}$ . Since  $C_y \subset \text{supp}(y)$ ,  $C_y$  is finite for every  $y \in Y$ . Now define  $b: Y \to \mathbb{R}$  by  $b(y) = \sum_{x \in B} |g_x(y)|$ . Notice that for every  $y \in Y$ ,  $b(y) = \sum_{x \in C_y} |g_x(y)|$ , hence b is well-defined.

#### CLAIM 1: $||b|| \le 2k$ .

For  $y \in Y$ , let  $C_y^+ = \{x \in B : g_x(y) > 0\}$  and  $C_y^- = \{x \in B : g_x(y) < 0\}$ . Notice that  $\|\sum_{x \in C_y^+} g_x\| = \|\phi(\sum_{x \in C_y^+} f_x)\| \le k \cdot \|\sum_{x \in C_y^+} f_x\| = k$ . Similarly we can prove that  $\|\sum_{x \in C_y^-} g_x\| \le k$ . So

$$|b(y)| = |\sum_{x \in C_y^+} g_x(y) - \sum_{x \in C_y^-} g_x(y)| \le |\sum_{x \in C_y^+} g_x(y)| + |\sum_{x \in C_y^-} g_x(y)| \le 2k,$$

which proves the claim.

Now for  $P \subset B$  finite let  $\mathcal{M}_P = \{ \sum_{x \in P} \alpha_x f_x : |\alpha_x| \le k \text{ for } x \in P \}$ . Notice that  $\mathcal{M}_P = \prod_{x \in P} [-k, k] \times \prod_{x \in X \setminus P} \{0\}.$ 

CLAIM 2: For every  $y \in Y$ ,  $P \subset B$  finite and  $\varepsilon > 0$ , there is a neighborhood  $U(y, P, \varepsilon)$  of y in Y such that for each  $z \in U(y, P, \varepsilon)$  and  $f \in \phi(\mathcal{M}_P)$ ,  $|f(y) - f(z)| < \varepsilon$ .

Notice that  $\mathcal{M}_P$  is compact in  $C_{p,A}^*(X)$ . It is easily seen that for every  $f \in \mathcal{M}_P$  and  $\varepsilon > 0, B(f, \varepsilon) \cap \mathcal{M}_P = \langle f, P, \varepsilon \rangle \cap \mathcal{M}_P$ . Since P is finite it now follows that  $\mathcal{M}_P$  is compact in  $C_{u,A}^*(X)$  and so  $\phi(\mathcal{M}_P)$  is compact in  $C_u^*(Y)$ . Hence by proposition 4.6.1,  $\phi(\mathcal{M}_P)$  is equicontinuous, from which the claim follows.

Now find  $N \in \mathbb{N}$  such that  $\frac{3}{4k}(N+1) \ge 2k$ .

CLAIM 3: There are  $y_0, \ldots, y_N \in Y, P_0, \ldots, P_N \subset B$  finite and  $U_0, \ldots, U_N$  neighborhoods of respectively  $y_0, \ldots, y_N$ , such that

(1) for every  $i \le N$ :  $C_{y_i} \subset P_i$ , (2)  $P_0 \subset P_1 \subset \cdots \subset P_N$ , (3)  $U_0 \supset U_1 \supset \cdots \supset U_N$ , (4) for every  $i \le N$ :  $U_i \subset U(y_i, P_i, 1/4)$ , and (5) for every  $i \le N$ :  $y_i \in Y^{(N-i)}$ .

190

We will prove this claim by induction. Since  $\kappa(Y) \ge \omega$ , we can find  $y_0 \in Y^{(N)}$ . Let  $P_0 = C_{y_0}$  and  $U_0 = U(y_0, P_0, 1/4)$ . Suppose  $y_0, \ldots, y_n, P_0, \ldots, P_n$  and  $U_0, \ldots, U_n$  are found for  $0 \le n < N$ . Since  $y_n \in Y^{(N-n)}$  and  $N-n \ge 1$ , we can find  $y_{n+1} \in U_n \setminus \{y_i : i \le n\} \cap Y^{(N-(n+1))}$ . Let  $P_{n+1} = P_n \cup C_{y_{n+1}}$  and

$$U_{n+1} = U_n \cap U(y_{n+1}, P_{n+1}, \frac{1}{4}).$$

This completes the inductive construction and hence the proof of the claim.

Now let  $g: Y \to [-1, 1]$  be a continuous function such that  $g(y_i) = (-1)^i$  for  $0 \le i \le N$ . Then ||g|| = 1, so  $||\phi^{-1}(g)|| \le k$ . So  $\phi^{-1}(g) = \sum_{x \in B} \alpha_x f_x$  with  $|\alpha_x| \le k$ . Notice that  $\sum_{x \in P_i} \alpha_x f_x \in \mathcal{M}_{P_i}$  for every  $0 \le i \le N$ .

CLAIM 4:  $g = \sum_{x \in B} \alpha_x g_x$ 

Indeed, let  $\mathcal{G} = \{S \subset B : S \text{ is finite}\}$  and for every  $S \in \mathcal{G}$  let  $f_S = \sum_{x \in S} \alpha_x f_x$ . By lemma 4.6.4  $\phi^{-1}(g) = \lim_{S \in \mathcal{G}} f_S$  and  $\sum_{x \in B} \alpha_x g_x = \lim_{S \in \mathcal{G}} \sum_{x \in S} \alpha_x g_x$ . So

$$g = \phi(\phi^{-1}(g)) = \phi(\lim_{S \in \mathscr{G}} f_S) = \lim_{S \in \mathscr{G}} \phi(f_S) = \lim_{S \in \mathscr{G}} \Sigma_{x \in S} \alpha_x g_x = \Sigma_{x \in B} \alpha_x g_x,$$

and the claim is proved.

Let  $0 \le i \le N$ . Since  $C_{y_i} \subset P_i$  (claim 3 (1)), we have by claim 4,

$$(-1)^{l} = \sum_{x \in B} \alpha_{x} g_{x}(y_{i}) = \sum_{x \in P_{i}} \alpha_{x} g_{x}(y_{i}).$$

By claim 3 (3) and (4),  $y_N \in U(y_i, P_i, 1/4)$ . Furthermore  $\sum_{x \in P_i} \alpha_x g_x \in \phi(\mathcal{M}_{P_i})$ , so by claim 2,

$$|\Sigma_{x \in P_i} \alpha_x g_x(y_N) - \Sigma_{x \in P_i} \alpha_x g_x(y_i)| < \frac{1}{4}.$$

If i > 0, we have by claim 3 (2)

$$\begin{aligned} |\Sigma_{x \in P_{i} \setminus P_{i-1}} \alpha_{x} g_{x}(y_{N})| &= |\Sigma_{x \in P_{i}} \alpha_{x} g_{x}(y_{N}) - \Sigma_{x \in P_{i-1}} \alpha_{x} g_{x}(y_{N})| \\ &= |\Sigma_{x \in P_{i}} \alpha_{x} g_{x}(y_{N}) - (-1)^{i} + (-1)^{i-1} - \Sigma_{x \in P_{i-1}} \alpha_{x} g_{x}(y_{N}) \pm 2| \\ &\geq 2 - |\Sigma_{x \in P_{i}} \alpha_{x} g_{x}(y_{N}) - \Sigma_{x \in P_{i}} \alpha_{x} g_{x}(y_{i})| - \\ &|\Sigma_{x \in P_{i-1}} \alpha_{x} g_{x}(y_{N}) - \Sigma_{x \in P_{i-1}} \alpha_{x} g_{x}(y_{i-1})| > \frac{3}{4}. \end{aligned}$$

If i = 0 and  $P_{-1} = \emptyset$ , then

$$|\sum_{x \in P_i \setminus P_{i-1}} \alpha_x g_x(y_N)| = |\sum_{x \in P_0} \alpha_x g_x(y_N) - \sum_{x \in P_0} \alpha_x g_x(y_0) + 1| > 1 - \frac{1}{4} = \frac{3}{4}.$$

So by claim 3 (2)

$$\sum_{x \in P_N} |\alpha_x g_x(y_N)| \geq \sum_{i=0}^N |\sum_{x \in P_i \setminus P_{i-1}} \alpha_x g_x(y_N)| > \frac{3}{4} (N+1),$$

hence

$$b(y_N) = \sum_{x \in P_N} |g_x(y_N)| \ge \sum_{x \in P_N} |\frac{\alpha_x}{k} g_x(y_N)| > \frac{3}{4k} (N+1) \ge 2k,$$

which gives a contradiction.  $\Box$ 

The isomorphical classification of the function spaces in section 2.6 and the first part of lemma 2.3.7 are not valid for the function spaces of bounded continuous functions as is shown in the following

**4.6.6 EXAMPLE:** There are  $l_p$ -equivalent countable metric locally compact spaces which are not  $l_p^*$ -equivalent.

**PROOF:** Let  $X = [1, \omega^2)$  and  $Y = [1, \omega^{\omega})$ . Since X is an open subspace of  $[1, \omega^2]$ ,  $X^{(\alpha)} = X \cap [1, \omega^2]^{(\alpha)}$  (proposition 2.2.4). Hence by proposition 2.2.5  $\kappa(X) = 2$ . Similarly,  $\kappa(Y) = \omega$ , so by theorem 4.6.5, X and Y are not  $\ell_p^*$ -equivalent spaces. However, by example 2.6.18, X and Y are  $l_p$ -equivalent.  $\Box$ 

**4.6.7 REMARK:** Let  $2 < n < \omega$ . By proposition 2.4.4 there is  $k \in \mathbb{N}$  such that

$$C_p^*([1, \omega]) = C_p([1, \omega]) \stackrel{k}{\sim} C_p([1, \omega^{n-1}]) = C_p^*([1, \omega^{n-1}]).$$

So by lemma 2.3.7 (and the remark just before lemma 2.3.7),

$$C_p^*([1,\,\omega^2))=C_p^*(\bigoplus_{i=0}^\infty[1,\,\omega])\stackrel{k}{\sim} C_p^*(\bigoplus_{i=0}^\infty[1,\,\omega^n))=C_p^*([1,\,\omega^n)).$$

This implies that the following is not true: If X and Y are  $l_p^*$ -equivalent spaces and n > 2, then  $\kappa(X) < n$  if and only if  $\kappa(Y) < n$ .

Motivated by theorems 4.6.2 and 4.6.5 and remark 4.6.7 we state the following:

**4.6.8 CONJECTURE:** Let X and Y be  $l_p^*$ -equivalent metric spaces and let  $\alpha$  be a prime component. Then

(a)  $\kappa(X) < \alpha$  if and only if  $\kappa(Y) < \alpha$ , and (b)  $\kappa(X) < \alpha + 1$  if and only if  $\kappa(Y) < \alpha + 1$ .

192

### References

- [1] A.V. Arhangel'skil, On linear homeomorphisms of function spaces, Soviet Math. Dokl. 25 (1982) 852-855.
- [2] A.V. ArhangelskiI, A survey of C<sub>p</sub>-theory, Questions and Answers in General Topology 5 (1987) 1-109.
- [3] J. Baars and J. de Groot, An isomorphical classification of function spaces of zerodimensional locally compact separable metric spaces, Comm. Math. Univ. Carolinae 29 (1988) 577-595.
- [4] J. Baars and J. de Groot, On the *l*-equivalence of metric spaces, Fund. Math. 137 (1991) 25-43.
- [5] J. Baars, J. de Groot and J. van Mill, A theorem on function spaces, Proc. Amer. Math. Soc. 105 (1989) 1020-1024.
- [6] J. Baars, J. de Groot, J. van Mill and J. Pelant, On topological and linear homeomorphism of certain function spaces, Top. and its Appl. 32 (1989) 267-277.
- [7] J. Baars, J. de Groot and J. Pelant, Function spaces on completely metrizable spaces, to appear in Trans. Amer. Math. Soc..
- [8] J.W. Baker, Compact spaces homeomorphic to a ray of ordinals, Fund. Math. 76 (1972) 19-27.
- [9] J.W. Baker, Ordinal subspaces of topological spaces, Gen. Top. and its Appl. 3 (1973) 85-91.
- [10] C. Bessaga and A. Pelczyński, Spaces of continuous functions IV (on isomorphical classification of spaces of continuous functions), Studia Math. 19 (1960) 53-62.
- [11] C. Bessaga and A. Pelczyński, Selected topics in infinite-dimensional topology, PWN Warszawa (1975).
- [12] M. Bestvina, P.L. Bowers, J. Mogilski and J.J. Walsh, *Characterizing Hilbert space topology revisited*, Top. and its Appl. 24 (1986) 53-69.
- [13] M. Bestvina and J. Mogilski, Characterizing certain incomplete infinite-dimensional ab-

solute retracts, Michigan Math. J. 33 (1986) 291-313

- [14] K. Borsuk, Theory of Retracts, PWN Warszawa (1967).
- [15] L.E.J. Brouwer, On the structure of perfect sets of points, Proc. Akad. Amsterdam, 12 (1910) 785-794.
- [16] R. Cauty, L'espace des fonctions continues d'un espace métrique dénombrable, Proc. Amer. Math. Soc. 113 (1991) 493-501.
- [17] T.A. Chapman, Dense sigma-compact subsets of infinite-dimensional manifolds, Trans. Amer. Math. Soc. 154 (1971) 399-426.
- [18] T.A. Chapman, Lectures on Hilbert cube manifolds, CMBS Regional Conf. Series in Math. No. 28, Amer. Math. Soc., Providence, R.I. (1976).
- [19] J.J. Dijkstra, T. Grilliot, D.J. Lutzer and J. van Mill, Function spaces of low Borelcomplexity, Proc. Amer. Math. Soc. 94 (1985) 703-710.
- [20] T. Dobrowolski, S.P. Gulko and J. Mogilski, Function spaces homeomorphic to the countable product of  $l_t^2$ , Top. and its Appl. 34 (1990) 153-160.
- [21] E.K. van Douwen, *The integers and topology*, Handbook of Set-theoretical topology, North-Holland Publishing Company, Amsterdam (1984) 111-167.
- [22] J. Dugundji, An extension of Tietze's theorem, Pac. J. Math. 1 (1951) 353-367.
- [23] J. Dugundji, Topology, Allyn and Bacon Inc, Boston (1966).
- [24] R. Engelking, General Topology, PWN Warszawa (1977).
- [25] L. Gillman and M. Jerison, *Rings of continuous functions*, Graduate texts in Mathematics, Springer Verlag, New York (1976).
- [26] I. Gohberg and S. Goldberg, Basic Operator Theory, Birkhäuser (1981).
- [27] S. Gulko, The space  $C_p(X)$  for countable infinite compact X is uniformly homeomorphic to  $c_0$ , preprint.
- [28] S.P. Gulko and A.V. Oskin, Isomorphic classification of spaces of continuous functions on totally ordered compact sets, Functional Analysis and its Applications 9 (1975) 56-57.

#### References

- [29] J.A. Guthrie, Ascoli theorems and the pseudocharacter of mapping spaces, Bull. Austral. Math. Soc. 10, No.3 (1974) 403-408.
- [30] D.W. Henderson, Infinite dimensional manifolds are open subsets of Hilbert space, Topology 9 (1970) 25-33.
- [31] S.T. Hu, Theory of Retracts, Wayne State University Press, Detroit (1965).
- [32] T. Jech, Set Theory, Academic Press New York, (1965).
- [33] O.H. Keller, Die Homoiomorphie der kompakten konvexen mengen in Hilbertschen raum, Math. Ann. 105 (1931) 748-758.
- [34] S.V. Kislyakov, Classification of spaces of continuous functions of ordinals, Siber. Math. J. 16 (1975) 226-231.
- [35] K. Kuratowski and A. Mostowski, *Set theory*, North-Holland Publicing Company, Amsterdam (1976).
- [36] S. Mazurkiewicz and W. Sierpiński, Contributions à la topologie des ensembles dénomerables, Fund. Math. 1 (1920) 17-27.
- [37] R.A. McCoy and I. Ntantu, *Topological properties of spaces of continuous functions*, Lecture Notes in Mathematics No. 1315, Springer Verlag, Berlin Heidelberg (1988).
- [38] J. van Mill, *Topological equivalence of certain function spaces*, Comp. Math. 63 (1987) 159-188.
- [39] J. van Mill, Infinite Dimensional Topology: prerequisites and introduction, North-Holland Publishing Company, Amsterdam (1989).
- [40] J. Nagata, On lattices of functions on topological spaces and functions on uniform spaces, Osaka Math. J. 1, No. 2 (1949) 166-181.
- [41] D. Pavlovskil, On spaces of continuous functions, Soviet Math. Dokl. 22 (1980) 34-37.
- [42] J. Pelant, A remark on spaces of bounded continuous functions, Indag. Math. 91 (1988) 335-338.
- [43] A. Pelczyński and Z. Semadeni, Spaces of continuous functions III (Spaces  $C(\Omega)$  for  $\Omega$  without perfect subsets), Studia. Math. 18 (1959) 211-222.

- [44] V.G. Pestov, *The coincidence of the dimension* dim *of l-equivalent topological spaces*, Soviet Math. Dokl. 26 (1982) 380-383.
- [45] H.H. Schaeffer, *Topological Vector Spaces*, Macmillan series in advanced mathematics and theoretical physics, New York (1966).
- [46] Z. Semadeni, Banach spaces non-isomorphic to their Cartesian squares II, Bull. Acad. Pol. Sci 8 (1960) 81-84
- [47] Z. Semadeni, Banach spaces of continuous functions, PWN Warszawa (1971).
- [48] W. Sierpiński, Cardinal and ordinal numbers, PWN Warszawa (1958).
- [49] A.H. Stone, Kernel constructions and Borel sets, Trans. of the Am. Math. Soc. 107 (1963) 58-70.
- [50] H. Toruńczyk, On cartesian factors and the topological classification of linear metric spaces, Fund. Math. 88 (1975) 71-86.
- [51] H. Toruńczyk, (G,K)-absorbing and skeletonized sets in metric spaces, unpublished manuscript.
- [52] V.V Uspenskii, On the topology of a free locally convex space. Soviet Math. Dokl. 27 (1983) 781-785.

# **Special Symbols**

N	8	<b>א</b> <sub>0</sub>	8
Q	8	C(X)	9
R	8	$C^{*}(X)$	9
С	8	<f, ε="" κ,=""></f,>	9
$X \approx Y$	8	$C_p(X)$	10
$X \sim Y$	8	$C_0(X)$	10
$\overline{A}$	8	$C_b(X)$	10
Int A	8	A(K, U)	11
minA	8	$C_p^*(X)$	12
maxA	8	$C_0^*(X)$	12
inf A	8	$C_b^*(X)$	12
sup A	8	$C_u^*(X)$	12
$B(x, \varepsilon)$	8	supp(y)	12
diam A	8	supp A	12
d(X)	8	ξ <sub>x</sub>	24
<i>X</i>	8	L(X)	24
$\mathcal{P}(X)$	8	L(f)	26
$X \oplus Y$	8	$L^{*}(X)$	28
$\bigoplus_{i=1}^{\infty} X_i$	8	$l_p$	31
$f \circ g$	8	40	31
$id_X$ , 1 <sub>X</sub> , 1	8	$l_b$	31
$\pi_i \colon \prod_{i \in I} X_i \to X_i$	8	$\ell_p^*$	32
$\chi_A$	8	$\ell_0^*$	32
11.11	8	$\ell_b^*$	32
Ker F	8	$\operatorname{cof} \mathcal{K}(X)$	33
span $\{v_1,\ldots,v_n\}$	8	$\overline{\alpha}$	45
$\operatorname{conv}\left\{v_1,\ldots,v_n\right\}$	8	i ( <b>φ</b> )	45
S <sup>n</sup>	8	$cf(\lambda)$	46
$B^{n+1}$	8	$\omega_{lpha}$	46
$W(\alpha)$	8	$\lim_{\xi \to \infty} \phi(\xi)$	46
[1, α]	8	ς<α α'	49
[1, α)	8	$X^{(\alpha)}$	51
ω	8	Ad	51
$\omega_1$	8	 κ(X)	54

Special Symbols

X/A	54
$C_{p,A}(X)$	56
$C_{p,a}(X)$	56
$\prod_{t \in T} E_t \stackrel{k}{\sim} \prod_{s \in S} F_s$	56
$C_{p, 0}([1, \alpha])$	61
$\Pi_{t\in T}^* E_t$	70
$\Pi_{\mathfrak{m}} E$	70
$\Pi_{t\in T}^*\Pi_{i\in I}^*E_{t,i}$	72
$C_{p,0}(X)$	99
$C_{p,0}^{*}(X)$	99
$l^2$	99
$l_f^2$	99
σω	99
C(X, Y)	101
$\mathcal{H}(X, Y)$	101
$\mathcal{H}(X)$	101
$\hat{d}(f, g)$	101
$\mathcal{I}(X)$	101
$\mathcal{I}_{\sigma}(X)$	101
Σ	103
$\Sigma_n$	103
ker A	105
ΣΕ	127
W(X, x)	130
$X^{\{\alpha\}}$	136
$X^{(\alpha,\beta)}$	138
<i>X</i> ≪α,β≫	138
$X(\alpha, \beta)$	149
<i>X</i> ≪α,β,γ≫	164
<i>Χ</i> ≪α, β, γ≫	164
$X^{<\alpha,\beta>}$	174
$X^{<\alpha>}$	174
<i>X</i> <α, β>	174

198

## **Author Index**

A.V. Arhangelskii	6,9,12,13,24,31,32		
J.W. Baker	142		
C. Bessaga	43,62,63,66,69		
M. Bestvina	130,131		
R. Cauty	100,129		
T. Dobrowolski	100,129		
J. Dugundji	57		
S.P. Gulko	37,81,100,129,133		
J.A. Guthrie	6		
D.W. Henderson	108		
S.V. Kislyakov	43,68,69,81		
S. Mazurkiewicz	54		
A.A. Miljutin	63		
J. van Mill	6,7,99,104,106,124,127,129		
J. Mogilski	100,129,130,131		
J. Nagata	6		
O.G. Okunev	37		
A.V. Oskin	81		
J. Pelant	6,7,31,187		
A. Pelczyński	43,62,63,66,69		
V.G. Pestov	6,23		
Z. Semadeni	69		
W. Sierpiński	54		
H. Toruńczyk	100,106,124,127		
V.V Uspenskiĭ	41		

# Subject Index

absolute $F_{\sigma\delta}$	128	index of an ordinal	45
absolute neighborhood retract	108	initial ordinal	45
absolute retract	108		
α-th derivative	51	Keller space	101
α-sequence	46	kernel of a $\mathcal{I}$ -matrix	105
bounded subset	9	<i>l<sub>b</sub></i> -equivalent	31
bounded type	13	$l_0$ -equivalent	31
		$l_p$ -equivalent	31
Cantor Bendixson Theorem	53	limit of an increasing $\alpha$ -sequence	46
Closed Graph Theorem	23	linear k-homeomorphism	56
cofinal	46	linear k-mapping	56
compact-open topology	10	linear <i>m</i> -embedding	170
compact ordinal space	68	Lower Semi Continuous (LSC)	15
contractible	109		
- -		Miljutin's Theorem	23
dense in itself	53		
derived set	51	power of an ordinal	45
		preserved by $l_b$ -equivalence	32
effective	13	preserved by $l_b^*$ -equivalence	32
equicontinuous	187	preserved by $l_0$ -equivalence	32
evaluation mapping	24	preserved by $l_0^*$ -equivalence	32
eventually constant	14	preserved by $l_p$ -equivalence	32
exponentation of ordinals	45	preserved by $l_p^*$ -equivalence	32
		prime component	48
$(\phi, \alpha)$ -relative pair	139	pseudo-boundary	101
		pseudo-interior	101
Hamel basis	25		
hemicompact	33	<i>Q</i> -matrix	105
Hilbert cube	100		
homotopically trivial	109	regular ordinal	46
homotopy type	108	relatively bounded	167
increasing $\alpha$ -sequence	46	same linear dimension	69

### Subject Index

	н. Н
scattered height	54
scattered space	53
Sierpinski-Mazurkiewicz Theore	em 54
singular ordinal	46
skeletoid	103
skeleton	102
smaller linear dimension	69
$\sigma_{\omega}$ -manifold	108
strong $\mathcal{I}$ -set	131
strongly $\mathcal{F}_{\sigma\delta}$ -universal	131
support	12
$\sigma$ -initial ordinal	82
$\sigma$ -ordinal	82
$\sigma$ -prime component	82
$\sigma \mathcal{I}$ -set	101
topology of pointwise convergen	ice 10
topology of uniform convergence	e 12
	 ,
U-close functions	130
Upper Semi Continuous (USC)	15
$\mathcal{I}$ -embedding	101
<i>I</i> -matrix	104
$\mathcal{Z}$ -set	101, 130

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

2 A.R. Bloemena. Sampling from a graph. 1964.

3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

Todat una menuoa. 1904.
 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.
 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

W.R. van Zwet. Convex transformations of random variables. 1964

8 J.A. Zonneveld. Automatic numerical integration. 1964. 9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964. 11 A.B. Paalman-de Miranda. Topological semigroups. 1964.

12 J.A. Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965.

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print; replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1. 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part I. 1968

27 European Meeting 1968. Selected statistical papers, part II. 1968

28 J. Oosterhoff. Combination of one-sided statistical tests.

1969 29 J. Verhoeff. Error detecting decimal codes. 1969

30 H. Brandt Corstius. Exercises in computational linguistics.

1970

31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.

33 F.W. Steutel. Preservation of infinite divisibility under mix-ing and related topics. 1970.

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary layers. 1971.

23. Simanan. On the orth of boundary layers. 1971.
37. J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvestijn, E.W. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972. 42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973. 45 A.A. Balkema. Monotone transformations and limit laws. 1973.

1913.
46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.
47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973.
48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok. Connected orderable spaces. 1974. 50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.), *Revised report on the algorithmic language ALGOL* 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory 1974

52 P.C. Baayen (ed.). Topological structures. 1974.

53 M.J. Faber. Metrizability in generalized ordered spaces. 1974.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974. 56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency concept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975.

63 J.W. de Bakker (ed.). Foundations of computer science. 1975.

64 W.J. de Schipper. Symmetric closed categories. 1975. 65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979. 74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977.

76 M. Rem. Associons and the closure statement. 1976 77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Emperical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary

problems. 1977 81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976.

82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

83 L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.

84 H.L.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallman spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and analytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen, Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteris-tic. 1979.

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. (eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bice 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bice 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. Go-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979.

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part 1: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part 11: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments and renewal theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979.

115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 1. 1979.

116 P.C. Baayen, J. van Mill (eds.). Topological structures 11, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980. 121 W.H. Haemers. Eigenvalue techniques in design and graph

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

123 I. Yuhász. Cardinal functions in topology - ten years later. 1980.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980. 126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures.* 1980

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.

132 H.M. Mulder. The interval function of a graph. 1980. 133 C.A.J. Klaassen. Statistical performance of location estirs. 1981.

134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part I. 1981. 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part II. 1981.

137 J. Telgen. Redundancy and linear programs. 1981. 138 H.A. Lauwerier. Mathematical models of epidemics. 1981.

139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis.

1981

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981. 142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a symposium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes. 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequences. 1982

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired ser-vices. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.
156 D.M.C. Appenderung representation and deta reliance in a set of the reliance of the re

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two dimensional smooth commutative formal groups over an algebraically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983. 161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

162 J.J. Dik. Tests for preference. 1983.

163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.

164 F.A. van der Duyn Schouten. Markov decision pro-with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.

#### CWI TRACTS

1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984

2 J.J. Dijkstra. Fake topological Hilbert spaces and characteri-zations of dimension in terms of negligibility. 1984.

3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984

4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984. 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.

6 A.P.W. Böhm. Dataflow computation. 1984. 7 A. Blokhuis. Few-distance sets. 1984.

8 M.H. van Hoorn. Algorithms and approximations for queue-ing systems. 1984.

9 C.P.J. Koymans. Models of the lambda calculus. 1984.

10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.

11 N.M. van Dijk. Controlled Markov processes; time-discretization. 1984.

12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985

13 D. Grune. On the design of ALEPH. 1985.

14 J.G.F. Thiemann. Analytic spaces and dynamic program-ming: a measure theoretic approach. 1985.

15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.

16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial dif-ferential operators: a case-study in Fourier integral operators. 1985.

17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.

18 A.D.M. Kester. Some large deviation results in statistics. 1985

19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.

20 B.F. Schriever. Order dependence. 1986.

21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986. 22 J.C.S.P. van der Woude. Topological dynamix. 1986.

23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.

24 J.C.M. Baeten. Filters and ultrafilters over definable sub-sets of admissible ordinals. 1986.

25 A.W.J. Kolen. Tree network and planar rectilinear location

theory, 1986. 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.

27 A.J.M. van Engelen. Homogeneous zero-dimensional abso-lute Borel sets. 1986.

28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.

29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.

30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.

32 J.H.J. Einmahl. Multivariate empirical processes. 1987.

33 O.J. Vrieze. Stochastic games with finite state and action spaces. 1987.

34 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.

35 M.L. Eaton. Lectures on topics in probability inequalities. 1987

36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.

37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.

38 J.B.G. Frenk. On Banach algebras, renewal measures and

regenerative processes. 1987.

39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.

40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems, 1987.

41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.

42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, for-malisms, and tools. 1987.

43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.

44 A.W. van der Vaart. Statistical estimation in large parame-ter spaces. 1988.

45 S.A. van de Geer. Regression analysis and empirical processes. 1988.

46 S.P. Spekreijse. Multigrid solution of the steady Euler equa-tions. 1988.

47 J.B. Dijkstra. Analysis of means in some non-standard situations. 1988.

48 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988.

49 F.W. Wubs. Numerical solution of the shallow-water equations 1988.

50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.

51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing, 1988.

52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.

53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic pro-gramming in function space. 1988.

54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.

55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.

56 J.C. Ebergen. Translating programs into delay-insensitive circuits. 1989.

57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.

58 M.C.M. de Gunst. A random model for plant cell popula-tion growth. 1989.

59 D. van Dulst. Characterizations of Banach spaces not con-taining l<sup>1</sup>. 1989.

60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

61 P. de Jong. Central limit theorems for generalized multil-inear forms, 1989.

62 V.J. de Jong. A specification system for statistical software. 1989

63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.

64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989.

65 B.M.M. de Weger. Algorithms for diophantine equations. 1989

66 A. Jung. Cartesian closed categories of domains. 1989. 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.

68 H.J. Woerdeman. Matrix and operator extensions. 1989.

69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.

70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedi-cated to Gijs de Leve. 1990.

71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.

72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990

73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.

74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991. 75 M.W.P. Savelsbergh. Computer aided routing. 1992.

76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991.
77 A.J. van Es. Aspects of nonparametric density estimation. 1991.

78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.

O.A.F. Kindervalet. Electorses in particle combinational computing 1992.
 J. Lodder. Towards a symmetrical theory of generalized functions. 1991.
 S.A. Smulders. Control of freeway traffic flow. 1992.
 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992.
 F. Thuijsman. Optimality and equilibria in stochastic games. 1992.
 R.J. Kooman. Convergence properties of recurrence sequences. 1992.
 A.M. Cohen (ed.). Computational aspects of Lie group representational Algebra Seminar at CWJ, Amsterdam. 1991.
 V. de Valk. One-dependent processes. 1992.
 J.A. Baars, J.A.M. de Groot. On topological and linear equivalence of certain function spaces. 1992.

,